Registration	No.	33-	

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form S-4 REGISTRATION STATEMENT

UNDER
THE SECURITIES ACT OF 1933

THE PEOPLES HOLDING COMPANY

(Exact name of registrant as specified in its charter)

Mississippi (State or other jurisdiction of incorporation or organization) 6022 (Primary Standard Industrial Classification Code Number) 64-0676974 (I.R.S. Employer Identification Number)

209 Troy Street Tupelo, Mississippi 38802-0709 (662) 680-1001

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Stephen M. Corban Executive Vice President 209 Troy Street Tupelo, Mississippi 38802-0709 (662) 680-1001

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Mark A. Fullmer Phelps Dunbar LLP Suite 2000 365 Canal Street New Orleans, LA 70130 (504) 566-1311 Robert Walker
Baker, Donelson, Bearman, Caldwell &
Berkowitz, P.C.
165 Madison Avenue
Memphis, TN 38103
(901) 577-0785

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this registration statement.

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box. \Box

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

CALCULATION OF REGISTRATION FEE

Title of each class of securities to be registered	Amount to be registered	Proposed maximum offering price per unit	Proposed maximum aggregate offering price	Amount of registration fee
Common Stock, \$5.00 par value	1,478,000(1)	Not applicable	\$12,307,212.27(2)	\$1,560.00(2)

- (1) This amount is based upon the number of shares of common stock anticipated to be issued upon completion of the transactions contemplated in the Agreement and Plan of Merger dated as of February 17, 2004 (the "Merger Agreement"), by and among The Peoples Holding Company ("Peoples"), Peoples Merger Corporation and Renasant Bancshares, Inc. ("Renasant").
- (2) Determined pursuant to Rule 457(f)(2) under the Securities Act of 1933, as amended, solely for the purpose of calculating the registration fee, based on the book value of common stock, \$1.00 par value, of Renasant as of the latest practicable date prior to the filing date (February 29, 2004). Pursuant to Rule 457(f)(3), the cash portion of the consideration to be paid by Peoples pursuant to the Merger Agreement has been deducted from the book value of the securities to be received by Peoples or cancelled in the transaction.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

Information contained in this proxy statement/prospectus is subject to completion or amendment. A registration statement relating to these securities has been filed with the Securities and Exchange Commission. These securities may not be sold nor may offers to buy be accepted prior to the time the registration statement becomes effective. This proxy statement/prospectus shall not constitute an offer to sell or the solicitation of an offer to buy nor shall there be any sale of these securities in any State in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of such State.

Neither the Securities and Exchange Commission nor any state securities regulator has approved or disapproved the shares of common stock to be issued by Peoples in the merger, as described in this proxy statement/prospectus, or passed upon the adequacy or accuracy of this proxy statement/prospectus. Any representation to the contrary is a criminal offense.

The shares of Peoples common stock to be issued in the merger are not savings or deposit accounts or other obligations of any bank or savings association or non-bank subsidiary of Peoples and are not insured by the Federal Deposit Insurance Corporation, the Savings Association Insurance Fund, the Bank Insurance Fund or any other governmental agency.

[GRAPHIC]

Dear Renasant Stockholders:

The Peoples Holding Company

Renasant Bancshares, Inc.
Proxy Statement
The Peoples Holding Company
Prospectus

You are cordially invited to attend the special meeting of stockholders of Renasant Bancshares, Inc. ("Renasant") which will be held at Renasant's main office, 2177 Germantown Road South, Germantown, Tennessee 38138, on ________, 2004, at 5:30 p.m. local time. The accompanying notice of the special meeting, proxy statement/prospectus and proxy card set forth the formal business to be transacted at the special meeting. Please review these materials carefully and attend the special meeting.

At the special meeting, you will be asked to vote upon a proposal to adopt and approve a merger agreement, a related plan of merger and a merger of Renasant into a wholly owned subsidiary of The Peoples Holding Company ("Peoples"). If the merger is completed, each share of Renasant common stock you hold will be exchanged for either (1) \$36.37 in cash, (2) 1.117015 shares of Peoples common stock, \$5.00 per share par value, or (3) a combination consisting of cash for 45% of your common stock and shares of Peoples common stock for 55% of your common stock. You will be asked to elect your form of payment. Regardless of your election, however, elections will be limited by the requirements that not less than 50% or more than 55% of the aggregate shares of Renasant common stock owned by Renasant stockholders be exchanged for Peoples common stock and that not less than 45% or more than 50% of the aggregate shares of Renasant common stock owned by Renasant stockholders be exchanged for cash. Your election is also subject to further adjustment by Peoples in order for the merger to qualify as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock that you receive in exchange for your shares of Renasant common stock.

If you wish, you may dissent from the merger agreement and obtain a cash payment for the fair value of your shares. To exercise dissenters' rights, you must not vote in favor of the adoption and approval of the merger agreement, and you must strictly comply with all of the applicable requirements of Tennessee law summarized in the accompanying proxy statement/prospectus under the heading "The Merger — Dissenters' Rights."

Peoples common stock is listed on the American Stock Exchange under the symbol "PHC". On April 5, 2004, the closing price of a share of Peoples common stock was \$35.40.

Your vote is very important. The Renasant board of directors has unanimously determined that the merger agreement, the related plan of merger and the merger are in the best interests of Renasant and its stockholders and unanimously recommends that you vote "FOR" the adoption and approval of the merger agreement, the related plan of merger and the merger. The merger cannot be completed unless the holders of a majority of the outstanding shares of Renasant common stock vote in favor of the adoption and approval of the merger agreement, the related plan of merger and the merger.

Whether or not you plan to attend the special meeting, please take the time to vote by promptly submitting the enclosed form of proxy. If you sign, date and mail your proxy card without indicating how you want to vote, your Renasant shares will be counted as a vote in favor of adoption and approval of the merger agreement, the related plan of merger and the merger. If you do not submit your proxy, the effect will be a vote against the merger agreement, the related plan of merger and the merger. If you are the record owner of your shares of Renasant common stock, returning your proxy does not deprive you of your right to attend the meeting and to vote your shares in person if you should decide to do so. Granting your proxy will impact your dissenters' rights as discussed in the accompanying proxy statement/prospectus.

The proposed merger is discussed in detail in the accompanying proxy statement/prospectus. We encourage you to read this entire document carefully. You can also obtain more information about Peoples in documents that it has filed with the Securities and Exchange Commission.

On behalf of your board of directors, we encourage you to vote "FOR" the adoption and approval of the merger agreement, the related plan of merger and the merger.

Chairman	Vice Chairman and Chief Executive Officer
/s/ JACK C. JOHNSON	/s/ FRANK J. CIANCIOLA

Table of Contents This proxy statement/prospectus is dated _______, 2004, and it is first being mailed to Renasant stockholders, along with the enclosed form of proxy card, on or about _______, 2004.

YOU SHOULD READ "RISK FACTORS" BEGINNING ON PAGE 17 FOR A DESCRIPTION OF THE FACTORS THAT MAY AFFECT THE VALUE OF THE PEOPLES COMMON STOCK TO BE ISSUED IN THE MERGER AND OTHER RISK FACTORS THAT SHOULD BE CONSIDERED WITH RESPECT TO THE MERGER.

REFERENCES TO ADDITIONAL INFORMATION

This proxy statement/prospectus incorporates important business and financial information about Peoples from documents that Peoples has filed with the Securities and Exchange Commission and that have not been included in or delivered with this proxy statement/prospectus. This information is available to you without charge upon your written or oral request. You can obtain documents incorporated by reference in this proxy statement/prospectus, other than exhibits to those documents, by requesting them in writing or by telephone from Peoples at the following address:

The Peoples Holding Company 209 Troy Street Tupelo, Mississippi 38802 Attention: James W. Gray Telephone: (662) 680-1001

IF YOU WOULD LIKE TO REQUEST DOCUMENTS, PLEASE DO SO PRIOR TO ______, 2004, IN ORDER TO RECEIVE THEM BEFORE THE SPECIAL MEETING.

See "Where You Can Find More Information" for more information about the documents referred to in this proxy statement/prospectus.

[Renasant Bancshares, Inc. letterhead] Renasant Bancshares, Inc. Notice of Special Meeting . 2004

Notice of Special Meeting, 2004
To the Stockholders of Renasant Bancshares, Inc.:
A special meeting of stockholders of Renasant Bancshares, Inc. ("Renasant") will be held at Renasant's main office, 2177 Germantown Road South, Germantown, Tennessee 38138, on, 2004 at 5:30 p.m., local time, and at any adjournments or postponements thereof, to consider and act upon the following matters:
• To consider and vote upon a proposal to approve and adopt (a) the Agreement and Plan of Merger dated as of February 17, 2004, by and among Renasant Bancshares, Inc., The Peoples Holding Company ("Peoples") and Peoples Merger Corporation, a wholly owned subsidiary of Peoples, pursuant to which Renasant will, upon satisfaction of certain conditions, merge with and into Peoples Merger Corporation, with Peoples Merger Corporation surviving the merger as a wholly owned subsidiary of Peoples, (b) the related plan of merger contemplated by the Plan of Merger and (c) the merger. As a result of the merger, you, as a holder of Renasant common stock, will have the right to receive for each share of your Renasant common stock either (i) \$36.37 in cash, (ii) 1.117015 shares of Peoples common stock or (iii) a combination consisting of cash and shares of Peoples common stock.
 Any other business properly brought before the special meeting or any adjournment or postponement thereof.
Regardless of your election, however, elections will be limited by the requirements that not less than 50% or more than 55% of the aggregate shares of Renasant common stock owned by Renasant stockholders be exchanged for Peoples common stock and not less than 45% or more than 50% of the aggregate shares of Renasant common stock owned by Renasant stockholders be exchanged for cash. Your election is also subject to further adjustment by Peoples in order for the merger to qualify as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock that you receive in exchange for your Renasant common stock.
If you wish, you may dissent from the merger agreement and obtain a cash payment for the fair value of your shares. To exercise dissenters' rights, you must not vote in favor of the adoption and approval of the merger agreement, and you must strictly comply with all of the applicable requirements of Tennessee law summarized under the heading "The Merger — Dissenters' Rights." The fair value of your shares, as determined by a court, may be more or less than the consideration to be paid in the merger.
The Renasant board of directors has fixed the close of business on, 2004 as the record date for the determination of stockholders entitled to notice of, and to vote at, the special meeting and any adjournments or postponements of the special meeting. Therefore, only stockholders of record on, 2004 are entitled to notice of, and to vote at, the special meeting. A list of stockholders entitled to vote will be available at Renasant's offices through the date of the special meeting as well as at the special meeting for examination by any stockholder, his agent or his attorney.
The accompanying proxy statement/prospectus describes the terms and conditions of the merger agreement and includes a complete text of the merger agreement and the related plan of merger as Annex A-1 and Annex A-2, respectively. We urge you to read the enclosed materials carefully for a complete description of the merger agreement, the plan of merger, and the merger. The accompanying proxy statement/prospectus forms a part of this notice.
Your vote is very important. The merger agreement, the related plan of merger and the merger must be adopted and approved by the holders of a majority of the outstanding shares of Renasant common stock. Even if you plan to attend the special meeting, we urge you to submit a valid proxy promptly so that your shares will be voted.
Your board of directors unanimously recommends that you vote "FOR" the adoption and approval of the merger agreement, the related plan of merger and the merger.
By Order of the Board of Directors
/s/ MICHAEL J. MCCARVER
- Its Secretary
Memphis, Tennessee

TABLE OF CONTENTS

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS	1
QUESTIONS AND ANSWERS ABOUT THE MERGER	2
SUMMARY	8
The Companies	8
The Merger	8
What You Will Receive in the Merger	9
The Special Meeting	11
Vote of Management-Owned Shares	11
Renasant's Reasons for the Merger; Recommendation of the Renasant Board	11
Conditions to the Merger	11
Certain Covenants and Agreements	13
Termination of the Merger Agreement	14
<u>Termination Fees</u>	15
Interests of Certain Persons in the Merger	16
<u>Dissenters' Rights</u>	17
Tax Consequences of the Merger	17
Regulatory and Third Party Approvals	17
RISK FACTORS	17
SELECTED HISTORICAL FINANCIAL DATA OF PEOPLES	18
SELECTED HISTORICAL FINANCIAL DATA OF RENASANT	20
UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL DATA OF PEOPLES	21
COMPARATIVE PER SHARE DATA	23
COMPARATIVE STOCK PRICES AND DIVIDENDS	24
THE SPECIAL MEETING	24
Purpose, Time and Place	24
Record Date; Voting Power	24
Quorum	24
Votes Required	25
Share Ownership of Management and Certain Stockholders	25
Voting of Proxies	25
Revocability of Proxies	26
Solicitation of Proxies	26
<u>Directors and Officers of Surviving Corporation</u>	26
THE MERGER	26
<u>General</u>	26
Background of the Merger	27
Renasant's Reasons for the Merger; Recommendation of the Board	29
<u>Peoples' Reasons for the Merger</u>	30
Opinion of Alex Sheshunoff & Co. Investment Banking, LP	30
Material United States Federal Income Tax Consequences	36
Accounting Treatment	38
Regulatory and Third-Party Approvals	38
Dissenters' Rights	40
Interests of Certain Persons in the Merger	43
Restrictions on Resales by Affiliates	45
THE MERGER AGREEMENT	45
General	45
Form of the Merger	45
Timing of Closing	45
Merger Consideration	46
Election and Election Procedures	46
Redesignation Procedures Redesignation Procedures	47
Adjustment Procedures	48
Aujustinent Frocedures	48

Procedures for Exchanging Renasant Common Stock Certificates	49
Dissenting Shares	50
Assumption of Renasant Stock Options	50
Treatment of Renasant Warrants	50
Representations and Warranties	51
Certain Covenants and Agreements	55
Conditions to the Completion of the Merger	61
Termination of the Merger Agreement	63
<u>Termination Fee</u>	64
COMPARISON OF RIGHTS OF STOCKHOLDERS OF RENASANT AND PEOPLES	65
BUSINESS OF RENASANT BANCSHARES, INC.	77
Renasant Bancshares	77
Renasant Bank	77
Market Area	79
<u>Competition</u>	79
<u>Facilities</u>	79
<u>Employees</u>	80
<u>Legal Proceedings</u>	80
<u>Liability and Asset Management</u>	80
Supervision and Regulation	80
Effects of Governmental Policies	86
OWNERSHIP OF RENASANT COMMON STOCK BY CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	87
<u>EXPERTS</u>	88
<u>LEGAL MATTERS</u>	88
WHERE YOU CAN FIND MORE INFORMATION	89
Annex A-1 Agreement and Plan of Merger by and among The Peoples Holding Company, Peoples Merger Corporation and Renasant Bar	ncshares, Inc. dated

as of February 17, 2004

Plan of Merger by and among The Peoples Holding Company, Peoples Merger Corporation, and Renasant Bancshares, Inc. Annex A-2

Form of Lock-Up and Non-Competition Agreement Form of Lock-Up Agreement Annex B-1

Annex B-2

Annex C Opinion of Alex Sheshunoff & Co. Investment Banking, LP Chapter 23 of the Tennessee Business Corporation Act Annex D

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS

This proxy statement/prospectus and the documents that are made part of this proxy statement/prospectus by reference to other documents filed with the Securities and Exchange Commission include various forward-looking statements about The Peoples Holding Company ("Peoples") and Renasant Bancshares, Inc. ("Renasant") that are subject to risks and uncertainties. Forward-looking statements include information concerning future financial performance, business strategy, projected plans and objectives of Peoples and Renasant.

Statements preceded by, followed by or that otherwise include the words "believes," "expects," "anticipates," "intends," "estimates," "plans," "may increase," "may fluctuate," "will likely result," and similar expressions, or future or conditional verbs such as "will," "should," "would," and "could" are generally forward-looking in nature and not historical facts. You should understand that the following important factors, in addition to those discussed elsewhere in this proxy statement/prospectus and in the documents which are incorporated by reference into this proxy statement/prospectus, could affect the future results of the combined company following the merger, and could cause results to differ materially from those expressed in such forward-looking statements:

- the effect of economic conditions and interest rates on a national, regional or international basis;
- the performance of Peoples' businesses following the merger;
- · the timing of the implementation of changes in operations to achieve enhanced earnings or effect cost savings;
- the ability of Peoples and Renasant to successfully integrate their operations, the compatibility of the operating systems of the combining companies, and the degree to which existing administrative and back-office functions and costs of Peoples and Renasant are complementary or redundant;
- the ability to satisfy all conditions precedent to the merger (including stockholder and various regulatory approvals);
- competitive pressures in the consumer finance, commercial finance, insurance, financial services, asset management, retail banking, mortgage lending and auto lending industries;
- the financial resources of, and products available to, competitors;
- changes in laws and regulations, including changes in accounting standards;
- · changes in policy by regulatory agencies;
- · changes in the securities and foreign exchange markets; and
- opportunities that may be presented to and pursued by the combined company following the merger.

Management of each of Peoples and Renasant believes the forward-looking statements about its company are reasonable. However, you should not place undue reliance on them. Any forward-looking

statements in the proxy statement/prospectus are not guarantees of future performance. They involve risks, uncertainties and assumptions, and actual results, developments and business decisions may differ from those contemplated by those forward-looking statements. Many of the factors that will determine these results are beyond Peoples' and Renasant's ability to control or predict. Peoples and Renasant disclaim any duty to update any forward-looking statements, all of which are expressly qualified by the statements in this section.

QUESTIONS AND ANSWERS ABOUT THE MERGER

Q: What is the proposed transaction for which I am being asked to vote?

A: You are being asked to vote to adopt and approve an agreement and plan of merger among Peoples, Peoples Merger Corporation and Renasant, a related plan of merger and the merger contemplated thereby. In this proxy statement/prospectus, we refer to the agreement and plan of merger and the related plan of merger as the "merger agreement". In the merger, Renasant will be merged with and into Peoples Merger Corporation, a wholly owned subsidiary of Peoples. After the merger, Peoples Merger Corporation will be the surviving corporation and will remain a wholly owned subsidiary of Peoples.

Q: Who is Peoples?

A: The Peoples Holding Company is a Mississippi corporation incorporated in 1982 that is the owner of the fourth largest bank headquartered in Mississippi, The Peoples Bank & Trust Company ("Peoples Bank"), a Mississippi-chartered bank incorporated in 1904. Peoples and Peoples Bank are headquartered in Tupelo, Mississippi. Through Peoples Bank, Peoples also owns of The Peoples Insurance Agency. As of December 31, 2003, Peoples had total assets of approximately \$1.4 billion, deposits of approximately \$1.1 billion and total shareholders' equity of approximately \$138 million. Peoples operates 44 community bank, insurance and financial services offices in 27 cities throughout north and north central Mississippi. Peoples Bank's deposits are insured by the Federal Deposit Insurance Corporation.

Q: What will I receive in exchange for my Renasant common stock in the merger?

A: In the merger, each share of Renasant common stock, par value \$1.00 per share, will be converted into the right to receive either (i) \$36.37 in cash, (ii) 1.117015 shares of Peoples common stock, \$5.00 par value per share, or (iii) a combination of cash for 45% of your shares of Renasant common stock and Peoples common stock for 55% of your shares of Renasant common stock.

Q: Can I elect the type of consideration I will receive in the merger?

A: Yes. Subject to the redesignation and adjustment procedures described in this proxy statement/prospectus, you may elect to receive all cash, all shares of Peoples common stock or a combination of cash and Peoples common stock in exchange for your shares of Renasant common stock.

Under the merger agreement, the aggregate number of shares of Renasant common stock to be converted into the right to receive cash shall not be less than 45% or more than 50% of the total number of shares of Renasant common stock outstanding immediately prior to the closing date of the merger (excluding shares owned by Renasant, Peoples or any subsidiary of Renasant or Peoples (other than in a fiduciary capacity)). The aggregate number of shares of Renasant common stock to be converted into the right to receive shares of Peoples common stock shall not be less than 50% or more than 55% of the total number of shares of Renasant common stock outstanding immediately prior to the closing date of the merger (excluding shares owned by Renasant, Peoples or any subsidiary of Renasant or Peoples (other than in a fiduciary capacity)).

Regardless of your election, however, elections will be limited by two requirements:

- not less than 50% or more than 55% of the aggregate shares of Renasant common stock owned by Renasant stockholders must be exchanged for Peoples common stock; and
- not less than 45% or more than 50% of the aggregate shares of Renasant common stock owned by Renasant stockholders must be exchanged for cash.

Your election is also subject to further adjustment by Peoples in order for the merger to qualify as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock that you receive in exchange for your Renasant common stock.

Q: If I elect to receive Peoples common stock in the merger, how many shares will I receive?

A: Subject to the redesignation and adjustment procedures described in this proxy statement/prospectus, if you elect to receive Peoples common stock for all or a portion of your Renasant common stock, you will receive 1.117015 shares of Peoples common stock for each share of Renasant common stock that you own.

You will not receive a fractional share of Peoples common stock. Instead, you will be paid cash in an amount equal to the fraction of a share of Peoples common stock otherwise issuable multiplied by the average closing price as reported by the American Stock Exchange of one share of Peoples common stock for the ten trading days immediately preceding the last trading day immediately prior to the closing date of the merger (the closing date is described in more detail in this proxy/prospectus).

For instance, if you own 100 shares of Renasant common stock and the ten-day average closing price of Peoples common stock is \$32.00 per share, a Renasant stockholder who elects to receive Peoples common stock in exchange for all 100 shares of Renasant common stock would receive 111 shares of Peoples common stock, plus \$22.45 in cash instead of a fractional share.

Q: How do I elect the form of consideration I prefer to receive?

A: A form of election is being mailed to you concurrently with the mailing of this proxy statement/prospectus. If your shares of Renasant common stock are registered in your own name, complete and sign the form of election and send it to Registrar and Transfer Company, 10 Commerce Drive, Cranford, New Jersey 07016-3572, the exchange agent for the merger. If your shares of Renasant common stock are held in the name of your nominee or other representative, such as the trustee of a trust of which you are the beneficiary, you must have such nominee or other representative submit the form of election on your behalf.

Q: Is there a deadline for making an election?

A: Yes. Your completed election form must be received by the exchange agent not later than 5:00 p.m. eastern time on the third business day immediately prior to the closing date of the merger.

Q: What if I do not send an election form, it is not received before the deadline or I improperly complete or sign my election form?

A: If the exchange agent does not receive from you a properly completed and signed election form before the deadline, then it will be assumed that you have elected to receive a combination of cash for 45% of your shares of Renasant common stock and Peoples common stock for the remaining 55% of your shares of Renasant common stock.

Q: When should I send in my stock certificate?

A: After the closing date of the merger, the exchange agent will deliver the necessary stock certificate transmittal materials to you. You should complete and sign the transmittal letter and deliver it, together with your stock certificate or certificates (and any other documents required by the exchange agent or Peoples), to the exchange agent at the address listed in the transmittal materials.

The exchange agent, upon receipt of your stock certificates and other documents, will deliver to you the cash, stock or combination thereof which you have elected to receive, as modified by any redesignation or adjustment procedures described in this proxy statement/prospectus. This delivery is subject to the payment of any transfer taxes that may arise in connection with the transfer of your shares of Renasant common stock.

Q: What happens if the number of shares elected to be converted into cash exceeds 50% of the outstanding shares of Renasant common stock or if the number of shares elected to be converted into shares of Peoples common stock exceeds 55% of the outstanding shares of Renasant common stock?

A: If the aggregate number of shares elected to be converted into cash exceeds 50% of the outstanding shares of Renasant common stock, then shares of Renasant common stock to be converted into cash will be redesignated on a pro rata basis into shares to be converted into shares of Peoples common stock so that the total number of Renasant shares to be converted into cash does not exceed 50% of the outstanding shares of Renasant common stock.

If the aggregate number of shares elected to be converted into shares of Peoples common stock exceeds 55% of the outstanding shares of Renasant common stock, then shares of Renasant common stock to be converted into shares of Peoples common stock will be redesignated on a pro rata basis into shares to be converted into cash so that the total number of Renasant shares to be converted into shares of Peoples common stock does not exceed 55% of the outstanding shares of Renasant common stock.

Holders of shares of Renasant common stock who elect to receive a combination of cash for 45% of their Renasant common stock and shares of Peoples common stock for 55% of their Renasant common stock will not be subject to these redesignation procedures. Also, a holder who has elected to receive cash for all of his or her shares of Renasant common stock and would receive less than 10 shares of Peoples common stock if his or her shares were redesignated is not subject to the redesignation procedures. Both of these types of holders, however, are subject to the adjustment procedures described immediately below.

Q: Can my election be affected in any other way?

A: Yes. Renasant and Peoples intend for the merger to qualify as a tax-free reorganization for federal income tax purposes with respect to shares of Peoples common stock received in exchange for shares of Renasant common stock. In order to satisfy the continuity of interest requirement for tax-free reorganizations, the merger agreement requires that the aggregate value of the Peoples common stock delivered to holders of Renasant common stock equal at least 40% of the total value of Renasant.

If Peoples determines that the value of its common stock to be delivered to holders of Renasant common stock does not satisfy this 40% requirement, then Peoples will take two actions. Peoples will reduce the amount of cash into which each share of Renasant common stock is converted and Peoples will

increase the number of shares of Peoples common stock into which each share of Renasant common stock is converted. The reduction in the cash payable to shareholders of Renasant will equal the increase in the value of the Peoples common stock that will be delivered to the shareholders of Renasant. This adjustment is mandatory unless the adjustment would cause Peoples to issue more than 85,000 additional shares of its common stock. If the adjustment would cause Peoples to issue more than 85,000 additional shares, then either Peoples or Renasant may terminate the merger.

Q: What are the United States federal income tax consequences of the merger to Renasant stockholders?

A: The tax consequences of the transaction to you will depend on your particular facts and circumstances and the form of merger consideration you receive. You should consult your tax advisor for a full understanding of the tax consequences of the merger.

Assuming that the merger is completed as currently contemplated, you will not recognize any gain or loss for United States federal income tax purposes on any of your Renasant shares exchanged for Peoples shares in the merger, except with respect to cash received in lieu of a fractional Peoples share. You may recognize gain or loss if you exchange your Renasant shares solely for cash in the merger. You may recognize gain, but not loss, if you exchange your Renasant shares for a combination of Peoples shares and cash, but not in excess of the cash you receive in the merger.

Q: Am I entitled to dissenters' rights?

A: Yes. If you wish, you may dissent from the merger agreement and obtain a cash payment for the fair value of your shares. To exercise dissenters' rights, you must not vote in favor of the adoption and approval of the merger agreement, and you must strictly comply with all of the applicable requirements of Tennessee law summarized under the heading "The Merger — Dissenters' Rights." The fair value of your shares may be more or less than the consideration to be paid in the merger.

We have included a copy of Chapter 23 of the Tennessee Business Corporation Act (the "TBCA") as Annex D to this proxy statement/prospectus.

Q: When and where is the special meeting?

A: The Renasant special meeting is scheduled to take place at Renasant's main office, 2177 Germantown Road South, Germantown, Tennessee 38138 on _______ 2004 at 5:30 p.m. local time.

Q: Who can vote on the merger?

A: Holders of record of Renasant common stock at the close of business on ______, 2004 can vote at the special meeting. On that date, 1,429,697 shares were outstanding and entitled to vote.

Q: What vote is required for approval?

A: The merger agreement must be adopted and approved by a majority of the outstanding shares of Renasant common stock. Therefore, if you abstain or fail to vote, it will be the same as voting against the merger agreement.

If you hold your shares of Renasant in a broker's name (sometimes called "street name" or "nominee name"), then you must provide voting instructions to your broker. If you do not provide instructions to the broker, your shares will not be voted on any matter on which the broker does not have

discretionary authority to vote, which includes the vote on the merger. A vote that is not cast for this reason is called a "broker non-vote." Broker non-votes will be treated as shares present for the purpose of determining whether a quorum is present at the meeting. For purposes of the vote on the merger agreement, a broker non-vote has the same effect as a vote AGAINST the merger agreement. For purposes of the vote on other matters properly brought at the special meeting, broker non-votes will not be counted as a vote FOR or AGAINST such matters or as an abstention on such matters.

Q: When do you expect the merger to be completed?

A: We are working to complete the merger as promptly as possible. We expect to complete the merger promptly after we receive both Renasant stockholder approval at the special meeting and all necessary regulatory approvals. We currently expect this to occur during the third quarter of 2004. Fulfilling some of the conditions to closing the merger, such as receiving certain governmental clearances or regulatory approvals is not entirely within our control.

Q: What do I need to do now?

A: After carefully reading and considering the information contained in this proxy statement/prospectus, please complete and mail your proxy card as soon as possible so that your shares may be voted at the special meeting. Your proxy card will instruct the persons named on the proxy card to vote your shares at the special meeting as you direct. If you sign and send in your proxy card and do not indicate how you want to vote, your proxy will be voted FOR the adoption and approval of the merger agreement and the merger. If you do not vote or if you abstain, the effect will be a vote against the merger agreement and the merger. Your vote is very important.

You should also complete the form of election accompanying this proxy statement/prospectus and submit it to Registrar and Transfer Company, 10 Commerce Drive, Cranford, New Jersey 07016-3572, the exchange agent for the merger.

Q: May I change my vote after I have mailed my signed proxy card?

A: You may change your vote at any time before your proxy is voted at the special meeting. You can do this in one of three ways:

- first, you can send a written notice stating that you want to revoke your proxy;
- second, you can complete and submit a new proxy card; or
- third, if you are the record owner of your shares of Renasant common stock, you can attend the Renasant special meeting and vote in person. Simply attending the meeting, however, will not revoke your proxy; you must vote at the meeting.

If you choose either of the first two methods, you must submit your notice of revocation or your new proxy card to:

Renasant Bancshares, Inc. Attention: Michael J. McCarver, Secretary 2177 Germantown Road South Germantown, Tennessee 38138

Q: If I plan to attend the Renasant special meeting in person, should I still grant my proxy?

A: Yes. Whether or not you plan to attend the special meeting, you should grant your proxy as described above. The failure of a Renasant stockholder to vote in person or by proxy will have the same effect as a vote against the adoption and approval of the merger agreement. The failure to give voting instructions to your broker will have the same effect as a vote against the adoption and approval of the merger agreement.

Q: What does Renasant's board of directors recommend?

A: Renasant's board of directors has unanimously determined that the proposed merger is advisable and in the best interests of Renasant and its stockholders and unanimously recommends that you vote FOR the proposal to adopt and approve the merger agreement.

Q: Who can help answer my questions?

A: If you have any questions about the merger or if you need additional copies of this proxy statement/prospectus or the enclosed proxy card or form of election, you should contact:

Francis J. Cianciola Vice Chairman 2177 Germantown Road South Germantown, Tennessee 38138 (901) 312-4000

SUMMARY

This summary highlights selected information from this proxy statement/prospectus and may not contain all the information that is important to you. For a more complete understanding of the merger and for a more complete description of the legal terms of the merger and the merger agreement, you should read this entire document carefully, as well as the additional documents to which we refer you. See "Where You Can Find More Information."

The Companies

The Peoples Holding Company 209 Troy Street Tupelo, Mississippi 38802 (662) 680-1001

Peoples is a Mississippi corporation incorporated in 1982 that is the owner of the fourth largest bank headquartered in Mississippi, The Peoples Bank & Trust Company ("Peoples Bank"), a Mississippi-chartered bank incorporated in 1904. Peoples Bank is headquartered in Tupelo, Mississippi. Through Peoples Bank, Peoples is also the owner of The Peoples Insurance Agency. As of December 31, 2003, Peoples had total assets of approximately \$1.4 billion, deposits of approximately \$1.1 billion and total shareholders' equity of approximately \$138 million. Peoples operates 44 community bank, insurance and financial services offices in 27 cities throughout north and north central Mississippi. Peoples Bank's deposits are insured by the Federal Deposit Insurance Corporation.

For financial statements and a discussion of Peoples' recent results of operations, see Peoples' Annual Report on Form 10-K for the year ended December 31, 2003, which is incorporated by reference in this proxy statement/prospectus.

Peoples Merger Corporation is a Tennessee corporation and a wholly owned subsidiary of Peoples. It was incorporated in 2004 for the purpose of effecting the merger between Peoples and Renasant.

Renasant Bancshares, Inc. 2177 Germantown Road South Germantown, Tennessee 38138 (901) 312-4000

Renasant Bancshares ("Renasant") is a Tennessee corporation incorporated in 1998 that is the owner of Renasant Bank. Renasant's activities are conducted primarily through Renasant Bank. Renasant Bank conducts a full-service banking business in its service area, emphasizing the banking needs of individuals and small to medium-sized businesses. As of December 31, 2003, Renasant had total assets of approximately \$225 million, deposits of approximately \$186 million and total shareholders' equity of approximately \$17 million. Renasant Bank draws most of its customer deposits and conducts most of its lending transactions from and within a primary service area in the eastern area of Shelby County and western area of Fayette County, Tennessee, as well as the northern area of DeSoto County, Mississippi. Renasant Bank's deposits are insured by the Federal Deposit Insurance Corporation.

The Merger (pages 26 through 29)

Under the terms of the merger agreement, Renasant will be merged into Peoples Merger Corporation. After the merger, Peoples Merger Corporation will be the surviving corporation and will

continue its corporate existence under Tennessee law as a wholly owned subsidiary of Peoples. The merger agreement and the plan of merger are attached to this document as Annex A-1 and Annex A-2, respectively, and are incorporated in this proxy statement/prospectus by reference. We encourage you to read the merger agreement carefully, as it is the legal document that governs the merger.

What You Will Receive in the Merger (pages 45 through 50)

The merger agreement provides that each share of Renasant common stock, other than treasury shares, shares owned by Peoples or any of the subsidiaries of Peoples or Renasant (other than in a fiduciary capacity) or by any person who has perfected dissenters' rights with respect to shares of Renasant common stock, will be converted on the closing date of the merger into the right to receive the merger consideration. The merger consideration is either:

- · for each share of Renasant common stock, cash in an amount equal to \$36.37, without interest;
- for each share of Renasant common stock, 1.117015 shares of Peoples common stock; or
- cash consideration for 45% of such holder's shares of Renasant common stock and stock consideration for 55% of such holder's shares of Renasant common stock.

Subject to the redesignation and adjustment procedures described below, as a holder of record of shares of Renasant common stock, you may elect to receive all cash, all shares of Peoples common stock or the combination of cash and Peoples common stock described above as consideration in exchange for your shares of Renasant common stock. You will not receive any fractional shares of Peoples common stock if you elect to receive all or a portion of the merger consideration as shares of Peoples common stock. Instead, you will be paid cash in an amount equal to the fraction of a share of Peoples common stock otherwise issuable upon conversion multiplied by the average closing price of one share of Peoples common stock as reported by the American Stock Exchange for the ten trading days immediately preceding the last trading day immediately prior to the closing date of the merger.

A form of election is being mailed to you concurrently with the mailing of this proxy statement/prospectus. If your shares of Renasant common stock are registered in your own name, you must complete and sign the form of election and send it to Registrar and Transfer Company, 10 Commerce Drive, Cranford, New Jersey 07016-3572, the exchange agent for the merger. If your shares of Renasant common stock are held in the name of your nominee or other representative, such as the trustee of a trust of which you are the beneficiary, you must have such nominee or other representative submit the form of election on your behalf. The form of election must be received by the exchange agent not later than 5:00 p.m. eastern time on the third business day immediately prior to the closing date of the merger.

The merger agreement contains redesignation procedures which may affect your election. Under the merger agreement, the number of shares of Renasant common stock to be converted into the right to receive cash must not be less than 45% or more than 50% of the total number of shares of Renasant common stock outstanding immediately prior to the closing date of the merger. Also, the number of shares of Renasant common stock to be converted into the right to receive shares of Peoples common stock must not be less than 50% or more than 55% of the total number of shares of Renasant common stock outstanding immediately prior to the closing date of the merger.

If the number of shares to be converted into the right to receive cash exceeds 50% of the outstanding shares of Renasant common stock, then all shares of Renasant common stock to be converted

into the right to receive cash will be redesignated on a pro rata basis into a combination of shares to be converted into cash and shares to be converted into shares of Peoples common stock. Shares will be redesignated such that the total number of Renasant shares converted into cash does not exceed 50% of the outstanding shares of Renasant common stock. If the number of shares to be converted into shares of Peoples common stock exceeds 55% of the outstanding shares of Renasant common stock, then all shares of Renasant common stock to be converted into shares of Peoples common stock will be redesignated on a pro rata basis for a combination of shares to be converted into cash and shares to be converted into shares of Peoples common stock. Shares will be redesignated such that the total number of Renasant shares exchanged for stock does not exceed 55% of the outstanding shares of Renasant common stock. Holders who chose to receive a combination of cash for 45% of their Renasant common stock and shares of Peoples common stock for 55% of their Renasant common stock are not subject to the redesignation procedures. Also, any holder who elected to receive cash but after the redesignation procedures would receive less than ten shares of Peoples common stock for his or her shares of Renasant common stock is not subject to the redesignation procedures.

The merger agreement also contains certain readjustment procedures which may affect your election. Under the merger agreement, the value of the shares of Peoples common stock to be delivered to the holders of Renasant common stock pursuant to their elections to receive the merger consideration must have an aggregate value of at least 40% of the value of Renasant as a whole. This is required so that the merger qualifies as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock received in exchange for shares of Renasant common stock. If, after the redesignation procedures described immediately above are completed, Peoples determines that the total value of its common stock included as part of the merger consideration does not satisfy the 40% threshold described above, then Peoples will take two actions. Peoples will reduce the amount of cash into which shares of Renasant common stock will be converted, and Peoples will increase the number of shares of Peoples common stock into which each share of Renasant common stock will be converted. The reduction in the cash payable to shareholders of Renasant will equal the increase in the value of the Peoples common stock that will be delivered to the shareholders of Renasant. This adjustment is mandatory unless the adjustment would cause Peoples to issue more than 85,000 additional shares of its common stock and Peoples decides not to issue the additional shares, then either Peoples or Renasant may terminate the merger. All holders of Renasant common stock are subject to these adjustment procedures, except for holders who perfected their dissenters' rights under Chapter 23 of the Tennessee Business Corporations Act (the "TBCA").

If the exchange agent does not receive from you a properly completed election form before the third business day immediately prior to the closing date of the merger, then it will be assumed that you have elected to receive a combination of cash for 45% of your shares of Renasant common stock and Peoples common stock for the remaining 55% of your shares of Renasant common stock.

After the closing date of the merger, the exchange agent will deliver to the former holders of Renasant common stock the necessary stock certificate transmittal materials and instructions for use so that the certificates representing shares of Renasant common stock may be properly surrendered. After receipt of such materials, each former holder of Renasant common stock shall surrender for cancellation the certificate or certificates representing such stock, together with a properly executed and completed letter of transmittal (and any other documents required by the exchange agent or Peoples). Upon receipt of such certificate(s) and other materials, and subject to the payment of any transfer taxes that may arise in connection with the transfer of your shares of Renasant common stock, the exchange agent will deliver to the former holder of Renasant common stock the merger consideration such holder elected to receive.

The Special Meeting

The Renasant special meeting will be held at Renasant's main office, 2177 Germantown Road South, Germantown, Tennessee 38138 on _______, 2004 at 5:30 p.m. local time. At the meeting, the holders of Renasant common stock will be asked to vote upon a proposal to adopt and approve the merger agreement and the merger. The Renasant board of directors has fixed the close of business on _______, 2004 as the record date for the determination of stockholders entitled to notice of, and to vote at, the special meeting. At the record date, approximately 1,429,697 shares of Renasant common stock were issued and outstanding and entitled to vote. Each share of Renasant common stock is entitled to one vote on any matter that may properly come before the meeting. The affirmative vote of a majority of the outstanding shares of Renasant common stock is required to adopt and approve the merger agreement and the merger.

Vote of Management-Owned Shares

As of the record date, the directors and executive officers of Renasant and their respective affiliates collectively owned approximately 21.67% of the outstanding shares of Renasant common stock, including shares subject to options currently exercisable but not exercised. All of the directors and executive officers of Renasant have entered into agreements with Peoples pursuant to which they have agreed to vote all of their shares in favor of the adoption and approval of the merger agreement and the merger. A form of the agreement for non-employee directors is attached as Annex B-1 to this proxy statement/prospectus and a form of the agreement for employee directors is attached as Annex B-2 to this proxy statement/prospectus.

Renasant's Reasons for the Merger; Recommendation of the Renasant Board (pages 29 and 30)

Renasant's board of directors has unanimously approved the merger agreement and the merger. Renasant's board of directors believes that the merger is advisable and in the best interests of Renasant and its stockholders and unanimously recommends that Renasant's stockholders vote for the adoption and approval of the merger agreement and the merger. In reaching its decision, the Renasant board considered a number of factors, which are described in more detail in "The Merger — Renasant's Reasons for the Merger; Recommendation of the Renasant Board" beginning on page 29. The Renasant board of directors did not assign relative weights to the factors described in that section or the other factors considered by it. In addition, the Renasant board did not reach any specific conclusion on each factor considered, but conducted an overall analysis of these factors. Individual members of the Renasant board of directors may have given different weights to different factors.

Conditions to the Merger (pages 61 through 63)

The obligations of both Peoples and Renasant to complete the merger are conditioned on the following conditions being fulfilled:

- receipt of all necessary regulatory governmental consents and approvals required to complete the merger and satisfaction of all conditions required to
 be satisfied prior to the closing date of the merger by the terms and conditions of such consents and approvals, and delivery of all notices, reports and
 other filings required to be delivered prior to the closing date;
- adoption and approval of the merger agreement and the merger by the Renasant stockholders;
- the registration statement, of which this document is a part, having become effective under the Securities Act, and no stop order suspending the
 effectiveness of the registration statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the
 SEC:

- receipt of all consents and approvals of persons other than governmental entities, except those consents which would not reasonably be expected to have a material adverse effect on Renasant or Peoples;
- absence of any governmental or judicial enactment or order prohibiting, restricting or making illegal completion of the merger;
- receipt of an opinion of Phelps Dunbar LLP, Peoples' outside counsel, that the merger will qualify as a tax-free reorganization under Section 368 of the Internal Revenue Code and that the exchange of shares of Renasant common stock for shares of Peoples common stock will not give rise to gain or loss to the holders of Renasant common stock; and
- the shares of Peoples common stock issuable to the holders of shares of Renasant common stock shall have been approved for listing on the American Stock Exchange.

In addition, Peoples' obligation to complete the merger is subject to, among other things:

- Renasant's performance of and compliance with in all material respects all obligations required by the merger agreement;
- the representations and warranties of Renasant in the merger agreement being true and correct, as of the date of the merger agreement and as of the closing date of the merger (except those that relate specifically to another date, which shall be true and correct as of that date), with some exceptions;
- the receipt of all permits, consents, waivers, clearances, approvals and authorizations necessary in connection with the completion of the merger, none of which contain any terms or conditions which would materially impair the value of Renasant to Peoples;
- Renasant stockholders who dissent from the merger shall not hold more than 15% of the outstanding shares of Renasant common stock immediately prior to the merger; and
- two qualified people selected by Renasant from its board of directors and who are acceptable to Peoples having been appointed by Peoples to the board of directors of Peoples as of the closing date.

In addition, Renasant's obligation to complete the merger is subject to, among other things:

- Peoples' and Peoples Merger Corporation's performance of and compliance in all material respects with all obligations required by the merger agreement;
- the representations and warranties of Peoples and Peoples Merger Corporation being true and correct as of the date of the merger agreement and as of the closing date of the merger (except those that relate specifically to another date, which shall be true and correct as of that date), with some exceptions; and

the receipt of all permits, consents, waivers, clearances, approvals and authorizations necessary in connection with the completion of the merger, none
of which adversely affect the merger consideration.

The merger is expected to be completed promptly after Renasant stockholder approval is received at the special meeting, all necessary regulatory approvals are received and other conditions to the closing described above are fulfilled. This is expected to occur during the third quarter of 2004, although fulfilling some of the conditions to closing the merger, such as receiving regulatory approvals, is not within the control of Peoples or Renasant.

Certain Covenants and Agreements (pages 55 through 61)

Renasant has agreed that neither it nor Renasant Bank, nor any person on either's behalf, will solicit or hold discussions with any third party regarding a merger, tender offer, recapitalization, consolidation or any similar transaction, sale or lease or other acquisition or assumption of all or a substantial portion of Renasant's or Renasant Bank's assets, purchase or acquisition of more than 20% of the voting power of Renasant or any similar transaction. Renasant may take the following actions:

- provide information to a third party regarding a proposal to engage in any of the above-described transactions;
- · negotiate and discuss such a transaction with a third party;
- recommend to the stockholders of Renasant the approval of such a transaction with a third party; or
- withdraw a recommendation regarding the merger with Peoples.

Renasant may take these actions only if (i) the board of directors determines in good faith (after consultation with outside legal counsel) that any of the above-described actions are necessary in order for its directors to comply with their fiduciary duties under applicable law and (ii) the board of directors determines in good faith (after consultation with its financial advisor) that the transaction with the third party is likely to be consummated and to result in a transaction more favorable to Renasant stockholders than the merger with Peoples.

Peoples has the right to match or better any acquisition proposal from a third party, and the merger agreement will be amended to reflect any new terms offered by Peoples. If Peoples does match or better such proposal, Renasant must cease, and cause Renasant Bank or its representative to cease, all discussions with the third party.

The merger agreement requires Peoples to provide certain indemnification for a period of 6 years following the closing date of the merger. Peoples must indemnify and hold harmless from liability for acts or omissions occurring at or prior to the closing date of the merger certain current or former directors and officers of Renasant or Renasant Bank to the same extent as such directors or officers would be indemnified under the articles of incorporation or bylaws of Peoples as if they were directors or officers of Peoples. The merger agreement also provides that Peoples shall use reasonable efforts to cause Peoples Merger Corporation or Peoples to obtain for a period of six years after the closing date of the merger policies of directors' and officers' liability insurance covering acts or omissions occurring prior to the closing date of the merger for certain directors and officers of Renasant on terms and in amounts no less favorable than those in effect on the date of the merger agreement. However, neither Peoples nor

Peoples Merger Corporation are required to pay an aggregate premium for such insurance coverage in excess of 200% of the amount for such coverage as currently held by Renasant but in such case shall purchase as much coverage as reasonably practicable for such amount.

Termination of the Merger Agreement (pages 63 and 64)

The merger agreement may be terminated and the merger may be abandoned at any time prior to the closing date of the merger:

- by the mutual written consent of Peoples and Renasant;
- by either Peoples or Renasant if:
 - (i) the closing date of the merger shall not have occurred on or prior to August 31, 2004 (or on or prior to October 31, 2004 where a governmental approval is pending and has not been finally resolved) or (ii) the merger agreement and the merger is not approved by Renasant's shareholders, unless the failure of either (i) or (ii) is caused by the party seeking to terminate to perform or observe its agreements at or before the closing date or the shareholders vote, as the case may be;
 - there has been a breach by the other party of (i) any covenant or undertaking contained in the merger agreement or (ii) any representation or
 warranty of the other party contained in the merger agreement preventing such party from satisfying a condition to closing in the merger
 agreement, and such breach has not been cured within thirty days following delivery of written notice of the breach;
 - any application for regulatory or governmental approval is denied or withdrawn at the request or recommendation of the governmental entity, but termination may not occur earlier than 30 days after such denial or withdrawal, unless within such 30-day period a petition for rehearing or an amended application is filed. A party may terminate 30 or more days after a petition for rehearing or an amended application is denied. No party may terminate when the denial or withdrawal is due to that party's failure to observe or performs its covenants or agreements;
 - any governmental entity shall have issued a final, non-appealable order prohibiting the completion of the merger; or
 - Peoples shall have elected not to make the adjustments necessary in the cash portion and stock portion of the merger consideration so that the
 merger qualifies as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock to be
 exchanged for shares of Renasant common stock;
- · by Peoples if:
 - Renasant's board of directors fails to make, or withdraws, modifies or qualifies, or proposes publicly to withdraw, modify or qualify, the
 recommendation in this proxy statement/prospectus that Renasant's stockholders vote to adopt and approve the merger agreement and the
 merger;

- the special meeting to approve the merger agreement and plan of merger is not called or convened by Renasant;
- Renasant approves or recommends, or publicly proposes to approve or recommend, an acquisition proposal by a third party; or
- Renasant stockholders dissenting from the merger hold more than 15% of the outstanding shares of Renasant common stock immediately prior to the merger.
- by Renasant if:
 - the board of directors of Renasant determines in good faith, after consultation with outside counsel, that the failure to terminate the merger
 agreement and accept an acquisition proposal from a third party would constitute a breach of its fiduciary duties; or
 - (i) Peoples makes the adjustments necessary in the cash portion and stock portion of the merger consideration so that the merger qualifies as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock to be exchanged for shares of Renasant common stock in the merger and (ii) the shares of Peoples' common stock do not meet the trading price target described on page 64 of this proxy statement/prospectus.

Termination Fees (pages 64 and 65)

Renasant must pay to Peoples a termination fee of \$5,000,000 if:

- (i) prior to any event allowing either party to terminate the merger agreement, an acquisition proposal from a third party is publicly announced or otherwise made known to Renasant's senior management, board of directors or stockholders generally and not irrevocably withdrawn more than five business days prior to the special meeting, (ii) the merger agreement is then terminated (y) by either Peoples or Renasant, because Renasant's stockholders failed to approve the merger agreement and the plan of merger or (z) by Peoples, because of a willful breach by Renasant of any covenant, undertaking, representation or warranty contained in the merger agreement, and (iii) the acquisition contained in the acquisition proposal is consummated within 12 months of the termination of the merger agreement;
- Peoples terminates the merger agreement because Renasant either (i) failed to recommend to its stockholders the approval of the merger agreement and the merger, (ii) effected a change in such recommendation, (iii) failed to call or convene the special meeting, or (iv) approved or recommended, or proposed publicly to approve or recommend any acquisition transaction; or
- Renasant terminates the merger agreement because its board of directors determines in good faith after consultation with outside counsel that the
 failure to terminate the merger agreement and accept an acquisition proposal from a third party would constitute a breach of its fiduciary duties under
 applicable laws.

Interests of Certain Persons in the Merger (pages 43 and 44)

In addition to their interests as stockholders, the directors and executive officers of Renasant may have interests in the merger that are different from, or in addition to, your interests. These interests exist because of rights they may have under individual employment agreements, under compensation and benefit plans, including the Renasant stock option plan, and under the merger agreement. These interests include, among other things:

- an employment agreement to be entered into by Peoples, Renasant Bank and Frank J. Cianciola on the closing date, pursuant to which Frank J. Cianciola will serve as President and Chief Executive Officer of Renasant Bank, for a period commencing on the closing date and ending five years after the closing date;
- an employment agreement to be entered into by Peoples, Renasant Bank and Michael J. McCarver on the closing date, pursuant to which Michael J. McCarver will serve as Executive Vice President of Renasant Bank, for a period commencing on the closing date and ending two years after the closing date.
- an employment agreement to be entered into by Peoples, Renasant Bank and Donald E. Russell on the closing date, pursuant to which Donald E. Russell will serve as Executive Vice President of Renasant Bank, for a period commencing on the closing date and ending two years after the closing date.
- the agreement between Renasant and Frank J. Cianciola which provides for a cash payment in the amount of \$500,000 in the event of a change of
 control of Renasant;
- the agreement between Renasant and Jack C. Johnson which provides for a cash payment in the amount of \$250,000 in the event of a change of control of Renasant:
- the agreements in the form attached to this proxy statement/prospectus as Annex B-1 entered into between Peoples and non-employee directors of Renasant, in which Peoples has agreed that for two years after the closing date Peoples will cause such directors to be appointed to the board of directors of Renasant Bank and will pay such directors the fees in accordance with the policy of Renasant Bank on directors' fees in place on the closing date;
- the agreements between Peoples and the directors and officers of Renasant who hold warrants to purchase Renasant common stock, pursuant to which Peoples has offered to purchase such warrants for an amount equal to the number of shares of Renasant common stock specified in such warrant multiplied by \$36.37, less the aggregate exercise price for all shares of Renasant common stock specified in such warrant. The purchase price for the warrants will be paid entirely in cash. Any warrants that are not sold to Peoples will be amended such that each warrant will entitle the holder to purchase shares of Peoples common stock equal to the number of shares of Renasant common stock that the warrant allowed the holder to purchase multiplied by 1.117015, at an exercise price equal to the exercise price specified in such warrant divided by 1.117015; and
- · Peoples' agreement to indemnify and hold harmless certain present and former directors and officers of Renasant and Renasant Bank.

The members of the Renasant board of directors knew of these additional interests, and considered them when they approved the merger agreement.

Dissenters' Rights (pages 40 through 43)

Under Tennessee law, Renasant stockholders have the right to dissent from the merger agreement and obtain payment for the fair value of their shares of Renasant common stock in connection with the merger. To exercise dissenters' rights, a holder of shares must not vote in favor of the adoption and approval of the merger agreement and the merger, and must otherwise strictly comply with all of the applicable requirements of Tennessee law. Dissenters' rights are described under the heading "The Merger – Dissenters' Rights." The relevant provisions of the applicable requirements of Tennessee law are included as Annex D to this proxy statement/prospectus.

Tax Consequences of the Merger (pages 36 through 38)

Assuming that the merger is completed as currently contemplated, a holder of Renasant common stock will not recognize any gain or loss for United States federal income tax purposes on any of the Renasant shares exchanged for Peoples shares in the merger, except with respect to cash received in lieu of a fractional Peoples share. A holder of Renasant common stock may recognize gain or loss if Renasant shares are exchanged solely for cash in the merger. Further, a holder of Renasant common stock may recognize gain, but not loss, if the Renasant shares are exchanged for a combination of Peoples shares and cash, but not in excess of the cash received in the merger.

Regulatory and Third-Party Approvals (pages 38 and 39)

Under the merger agreement, Peoples and Renasant have agreed to use their best efforts to obtain all necessary actions or nonactions, waivers, consents and approvals from any governmental authority necessary to complete and make effective the merger and other transactions contemplated by the merger agreement. The required regulatory approvals include approval from the Federal Reserve Board and the Tennessee Commissioner of Financial Institutions. All applications and notices have been filed, or are in the process of being filed.

RISK FACTORS

In addition to the other information included in this proxy statement/prospectus, you should carefully consider the matters described below in determining whether to adopt and approve the merger agreement and the merger.

Redesignation/Adjustment Risk. You may receive a form of consideration different from the form of consideration you elect. The consideration to be received by Renasant stockholders in the merger is subject to the requirement that not less than 50% or more than 55% of the shares of Renasant common stock be converted into the right to receive Peoples common stock and that not less than 45% or more than 50% of the shares of Renasant common stock be converted into the right to receive cash. The merger agreement contains redesignation procedures to achieve this desired result. If you elect to receive all cash and the available cash is oversubscribed, then a portion of your merger consideration will be paid in Peoples common stock. If you elect to receive all stock and the available stock is oversubscribed, then a portion of the merger consideration you receive will be paid in cash. Therefore, you may not receive exactly the form of consideration that you elect.

Your election is also subject to further adjustments by Peoples in order for the merger to qualify as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common

stock received in exchange for shares of Renasant common stock. These adjustment procedures allow Peoples to alter the amount of cash or the number of shares of Peoples common stock to be received for each share of Renasant common stock. On account of these procedures, you may receive less cash or more Peoples common stock than you elected to receive.

Stock Price Fluctuation Risk. Upon the closing of the merger, each share of Renasant common stock you own will automatically be converted into the right to receive either 1.117015 shares of Peoples common stock, \$36.37 in cash or a combination of both Peoples common stock and cash. Because the market price of Peoples common stock may fluctuate, you cannot be sure of the market value of the Peoples common stock that you elect to receive in the merger. Stock price changes may result from a variety of factors, including general market and economic conditions, changes in Peoples' businesses, operations and prospects, and regulatory considerations. Many of these factors are beyond Peoples' control. In addition, there will be a time period between the completion of the merger and the time when Renasant stockholders receiving stock consideration actually receive certificates evidencing Peoples common stock. Until stock certificates are received, Renasant stockholders will not be able to sell their Peoples shares in the open market and, thus, will not be able to avoid losses resulting from any decline in the trading price of Peoples common stock during this period.

Trading Volume Risk. For the three months ended March 31, 2004, the average daily trading volume for Peoples common stock was 4,090 shares per day. Because Peoples common stock has a low daily trading volume, significant purchases or sales of Peoples common stock may cause volatility in the price of Peoples common stock.

Interested Party Risk. You should be aware that the directors and some executive officers of Renasant have interests in the merger that are different from, or in addition to, the interests of stockholders generally. For example, on the closing date, the chief executive officer, the chief financial officer and an executive vice president of Renasant will enter into agreements that provide for their continued employment following the merger. Certain other interests of Renasant's directors and executive officers may cause some of these persons to view the proposed transaction differently than you view it. For a discussion of these interests, see "The Merger-Interests of Certain Persons in the Merger." Despite these additional or different interests, the directors of Renasant believe that the merger is in the best interests of Renasant and its stockholders.

SELECTED HISTORICAL FINANCIAL DATA OF PEOPLES

The following selected financial data for each of the five years in the period ended December 31, 2003 have been derived from Peoples' audited consolidated financial statements. This data should be read together with the audited consolidated financial statements of Peoples, including the notes to such financial statements, incorporated into this proxy statement/prospectus by reference and with the "Management's Discussion and Analysis of Financial Condition and Results of Operations" of Peoples contained in, or incorporated in, the annual reports and other information that Peoples has filed with the Securities and Exchange Commission.

PEOPLES SELECTED CONSOLIDATED HISTORICAL FINANCIAL DATA

(Unaudited)

(In Thousands, Except Share Data)

	2003	2002	2002 2001		2001 2000		1999
Year ended December 31							
Interest income	\$ 70,810	\$ 78,418	\$ 87,766	\$ 89,434	\$ 83,500		
Interest expense	21,777	26,525	40,922	44,132	37,342		
Provision for loan losses	2,713	4,350	4,790	6,373	3,192		
Noninterest income	31,223	27,442	24,389	18,529	19,476		
Noninterest expense	52,523	50,496	46,747	42,474	41,480		
Income before income taxes	25,020	24,489	19,696	14,984	20,962		
Income taxes	6,839	6,819	5,109	3,800	6,182		
				-			
Income before cumulative effect of							
accounting change	18,181	17,670	14,587	11,184	14,780		
Cumulative effect of accounting change		(1,300)					
Net income	\$ 18,181	\$ 16,370	\$ 14,587	\$ 11,184	\$ 14,780		
Net Income Per Common Share-Basic	\$ 2.20	\$ 1.95	\$ 1.66	\$ 1.22	\$ 1.59		
Net Income Per Common Share-							
Diluted	\$ 2.19	\$ 1.94	\$ 1.66	\$ 1.22	\$ 1.59		
Book value at December 31	\$ 16.79	\$ 15.88	\$ 14.44	\$ 13.39	\$ 12.47		
Closing Price on the AMEX at December							
31	\$ 33.00	27.17	24.67	12.00	19.25		
Cash dividends declared and paid	\$ 0.75	0.69	0.64	0.59	0.56		
At December 31							
Loans, net of unearned income	\$ 862,652	\$ 859,684	\$ 818,036	\$ 812,701	\$ 798,083		
Securities	414,270	344,781	277,293	278,574	266,744		
Assets	1,415,214	1,344,512	1,254,727	1,211,940	1,162,959		
Deposits	1,133,931	1,099,048	1,063,055	1,046,605	978,958		
Long term debt	95,918	64,696	32,254	15,087	17,424		
Shareholders' equity	137,625	132,778	123,582	121,661	116,089		

PEOPLES SELECTED CONSOLIDATED HISTORICAL FINANCIAL DATA (continued)

(In Thousands, Except Share Data)

	2003	2002	2001	2000	1999
Selected Ratios					
Return on average:					
Total assets	1.33%	1.25%	1.18%	.93%	1.29%
Shareholders' equity	13.41%	12.85%	11.70%	9.49%	13.19%
Before cumulative effect of accounting change, return on average:					
Total assets	1.33%	1.35%	1.18%	.93%	1.29%
Shareholders' equity	13.41%	13.87%	11.70%	9.49%	13.19%
Average shareholders' equity to average assets	9.89%	9.75%	10.07%	9.85%	9.77%
At December 31					
Shareholders' equity to assets	9.72%	9.88%	9.85%	10.04%	9.98%
Allowance for loan losses to total loans, net of unearned Income	1.53%	1.42%	1.39%	1.30%	1.26%
Allowance for loan losses to non-performing loans	181.09%	338.22%	178.63%	147.89%	126.47%
Non-performing loans to total loans, net of unearned income	0.85%	0.42%	0.78%	0.88%	1.00%
Dividend payout	34.25%	35.59%	38.52%	47.76%	35.24%

All per share information listed above has been restated to reflect the three-for-two stock split effected in the form of a share dividend on December 1, 2003, and any other stock splits or stock dividends declared during the five-year period covered by the above table.

SELECTED HISTORICAL FINANCIAL DATA OF RENASANT

The following selected financial data for each of the five years in the period ended December 31, 2003 have been derived from Renasant's audited financial statements.

RENASANT SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA

(Unaudited)

(In Thousands, Except Share Data)

	2003	2002	2001	2000	1999
Year ended December 31					
Interest income	\$ 11,177	\$ 11,134	\$ 10,133	\$ 8,915	\$ 2,263
Interest expense	4,057	4,443	5,486	5,230	1,079
Provision for loan losses	506	802	630	364	740
Noninterest income	992	624	574	317	87
Noninterest expense	5,541	4,875	3,683	3,085	1,976
Income before income taxes	2,065	1,638	908	553	(1,445)
Income taxes	<u>750</u>	594	106		
Net income	\$ 1,315	\$ 1,044	\$ 802	\$ 553	\$ (1,445)
Net Income Per Common Share-Basic	\$ 0.92	\$ 0.73	\$ 0.56	\$ 0.46	\$ (1.36)
Net Income Per Common Share-Diluted	\$ 0.81	\$ 0.66	\$ 0.52	\$ 0.42	\$ (1.22)
Book value at December 31	\$ 12.07	\$ 11.38	\$ 10.37	\$ 9.84	\$ 8.60
At December 31					
Loans, net of unearned income	\$ 173,211	\$ 163,309	\$ 133,198	\$ 88,591	\$59,129
Securities	39,881	40,191	20,172	27,136	19,304
Assets	224,882	208,968	163,357	121,620	82,666
Deposits	185,739	172,967	134,012	101,509	65,373
Long term debt	8,178	3,226	4,092	3,000	4,000
Shareholders' equity	17,267	16,260	14,822	13,970	9,459

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL DATA OF PEOPLES

The following table shows information about Peoples' financial condition and operations, including per share data and financial ratios, after giving effect to the merger. This information is called pro forma information. The table below sets forth the information as if the merger had become effective on December 31, 2003, with respect to financial condition, and January 1, 2003, with respect to operations data. The pro forma data in the table assumes that the merger is accounted for as an acquisition by Peoples of Renasant using the purchase method of accounting. See "The Merger – Accounting Treatment" on page 38 of this proxy statement/prospectus. Under this method of accounting, the assets and liabilities of Renasant will be recorded by Peoples at their estimated fair values as of the date the merger is completed. These estimates are subject to further adjustment as additional information becomes available and as additional analyses are performed. The pro forma statement of income includes amortization/accretion of fair value adjustments related to the purchase price allocation, but does not include the impact of merger related costs incurred subsequent to the

merger. This table should be read in conjunction with, and is qualified in its entirety by, the historical financial statements, including the notes thereto, of Peoples. See "Where You Can Find More Information" on page 89 of this proxy statement/prospectus in order to obtain copies of such historical financial statements.

The pro forma information, while helpful in illustrating the financial characteristics of the combined company under one set of assumptions, does not reflect the benefits of expected cost savings or opportunities to earn additional revenue, among other factors, that may result as a consequence of the merger, and, accordingly, does not attempt to predict or suggest future results. It also does not necessarily reflect what the historical results of the combined company would have been had Peoples and Renasant been combined during this period. Upon completion of the merger, the operating results of Renasant will be reflected in the consolidated financial statements of Peoples on a prospective basis.

PRO FORMA COMBINED SELECTED FINANCIAL DATA

(Unaudited) (In Thousands, Except Share Data)

	2003
Year ended December 31	
Interest income	\$ 81,461
Interest expense	25,629
Provision for loan losses	3,219
Noninterest income	32,215
Noninterest expense	59,001
Income before income taxes	25,827
Income taxes	7,149
Net income	\$ 18,678
Net Income Per Common Share-Basic	\$ 2.06
Net Income Per Common Share-Diluted	\$ 2.02
Book value at December 31	\$ 18.15
Cash dividends declared and paid	\$ 0.75
At December 31	
Loans, net of unearned income	\$1,038,945
Securities	433,292
Assets	1,657,124
Deposits	1,319,931
Long term debt	104,096
Shareholders' equity	168,334

Pro forma adjustments consisted of recording the impact of: stock and cash issued to effect the merger assuming a 50-50 ratio between such cash and stock, estimated fair value adjustments to the assets and liabilities of Renasant, estimated core deposit intangible of \$4.3 million, estimated goodwill of \$37 million and tax effects of the adjustments.

COMPARATIVE PER SHARE DATA

The following table sets forth for Peoples common stock and Renasant common stock certain historical, pro forma and pro forma-equivalent per share financial information. The pro forma and pro forma-equivalent per share information gives effect to the merger as if the merger had been effective as of December 31, 2003, with respect to book value information, and January 1, 2003, with respect to net income per share data. The pro forma data in the tables assume that the merger is accounted for using the purchase method of accounting and represents a current estimate based on available information of the combined company's results of operations. See "The Merger — Accounting Treatment" on page 38. The pro forma financial adjustments record the assets and liabilities of Renasant at their estimated fair values and are subject to adjustment as additional information becomes available and as additional analyses are performed. This table should be read in conjunction with, and is qualified in its entirety by, the historical financial statements, including the notes thereto, of Peoples. See "Where You Can Find More Information" on page 89 of this proxy statement/prospectus in order to obtain copies of such historical financial statements.

The pro forma information, while helpful in illustrating the financial characteristics of the combined company under one set of assumptions, does not reflect the benefits of expected cost savings or opportunities to earn additional revenue, among other factors, that may result as a consequence of the merger and, accordingly, does not attempt to predict or suggest future results. It also does not necessarily reflect what the historical results of the combined company would have been had Peoples and Renasant been combined during this period. Upon completion of the merger, the operating results of Renasant will be reflected in the consolidated financial statements of Peoples on a prospective basis.

Renasant historically has not paid cash dividends on its common stock.

	Per Share				
Year ended December 31, 2003	Net Income*	Bool	« Value **	Div	ridends
Peoples Historical	\$ 2.19	\$	16.79	\$	0.75
Renasant Historical	\$ 0.81	\$	12.07	\$	0.00
Pro Forma Combined	\$ 2.02	\$	18.15	\$	0.75
Per Equivalent Peoples Share***	\$ 2.26	\$	20.27	\$	0.75

^{*} Net income per share calculated on a fully diluted share basis.

^{**} Book value per share is calculated on the number of shares outstanding as of the end of the period.

^{***} Per Equivalent Peoples Share is pro forma combined multiplied by 1.117015, except for dividends per share which is based on Peoples' prior year dividend payout.

COMPARATIVE STOCK PRICES AND DIVIDENDS

On February 13, 2004, the last trading day prior to the public announcement of the execution of the merger agreement, the last sales price of Peoples common stock was \$31.85 per share. Shares of Renasant common stock do not trade in any established public market, and thus no market value for Renasant shares on the last day prior to the announcement of the merger can be provided.

On April 5, 2004, the most recent practicable trading day prior to the printing of this proxy statement/ prospectus, the last sales price of Peoples common stock was \$35.40 per share. The market price of shares of Peoples common stock is subject to fluctuation. As a result, Renasant stockholders are urged to obtain current market quotations. On April 5, 2004, there were approximately 8,187,000 shares of Peoples common stock outstanding held by approximately 2,600 holders of record.

THE SPECIAL MEETING

Purpose, Time and Place

This proxy statement/prospectus is being furnished to you in connection with the solicitation of proxies by the Renasant board of directors from holders of
Renasant common stock, the only class of Renasant capital stock outstanding, for use at the special meeting to be held at Renasant's main office, 2177
Germantown Road South, Germantown, Tennessee 38138 on, 2004, at 5:30 p.m. local time and at any adjournments or postponements of the special
meeting. The members of the Renasant board of directors are listed under the heading "Security Ownership of Certain Beneficial Owners and Management"
beginning on page 87 of this proxy statement/prospectus. This proxy statement/prospectus and the form of election are first being distributed to Renasant
stockholders on or about, 2004.

At the special meeting, holders of Renasant common stock will be asked to consider and vote upon:

- a proposal to adopt and approve the merger agreement and the merger;
- · such other matters as may properly come before the meeting.

Record Date: Voting Power

The Renasant board of directors has fixed the close of business on _______, 2004 as the record date for determining the holders of Renasant common stock entitled to notice of, and to vote at, the special meeting. Only holders of record of Renasant common stock at the close of business on the record date will be entitled to notice of, and to vote at, the special meeting.

On the record date, 1,429,697 shares of Renasant common stock were issued and outstanding and entitled to vote at the special meeting. Each share of Renasant common stock is entitled to one vote on any matter which may properly come before the special meeting. Votes may be cast at the special meeting in person or by proxy.

Quorum

The presence at the special meeting, either in person or by proxy, of the holders of a majority of the outstanding Renasant common stock entitled to vote is necessary to constitute a quorum in order to transact business at the special meeting. However, if a quorum is not present at the special meeting, it is expected that the meeting will be adjourned or postponed in order to solicit additional proxies.

Votes Required

Approval of the proposal to adopt and approve the merger agreement and the merger will require the affirmative vote of a majority of the outstanding shares of Renasant common stock. Under applicable Tennessee law, in determining whether the proposal to adopt and approve the merger agreement and the merger has received the requisite number of affirmative votes, abstentions and failures to vote will have the same effect as a vote against the proposal.

Renasant stockholders may not cumulate votes on the proposal to adopt and approve the merger agreement and the merger.

Share Ownership of Management and Certain Stockholders

As of the date hereof, Renasant's directors and executive officers and their affiliates may be deemed to be the beneficial owners of approximately 321,000 outstanding shares of Renasant common stock, including shares subject to options not exercised but currently exercisable (collectively representing approximately 21.67% of the voting power of the common stock). The directors and executive officers of Renasant are parties to agreements with Peoples whereby they agreed to vote their shares for adoption and approval of the merger agreement and the merger. Peoples and Renasant have been informed that all of the approximately 21.67% of the outstanding shares of Renasant common stock owned by the directors and executive officers of Renasant and their respective affiliates will be voted in favor of the approval and adoption of the merger agreement and the merger.

Voting of Proxies

Shares represented by properly executed proxies received in time for the special meeting will be voted at the special meeting in the manner specified by such proxies. If your proxy is properly executed but does not contain voting instructions, your proxy will be voted FOR adoption and approval of the merger agreement and the merger. If other matters are properly presented before the special meeting, the persons named in such proxy will have authority to vote in accordance with their judgment on any other such matter, including without limitation, any proposal to adjourn or postpone the meeting or otherwise concerning the conduct of the special meeting. Please note, however, that a proxy that has been designated to vote against the adoption and approval of the merger agreement and the merger will not be voted, either directly or through a separate proposal, to adjourn the meeting to solicit additional votes. It is not expected that any matter other than as described in this proxy statement/prospectus will be brought before the special meeting.

If a shareholder holds shares of Renasant in a broker's name (sometimes called "street name" or "nominee name"), then the shareholder must provide voting instructions to the broker. If the shareholder does not provide instructions to the broker, the shares will not be voted on any matter on which the broker does not have discretionary authority to vote, which includes the vote on the merger. A vote that is not cast for this reason is called a "broker non-vote." Broker non-votes will be treated as shares present for the purpose of determining whether a quorum is present at the meeting. For purposes of the vote on the merger agreement, a broker non-vote is the same as a vote AGAINST the merger agreement. For purposes of the vote on other matters properly brought at the special meeting, broker non-votes will not be counted as a vote FOR or AGAINST such matter or as an abstention on such matter.

Revocability of Proxies

The grant of a proxy on the enclosed proxy card does not preclude a stockholder from voting in person at the special meeting. You may revoke a proxy at any time prior to your proxy being voted at the special meeting by:

- delivering, prior to the special meeting, a written notice of revocation bearing a later date or time than the proxy to the Secretary of Renasant at 2177 Germantown Road South, Germantown, Tennessee 38138;
- submitting another proxy by mail that is later dated and properly signed; or
- · if you are the record owner of shares of Renasant common stock, attending the special meeting and voting such shares in person.

Attendance at the special meeting will not by itself constitute revocation of a proxy. If an adjournment or postponement occurs, it will have no effect on the ability of stockholders of record as of the record date to exercise their voting rights or to revoke any previously delivered proxies.

Solicitation of Proxies

Renasant generally will bear the cost of solicitation of proxies. In addition to solicitation by mail, the directors, officers and employees of Renasant and its subsidiaries may solicit proxies from stockholders by telephone, facsimile or in person.

Directors and Officers of Surviving Corporation

If you approve the merger, which is described below, Renasant will merge into Peoples Merger Corporation, which will be the surviving corporation. E. Robinson McGraw is the sole director of Peoples Merger Corporation, and E. Robinson McGraw and Stuart R. Johnson are the sole executive officers of Peoples Merger Corporation. Neither E. Robinson McGraw nor Stuart R. Johnson are paid any compensation in connection with their respective positions with Peoples Merger Corporation. E. Robinson McGraw is also a director and executive officer of Peoples and Stuart R. Johnson is an executive officer of Peoples, the parent of Peoples Merger Corporation. Information with respect to these individuals' duties, compensation and transactions, if any, with Peoples can be found under the headings "Directors and Executive Officers of the Registrant," "Executive Compensation" and "Certain Relationships and Related Transactions" in the Company's most recent annual report on Form 10-K, dated March 11, 2004 and filed with the Securities and Exchange Commission. Such annual report is incorporated by reference into this proxy statement/prospectus.

THE MERGER

The discussion in this proxy statement/prospectus of the merger and the principal terms of the merger agreement is subject to, and qualified in its entirety by reference to, the merger agreement and the related plan of merger, copies of which are attached to this proxy statement/prospectus as Annex A-1 and A-2, respectively, and are incorporated into this proxy statement/prospectus by reference.

General

On February 13, 2004, the Renasant board of directors unanimously approved the merger agreement and the merger, subject to the resolution of outstanding issues to be resolved by a committee

appointed by the board of directors, which issues have since been resolved. If all of the conditions set forth in the merger agreement are satisfied or waived (to the extent permitted by law) and if the merger is otherwise completed, Renasant will merge into Peoples Merger Corporation, a wholly owned subsidiary of Peoples. After completion of the merger, Peoples Merger Corporation will be the surviving corporation and will continue its corporate existence under Tennessee law as a wholly owned subsidiary of Peoples.

On the closing date of the merger each share of Renasant common stock outstanding immediately before the closing date of the merger (except as provided below) will, by virtue of the merger and without any action on the part of any stockholder, be converted into the right to receive (1) \$36.37 in cash, without interest, (2) 1.117015 shares of Peoples common stock, plus cash in lieu of any fractional share interest, or (3) a combination of Peoples common stock and cash.

Renasant stockholders will have the opportunity to elect the form of consideration to be received for all shares of Renasant common stock held by them, subject to redesignation and adjustment procedures set forth in the merger agreement. The redesignation procedures are intended to ensure that not less than 50% or more than 55% of the outstanding shares of Renasant common stock in the aggregate will be converted into the right to receive Peoples common stock and that not less than 45% or more than 50% of the outstanding shares of Renasant common stock in the aggregate will be converted into the right to receive cash. The adjustment procedures are intended to ensure that the merger qualifies as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock received in exchange for shares of Renasant common stock. The redesignation and adjustment procedures are described in more detail below. Shares of Renasant common stock held by Peoples or Renasant or their subsidiaries, other than in a fiduciary capacity, or by Renasant stockholders who have elected to exercise dissenters' rights will not be converted into the right to receive the merger consideration upon completion of the merger.

Background of the Merger

Renasant was incorporated in September 1998. Renasant Bank, a wholly-owned subsidiary of Renasant, commenced operations as a Tennessee state bank in May 1999. Since its formation, Renasant Bank's primary service area has been the eastern area of Shelby County, Tennessee and the immediately surrounding areas. In early 2003, Renasant Bank began investigating the possibility of expanding its services into DeSoto County, Mississippi. Renasant Bank opened a loan production office in Hernando, Mississippi in January 2003.

On December 17, 2003, E. Robinson McGraw, the President of Peoples and Peoples Bank, met with Frank J. Cianciola, the Chairman of Renasant Bank and the Vice Chairman of Renasant. E. Robinson McGraw indicated that Peoples was interested in discussing the possibility of acquiring Renasant and Renasant Bank. On December 22, 2003, Frank J. Cianciola and Jack C. Johnson, the Vice Chairman of Renasant Bank and Chairman of Renasant, met with E. Robinson McGraw and discussed the potential strategic advantages of a combination of the two entities.

On January 6, 2004, Frank J. Cianciola and Michael J. McCarver, the Chief Financial Officer of Renasant Bank, met with E. Robinson McGraw and Stuart R. Johnson, the Chief Financial Officer of Peoples Bank, and held further discussions regarding each bank's philosophy, culture, and strategy. Renasant and Peoples executed a confidentiality agreement that provided that Peoples, E. Robinson McGraw and Stuart R. Johnson would maintain the confidentiality of the information provided to them by Renasant. On this date, Frank J. Cianciola and Michael J. McCarver provided E. Robinson McGraw and Stuart R. Johnson with preliminary due diligence information.

On January 21, 2004, Frank J. Cianciola and Jack C. Johnson again met with E. Robinson McGraw, discussing the proposed structure of an acquisition.

On January 22, 2004, Peoples made a preliminary offer of \$36.00 per share of Renasant common stock, with 50% of the consideration to be paid in cash and the remaining consideration to be paid with shares of Peoples common stock. The offer price was subject to the completion of due diligence and the negotiation of the terms of a definitive agreement.

On January 23, 2004, the Renasant Board of Directors met and discussed the possibility of being acquired by Peoples. The Board of Directors authorized Frank J. Cianciola and Jack C. Johnson to continue negotiations with Peoples within a given stock price range, subject to the Board's approval of the terms and conditions of the acquisition to be set forth in a definitive agreement. Frank J. Cianciola and Jack C. Johnson then responded to Peoples' preliminary offer with a counteroffer of \$37.25 per share of Renasant common stock.

During the week of January 26, 2004, Peoples responded to Renasant's counteroffer by reiterating its original offer of \$36.00 per share of Renasant common stock but at the same time indicating its willingness to increase the price per share to \$36.62, subject to the results of its due diligence.

From January 28, 2004 until February 2, 2004, representatives of Peoples conducted due diligence at Renasant Bank's offices, reviewing Renasant Bank's records and conducting interviews with senior executive officers of Renasant Bank.

On February 2, 2004, the Renasant Board of Directors met via a telephonic meeting, and agreed to a sales price of the Renasant stock for \$36.62 per share, with approximately 50% of the consideration to be paid in cash and the remaining consideration to be paid with shares of Peoples common stock, subject to the Board's approval of a definitive agreement. The Board of Directors also approved an amendment to Frank J. Cianciola's employment agreement to provide for a cash payment in the amount of \$500,000 in the event of a change of control and approved a change of control agreement between Renasant and Jack C. Johnson which provides for a cash payment in the amount of \$250,000 in the event of a change in control.

On February 4, 2004, Frank J. Cianciola and Michael J. McCarver met with Peoples' financial advisers to discuss the possible structure of an acquisition and other factors concerning the proposed transaction.

On February 5, 2004, E. Robinson McGraw met with Jack C. Johnson and Frank J. Cianciola and proposed a purchase price of \$36.37 per share of Renasant stock based upon its due diligence review. Subject to the Renasant Board's approval and the negotiation of a definitive agreement, Frank J. Cianciola and Jack C. Johnson accepted the offer, and a stock exchange ratio of 1.117015 was agreed upon based upon the offer of \$36.37 per share of Renasant stock and the closing price of Peoples common stock as of January 27, 2004 of \$32.56 per share.

The parties negotiated the terms of a definitive agreement from February 5 to February 17, 2004. During this time, Renasant conferred with its legal and financial advisers regarding the terms of the transaction. Renasant conducted due diligence on Peoples and Peoples concluded its due diligence on Renasant and Renasant Bank during this time.

On February 13, 2004, the Board of Directors of Renasant met and considered the form of the definitive agreement, discussing the pricing terms as well as the few outstanding issues to finalize the definitive agreement. The Board of Directors also reviewed and considered the opinion of Alex

Sheshunoff & Co. Investment Banking, LP ("Sheshunoff") that the merger consideration to be received by Renasant shareholders was fair from a financial point of view. The Board of Directors established a committee to resolve the outstanding issues and finalize the negotiations. Directors Johnson, Cianciola, and Bobango were appointed to the committee. Subject to the revisions deemed necessary by the appointed committee to resolve the remaining outstanding issues, the Board of Directors unanimously approved the definitive agreement and recommended that the definitive agreement be approved by Renasant's shareholders.

The committee of the Board of Directors finalized the outstanding issues on February 17, 2004 and the definitive agreement between Renasant and Peoples was executed on that date.

Renasant's Reasons for the Merger; Recommendation of the Board

The Renasant Board of Directors believes that the terms of the merger are fair to, and in the best interest of, Renasant and its shareholders and unanimously recommends that Renasant's shareholders vote FOR approval of the merger and the merger agreement. In reaching its determination that the merger and merger agreement are fair to, and in the best interest of, Renasant and its shareholders, the Renasant Board of Directors consulted with its legal and financial advisers, as well as with Renasant and Renasant Bank management, and considered a number of factors, including, without limitation, the following:

- the consideration to be received by Renasant's shareholders in the merger, the premium over the market price represented by the merger consideration, and the opinion of Sheshunoff that the consideration to be received by Renasant's shareholders as a result of the merger is fair from a financial point of view:
- the fact that, immediately following the merger, Renasant Bank would continue to operate under its existing name and management team;
- the Board's familiarity with and review of Renasant's business, financial condition, results of operations, and prospects, including, without limitation, its potential growth and profitability and the general business risks of continued operation as a separate entity;
- information concerning the business, financial condition, and results of operations of Peoples, including the recent performance of Peoples common stock and the historical financial data of Peoples;
- the Board's belief that the terms of the merger agreement are attractive in that the merger agreement allows Renasant shareholders to elect to convert an illiquid asset into cash or into shares of an institution with publicly traded stock, whose capital base is strong and which has a history of paying dividends:
- the expectation that the merger will generally be a tax-free transaction to the extent Renasant's shareholders receive shares of Peoples common stock;
- the current and prospective economic and regulatory environment and competitive constraints facing the banking and financial institutions in the market area of Renasant Bank; and
- the recent business combinations involving financial institutions, either announced or completed, during the past year in the United States, the State of Tennessee and contiguous states and the effect of such combinations on competitive conditions in the market area of Renasant Bank.

In view of the wide variety of factors considered by the Board, the Board did not quantify or otherwise attempt to assign relative weights to the specific factors considered in making its determination.

FOR THE REASONS SET FORTH ABOVE, THE RENASANT BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS THAT THE RENASANT STOCKHOLDERS VOTE FOR THE APPROVAL AND ADOPTION OF THE MERGER AGREEMENT, THE PLAN OF MERGER AND THE MERGER.

Peoples' Reasons for the Merger

On February 17, 2004, the Peoples board of directors approved the merger agreement, the merger and the other transactions contemplated by those agreements. In connection with its approval of the merger, the board of directors recognized that:

- · the merger will expand Peoples' business into the high-growth Memphis, Tennessee region;
- the merger will increase Peoples' core deposit base by \$186 million, an important funding source; and
- the merger is expected to be initially dilutive to Peoples' earnings per share in 2004 and accretive in 2006 based on a generally accepted accounting principles basis.

The Peoples board of directors also considered the following risks associated with the merger in connection with its deliberations of the proposed transaction:

- · the challenges of integrating Renasant's businesses, operations and workforce with those of Peoples;
- the increased exposure to the Memphis, Tennessee market; and
- whether or not Peoples would be able to retain key management of Renasant.

The foregoing discussion of the factors considered by the board of directors is not intended to be exhaustive, but, rather, includes all principal factors considered by the board of directors. In reaching its decision to approve the merger agreement, the merger and the other transactions contemplated by the merger agreement, the board of directors did not quantify or assign any relative weights to the factors considered, and individual directors may have given different weights to different factors. The board of directors considered all these factors as a whole, and overall considered them to be favorable to, and to support, its determination.

Opinion of Alex Sheshunoff & Co. Investment Banking, LP

On February 2, 2004, Renasant retained Alex Sheshunoff & Co. Investment Banking, L.P. ("Sheshunoff") to provide its opinion as to the fairness from a financial viewpoint of the merger consideration to the shareholders of Renasant. As part of its investment banking business, Sheshunoff is regularly engaged in the valuation of securities in connection with mergers and acquisitions and valuations for estate, corporate and other purposes. The Board of Directors of Renasant retained Sheshunoff based upon its experience as a financial advisor in mergers and acquisitions of financial institutions and its knowledge of financial institutions. The Board selected Sheshunoff based on its

familiarity with the firm arising from its engagement of Sheshunoff in June 2003 to consult with the Board of Directors as to an appropriate price as of June 30, 2003 for the common stock of Renasant for the Board of Directors to use as a benchmark in determining the price at which to issue and redeem shares of Renasant common stock from time to time.

On February 17, 2004, Sheshunoff rendered its written opinion to Renasant's board of directors that, as of such date, the merger consideration was fair, from a financial point of view, to the shareholders of Renasant. Sheshunoff did not advise Renasant in connection with the proposed merger. The type and amount of consideration and the terms and conditions of the merger were negotiated directly by and between Peoples and Renasant.

The full text of the fairness opinion which sets forth, among other things, assumptions made, procedures followed and matters considered is attached as Annex C to this joint proxy statement/prospectus. No limitations were placed on Sheshunoff's review. The shareholders of Renasant are urged to read Sheshunoff's fairness opinion carefully and in its entirety. The fairness opinion is addressed to the Board of Directors of Renasant and does not constitute a recommendation to any shareholder of Renasant as to how such shareholder should vote at Renasant's special meeting.

In connection with the fairness opinion, Sheshunoff:

- 1. Reviewed the merger agreement;
- 2. Reviewed certain publicly available financial statements and other information about Peoples;
- 3. Reviewed internal financial statements, operating data, and financial forecasts for Renasant;
- 4. Conducted conversations with executive management of Renasant regarding recent and projected financial performance;
- 5. Analyzed the present value of the after-tax cash flows Renasant could produce on an independent basis through the year 2008, based on assumptions provided by management;
- 6. Compared the proposed financial terms of the merger with the financial terms of certain other transactions that Sheshunoff deemed to be relevant;
- 7. Compared the historical stock price data and trading volume of Peoples common stock with that of certain other comparable publicly traded companies;
- 8. Compared certain financial characteristics and performance measures of Peoples with that of certain other comparable publicly traded companies; and
- 9. Compared the historical stock price performance of Peoples common stock with that of selected indices Sheshunoff deemed relevant.

In connection with its review, Sheshunoff relied upon and assumed the accuracy and completeness of all of the foregoing information provided to it or made publicly available, and Sheshunoff did not assume any responsibility for independent verification of such information. Sheshunoff assumed that internal confidential financial projections provided by Renasant were reasonably prepared reflecting the best currently available estimates and judgments of the future financial performance of Renasant, and did not independently verify the validity of such assumptions.

Sheshunoff did not make any independent evaluation or appraisal of the assets or liabilities of Renasant or Peoples nor was Sheshunoff furnished with any such appraisals. Sheshunoff did not examine any individual loan files of Renasant or Peoples. Sheshunoff is not an expert in the evaluation of loan portfolios for the purposes of assessing the adequacy of the allowance for losses with respect thereto and has assumed that such allowances were, in the aggregate, adequate to cover such losses.

The fairness opinion is necessarily based on economic, market and other conditions as in effect on, and the information made available to Sheshunoff as of February 13, 2004.

In rendering the fairness opinion, Sheshunoff performed a variety of financial analyses. The preparation of an opinion involves various determinations as to the most appropriate and relevant methods of financial analysis and the application of those methods to the particular circumstances. Consequently, the fairness opinion is not readily susceptible to partial analysis or summary description. Moreover, the evaluation of fairness, from a financial point of view, of the merger consideration is to some extent subjective, based on the experience and judgment of Sheshunoff, and not merely the result of mathematical analysis of financial data. Sheshunoff did not attribute particular weight to any analysis or factor considered by it. Accordingly, notwithstanding the separate factors summarized below, Sheshunoff believes that its analyses must be considered as a whole and that selecting portions of its analyses and of the factors considered by it, without considering all analyses and factors, could create an incomplete view of the evaluation process underlying its opinion. The ranges of valuations resulting from any particular analysis described below should not be taken to be Sheshunoff's view of the actual value of Renasant, Peoples, or the combined entity.

In performing its analyses, Sheshunoff made numerous assumptions with respect to industry performance, business, and economic conditions and other matters, many of which are beyond the control of Renasant or Peoples. The analyses performed by Sheshunoff are not necessarily indicative of actual values or future results, which may be significantly more or less favorable than suggested by such analyses, nor are they appraisals. In addition, Sheshunoff's analyses should not be viewed as determinative of the opinion of the Board of Directors or the management of Renasant with respect to the value of Renasant or Peoples or to the fairness of the merger consideration.

The following is a summary of the analyses performed by Sheshunoff in connection with its opinion. The following discussion contains financial information concerning Renasant and Peoples as of December 31, 2003 and market information as of February 13, 2004.

For the purposes of the following analyses, Sheshunoff utilized a value of the merger consideration of \$36.37 per share and assumed Peoples' ten-day average stock price was \$32.00 per share. The merger consideration may have a higher or lower value depending on Peoples' stock price at the time of the merger.

Renasant Discounted Cash Flow Analysis. Using discounted cash flow analysis, Sheshunoff estimated the present value of the future after-tax cash flow streams that Renasant could produce on a stand-alone basis through the year 2008, under various circumstances, assuming that it performed in accordance with the projections provided by Renasant management.

Sheshunoff estimated the terminal value for Renasant at the end of 2008 by capitalizing the final period projected earnings using a discount rate that is the quotient of (i) the assumed annual long-term growth rate of the earnings of Renasant of 5% plus one and (ii) the difference between a range of required rates of return and the assumed annual long-term growth rate of earnings in (i) above. Sheshunoff discounted the annual cash flow streams (defined as all earnings in excess of that required to maintain a tangible equity to asset ratio of 7.0%) and the terminal values using discount rates ranging from 12% to

14%. The discount range was chosen to reflect different assumptions regarding the required rates of return of Renasant and the inherent risk surrounding the underlying projections. This discounted cash flow analysis indicated a range of values per share of \$14.55 to \$19.49 as shown in the table below compared to the merger consideration of \$36.37.

	DISCOUNT RATE		
	14.0%	13.0%	12.0%
Seller Present Value (in millions)	\$24,000	\$27,547	\$32,147
Seller Present Value (per share)	\$ 14.55	\$ 16.70	\$ 19.49

<u>Analysis of Selected Transactions</u>: Sheshunoff performed an analysis of premiums paid in selected recently completed acquisitions of banking organizations with comparable characteristics to the merger. Two sets of comparable transactions were selected to ensure a thorough analysis.

The first set of comparable transactions consisted of a group of transactions for banks in metropolitan areas in the Southeast Region (as defined by SNL Financial) of the United States for which pricing data were available. These comparable transactions consisted of eighteen mergers and acquisitions of banks with assets between \$100 million and \$300 million that were announced between January 1, 2003 and February 13, 2004. The analysis yielded multiples of the purchase prices in these transactions relative to:

- 1. Tangible book value ranging from 0.81 times to 4.8 times with an average of 2.6 times and a median of 2.6 times compared with the multiples implied in the merger of 3.3 times December 31, 2003 tangible book value for Renasant;
- 2. Last twelve months earnings ranging from 9.8 times to 41.7 times with an average of 26.4 times and a median of 26.5 times compared with the multiples implied in the merger of 43.2 times last twelve months earnings as of December 31, 2003 for Renasant;
- 3. Total assets ranging between 10.2% and 33.7% with an average of 22.0% and a median of 20.5% compared with the multiples implied in the merger of 25.1% of December 31, 2003 total assets for Renasant; and
- 4. Total deposits ranging from 11.8% to 41.2% with an average of 26.2% and a median of 25.1% compared with the multiples implied in the merger of 30.5% of deposits as of December 31, 2003 for Renasant.

The second set of comparable transactions consisted of banks in the United States with asset size and characteristics similar to Renasant for which pricing data were available. These comparable transactions consisted of thirty-four mergers and acquisitions of banks in the United States with total assets between \$100 million and \$300 million that were announced between July 1, 2003 and February 13, 2004. The analysis yielded multiples of the purchase prices in these transactions relative to:

- 1. Tangible book value ranging from 0.8 times to 4.8 times with an average of 2.5 times and a median of 2.5 times compared with the multiples implied in the merger of 3.3 times December 31, 2003 tangible book value for Renasant;
- 2. Last twelve months earnings ranging from 9.8 times to 42.1 times with an average of 25.1 times and a median of 23.7 times compared with the multiples implied in the merger of 43.2 times last twelve months earnings as of December 31, 2003 for Renasant;
- 3. Total assets ranging between 10.2% and 33.7% with an average of 20.3% and a median of 19.8% compared with the multiples implied in the merger of 25.1% of December 31, 2003 total assets for Renasant; and
- 4. Total deposits ranging from 11.8% to 41.2% with an average of 24.4% and a median of 23.3% compared with the multiples implied in the merger of 30.5% of deposits as of December 31, 2003 for Renasant.

Contribution Analysis: Sheshunoff reviewed the relative contributions of Renasant and Peoples to the combined company based on the December 31, 2003 financial data. Sheshunoff compared the pro forma ownership interests of Renasant and Peoples of 9.7% and 90.3%, respectively, to: (i) total assets of 13.8% and 86.2%, respectively; (ii) total loans of 16.7% and 83.3%, respectively; (iii) total deposits of 14.1% and 85.9%, respectively; (iv) total equity of 11.1% and 88.9%, respectively; (v) net interest income of 11.9% and 88.1%, respectively; (vi) non-interest income of 2.4% and 97.6%, respectively; (vii) non-interest expense of 9.2% and 90.8%, respectively; and (viii) earnings of 5.7% and 94.3%, respectively. The cash consideration paid by Peoples to Renasant shareholders reduced the equity position of Renasant shareholders in the combined company relative to their proportional contribution of the other criteria compared. The effect of the cash consideration paid was to increase the proportional ownership by Peoples' shareholders of the criteria compared.

Comparable Company Analysis: Sheshunoff compared the operating and market results of Peoples to the results of other publicly traded companies. The comparable publicly traded companies were selected primarily on the basis of two criteria: geographic location and total asset size. The geographic location of the banks was the Southeast Region of the United States (as defined by SNL Financial). Within the selected region, Peoples was compared to companies with total assets between \$1 billion and \$2 billion ("Peoples Peer Group"). The data for the following tables are based on information provided by SNL Financial. Some of the ratios presented are proprietary to SNL Financial and may not strictly conform to the common industry calculations or to the calculations by Peoples in its financial statements.

	Peoples	Peoples Peer Group Mean
Net Interest Margin	4.23%	4.22%
Efficiency Ratio	62.8%	62.9%
Return on Average Assets	1.33%	1.22%
Return on Average Equity	13.41%	13.26%
Equity to Asset Ratio	9.72%	9.35%
Tangible Equity to Tangible Asset Ratio	9.40%	7.81%
Ratio of Non-performing Assets to Total Assets	0.45%	0.64%
Ratio of Non-performing Loans to Total Loans	0.54%	0.64%
Ratio of Loan Loss Reserves to Loans	1.53%	1.37%
Ratio of Loan Loss Reserves to Non-performing Assets	206%	192%

Generally, Peoples' performance as measured by its net interest margin, efficiency ratio, and return on average equity were similar to the Peoples Peer Group Mean. Peoples' return on average assets was somewhat higher than the Peoples Peer Group Mean. Its capital levels were similar as shown in the equity to asset ratio and its tangible equity to tangible assets ratio was higher. Peoples' asset quality, as measured by its ratio of non-performing assets to total assets and its ratio of non-performing loans to total loans, was better than the Peoples Peer Group Mean. Its coverage of non-performing assets as shown by the ratio of loan loss reserves to non-performing assets was better.

The companies' market results based on market data as of December 31, 2003 are shown in the following table.

	Peoples	Peoples Peer Group Mean
Market Price as a Multiple of Stated Book Value	1.97x	1.69x
Market Price as a Multiple of Stated Tangible Book Value	2.04x	2.02x
Price as a Multiple of LTM Earnings	15.1x	14.1x
Dividend Yield per Share	2.42%	3.50%
Dividend Payout per Share	34.4%	39.1%

Peoples' market price as a multiple of stated book value was higher than its peer group and its price to stated tangible book was similar to the comparable peer group multiple. Its price to earnings multiple as shown in the price as a multiple of last twelve months earnings through December 31, 2003 was slightly above the comparable ratio for the Peoples Peer Group. Peoples' dividend yield and dividend payout were below the respective measures for its peer group.

Sheshunoff compared selected stock market results of Peoples to the publicly available corresponding data of other composites that Sheshunoff deemed to be relevant, including (i) the NASDAQ Bank Stock index, (ii) the SNL index of banks in the Southeastern Region of the U.S. (as defined by SNL Financial, L.C.), and (iii) the S&P 500. Peoples' common stock price has performed better than the selected indices for the period from January 1, 2002 through February 13, 2004.

No company or transaction used in the comparable company and comparable transaction analyses is identical to Renasant, Peoples, or Peoples Merger Corporation as the surviving corporation in the merger. Accordingly, an analysis of the results of the foregoing necessarily involves complex considerations and judgments concerning differences in financial and operating characteristics of Renasant and Peoples and other factors that could affect the public trading value of the companies to which they are being compared. Mathematical analysis (such as determining the average or median) is not in itself a meaningful method of using comparable transaction data or comparable company data.

Pursuant to an engagement letter dated February 2, 2004 between Renasant and Sheshunoff, Renasant agreed to pay Sheshunoff a professional fee of \$30,000. In addition, Renasant agreed to reimburse Sheshunoff for its reasonable out-of-pocket expenses. Renasant also agreed to indemnify and hold harmless Sheshunoff and its officers and employees against certain liabilities in connection with its services under the engagement letter, except for liabilities resulting from the negligence, violation of law or regulation or bad faith of Sheshunoff or any matter for which Sheshunoff may have strict liability. Sheshunoff has no other material contacts with Renasant, except that Renasant entered into an agreement with Alex Sheshunoff Management Services, L.P., an affiliate of Sheshunoff ("Management Services"), pursuant to which Management Services agreed to review the terms of all outstanding loans of Renasant in amounts of at least \$500,000. This engagement began in February, 2004 and was completed in March, 2004.

The fairness opinion is directed only to the question of whether the merger consideration is fair from a financial perspective and does not constitute a recommendation to any Renasant shareholder to vote in favor of the merger. No limitations were imposed on Sheshunoff regarding the scope of its investigation or otherwise by Renasant.

Based on the results of the various analyses described above, Sheshunoff concluded that the merger consideration to be paid by Renasant pursuant to the merger is fair to the Renasant shareholders, from a financial point of view.

Material United States Federal Income Tax Consequences

The following discussion is a summary of certain United States federal income tax consequences of the merger to a holder of shares of Renasant common stock who holds the shares as capital assets (referred to in this discussion as a "Holder"). The discussion is based on laws, regulations, rulings and decisions in effect on the date hereof, all of which are subject to change, possibly with retroactive effect. This discussion is for general information only and may not be applicable with respect to Holders subject to special treatment under the Internal Revenue Code (including, but not limited to, financial institutions, tax-exempt organizations, mutual funds, insurance companies, S corporations or other pass-through entities, dealers in securities or foreign currency, Holders who exercise dissenters' rights, foreign Holders, Holders who acquired shares of Renasant common stock pursuant to the exercise of employee stock options or otherwise as compensation or through tax-qualified retirement plans, traders in securities who elect the mark-to-market method of accounting for their securities holdings, Holders subject to the alternative minimum tax, Holders who have a functional currency other than the U.S. dollar, or Holders who hold Renasant common stock as part of a hedge against currency risk, straddle or a constructive sale or conversion transaction). In addition, the discussion does not address the state, local or foreign tax consequences of the merger.

Based on representations provided by Renasant and Peoples and on certain customary factual assumptions, all of which must continue to be accurate in all material respects as of the closing date of the merger, it is the opinion of Phelps Dunbar LLP, counsel to Peoples, that the material United States federal income tax consequences of the merger are as follows:

- the merger will be treated as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code;
- a Holder will not recognize gain or loss if the Holder exchanges Renasant common stock solely for Peoples common stock, except to the extent of any cash received in lieu of a fractional share;
- where a Holder exchanges Renasant common stock solely for cash in the merger, such cash will be treated as having been received by such holder as a
 distribution in redemption of such Holder's Renasant common stock, subject to the provisions and limitations of Section 302 of the Internal Revenue
 Code:
- a Holder will recognize gain (but not loss) if the Holder exchanges Renasant common stock for a combination of Peoples common stock and cash, and the Holder's gain will be equal to the lesser of:
 - (1) the excess, if any, of:
 - (a) the sum of the cash (excluding any cash received in lieu of a fractional share of Peoples common stock) and the fair market value of the Peoples common stock received (including any fractional share of Peoples common stock which is deemed to be distributed in the merger and then redeemed by Peoples), over
 - (b) the Holder's tax basis in Renasant common stock, or
 - (2) the amount of cash received;

- a Holder's tax basis in the Peoples common stock received in the merger will equal the Holder's tax basis in the Renasant common stock surrendered (less the tax basis allocable to any fractional share which is deemed to be distributed in the merger and then redeemed by Peoples), increased by (i) the amount treated as a dividend and (ii) the amount of gain to the Holder which was recognized on such exchange (not including any portion of such gain which was treated as a dividend), and decreased by the amount of cash received in the merger (excluding any cash received in lieu of a fractional share interest);
- a Holder's holding period for the Peoples common stock received in the merger will include the holding period for the shares of Renasant common stock surrendered, provided that the Renasant common stock was held as a capital asset at the time of the merger; and
- the receipt of cash in lieu of fractional shares of Peoples common stock will be treated as if the fractional shares were distributed in the merger and then redeemed by Peoples. Generally, these cash payments will be treated as having been received as distributions in full payment in exchange for the shares considered redeemed.

If a Holder acquired different blocks of Renasant common stock at different times and at different prices, any gain or loss will be determined separately with respect to each block of Renasant common stock, and the cash and Peoples common stock received will be allocated pro rata to each block of stock. In addition, a Holder's basis and holding period in the Peoples common stock received in the merger will be determined separately with reference to each block of Renasant common stock.

Taxation of Capital Gain. Subject to the discussion under "Possible Treatment of Gain as a Dividend" set forth below, gain or loss recognized in connection with the merger will constitute long-term capital gain or loss if a Holder's holding period in the Renasant common stock is greater than one year as of the date of the merger. If a Holder is not a corporation, this long-term capital gain generally will be taxed at a maximum United States federal income tax rate of 15%. The deductibility of capital losses is subject to limitations.

Possible Treatment of Gain as a Dividend. In general, in determining whether the gain recognized in the merger will be treated as capital gain or dividend income, a Holder (other than a Holder who exchanges Renasant common stock solely for cash) will be treated as if it first exchanged all of the Holder's shares of Renasant common stock solely for Peoples common stock and then Peoples immediately redeemed a portion of that Peoples common stock in exchange for the cash that the Holder actually received. Gain recognized in this deemed redemption of Peoples common stock will result in capital gain if there is a "meaningful reduction" in the Holder's deemed percentage stock ownership in Peoples. The Internal Revenue Service has ruled that a relatively minor reduction in the percentage stock ownership of a minority shareholder in a publicly held corporation whose relative stock interest is minimal and who exercises no control with respect to corporate affairs is a "meaningful reduction." Accordingly, in most circumstances, gain recognized by a Holder that exchanges its shares of Renasant common stock for a combination of Peoples common stock and cash will be capital gain. Each Holder that may be subject to these rules should consult its tax advisor.

Cash Received in Lieu of a Fractional Share. Cash received in lieu of a fractional share of Peoples common stock will be treated as received in redemption of the fractional share. Generally, a Holder of a fractional share will recognize gain or loss equal to the difference between the amount of cash received and the portion of the basis of the Holder's shares of Renasant common stock allocable to the fractional share. Such gain or loss generally will be capital gain or loss, and will be long-term capital gain or loss if the holding period for the Holder's shares of Renasant common stock was greater than one year as of the date of the exchange.

Backup Withholding. Unless a Holder complies with certain reporting or certification procedures or is an "exempt recipient" (i.e., in general, corporations and certain other entities), the Holder may be subject to a backup withholding tax of 28% with respect to any cash payments received pursuant to the merger. A foreign shareholder should consult its tax advisor with respect to the application of withholding rules to any cash payments received by it pursuant to the merger.

Reporting Requirements. If a Holder receives Peoples common stock as a result of the merger, such Holder will be required to retain records pertaining to the merger and will be required to file with such Holder's United States federal income tax return for the year in which the merger takes place a statement setting forth certain facts relating to the merger.

Each Renasant stockholder is urged to consult his or her own tax advisor with respect to the federal, state, local and foreign tax consequences of the merger.

Accounting Treatment

Peoples will account for the merger as a purchase transaction under generally accepted accounting principles. Under the purchase method of accounting, the assets and liabilities of Renasant will be recorded, as of completion of the merger, at their respective fair values and added to those of Peoples. Financial statements and reported results of operations of Peoples issued after completion of the merger will reflect these values, but will not be restated retroactively to reflect the historical financial position or results of operations of Renasant.

Regulatory and Third-Party Approvals

Under the merger agreement, Peoples and Renasant have agreed to use their best efforts to obtain all necessary actions or nonactions, waivers, consents and approvals from any governmental authority necessary to complete and make effective the merger and other transactions contemplated by the merger agreement. The required regulatory approvals include approvals of federal and state agencies as described below. All other applications and notices have been filed, or are in the process of being filed.

Federal Bank Regulatory Approvals. The merger is subject to the prior approval of the Federal Reserve Board (the "Federal Reserve") under Sections 3(a) (1) and 3(a)(3) of the Bank Holding Company Act (the "BHC Act"), and related federal regulations. In reviewing an application for such approvals under the BHC Act, the Federal Reserve is required to consider the following:

- competitive factors, such as whether the merger will result in a monopoly or whether the benefits of the merger to the public in meeting the needs and convenience of the community clearly outweigh the merger's anticompetitive effects or restraints on trade; and
- · banking and community factors, which includes an evaluation of:
 - the financial and managerial resources of Peoples, including its subsidiaries, and of Renasant, and the effect of the proposed transaction on those resources;
 - management expertise;
 - internal-control and risk-management systems;

- the capital of Peoples; and
- the convenience and needs of the communities to be served; and
- the effectiveness of Peoples and Renasant in combating money laundering activities.

The application process includes publication and opportunity for comment by the public. The Federal Reserve may receive, and must consider, properly filed comments and protests from community groups and others regarding (among other issues) each institution's performance under the Community Reinvestment Act of 1977, as amended. The Federal Reserve is also required to ensure that the proposed transaction would not violate Tennessee law regarding the number of years a bank must be in operation before it can be acquired, deposit concentration limits, Tennessee community reinvestment laws and any Tennessee antitrust statutes

State Bank Regulatory Approvals. Peoples or its subsidiaries have filed or will file applications or notices with the Tennessee Commissioner of Financial Institutions for approval of the merger. The standards that are required to be considered by this state bank regulatory authority are similar to those described above with regard to the Federal Reserve.

Other Regulatory Approvals. In connection with or as a result of the merger, Peoples or Renasant may be required, pursuant to other laws and regulations, either to notify or obtain the consent of other regulatory authorities and organizations to which such companies or subsidiaries of either or both companies may be subject.

If the approval of the merger by any of the authorities mentioned above is subject to compliance with certain conditions, there can be no assurance that the parties or their subsidiaries will be able to comply with such conditions or that compliance or non-compliance will not have adverse consequences for the combined company after consummation of the merger. The parties believe that the proposed merger is compatible with such regulatory requirements.

While Peoples and Renasant believe that they will receive the requisite regulatory approvals for the merger, there can be no assurance regarding the timing of the approvals or the ability of the companies to obtain the approvals on satisfactory terms or the absence of litigation challenging such approvals or otherwise. There can likewise be no assurance that any state attorney general or other domestic regulatory authority will not attempt to challenge the merger on antitrust grounds or for other reasons, or, if such a challenge is made, as to the result thereof. The merger is conditioned upon the receipt of all consents, approvals and actions of governmental authorities and the filing of all other notices with such authorities in respect of the merger. See "The Merger Agreement — Conditions to the Completion of the Merger" on pages 61 through 63.

Peoples is not aware of any regulatory approvals that would be required for completion of the transactions contemplated by the merger agreement other than as described above. Should any other approvals be required, it is presently contemplated that such approvals would be sought. There can be no assurance that any other approvals, if required, will be obtained.

Third-Party Approvals. The merger is conditioned upon the receipt of all consents and approvals of third parties with respect to certain agreements, such as real property leases, unless the failure to obtain any such consent or approval would not reasonably be expected to have a material adverse effect on Peoples or Renasant. Pursuant to the merger agreement, Renasant and Peoples have agreed to use their best efforts to obtain all consents, approvals and waivers from third parties necessary in connection with the completion of the merger.

Dissenters' Rights

THIS DISCUSSION IS NOT A COMPLETE STATEMENT OF THE LAW PERTAINING TO DISSENTERS' RIGHTS UNDER THE TBCA AND IS QUALIFIED IN FULL BY THE FULL TEXT OF CHAPTER 23 OF THE TBCA, WHICH IS REPRINTED IN ITS ENTIRETY AS ANNEX D TO THIS PROXY STATEMENT/PROSPECTUS. ALL REFERENCES IN CHAPTER 23 AND IN THIS SUMMARY TO A "STOCKHOLDER" OR "HOLDER" ARE TO THE RECORD HOLDER OF THE SHARES OF RENASANT COMMON STOCK AS TO WHICH DISSENTERS' RIGHTS ARE ASSERTED.

Holders of Renasant common stock are entitled to appraisal rights under Chapter 23 of the TBCA. Under the TBCA, record holders of Renasant common stock who continuously hold such shares through the closing date of the merger, who follow the procedures set forth in Chapter 23 and who do not vote in favor of the merger agreement will be entitled to receive payment of the "fair value" of their shares of Renasant common stock as determined by Renasant. In the event that such record holder believes Renasant's payment is less than the fair value of the shares, such holder has the right to have the fair value of the shares determined by a Tennessee court, which will include a fair rate of interest, if any, as determined by the court. "Fair value" is exclusive of any element of value arising from the accomplishment or expectation of the merger. The holders are, in such circumstances, entitled to dissenters' rights because they hold stock of constituent corporations to the merger, and may be required by the merger agreement to accept merger consideration in the form of cash consideration.

A person having a beneficial interest in shares of Renasant common stock held of record in the name of another person, such as a broker or nominee, must act promptly to cause the record holder to follow the steps summarized below properly and in a timely manner to perfect the dissenters' rights provided under Chapter 23.

Under Chapter 23, where a proposed merger is to be submitted for approval at a meeting of stockholders, as in the case of the Renasant special meeting, the notice of such special meeting must state that shareholders are or may be entitled to assert dissenters' rights under Chapter 23, and be accompanied by a copy of Chapter 23. This proxy statement/prospectus constitutes the required notice to the record holders of shares of Renasant common stock, and a copy of Chapter 23 is attached to this proxy statement/prospectus as Annex D. Any stockholder who wishes to exercise appraisal rights should review the following discussion and Annex D carefully, because failure to timely and properly comply with the procedures specified will result in the loss of dissenters' rights under the TBCA.

A holder of shares of Renasant common stock wishing to exercise his or her dissenters' rights must take both of the following actions. The holder must deliver to Renasant, before the vote on the merger agreement at the Renasant special meeting is taken, a written notice that reasonably informs Renasant of the identity of the record holder and the record holder's intent to demand payment of the fair value of the record holder's Renasant common stock if the merger is effected. The holder also must not vote in favor of the adoption of the merger agreement or the approval of the merger. A proxy or vote against the merger shall not constitute a demand. In addition, mere failure to execute and return a Form of Election to the exchange agent does not constitute a demand. All notices and written demands for payment should be delivered to Renasant at Renasant Bancshares, Inc., 2177 Germantown Road South, Germantown, Tennessee 38138, Attention: Secretary. A HOLDER OF RECORD WHO DOES NOT DELIVER NOTICE TO RENASANT BEFORE THE SPECIAL SHAREHOLDERS MEETING IS NOT ENTITLED TO PAYMENT FOR HIS OR HER SHARES PURSUANT TO CHAPTER 23. INSTEAD, SUCH HOLDER WILL BE DEEMED TO HAVE ELECTED TO RECEIVE CASH FOR 45% OF THE SHARES OF RENASANT COMMON STOCK HE OR SHE OWNS AND SHARES OF PEOPLES COMMON STOCK FOR 55% OF THE SHARES OF RENASANT COMMON STOCK HE OR SHE OWNS.

A holder of shares of Renasant common stock wishing to exercise his or her dissenters' rights must hold his or her shares of record on the date the written demand for appraisal is made and must hold his or her shares continuously through the closing date of the merger. Accordingly, a record holder of shares of Renasant common stock who is the record holder of Renasant common stock on the date the written demand for payment is made, but who after such date transfers such stock prior to the completion of the merger, will lose any right to payment in respect of such shares.

Only a holder of record of shares of Renasant common stock is entitled to assert dissenters' rights for the shares of Renasant common stock registered in that holder's name. A demand for payment should be executed by or on behalf of the holder of record, fully and correctly, as such holder's name appears on such holder's stock certificates. If the shares of Renasant common stock are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of the demand should be made in that capacity. If the shares of Renasant common stock are owned of record by more than one person, as in a joint tenancy or tenancy in common, the demand should be executed by or on behalf of all joint owners. An authorized agent, including one or more joint owners, may execute a demand for payment on behalf of a holder of record; however, the agent must identify the record owner or owners and expressly disclose the fact that, in executing the demand, the agent is agent for such owner or owners. A holder of record may assert dissenters' rights as to fewer than all the shares registered in his or her name only if the holder of record dissents with respect to all shares beneficially owned by any one person and notifies Renasant in writing of the name and address of each person on whose behalf the holder of record asserts dissenters' rights.

If the merger and the merger agreement are approved at the special meeting, within 10 days after such special meeting, Renasant (or Peoples Merger Corporation, in the event the merger has been completed before this notice is sent) must send a notice of the approval and effectiveness of the merger to each person who has properly asserted dissenters' rights under Chapter 23 and has not voted in favor of or consented to the merger. The notice must:

- · state where a payment demand must be sent and where and when certificates for shares of Renasant common stock must be deposited;
- provide a form for demanding payment that requires any person asserting dissenters' rights to certify that he or she acquired beneficial ownership of his or her shares of Renasant common stock before February 17, 2004, the date when the principal terms of the merger were publicly announced;
- set a date by which Renasant (or Peoples Merger Corporation, if the merger will be completed by the date to be set) must receive the demand for payment from the holder of record, which date must be within 60 days of the date of the notice; and
- · contain a copy of Renasant's charter.

A holder of record sent the notice described in the paragraph above must demand payment, certify that he or she was the beneficial owner of shares before February 17, 2004 and deposit his or her certificates for shares of Renasant stock in accordance with the notice. A holder taking such action does not lose any other of his or her rights as a holder of shares of Renasant common stock until those rights are cancelled pursuant to the merger agreement. In the event that the merger is not consummated with 60

days after the date set for demanding payment and depositing share certificates, Renasant will return the share certificates. A HOLDER OF RECORD WHO DOES NOT COMPLY WITH THE TERMS OF THE NOTICE FROM RENASANT (OR PEOPLES MERGER CORPORATION) IS NOT ENTITLED TO PAYMENT FOR HIS OR HER SHARES PURSUANT TO CHAPTER 23. INSTEAD, SUCH HOLDER WILL BE DEEMED TO HAVE ELECTED TO RECEIVE CASH FOR 45% OF THE SHARES OF RENASANT COMMON STOCK HE OR SHE OWNS AND SHARES OF PEOPLES COMMON STOCK FOR 55% OF THE SHARES OF RENASANT COMMON STOCK HE OR SHE OWNS. A demand for payment may not be withdrawn without Renasant's (or Peoples Merger Corporation's, if the merger has been completed) consent.

Under Chapter 23, after receipt of a demand for payment or the effectuation of the merger, whichever is later, Peoples Merger Corporation (on behalf of Renasant, since the merger will have been effected) must pay each holder of record who properly complied with the requirements of the notice from Renasant the fair value of such holder's shares, with interest accrued from the date of the approval of the merger. The payment, when made, must be accompanied by:

- Renasant's balance sheet, income statement and a statement of changes in shareholder's equity, each as of December 31, 2003, together with the latest available interim financial statements:
- a statement of Peoples Merger Corporation's estimate of the fair value of a share of its common stock;
- · an explanation as to how interest was calculated;
- a statement that a holder dissatisfied with his or her payment may seek a judicial determination of the fair value of his or her shares in accordance with Chapter 23; and
- a copy of Chapter 23, if not previously delivered to such shareholder.

If a holder of shares of Renasant common stock did not own his or her shares on February 17, 2004, Renasant (or Peoples Merger Corporation, if the merger has been completed) may refuse to make any payments pursuant to Chapter 23. If payment is refused, Renasant (or Peoples Merger Corporation) may still pay a holder of record who did not own his or her shares on February 17, 2004 yet wishes to exercise his or her dissenters' rights the fair value of such persons shares, with accrued interest, provided that such holder accepts such payment in full satisfaction of his or her demand for payment.

A holder of record who has exercised his or her dissenters' rights may notify Peoples Merger Corporation (rather than Renasant, since the merger will have been completed) in writing of his or her own estimate of the fair value for his or her shares of Renasant common stock and the amount of interest due and demand payment of any amounts above what Peoples Merger Corporation has paid already under any of the following circumstances:

- a holder of record who has exercised his or her dissenters' rights believes that the amount paid by Peoples Merger Corporation is less than the actual fair value of his or her shares or that the interest is incorrectly calculated;
- · Peoples Merger Corporation fails to pay the holder of record within 60 days after the date by which a holder of record must demand payment; or

· the merger is not consummated and Renasant fails to return the holder's stock certificates

A holder of record waives this right to be paid his or her own estimate of fair value if Peoples Merger Corporation is not notified of such demand within 30 days after Peoples Merger Corporation has paid, or offered to pay, such holder of record.

If the holder of record's demand for payment of his or her own estimate of fair value plus interest has not been settled within 60 days after receiving a demand for payment, Peoples Merger Corporation (rather than Renasant, since the merger will have been completed) must bring a proceeding in a court of record having equity jurisdiction in Shelby County, Tennessee to determine the fair value of the holder of record's shares of Renasant common stock. If Peoples Merger Corporation fails to bring such proceeding at all or within the required time period, it must pay the holder's demand for payment. All holders of record whose demands have not been settled must be parties to the proceeding and served with a copy of the petition.

The court hearing the case has exclusive and plenary jurisdiction. It may appoint one or more persons to receive evidence regarding the fair value of shares of Renasant common stock and to recommend a decision on fair value to the court. A holder of record party to such proceeding has the same discovery rights as parties in other civil proceedings. Each holder of record party to such proceeding is entitled to judgment in the amount, if any, that the court finds such holder's estimate of fair value plus accrued interest exceeds the amount paid by Renasant. The judgment may be the fair value plus accrued interest of the shares of Renasant common stock owned by a holder of record who became a beneficial owner after February 17, 2004 and whom Renasant (or Peoples Merger Corporation) determined to withhold payment as described above.

The court shall determine all costs of the proceeding, including compensation and expenses of persons appointed by the court to receive evidence. Peoples Merger Corporation shall pay the costs of the proceeding, except that the court may assess costs against all or some of the holders of record party to the proceeding to the extent the court finds such individuals acted arbitrarily, vexatiously or not in good faith in demanding payment. The court may also assess legal and expert fees against either Peoples Merger Corporation or the holders or record. If the court finds that the legal fees of one attorney were substantially beneficial to all holders of record party to the suit, but that Peoples Merger Corporation should not pay the attorneys' fees, the court may have these fees paid out of the amount awarded to all holders of record who benefited.

Interests of Certain Persons in the Merger

Certain members of Renasant's management and board of directors may be deemed to have interests in the merger that are in addition to their interests as stockholders of Renasant. The Renasant board was aware of these interests and considered them, among other matters, in approving the merger agreement.

On the closing date, Peoples and Renasant Bank will enter into an employment agreement with Francis J. Cianciola, the Chief Executive Officer of Renasant, pursuant to which Frank J. Cianciola will serve as President and Chief Executive Officer of Renasant Bank for a period commencing on the closing date and terminating five years after the closing date. Frank J. Cianciola will receive an annual salary of \$210,000, plus a bonus potential in accordance with Peoples' policies and employee benefits.

On the closing date, Peoples and Renasant Bank will enter into an employment agreement with Michael J. McCarver, the Chief Financial Officer of Renasant Bank, pursuant to which Michael J. McCarver will serve as Executive Vice President of Renasant Bank for a period commencing on the closing date and terminating two years after the closing date. Michael J. McCarver will receive an annual salary of \$106,000, plus a bonus potential in accordance with Peoples' policies and employee benefits.

On the closing date, Peoples and Renasant Bank will enter into an employment agreement with Donald E. Russell, an Executive Vice President of Renasant Bank, pursuant to which Donald E. Russell will serve as Executive Vice President of Renasant Bank for a period commencing on the closing date and terminating two years after the closing date. Donald E. Russell will receive an annual salary of \$116,000, plus a bonus potential in accordance with Peoples' policies and employee benefits.

Frank J. Cianciola and Jack C. Johnson have each entered into an agreement with Renasant requiring a cash payment by Renasant upon a change of control of Renasant. Renasant has agreed to pay Frank J. Cianciola \$500,000 and Jack C. Johnson \$250,000 in the event of a change of control of Renasant.

Peoples has also entered into agreements, in the form attached to this proxy statement/prospectus as Annex B-1, with the non-employee directors of Renasant. Under these agreements, Peoples has agreed that for a period of two years after the closing date Peoples will cause such directors to be appointed to the board of directors of Renasant Bank. These agreements also provide that Peoples will pay such directors the fees in accordance with the policy of Renasant Bank on directors' fees in place on the closing date. As of the date of this proxy statement, John A. Bobango, a director of Renasant, has taken a temporary leave of absence from the Renasant Board of Directors. Peoples will fulfill its obligations under the agreement with John A. Bobango at such time as he resumes his services as a director of Renasant.

Peoples and the directors and officers of Renasant who hold warrants to purchase Renasant common stock will have entered into an agreement regarding these warrants prior to the closing date. Peoples has offered to purchase such warrants for an amount equal to the number of shares of Renasant common stock specified in such warrant multiplied by \$36.37, less the aggregate exercise price for all shares of Renasant common stock specified in such warrant. The purchase price for the warrants will be paid entirely in cash. Any warrants not sold to Peoples will be amended such that each warrant will entitle the holder to purchase shares of Peoples common stock equal to the number of shares of Renasant common stock that the warrant allowed the holder to purchase multiplied by 1.117015. The exercise price will be an amount equal to the exercise price specified in the warrant divided by 1.117015.

The merger agreement provides that for a period of 6 years following the closing date of the merger Peoples will indemnify and hold harmless from liability those current or former directors and officers of Renasant or Renasant Bank who sign an agreement with Peoples. The indemnification applies to acts or omissions occurring at or prior to the closing date of the merger. Peoples will provide indemnification to the same extent as such directors or officers would be indemnified under the articles of incorporation or bylaws of Peoples as if they were directors or officers of Peoples. Peoples' liability for indemnification payments is limited to a total of \$2,000,000. Peoples has also agreed to use reasonable efforts to obtain for a period of six years after the closing date of the merger policies of directors' and officers' liability insurance for directors and officers of Renasant specified in the merger agreement. The insurance policies must cover acts or omissions occurring prior to the closing date of the merger. The policies must be on terms and in amounts no less favorable than those in effect on the date of the merger agreement. However, neither Peoples nor Peoples Merger Corporation are required to pay an aggregate premium for such insurance coverage in excess of 200% of the amount for such coverage as currently held by Renasant but in such case shall purchase as much coverage as reasonably practicable for such amount.

Restrictions on Resales by Affiliates

The shares of Peoples common stock to be issued to Renasant stockholders in the merger have been registered under the Securities Act. These shares may be traded freely and without restriction by those stockholders not deemed to be "affiliates" of Renasant as that term is defined under the Securities Act. An affiliate of a corporation, as defined by the rules promulgated under the Securities Act, is a person who directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, that corporation. Any subsequent transfer by an affiliate of Renasant must be one permitted by the resale provisions of Rule 144 promulgated under the Securities Act, in the case of such persons who become affiliates of Peoples, or Rule 145 promulgated under the Securities Act or as otherwise permitted under the Securities Act. These restrictions are expected to apply to the Renasant directors and certain executive officers of Renasant as well as to certain other related individuals or entities.

THE MERGER AGREEMENT

General

The Peoples board of directors and the Renasant board of directors have each unanimously approved the merger agreement, the merger and the other transactions contemplated by the merger agreement. This section of the proxy statement/prospectus describes material provisions of the merger agreement. This description does not purport to be complete and is qualified in its entirety by reference to the merger agreement and the related plan of merger, copies of which are attached as Annex A-1 and Annex A-2, respectively, to this proxy statement/prospectus. We urge you to read the merger agreement and the related plan of merger carefully and in their entirety.

Form of the Merger

If the holders of Renasant common stock adopt and approve the merger agreement and the merger and if all other conditions to the merger are satisfied or waived, Renasant will be merged into Peoples Merger Corporation, a wholly owned subsidiary of Peoples. After the merger, Peoples Merger Corporation will be the surviving corporation and will continue its corporate existence under Tennessee law as a wholly owned subsidiary of Peoples.

Timing of Closing

The closing of the merger will take place on the fifth business day, or such later date as the parties mutually agree, following the receipt of all necessary approvals and consents of all governmental entities, the expiration of all statutory waiting periods and the satisfaction or waiver of the conditions to the merger set forth in the merger agreement. Articles of Merger will be filed with the office of the Tennessee Secretary of State, as required under the corporation laws of Tennessee. The Articles of Merger will establish the closing date of the merger, which shall be no later than the day following the closing.

Merger Consideration

At the closing date of the merger, each share of Renasant common stock, except for treasury shares, shares held by Peoples or any of the subsidiaries of Peoples or Renasant (other than in a fiduciary capacity) and any dissenting shares, shall be converted into the right to receive either:

- cash in an amount equal to \$36.37, without interest;
- 1.117015 shares of Peoples common stock; or
- the cash consideration described above for 45% of the holder's shares of Renasant common stock and the stock consideration described above for 55% of the holder's shares of Renasant common stock.

Subject to the redesignation and adjustment procedures set forth in the merger agreement and described below, each holder of record of shares of Renasant common stock, except for treasury shares, shares held by Peoples or any of the subsidiaries of Peoples or Renasant (other than those held in a fiduciary capacity) and any dissenting shareholders, will be entitled to elect to receive either the cash consideration described above, the stock consideration described above, or the cash consideration for 45% of the holder's shares of Renasant common stock and the stock consideration for 55% of the holder's shares of Renasant common stock.

No fractional shares of Peoples common stock will be issued in connection with the merger. Instead, Peoples will make a cash payment without interest to each Renasant stockholder who would otherwise receive a fractional share. The amount of such cash payment will be determined by multiplying the fraction of a share of Peoples common stock otherwise issuable to such stockholder by the average closing price of one share of Peoples common stock for the ten trading days immediately preceding the last trading day immediately prior to the closing date of the merger as reported by the American Stock Exchange.

Election and Election Procedures

A form of election on which Renasant stockholders elect to receive cash, Peoples common stock or a combination of cash and Peoples common stock accompanies this proxy statement/prospectus. After the date of the mailing of this proxy statement/prospectus, Peoples and Renasant will each use their reasonable efforts to mail the form of election to all persons who become holders of record of Renasant common stock during the period between the record date of the special meeting and the date that is seven business days prior to the closing date of the merger and to otherwise make the form of election available to all persons who become holders of record of Renasant common stock between seven business days and four business days before the closing date of the merger.

To be effective, a form of election must be properly completed, signed and submitted (by mail, fax or other delivery) to Registrar and Transfer Company, the exchange agent, by the close of business on the third business day prior to the closing date of the merger. All elections will be irrevocable. Holders of record of shares of Renasant common stock who hold such shares in a representative capacity (for example, as a nominee or a trustee) may submit multiple forms of election, provided that such nominee or representative certifies that each form of election covers all of the shares of Renasant common stock held for a particular beneficial owner by the nominee or representative.

Peoples will have the discretion, which it may delegate in whole or in part to the exchange agent, to determine whether forms of election have been properly completed, signed and submitted and to disregard immaterial defects in forms of election. The decision of Peoples or the exchange agent in such matters will be conclusive and binding. Neither Peoples nor the exchange agent will be under any obligation to notify any person of any defect in a form of election.

If a Renasant stockholder does not submit a form of election to the exchange agent by 5:00 p.m., eastern time, on the third business day immediately prior to the closing date of the merger or if Peoples or the exchange agent determines that an election by a holder of Renasant common stock was not properly made, then such holder will be deemed to have elected to receive cash consideration for 45% of such holder's shares of Renasant common stock and stock consideration for the remaining 55% of such holder's shares.

Neither the Renasant board of directors nor its financial advisor makes any recommendation as to whether stockholders should elect to receive the cash consideration or the stock consideration in the merger, or a combination of the two. You must make your own decision with respect to such election, bearing in mind the tax consequences of the election you choose. See "The Merger — Material United States Federal Income Tax Consequences".

Redesignation Procedures

Under the merger agreement, the number of shares of Renasant common stock to be converted into the right to receive cash must not be less than 45% or more than 50% of the total number of shares of Renasant common stock outstanding immediately prior to the closing date of the merger, excluding treasury shares and shares held by Peoples or any of the subsidiaries of Peoples or Renasant (other than in a fiduciary capacity). If, after the results of the forms of election are calculated, the number of shares of Renasant common stock to be converted into cash exceeds this 50% threshold, the exchange agent will determine the number of shares convertible into cash that must be redesignated as shares convertible into Peoples common stock in order not to exceed the 50% threshold. After such determination, all holders who have elected to receive solely cash in exchange for their Renasant common stock shall, on a pro rata basis, have such number of their shares redesignated as shares convertible into Peoples common stock so that the total number of shares of Renasant common stock to be converted into the right to receive cash will not be greater than 50% of the outstanding shares of Renasant common stock immediately prior to the closing date of the merger.

Notwithstanding the foregoing, no redesignation will be effected for a holder who has made a cash election but, as a result of such redesignation, would receive fewer than 10 shares of Peoples common stock in exchange for all of such holder's shares of Renasant common stock, unless such redesignation is necessary for the merger to qualify as a tax-free reorganization as discussed below.

The number of shares of Renasant common stock to be converted into the right to receive shares of Peoples common stock may be not less than 50% or more than 55% of the number of shares of Renasant common stock outstanding immediately prior to the closing date of the merger, excluding treasury shares and shares held by Peoples or any of the subsidiaries of Peoples or Renasant (other than in a fiduciary capacity). If, after the results of the forms of election are calculated, the number of shares of Renasant common stock to be converted into shares of Peoples common stock exceeds this 55% threshold, the exchange agent will determine the number of shares convertible into shares of Peoples common stock that must be redesignated as shares convertible into cash in order not to exceed the 55% threshold. After such determination, all holders who have elected to receive solely Peoples common stock in exchange for their Renasant common stock will, on a pro rata basis, have a portion of their shares redesignated as shares convertible into shares of Peoples common stock so that the total number of shares of Renasant common stock to be converted into the right to receive shares of Peoples common stock will not be greater than 55% of the outstanding shares of Renasant common stock immediately prior to the closing date of the merger.

Holders who have elected to receive a combination of cash consideration with respect to 45% of their shares of Renasant common stock and stock consideration with respect to the remaining 55% of their

shares of Renasant common stock will not be subject to the redesignation procedures described above but will be subject to the adjustment procedures described below if such adjustment is necessary for the merger to qualify as a tax-free reorganization under Section 368 of the Internal Revenue Code.

Adjustment Procedures

Renasant and Peoples intend that the merger qualify as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock received in exchange for shares of Renasant common stock. To qualify, the merger must satisfy the continuity of interest requirement set forth in Treasury Regulation Section 1.368-1(e). To satisfy this requirement, the merger agreement requires that the total value of the shares of Peoples common stock to be delivered to holders of shares of Renasant common stock as part of their merger consideration be at least 40% of the total value of Renasant. If, after the exchange agent completes the redesignation procedures described above (to the extent any redesignation procedures are necessary), Peoples determines after good faith consultations with its tax counsel that the value of the Peoples common stock to be delivered as part of the merger consideration does not meet this 40% threshold, then Peoples may undertake the adjustment procedures. Under these procedures, Peoples shall increase the number of shares of Peoples common stock into which each share of Renasant common stock is convertible and decrease the cash to be paid for each share of Renasant common stock to the extent necessary to meet the 40% threshold. If the adjustment necessary to meet the 40% threshold requires Peoples to issue more than 85,000 additional shares of common stock, then the adjustment shall be at Peoples' option and not mandatory. Any increase in the number of shares of Peoples common stock to be received for each share of Renasant common stock and any decrease in the total amount of cash to be paid for each share of Renasant common stock will be done such that the reduction in the amount of cash to be paid as part of the merger consideration is equal to the increase in the value of the shares of Peoples common stock to be delivered as part of the merger.

After the redesignation and adjustment procedures (if any) are completed, all cash election shares and 45% of the shares of Renasant common stock which are subject to combination elections will be converted into the right to receive the cash consideration, and all stock election shares and 55% of the shares of Renasant common stock that are subject to combination elections will be converted into the right to receive the stock consideration. Certificates previously evidencing shares of Renasant Common Stock will, upon surrender, be exchanged, as applicable, for the cash consideration or for certificates evidencing the stock consideration, multiplied in each case by the number of shares previously evidenced by the canceled certificate. Cash will be paid for any fractional shares as described above. See "The Merger — Procedures for Exchanging Renasant Common Stock Certificates".

Each share of Renasant common stock held in the treasury of Renasant and each share of Renasant common stock owned by Peoples or any subsidiary of Peoples or Renasant, other than in a fiduciary capacity, immediately prior to the closing date of the merger will be canceled and extinguished without any conversion thereof. No payment will be made with respect to such shares.

Holders of shares of Renasant common stock who elect to exercise the dissenters' rights provided for in Chapter 23 of the TBCA will not have their shares converted into the right to receive merger consideration. In the event that a holder's dissenters' rights are lost or withdrawn, such holder will be deemed to have elected to receive a combination of cash consideration with respect to 45% of his or her shares of Renasant common stock and stock consideration with respect to the remaining 55% of his or her shares of Renasant common stock

Procedures for Exchanging Renasant Common Stock Certificates

Promptly after the closing date of the merger, the exchange agent will provide appropriate stock certificate transmittal materials to the holders of Renasant common stock. The transmittal materials will contain instructions for use in effecting the surrender to the exchange agent of Renasant common stock certificates in exchange for the merger consideration. After the closing date of the merger, each holder of shares of Renasant common stock issued and outstanding immediately prior to the closing date, other than dissenting shares, shall surrender for cancellation the certificate or certificates representing such shares to the exchange agent, together with a letter of transmittal duly executed and completed, in accordance with the instructions contained in the transmittal materials, and any other documents reasonably required by the exchange agent or Peoples. Each certificate for Renasant common stock shall be duly completed and endorsed as the exchange agent may require. Until you surrender your certificate or certificates representing your shares of Renasant common stock, Peoples will not be obligated to deliver the merger consideration to you.

If you surrender your stock certificates after the closing date of the merger, promptly upon surrender of your Renasant stock certificates and any other documents reasonably required by the exchange agent or Peoples, Peoples will deliver to you the merger consideration, consisting, as applicable, of Peoples common stock certificates, together with all withheld dividends or other distributions, but without interest thereon, and any cash payments due, including any cash payment for a fractional share, without interest. After the closing date of the merger, each certificate representing your outstanding shares of Renasant common stock immediately prior to the closing date of the merger will be deemed for all corporate purposes, other than the payment of dividends and other distributions to which you may be entitled as a former Renasant stockholder, to evidence only your right to receive the merger consideration in exchange for each such share.

Six months after the closing date of the merger, any merger consideration held by the exchange agent that remains undistributed to the former stockholders of Renasant will be delivered to Peoples upon demand. Any former Renasant stockholder who has not already complied with the surrender and exchange procedures at such time may look only to Peoples for payment of his or her claims for cash, Peoples common stock or any dividends or distributions with respect to Peoples common stock, all without any interest thereon.

From and after the closing date, the holders of certificates of Renasant common stock shall not have any rights with respect to such shares represented by such certificates. All rights to receive the merger consideration shall be deemed to have been paid or issued in full satisfaction of all rights pertaining to such shares of Renasant common stock. After the closing date, there shall be no further registration or transfers of shares of Renasant common stock.

In the event of any lost, stolen or destroyed certificates of Renasant common stock, you must make an affidavit of that fact and, if required by Peoples or the exchange agent, post a bond in such amount as either Peoples or the exchange agent reasonably direct as indemnity against any claim that may be made with respect to such certificates claimed to be lost, stolen or destroyed. Upon receipt of such affidavit and the posting of any bond required, the exchange agent will issue the merger consideration with respect to such lost, stolen or destroyed certificates.

If any merger consideration is to be issued or paid in the name of a person other than the person in whose name the Renasant common stock certificate being surrendered is registered, one condition to the payment and issuance of such merger consideration is that the certificate so surrendered be properly endorsed or otherwise be in proper form for transfer. Another condition will be that the person requesting the exchange pay or establish the prior payment or inapplicability of any transfer and other taxes required

by reason of the payment of the merger consideration in the name of a person other than the registered holder of the Renasant common stock certificate. Peoples or the exchange agent shall have the right to deduct and withhold from the merger consideration such amounts as Peoples or the exchange agent are required to deduct and withhold under any federal, state, local or foreign tax law with respect to the making of such payment. Any amounts withheld shall be treated as having been paid to the holder of shares of Renasant common stock in respect of whom such deduction and withholding was made.

If you receive shares of Peoples common stock as a result of the merger, you will have the right to vote after the closing date at any meeting of Peoples shareholders, according to the number of shares of Peoples common stock you received, regardless of whether you have exchanged your certificates for shares of Renasant common stock. You will also have the right to receive any dividends declared on Peoples' common stock, but such dividends will not be paid until stock certificates are physically exchanged.

Dissenting Shares

No share of Renasant common stock as to which the holder exercised his dissenters' rights as provided in Chapter 23 of the TBCA shall be converted into the right to receive the merger consideration. Peoples will direct all negotiations and proceedings with respect to any demands for payment of fair value according to the provisions of Chapter 23. Renasant will not, without the prior written consent of Peoples, make any payments or settle or otherwise negotiate with a holder who has exercised his dissenters' rights. If any holder withdraws or loses (through failure to perfect or otherwise) his or her right to payment, such holder will be deemed to have made a combination election and will receive the merger consideration according to such election. See "The Merger – Dissenters' Rights" discussed above.

Assumption of Renasant Stock Options

On the closing date of the merger, Peoples will assume the Stock Option Plan for Renasant Bancshares, Inc. and all of Renasant's obligations under this plan. After the closing date, each stock option granting such holder the right to purchase shares of Renasant common stock shall entitle the holder to purchase a number of shares of Peoples common stock equal to the number of shares of Renasant common stock specified in such stock option, multiplied by 1.117015. The exercise price will be equal to the quotient of the exercise price specified in the option to purchase Renasant common stock divided by 1.117015.

Any restriction on the exercise of any such Renasant stock option will continue in full force and effect, and the term and other provisions of such Renasant stock option will otherwise remain unchanged. However, each Renasant stock option assumed by Peoples will, in accordance with its terms, be subject to further adjustment as appropriate to reflect any stock split, division or subdivision of shares, stock dividend, recapitalization or other similar transaction subsequent to the closing date of the merger.

Eligibility to receive stock option grants following the closing date of the merger with respect to Peoples common stock will be determined by Peoples in accordance with its plans and procedures as in effect from time to time, and subject to any contractual obligations.

Treatment of Renasant Warrants

Prior to the closing date of the merger, Peoples shall cause each holder of a warrant to purchase shares of Renasant common stock to execute and deliver an agreement pursuant to which such holder may sell to Peoples, effective as of the closing date of the merger, such warrant to purchase shares of Renasant

common stock. The purchase price for the warrant will be an amount equal to the number of shares of Renasant common stock specified in such warrant multiplied by \$36.37, less the aggregate exercise price for all shares of Renasant common stock specified in such warrant, and subject to any required withholding taxes. Pursuant to the agreement, if the holder of the warrant does not sell such warrant to Peoples, the warrant will be amended so that from and after the closing date of the merger the warrant shall entitle the holder of such warrant to purchase a number of shares of Peoples common stock equal to the number of shares of Renasant common stock specified in the warrant multiplied by 1.117015, at an exercise price equal to the quotient of the exercise price specified in such warrant divided by 1.117015.

If a warrant holder elects to sell his or her warrant to Peoples, such holder must represent that he or she is an "accredited investor" as defined in Rule 501 of Regulation D, promulgated under the Securities Act of 1933, as amended.

Representations and Warranties

Renasant has made a number of representations and warranties in the merger agreement, the material aspects of which are described below. These descriptions are qualified in their entirety by reference to the merger agreement, a copy of which is attached to this proxy statement/prospectus as Annex A-1:

- Renasant is a corporation duly organized and validly existing under the laws of Tennessee and is registered as a bank holding company under the Bank Holding Company Act of 1956, as amended. Renasant has the power and authority, and has obtained all licenses and qualifications, to own or lease its property and assets and to carry on its business. Renasant Bank is the only subsidiary of Renasant, is also duly organized and validly existing under the laws of Tennessee, and has the power and authority, and has obtained all licenses and qualifications, to own or lease its property and assets and to carry on its business.
- The authorized capital stock of Renasant as of the date of the merger agreement consists of 2,000,000 shares of common stock (of which 1,429,697 are issued and outstanding) and 1,000,000 shares of preferred stock, none of which are issued or outstanding. Options to purchase 99,980 shares of Renasant common stock and warrants to purchase 120,000 shares of Renasant common stock are the only outstanding agreements or other commitments relating to the transfer, purchase or issuance of any of the capital stock of Renasant.
- Renasant has full power and authority to consummate the transactions called for by the merger agreement. Assuming all necessary consents and
 approvals are obtained, the consummation of the merger agreement will not violate any law, regulation or other governmental or judicial enactment or
 order or the terms of any agreement to which Renasant or Renasant Bank is a party.
- No takeover laws apply to the transactions contemplated by the merger agreement.
- The consolidated financial statements and associated materials of Renasant delivered to Peoples fairly present the financial condition of Renasant and were prepared in accordance with generally accepted accounting principles. Renasant has no obligations or liabilities material to Renasant or Renasant Bank taken as a whole which were not reflected in the financial statements or incurred since December 31, 2003 in the ordinary course of business consistent with past practices.

- There have not been any material adverse changes in the business, operations, prospects, assets or financial condition of Renasant and Renasant Bank taken as a whole since December 31, 2003.
- There are no legal proceedings pending or threatened against Renasant or Renasant Bank.
- Renasant and Renasant Bank have filed all tax returns and other information required to be filed with respect to any taxes of any kind and have paid all taxes due, other than those which are being contested in good faith and have not been finally determined. Proper reserves for all taxes have been established by Renasant and Renasant Bank. No audits or similar proceedings are presently pending. No tax liens exist except for those liens for taxes not yet due and payable.
- All employee benefit plans of Renasant have been operated in compliance with the applicable provisions of the Employee Retirement Income Security Act of 1974, as amended, and the Internal Revenue Code, and all regulations, rulings and announcements promulgated under either statute. There are no pending or threatened claims, administrative proceedings or litigation relating to any employee benefit plans. All contributions to such plans have been timely made and reflected on Renasant's financial statements. There are no obligations to provide health and life benefits to terminated employees except for Frank J. Cianciola and Don Russell, or as required by law, and, other than as disclosed to Peoples, there have been no amendments or other changes to employee benefits plans that would increase the expense incurred in maintaining such plan above the level incurred in 2003.
- All banking-related reports required to be filed with state or federal authorities have been filed, and such reports have been made available to Peoples.
- None of the information provided to Peoples in connection with this proxy statement/prospectus contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements made, in light of the circumstances under which they were made, not misleading.
- The deposit accounts of Renasant Bank are insured to the maximum extent permitted by federal law. All required premiums and assessments have been paid.
- Renasant and Renasant Bank have disclosed to Peoples, to the extent of their existence at all, all contracts relating to the borrowing of money, employment and consulting agreements, agreements with labor unions, agreements requiring payments to a director, officer or employee of Renasant or Renasant Bank in connection with this merger, agreements limiting Renasant or Renasant Bank's ability to carry on its business, any agreement or order with the Federal Reserve Board or any other regulatory agency, any joint venture or other profit-sharing agreement, or any other material agreement. Renasant and Renasant Bank are not in default, or will be in default after a lapse of time or the receipt of notice, of any of the above-listed agreements. All such agreements are enforceable.
- All real and personal property owned or leased by Renasant and Renasant Bank is in adequate condition and sufficient for each to carry on its business. The business operations and the properties and assets of Renasant and Renasant Bank are adequately insured in the judgment of Renasant.

- · Neither Renasant nor Renasant Bank, nor any of the properties owned by either, have been or are in violation of any environmental law.
- The allowance for loan losses maintained by Renasant is adequate to provide for anticipated losses. Other real estate is properly reflected on the balance sheet of Renasant.
- Except as disclosed to Peoples, there have been no transactions with "affiliates," as such term is defined under Rule 405 of the Securities Act of 1933, as amended.
- Renasant and Renasant Bank's internal accounting controls are sufficient to provide reasonable assurances that transactions are executed and recorded in accordance with management's authorizations and that access to assets is granted only in accordance with management's authorizations.
- All interest rate swaps and other risk management arrangements were entered into in accordance with prudent business practices and with financially responsible parties, and each such arrangement is the legally binding obligation of Renasant or Renasant Bank.
- All of the loans of Renasant and Renasant Bank are validly documented and were made in the ordinary course of business, the security for such loans is properly perfected, and no collateral has been released unless approved by Renasant or Renasant Bank. The terms of all loans are in compliance with applicable laws and such loans are enforceable against the obligors.
- None of the investments reflected on the balance sheets of Renasant are subject to any restrictions that materially impair the ability of Renasant or Renasant Bank to dispose of such investment.
- Neither Renasant nor Renasant Bank has taken any action which would prevent the merger from qualifying as a tax-free reorganization for federal
 income tax purposes with respect to the shares of Peoples common stock received in exchange for shares of Renasant common stock or would impede
 the receipt of any consent or approval required. No stock of Renasant or Renasant Bank has been redeemed, no dividends declared out of the ordinary
 course or other actions taken which would prevent Peoples from acquiring substantially all of Renasant's assets.
- Renasant and Renasant Bank have disclosed to Peoples all of their respective intellectual property, and have the right to use all intellectual property used in their business. The conduct of the business of Renasant and Renasant Bank does not violate any intellectual property owned by any third party, and no third party is violating Renasant or Renasant Bank's intellectual property rights.
- Renasant Bank is in compliance with all applicable provisions of the Community Reinvestment Act of 1977, as amended, and received an "Outstanding" rating as of its most recent evaluation.
- Neither Renasant nor Renasant Bank is currently in any discussions with any third party regarding a takeover by such third party of either Renasant or Renasant Bank.

- · No unlawful gifts or payments have been made by Renasant, Renasant Bank or any director, officer, employee or agent of either.
- Renasant operates at least one significant historic business line, or owns at least a significant portion of its historic business assets, in each case as defined under Treasury Regulation Section 1.368-1(d).

Peoples has made a number of representations and warranties in the merger agreement, the material aspects of which are described below. These descriptions are qualified in their entirety by reference to the merger agreement, a copy of which is attached to this proxy statement/prospectus as Annex A-1:

- Peoples is a corporation duly organized and validly existing under the laws of Mississippi and is registered as a bank holding company under the Bank Holding Company Act of 1956, as amended. Peoples has the power and authority, and has obtained all licenses and qualifications, to own or lease its property and assets and to carry on its business.
- Peoples Merger Corporation is a corporation duly organized and validly existing under the laws of Tennessee and has the power and authority to conduct its business as it is now being conducted. Peoples Merger Corporation was formed solely for the purpose of engaging in the merger and has not engaged in any business activity.
- Peoples and Peoples Merger Corporation have full power and authority to consummate the transactions called for by the merger agreement. Assuming all necessary consents and approvals are obtained, the consummation of the merger agreement will not violate any law, regulation or other governmental or judicial enactment or order or the terms of any agreement to which Peoples or Peoples Merger Corporation is a party.
- The consolidated financial statements and associated materials of Peoples delivered to Renasant fairly present the financial condition of Peoples and were prepared in accordance with generally accepted accounting principles. Peoples has no obligations or liabilities material to Peoples or Peoples Merger Corporation taken as a whole which were not reflected in the financial statements or incurred since September 30, 2003 in the ordinary course of business consistent with past practices.
- There have not been any material adverse changes in the business, operations, prospects, assets or financial condition of Peoples or any of its subsidiaries taken as a whole since September 30, 2003.
- · There are no material legal proceedings pending or threatened against Peoples or any of its subsidiaries.
- None of the information relating to Peoples in connection with this proxy statement/prospectus contains any untrue statement of a material fact or
 omits to state a material fact necessary to make the statements made, in light of the circumstances under which they were made, not misleading.
- The deposit accounts of Peoples Bank are insured to the maximum extent permitted by federal law, and all premiums and assessments required thereunder have been paid.

- Peoples Bank is "well capitalized" as such term is defined by Federal Deposit Insurance Corporation Regulations, and Peoples is "well capitalized" as
 such term is defined by Federal Reserve Board regulations. Peoples has sources of capital and financing, and a sufficient number of shares, to pay the
 merger consideration and any other amounts and to effect the transactions contemplated by the merger agreement.
- Peoples has made available to Renasant all reports or other documents filed by Peoples under the Securities Act of 1933, as amended, and the Securities Exchange Act of 1934, as amended, since January 1, 2002. All such reports and other documents comply with applicable federal rules.

Certain Covenants and Agreements

Renasant and Peoples have each entered into certain covenants and agreements relating to their respective actions prior to the consummation of the merger.

General. In addition to the covenants and agreements described in more detail below, the following covenants and agreements are applicable to Renasant and Renasant Bank (these descriptions are qualified in their entirety by reference to the merger agreement, a copy of which is attached to this proxy statement/prospectus as Annex A-1):

- Until the closing date, Renasant and Renasant Bank shall conduct their respective businesses in the ordinary course and consistent with past practices and prudent banking practices.
- · Until the closing date of the merger, and except as approved by Peoples in writing, Renasant shall not, and shall not permit Renasant Bank to:
 - change any provision of the Charter or other governing instrument or Bylaws of Renasant or Renasant Bank;
 - except for the issuance of the common stock pursuant to the present terms of the outstanding options and the outstanding warrants, change the
 number of shares of its authorized or issued stock, issue or grant any shares of its capital stock or any option, warrant, call, commitment,
 subscription, award, right to purchase or agreement, or any securities convertible into shares of such capital stock, or split, combine or reclassify
 any shares of its capital stock, or redeem or otherwise acquire any shares of such capital stock;
 - declare, set aside or pay any dividend or other distribution on the stock of Renasant or Renasant Bank;
 - grant any severance or termination pay (other than pursuant to binding contracts in effect on the date hereof and disclosed to Peoples), to, or enter into or amend any employment, consulting or compensation agreement with, any of its directors, officers, employees or consultants, or award any increase in compensation or benefits to any such person;
 - enter into or modify any employee benefit, incentive or welfare contract, plan or arrangement, or any trust agreement related to an employee benefit plan, in respect of any of its directors, officers, employees or consultants, other than any change that is required by law or is necessary or advisable to maintain the tax-qualified status of any tax-qualified

- plan, or make any contributions to any other defined contribution plan or any defined benefit pension or retirement plan other than in the ordinary course of business consistent with past practice or plan terms;
- sell or dispose of any assets or knowingly incur any liabilities other than in the ordinary course of business, or acquire (other than to realize upon collateral for a defaulted loan) any business or entity;
- make any capital expenditures in excess of \$50,000 in the aggregate, other than pursuant to binding commitments existing on the date of the merger agreement or to maintain existing assets in good repair or as disclosed to Peoples;
- file any applications or make any contract with respect to branching or site location;
- make any material change in its accounting methods or practices, other than changes required by GAAP, or change any of methods of reporting income and deductions for federal income tax purposes, except as required by law or regulation;
- change its banking policies in any respect except as may be required by law or regulation;
- · make, change or revoke any material tax election, amend any material return or settle or compromise any material liability for taxes;
- engage in any transaction with any person controlling, controlled by or under common control with Renasant or Renasant Bank, except as disclosed to Peoples;
- · enter into any leveraged arbitrage programs or take any other action for purposes of hedging exposure to changes in market rates of interest;
- originate or acquire any loans or other extensions of credit except for originations in accordance with existing lending policies and any current lending commitments outstanding;
- knowingly take, or fail to take, any action that would adversely affect or delay the ability of any party to perform its covenants and agreements under the merger agreement, to obtain any necessary approvals, consents or waivers of any governmental entity (or that would cause any of the approvals, consents or waivers to contain conditions or restrictions that would materially impair the value of the transaction) or to consummate the merger;
- knowingly take, or fail to take, any action that would result in any of its representations and warranties described above not being true and correct in any material respect on the closing date of the merger;
- merge in any manner or consolidate with any other corporation or bank;
- knowingly fail to comply with any laws, regulations, ordinances or governmental actions applicable to Renasant or Renasant Bank in a manner adverse to such business; or
- agree to do any of the foregoing.

Peoples has also agreed that until the closing date of the merger, Peoples shall not, and shall not permit any of its subsidiaries to:

- fail to conduct its business in the ordinary course consistent with past practices and prudent banking practices or fail to maintain any properties, leases, employees and advantageous business relationships, except as would not have a material adverse effect on the ability of Peoples and Peoples Merger Corporation to comply with the merger agreement;
- knowingly take any action that would adversely affect or delay the ability of Peoples or Peoples Merger Corporation to perform their respective
 covenants and agreements under the merger agreement, to obtain any necessary approvals, consents or waivers of any governmental entity (or that
 would cause any of the approvals, consents or waivers to contain conditions or restrictions that would materially impair the value of the transaction) or
 to consummate the merger;
- knowingly take, or fail to take, any action that would result in any of its representations and warranties not being true and correct in any material respect; or
- · agree to do any of the foregoing.

The descriptions of the foregoing covenants of Peoples are qualified in their entirety by reference to the merger agreement, a copy of which is attached to this proxy statement/prospectus as Annex A-1.

No Solicitation. Renasant has agreed that, until the earlier of the closing date of the merger or the date of termination of the merger agreement, neither Renasant nor Renasant Bank, nor any of their respective officers, directors, agents, representatives or affiliates, including any investment banker, attorney, financial advisor, accountant or other representative retained by Renasant or Renasant Bank, will solicit or hold discussions or negotiations with, or provide any information to, any person, entity or group concerning any "acquisition transaction." An "acquisition transaction" is defined in the merger agreement to mean an offer or proposal by a person or entity other than Peoples for (i) a merger, tender offer, recapitalization or consolidation, or similar transaction, involving Renasant or Renasant Bank, (ii) a purchase, lease or other acquisition or assumption of all or a substantial portion of the assets of Renasant or Renasant Bank, (iii) a purchase or other acquisition of beneficial ownership of securities representing 20% or more of the voting power of Renasant, or (iv) any substantially similar transaction.

The merger agreement does not prevent Renasant, or its board of directors, from, prior to the approval of the merger by the stockholders of Renasant:

- providing information in response to a request by a person who has made an unsolicited bona fide written proposal to engage in any acquisition transaction, after receipt from such person of an executed confidentiality agreement;
- engaging in any negotiations or discussions with a person who has made an unsolicited bona fide written proposal to engage in an acquisition transaction (an "acquisition proposal");
- failing to recommend, or withdrawing its recommendation of, the merger agreement and the merger and/or failing to hold the special meeting to consider this agreement; or
- · recommending an acquisition proposal to the stockholders of Renasant.

The board of directors of Renasant, however, may only undertake any of the foregoing actions after the board of directors has determined in good faith after consultation with outside legal counsel or its financial advisers, as appropriate, that (i) such action would be required in order for the directors to fulfill their fiduciary duties under applicable law and (ii) such acquisition proposal both is likely to be consummated, taking into account all aspects of the proposal and the person making the proposal, and, if consummated, would result in a transaction more favorable to Renasant's stockholders from a financial point of view than the merger. Any proposal complying with (i) and (ii) is a "superior proposal" under the merger agreement.

Renasant must communicate in writing to Peoples the terms of any proposal it receives to engage in an acquisition transaction no more than 48 hours after receipt. Within 10 days of delivery, Peoples has the right to match or better any superior proposal delivered to it. If Peoples notifies Renasant that it will match or better the superior proposal, the merger agreement and the plan of merger must be amended to reflect the matched or bettered terms within two days of Peoples' decision to so match or better the superior proposal. Upon such amendment, Renasant may not terminate the merger agreement and must notify the party making the superior proposal that such proposal has been matched or bettered and that the merger agreement has been amended to reflect this fact. After such amendment to the merger agreement, Renasant must, and must cause Renasant Bank and its representatives to, cease and terminate all discussions and negotiations regarding the superior proposal. New proposals from the third party may be made, and Peoples retains the same rights regarding such other proposals. If eleven days after Renasant has determined that the acquisition proposal is a superior proposal Peoples has not notified Renasant of its decision to match or better the superior proposal, Renasant may proceed with the superior proposal.

Board Recommendation. Except to the extent as it would cause the directors on Renasant's board of directors to breach their fiduciary duties under applicable law, the board of directors of Renasant, and any committee of the board, shall not

- withdraw, modify or qualify, or publicly propose to do any of the foregoing, in any manner adverse to Peoples, the approval of the merger agreement and the plan of merger or its recommendation to stockholders of the approval of the merger agreement and plan of merger, or take any action or make any statement in connection with the special meeting inconsistent with such approval or its recommendation to the stockholders of Renasant (collectively, a "change in recommendation");
- approve or recommend, or publicly propose to do either, any acquisition proposal; or
- cause Renasant to enter into any letter of intent, agreement in principal or similar agreement related to an acquisition transaction.

A change in recommendation shall also include the failure by Renasant's board of directors to recommend against an unsolicited bona fide written proposal to engage in an acquisition transaction.

Special Meeting. The merger agreement requires Renasant to call a meeting of its stockholders and to submit the merger agreement and the merger to its stockholders for approval. Subject to the board of directors adherence to its fiduciary duties under applicable law, the board shall recommend to Renasant stockholders the approval of the merger agreement and the plan of merger. Renasant will use its best efforts to obtain such shareholder approval. Renasant is not required to hold the special meeting or recommend the approval of the merger agreement and the plan of merger if the board of directors determines in good faith that holding such special meeting and making the recommendation (or failing to withdraw, modify or change such recommendation) would constitute a breach of its fiduciary duties under applicable law.

Access to Information. The merger agreement requires that, until completion of the merger, each of Renasant and Peoples must keep the other party advised of all material developments relevant to its business and the businesses of its subsidiaries. Each party shall provide to the other, upon request, reasonable details of any such development. Upon reasonable notice, Renasant shall afford to representatives of Peoples access, during normal business hours during the period prior to the closing date of the merger, to all of the properties, books, contracts, commitments and records of Renasant and the Renasant subsidiaries. During such period, Renasant also shall make available all information concerning its businesses as may be reasonably requested. Each of Renasant and Peoples agrees that it will, and will cause each of its directors, officers, attorneys and advisors to, maintain the confidentiality of all confidential information obtained from the other party. In the event of a termination of the merger agreement, each party will return to the other party upon request all confidential information previously furnished in connection with the transactions contemplated by the merger agreement.

Employee Matters. Each employee of Renasant or Renasant Bank who becomes an employee of Peoples or a Peoples subsidiary immediately following the closing date of the merger will be covered by the employee benefits plan of such employee's employer on substantially the same basis as other employees of the employer performing services in a comparable position. Notwithstanding the foregoing, such employer may

- continue any of the Renasant benefit plans for transferred employees in lieu of offering participation in the its benefit plans providing similar benefits (e.g., medical and hospitalization benefits);
- · terminate or amend any of Renasant's benefit plans; or
- merge any such benefit plans with its own employer benefit plans.

Under any such action, the transferred employees must receive benefits substantially similar to the benefits provided to employees of Peoples generally.

Transferred employees' service with Renasant or Renasant Bank shall be recognized as service with Peoples or a Peoples subsidiary, as the case may be, for purposes of eligibility to participate and vesting under the benefit plans, policies or arrangements. Peoples has agreed that any preexisting condition, limitation or exclusion in its medical, long-term disability and life insurance plans shall not apply to transferred employees or their covered dependents who are covered under a medical or hospitalization indemnity plan maintained by Renasant or Renasant Bank on the closing date of the merger and who then change coverage to the Peoples medical or hospitalization indemnity health plan. Nothing shall restrict the ability of Peoples Merger Corporation, or any other Peoples employer, to amend or terminate Renasant employee benefit plans in accordance with their terms. Such amendment or termination must affect all covered employees equally.

As of the closing date of the merger, Peoples shall assume and honor in accordance with their terms all severance agreements existing immediately prior to the closing date between Renasant or Renasant Bank and any officer which have been disclosed to Peoples in the merger agreement.

Following the closing date of the merger, Peoples shall maintain any deferred compensation plan, policy, program or arrangement disclosed to Peoples in the merger agreement, provided that Peoples may determine to continue, amend, or merge or consolidate any such plan, policy, program or arrangement

with any similar arrangement sponsored by Peoples. On the closing date, Renasant has agreed to terminate all qualified benefits plans and to distribute amounts under the plans in accordance with and at the times permitted under such benefits plans and under applicable law.

Directors' and Officers' Insurance and Indemnification. The merger agreement provides that for a period of 6 years following the closing date of the merger Peoples will indemnify and hold harmless the current and former directors and officers of Renasant and Renasant Bank disclosed to Peoples in the merger agreement, and their heirs, personal representatives and estates. Peoples will indemnify such individuals against, and shall advance or reimburse any and all costs and expenses of, any judgments, interest, fines, damages or other liabilities, or amounts paid in settlement, as such are incurred in connection with any claim, action, suit or proceeding based upon or arising from the indemnified party's capacity as an officer or director of Renasant or Renasant Bank. The indemnification will be provided to the same extent as such directors or officers would be indemnified under the articles of incorporation or bylaws of Peoples as if they were directors or officers of Peoples.

Peoples will only indemnify such individuals for indemnification claims aggregating \$2,000,000 (and Peoples has no responsibility as to how such total sum is allocated among that group). No director or officer will be indemnified by Peoples unless he or she has executed a joinder agreement with Peoples. The joinder agreement grants Peoples the right to participate in or completely assume the defense of such officer or director. The agreement also requires such officer or director to cooperate in the defense of any action for which indemnification is sought. Any amounts owed by Peoples pursuant to its indemnification obligations will be reduced by any amounts that an indemnified party receives from any third party.

Peoples has also agreed to indemnify and hold harmless Renasant, Renasant Bank and each of the directors, officers and controlling persons of either against any losses, claims, damages or liabilities arising under the Securities Act of 1933, as amended. Peoples will indemnify such individuals only insofar as such losses, claims, damages or liabilities arise out of or are based upon an untrue statement or alleged untrue statement of a material fact contained in this proxy statement/prospectus, or in any amendment or supplement, or arising out of or based upon the omission or alleged omission to state in any such document a material fact required to be stated or necessary to make the statements in such document not misleading. Peoples will pay or promptly reimburse such person for any legal or other expenses reasonably incurred in connection with investigating or defending any such action or claim. Peoples, however, is not obligated to indemnify such persons with respect to any such loss, claim, damage or liability (or actions in respect of any of the foregoing) which arises out of or is based upon any untrue statement or alleged untrue statement or omission or alleged omission made in this proxy statement/prospectus, or in any amendment or supplement, in reliance upon information furnished to Peoples by Renasant or Renasant Bank for use in this proxy statement/prospectus. Peoples is entitled to participate in the defense of any action. If Peoples assumes the defense with counsel satisfactory to the indemnified party, after notice to the indemnified party of its election to assume the defense, Peoples will not be liable for any legal or other expenses incurred by the indemnified party. The indemnification provided in connection with actions arising under the Securities Act of 1933, as amended, is not subject to the aggregate \$2,000,000 limit described above.

The merger agreement also provides that Peoples shall use reasonable efforts to cause Peoples Merger Corporation or Peoples to obtain for a period of six years after the closing date of the merger policies of directors' and officers' liability insurance. The policy must cover acts or omissions occurring prior to the closing date of the merger for certain directors and officers of Renasant on terms and in amounts no less favorable than those in effect on the date of the merger agreement. However, neither Peoples nor Peoples Merger Corporation is required to pay an aggregate premium for such insurance coverage in excess of 200% of the amount for such coverage as currently held by Renasant (which is \$8,000,000), but in such case shall purchase as much coverage as reasonably practicable for such amount.

Accounting Policies. As soon as all conditions to the consummation of the merger have been satisfied or waived, and consummation of the transactions contemplated by the merger agreement seems reasonably assured, and in any event prior to the closing date of the merger, Renasant shall, and it shall cause Renasant Bank to, adopt the accounting procedures and policies of Peoples.

Certain Agreements. Renasant has agreed to use its best efforts to cause each "affiliate" of Renasant to agree not to dispose of any shares of Peoples common stock received in the merger except in compliance with the Securities Act of 1933, as amended, and the rules and regulations under such statute. An "affiliate" of Renasant is any person controlling, controlled by or under common control with Renasant. Each director and executive officer of Renasant signed an agreement with Peoples on or prior to the closing date of the merger agreement obligating such person to vote his or her shares of Renasant common stock in favor of the merger agreement and the merger. The agreement signed by the non-employee directors contains two other features. The non-employee director is obligated not to compete with Peoples as provided in the agreement. Peoples also has agreed to have such directors appointed to the board of Peoples Merger Corporation for at least two years following the merger and to pay these directors the fees that they would have earned for serving on the Renasant board. Copies of the form of these agreements are annexed to this proxy statement/prospectus as Annex B-1 and Annex B-2.

Certain Adjustments. If prior to the closing date of the merger, the outstanding shares of Peoples common stock are increased, decreased or changed into a different number of shares by reason of any reorganization, reclassification, recapitalization or combination, stock split, reverse stock split, stock dividend or rights issued in respect of such stock, or any similar event occurs, the merger consideration will be adjusted accordingly to provide to the Renasant common stockholders merger consideration with the same economic value as contemplated by the merger agreement prior to such event.

Conditions to the Completion of the Merger

The respective obligations of the parties under the merger agreement to complete the merger shall be subject to the fulfillment at or prior to the closing date of the following conditions:

- All necessary regulatory or governmental approvals and consents required to complete the merger shall have been obtained, all conditions required to
 be satisfied prior to the closing date of the merger by the terms of such approvals and consents shall have been satisfied, and all waiting periods shall
 have expired.
- all notices, reports and other filings required to be made with any governmental entity prior to the closing date by Peoples or Renasant or any of their respective subsidiaries shall have been made and become final.
- Stockholders of Renasant shall have approved the merger agreement and the merger.
- None of Peoples, Peoples Bank, Renasant or Renasant Bank shall be subject to any governmental or judicial enactment or order which prohibits, restricts or makes illegal completion of the merger.
- All consents or approvals of all persons other than governmental entities required for consummation of the merger shall have been obtained (unless the
 failure to obtain any such consent or approval would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on
 Renasant or Peoples).

- This proxy statement/prospectus shall have become effective, no stop order suspending its effectiveness shall have been issued, and no proceedings for that purpose shall have been initiated or threatened by the Securities and Exchange Commission.
- Renasant and its stockholders shall have received an opinion of Phelps Dunbar LLP, satisfactory in form and substance to Peoples and Renasant, stating that if the merger is consummated in accordance with the terms described in this proxy statement/prospectus, (i) it will constitute a tax-free reorganization within the meaning of Section 368 of the Internal Revenue Code and (ii) the exchange of Renasant common stock to the extent exchanged for Peoples common stock will not give rise to gain or loss to the stockholders of Renasant with respect to such exchange.
- The shares of Peoples common stock issuable in the merger shall have been approved for listing on the American Stock Exchange on or before the closing date, subject to official notice of issuance.

The obligations of Peoples and Peoples Merger Corporation under the merger agreement to complete the merger are also subject to the fulfillment, at or prior to the closing date of the merger, of the following conditions (which may be waived by Peoples to the extent permitted by law):

- Peoples shall have received a certificate from an officer of Renasant that (i) all obligations of Renasant to be performed prior to the closing date of the merger have been performed and complied with in all material respects, and (ii) Renasant's representations and warranties are true and correct in all respects as of the closing date of the merger as though made at that time (except that those which specifically relate to an earlier date shall be true and correct as of such date), except for any violations or breaches which would not have a material adverse effect on Renasant.
- All permits, consents, waivers, clearances, approvals and authorizations of all governmental entities or third parties necessary to consummate the
 merger shall have been obtained, none of which shall contain any terms or conditions which would materially impair the value of Renasant to Peoples.
- Renasant stockholders who dissent from the merger shall not hold more than 15% of the outstanding shares of Renasant common stock immediately
 prior to the merger.
- · Two qualified directors of Renasant acceptable to Peoples shall be appointed by Peoples to Peoples board of directors on the closing date.

The obligations of Renasant under the merger agreement to complete the merger are also subject to the fulfillment, at or prior to the closing date of the merger, of the following conditions (any one or more of which may be waived by Renasant to the extent permitted by law):

• Renasant shall have received a certificate from an officer of Peoples that (i) all obligations of Peoples and Peoples Merger Corporation required to be performed prior to the closing date of the merger have been performed and complied with in all material respects, and (ii) the representations and warranties of Peoples and Peoples Merger Corporation are true and correct in all respects as of the closing date as though made at that time (except that those which specifically relate to an earlier date shall be true and correct as of such date), except for any violations or breaches which would not have a material adverse effect on Peoples.

All permits, consents, waivers, clearances, approvals and authorizations of all governmental entities or third parties necessary to consummate the
merger shall have been obtained, none of which shall adversely affect the merger consideration.

Termination of the Merger Agreement

The merger agreement may be terminated at any time prior to the closing date of the merger, whether before or after approval of the merger agreement and the merger by Renasant's stockholders:

- by mutual written consent of Peoples and Renasant;
- by Peoples or Renasant if:
 - (i) the closing date of the merger shall not have occurred on or prior to August 31, 2004, unless delayed because approval by a governmental entity is pending and has not been finally resolved, in which event such date shall be automatically extended to October 31, 2004; or (ii) Renasant's stockholders do not approve the merger agreement and merger at the special meeting. A party may not terminate the merger agreement if either of the above two events occurs because of that party's failure to perform or observe its agreements at or before the closing date or such vote, as the case may be;
 - any application for regulatory or governmental approval is denied or withdrawn at the request or recommendation of the governmental entity, but termination may not occur earlier than 30 days after such denial or withdrawal, unless within such 30-day period a petition for rehearing or an amended application is filed. A party may terminate 30 or more days after a petition for rehearing or an amended application is denied. No party may terminate when the denial or withdrawal is due to that party's failure to observe or performs its covenants or agreements;
 - any governmental entity shall have issued a final, non-appealable order prohibiting the completion of the merger;
 - Peoples shall have elected not to make the adjustments necessary in the cash portion and stock portion of the merger consideration so that the merger qualifies as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock to be exchanged for shares of Renasant common stock in the merger; or
 - there has been a breach by the other party of (i) any covenant or undertaking contained in the merger agreement or (ii) any representation or warranty of the other party contained in the merger agreement preventing such party from satisfying a condition to closing in the merger agreement, and such breach has not been cured within thirty days following delivery of written notice of the breach.
- · by Peoples if:
 - Renasant's board of directors fails to make, or withdraws, modifies or qualifies, or proposes publicly to withdraw, modify or qualify, the recommendation in this proxy statement/prospectus that Renasant's stockholders vote to adopt and approve the merger agreement and the merger;

- the special meeting to approve the merger agreement and plan of merger is not called or convened by Renasant;
- · Renasant approves or recommends, or publicly proposes to approve or recommend, an acquisition proposal by a third party; or
- Renasant stockholders dissenting from the merger hold more than 15% of the outstanding shares of Renasant common stock immediately prior to the merger.

· by Renasant:

- if the board of directors of Renasant determines in good faith, after consultation with outside counsel, that the failure to terminate the merger
 agreement and accept an acquisition proposal from a third party would constitute a breach of its fiduciary duties under applicable law;
- upon the occurrence of both of the following events:
 - An adjustment was made to qualify the merger as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock received in exchange for shares of Renasant common stock; and
 - (i) the quotient obtained by dividing the "average closing price" (defined below) by the closing price of a share of Peoples common stock on February 23, 2004 is less than (ii) the quotient obtained by dividing the index value for the NASDAQ Bank Index on the "determination date" (as defined below) by such index value on February 23, 2004, less 0.15.

Closing prices are determined on a per share basis. "Average closing price" means the average of the final sales prices of Peoples common stock on each of the 10 consecutive full trading days ending at the close of trading on the determination date. "Determination date" means the later of the date of (i) receipt by Peoples of the consent of the Federal Reserve Board to the merger (disregarding any waiting period) and (ii) the special meeting occurs.

Termination Fee

There are three sets of circumstances under which Renasant may owe Peoples a termination fee. In all cases, the termination fee is \$5,000,000.

Under the first set of circumstances, prior to any event allowing either party to terminate the merger agreement, an acquisition proposal must have been publicly announced or otherwise made known to Renasant's senior management, board of directors or stockholders generally and not have been irrevocably withdrawn more than five business days prior to the special meeting. Next, the merger agreement must have been terminated either (i) by Peoples or Renasant, because Renasant's stockholders failed to approve the merger agreement and the plan of merger or (ii) by Peoples, because of a willful breach by Renasant of any covenant, undertaking, representation or warranty contained in the merger agreement. In such event, if the acquisition transaction is consummated within 12 months of the termination of the merger agreement, then on the date of such consummation Renasant must pay Peoples a fee equal to \$5,000,000 by wire transfer of same-day funds.

Alternatively, if Peoples terminates the merger agreement because Renasant either (i) failed to recommend to its stockholders the approval of the merger agreement and the merger, (ii) effected a change in such recommendation, (iii) failed to call or convene the special meeting, or (iv) approved or recommended, or proposed publicly to do the foregoing, any acquisition transaction, then Renasant is required to pay Peoples \$5,000,000 by wire transfer of same-day funds. A termination under these circumstances is not effective until Peoples receives such funds.

Finally, if Renasant terminates the merger agreement because Peoples has not matched or bettered an acquisition proposal and Renasant's failure to terminate the merger agreement would constitute a breach of the fiduciary duties of its board of directors under applicable law, then Renasant must pay Peoples \$5,000,000 by wire transfer of same-day funds. As above, a termination under these circumstances is not effective until Peoples receives such funds. In no event shall Renasant be required to pay the \$5,000,000 fee under both these facts and the first set of circumstances.

If Renasant fails promptly to pay termination fee and Peoples sues for such fee and wins a judgment against Renasant, Renasant must also pay to Peoples its costs and expenses (including reasonable attorneys' fees and expenses) in connection with such suit.

COMPARISON OF RIGHTS OF STOCKHOLDERS OF RENASANT AND PEOPLES

This section of the proxy statement/prospectus describes material differences between the current rights of the holders of Renasant common stock and rights those shareholders will have as Peoples shareholders following the merger. The following discussion is intended only to highlight material differences between the rights of corporate shareholders under Mississippi law and Tennessee law generally and specifically with respect to Renasant stockholders and the holders of Peoples common stock pursuant to the respective charters and bylaws of Renasant and Peoples. The discussion does not constitute a complete comparison of the differences between the rights of such holders or the provisions of the TBCA, the Mississippi Business Corporation Act (the "MBCA"), Peoples' articles of incorporation, Peoples' bylaws, Renasant's charter and Renasant's bylaws.

The rights of the holders of Renasant common stock are governed by Tennessee law, Renasant's charter and Renasant's bylaws. Upon completion of the merger, the rights of the holders of Renasant common stock who become Peoples shareholders as a result of the merger will be governed by Mississippi law and by Peoples' articles of incorporation and Peoples' bylaws.

Renasant

board of directors possesses discretion to designate the series of

any preferred stock, the number of shares of such

Description of common stock	Peoples is authorized to issue 15,000,000 shares of common stock, par value \$5.00 per share. Holders of Peoples common stock are entitled to one vote per share.	Renasant is authorized to issue 2,000,000 shares of common stock, par value \$1.00 per share. Holders of Renasant common stock are entitled to one vote per share.
Description of preferred stock	Peoples' articles of incorporation do not contain any provision authorizing preferred stock.	Renasant is authorized to issue 1,000,000 shares of preferred stock. No preferred stock has been issued by Renasant. The

Peoples Renasant

Liability of shareholders

There is no provision in the MBCA regarding the liability of an assignee or transferee of shares or of a person holding shares as collateral security.

the voting rights, if any; and conversion rights, if any.

Under the TBCA, any person becoming an assignee or transferee of shares or of a subscription for shares in good faith and without knowledge or notice that the full consideration for such shares has not been paid shall not be personally liable for any unpaid portion of such consideration. The transferor shall remain liable for the consideration, whether such assignment or transfer be voluntary or involuntary. No person holding shares in any corporation as collateral security shall be personally liable as a shareholder, but the person pledging such shares shall be considered the holder thereof and shall be so liable. No executor,

administrator, guardian, trustee, or other fiduciary shall be personally liable as a shareholder, but the estate and funds in the hands of such executor, administrator, guardian, trustee, or other

fiduciary shall be liable.

series, and the relative rights, preferences and limitations of all shares of preferred stock, including: the annual dividend rate and whether such dividends are cumulative; the terms, conditions and price for any redemption; the liquidation preferences, if any;

Voting by Holders of Debt Obligations

The MBCA contains no provisions regarding the grant of voting rights to holders of debt obligations.

A corporation may in its charter confer upon the holders of any bonds, debentures or other debt obligations the power to vote in respect of its corporate affairs and management of the corporation to the extent and in the manner provided in the charter. A corporation may confer upon such holders of bonds, debentures or other debt obligations the same right of inspection of its books, accounts and other records, and also any other rights, which the shareholders of the corporation have. If and to the extent the charter so provides, such holders shall be deemed to be shareholders, and their bonds, debentures or other debt obligations shall be deemed to be shares of stock, for the

Voting Agreements

An agreement among the shareholders of a corporation is effective among the shareholders and the corporation even though it is inconsistent with the MBCA in that it (i) eliminates the board of directors or restricts the discretion or powers of the board of directors, (ii) governs the authorization or making of distributions whether or not in proportion to ownership of shares, (iii) establishes who shall be directors or officers of the corporation, or their terms of office or manner of selection or removal, (iv) governs, in general or in regard to specific matters, the exercise or division of voting power by or between the shareholders and directors or by or among any of them, including use of weighted voting rights or director proxies, (v) establishes the terms and conditions of any agreement for the transfer or use of property or the provision of services between the corporation and any shareholder, director, officer or employee of the corporation or among any of them, (vi) transfers to one or more shareholders or other persons all or part of the authority to exercise the corporate powers or to manage the business and affairs of the corporation, (vii) requires dissolution of the corporation at the request of one or more of the shareholders or upon the occurrence of a specified event or contingency, or (viii) otherwise governs the exercise of the corporate powers or the management of the business and affairs of the corporation or the

purpose of any provision which requires the vote of shareholders as a prerequisite to any corporate action. The charter may divest the holders of shares of capital stock, in whole or in part, of their right to vote on any corporate matter whatsoever.

Renasant's charter currently has no provision granting voting rights to holders of bonds, debentures or other debt obligations.

No written agreement to which all or less than all the shareholders have actually assented, whether embodied in the charter or bylaws or in any agreement in writing signed by all the parties, shall be invalid as between the parties to such agreement on the ground that it is an attempt by the parties to restrict the discretion of the board of directors in its management of the business of the corporation, to treat the corporation as if it were a partnership or to arrange their relationships in a manner that would be appropriate only between partners.

relationship among the shareholders, the directors and the corporation, or among any of them, and is not contrary to public policy.

Such an agreement shall be set forth in the articles of incorporation or bylaws and approved by all persons who are shareholders at the time of the agreement, or in a written agreement that is signed by all persons who are shareholders at the time of the agreement and is made known to the corporation. The agreement must provide that it is subject to amendment only by all persons who are shareholders at the time of the amendment, unless the agreement provides otherwise. An agreement shall be valid for ten (10) years, unless the agreement provides otherwise.

If at the time of the agreement the corporation has shares outstanding represented by certificates, the corporation shall recall the outstanding certificates and issue substitute certificates that comply with this subsection. The failure to note the existence of the agreement on the certificate or information statement shall not affect the validity of the agreement or any action taken pursuant to it.

A shareholders agreement shall cease to be effective when shares of the corporation are listed on a national securities exchange or regularly traded in a market maintained by one or more members of a national or affiliated securities association.

The existence or performance of a shareholders agreement shall not be a ground for imposing personal liability on any shareholder for the acts or debts of the corporation even if the agreement or its performance treats the corporation as if it were a partnership or results in failure to observe the corporate formalities otherwise applicable to the matters governed by the agreement.

Peoples Renasant

Nomination of Directors

The charter of the Nominating Committee of Peoples provides that shareholders may recommend to the nominating committee individuals to be nominated for election to the board of directors. The shareholder must submit such recommendation to the Secretary of Peoples at the address of Peoples' headquarters. Shareholder recommendations of candidates for election of directors must be submitted to Peoples by the date by which shareholder proposals must be submitted to Peoples for inclusion in its definitive proxy statement. Such recommendations must set forth the name, age, address, occupation, ownership of Peoples common stock, any other information required by the proxy rules under the federal securities laws, and the nominee's consent to be nominated as a candidate and serve if elected. The Peoples nominating committee will apply the same criteria regarding whether a shareholder-recommended candidate should be accepted as it applies to candidates recommended from other

Neither the charter nor the bylaws of Renasant address the issue of nominations for directors.

Number of directors

The articles of incorporation of Peoples provides that the number of directors of Peoples for each year shall be fixed by resolution adopted by the affirmative vote of a majority of the entire board of directors. The number of directors must be fixed between seven and twenty directors. The number of directors is currently fixed at sixteen.

The bylaws of Renasant provide that the number of directors shall be determined from time to time by resolution of the board of directors. The number of directors is currently fixed at ten.

Classified board of directors

The articles of incorporation of Peoples divides the board of directors into three classes (designated Class 1, Class 2 and Class 3), each class to be equal in number as nearly as possible. Elections of directors are staggered such that only one class of directors is up for election at each year's annual meeting. Each class of directors serves for a three year term.

The board of directors of Renasant is not classified. All directors must be elected at each year's annual meeting.

Classification of directors makes it more difficult for stockholders to change the composition of the board of directors.

The affirmative vote of at least 80% of all outstanding shares of Peoples common stock is required to alter, amend, repeal or adopt any provision affecting the classification of Peoples' board of directors.

Election of directors

Because only one class of Peoples directors comes up for election at any year's annual meeting, at least two annual meetings of shareholders will generally be required to change a majority of the board of directors. If Peoples was confronted by a shareholder attempting to force a proxy contest, a tender or exchange offer or other extraordinary corporate transaction, the extended time period required to replace a majority of the board of directors would allow the board sufficient time to review the proposal, to review available alternatives to the proposal, and to act in what it believes to be the best interest of Peoples' shareholders. At the same time, though, a classified board of directors may have the effect of deterring such proposals or make them less likely to succeed, making it more difficult to replace management.

The MBCA provides that shareholders have a right to cumulate their votes for directors for any corporation incorporated prior to July 1, 2002. Peoples was incorporated before such date, and shareholders have the right to cumulate their votes for directors.

Indemnification

Under Peoples' bylaws, any person, his heirs, executors or administrators, may be indemnified by Peoples for reasonable expenses (including judgments and compromise settlements, except as in a derivative suit situation where any judgment in the matter would run in favor of the corporation) actually incurred in Since all of Renasant's directors are elected at each year's annual meeting, a majority of the board of directors can be replaced at one annual meeting.

The TBCA provides that shareholders do not have the right to cumulate their votes for directors unless such right is contained in the charter of the corporation. Renasant's charter is silent regarding cumulative voting, and thus holders of Renasant common stock do not have the right to cumulate their votes for directors.

The provisions of the TBCA regarding the indemnification of directors and officers of a corporation are not materially different from the provisions in the MBCA regarding the indemnification of directors and officers, as described with respect to Peoples.

Peoples

connection with any action, suit or proceeding, civil or criminal, to which he was made a party by reason of service as a director, officer, or employee of Peoples. No person shall be so indemnified or reimbursed as to any matter as to which he shall finally be adjudged to have been guilty of gross negligence, willful misconduct or criminal acts in the performance of his duty to Peoples. Further, no person shall be so indemnified or reimbursed as to any matter in such action or suit which has been the subject of a compromise settlement except with the approval (i) of a court of competent jurisdiction, or (ii) the holders of record of a majority of the outstanding shares of the corporation, or (iii) a majority of the corporation's board of directors, excluding members who are parties to the same or substantially the same suit or proceeding.

These rights are not exclusive of other rights to which such person may be entitled under the MBCA. The board of directors of the corporation may, in its discretion, purchase directors' and officers' liability insurance coverage to provide, in whole or in part, for such indemnification or reimbursement.

The MBCA empowers Peoples to indemnify an individual who is a party to a proceeding because he is a director against liability incurred in the proceeding if (i) he conducted himself in good faith, and he reasonably believed, in the case of conduct in his official capacity, that his conduct was in the best interest, and in all other cases, that his conduct was at least not opposed to the best interests of the corporation and, in the case of any criminal proceeding, he had no reasonable cause to believe his conduct was unlawful, or (ii) he engaged in conduct which broader indemnification has been made permissible or obligatory under a provision of the articles of incorporation as authorized by the MBCA.

Renasan

Renasant's charter provides that Renasant shall have the power to indemnify any director, officer, employee, agent or any other person serving at the request of Renasant in any capacity with another corporation, partnership, joint venture, trust or other enterprise to the fullest extent permitted by the TBCA, as it existed on the date the charter was filed and as it may be amended in the future. Any indemnification may continue as to any person who has ceased to be a director, officer, employee or agent and may inure to the benefit of the heirs, executors and administrators of such person.

The termination of a proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent is not, of itself, determinative that the director did not meet the relevant standard of conduct.

Unless ordered by a court pursuant to the MBCA, Peoples may not indemnify a director (i) in connection with a proceeding by or in the right of the corporation except for reasonable expenses incurred in connection with the proceeding if it is determined that the director has met the relevant standard of conduct under the MBCA or (ii) in connection with any proceeding with respect to conduct for which he was adjudged liable on the basis that he received a financial benefit to which he was not entitled, whether or not involving action in his official capacity.

The MBCA further provides that Peoples shall indemnify a director who was wholly successful, on the merits or otherwise, in the defense of any proceeding to which he was a party because he was a director of the corporation against reasonable expenses incurred by him in connection with the proceeding.

The MBCA also provides that Peoples may, before final disposition of a proceeding, advance funds to pay for or reimburse the reasonable expenses incurred by a director who is a party to a proceeding because he is a director. The director must deliver to Peoples (i) a written affirmation of his good faith belief that he has met the relevant standard of conduct described in the MBCA or that the proceeding involves conduct for which liability has been eliminated under a provision of the articles of incorporation as authorized by the MBCA and (ii) his written

undertaking to repay any funds advanced if he is not entitled to mandatory indemnification under the MBCA and is ultimately determined under the MBCA that he has not met the relevant standard of conduct described in the MBCA. The undertaking required must be an unlimited general obligation of the director but need not be secured. The undertaking may be accepted without reference to the financial ability of the director to make repayment. A corporation may not indemnify a director as described unless authorized (i) by the board of directors (v) if there are two or more disinterested directors, by a majority vote of all the disinterested directors (a majority of whom shall for such purpose constitute a quorum) or by a majority of the members of a committee of two or more disinterested directors appointed by such a vote, or (z) if there are fewer than two disinterested directors, by the vote necessary for action by the board in accordance with the MBCA, in which authorization directors who do not qualify as disinterested directors may participate, or (ii) by the shareholders, but shares owned by or voted under the control of a director who at the time does not qualify as a disinterested director may not be voted on the authorization.

Peoples may also indemnify and advance expenses to an officer of the corporation who is a party to a proceeding because he is an officer to the same extent as for a director.

Limitations on liability of directors

The articles of incorporation and bylaws of Peoples are silent on the subject of the limitation of a director's liability. The MBCA provides that a director shall not be liable to Peoples or its shareholders for any decision to take or not to take action, or any failure to take any action, as a director, unless the party asserting liability in a proceeding establishes certain information. The party must show that (A) the director was a party to

The charter of Renasant provides that no director of Renasant shall be personally liable to Renasant for monetary damages for breach of fiduciary duty as a director, to the fullest extent provided under Tennessee law, existing now or in the future.

Under the TBCA, a director is not liable for any action taken as a director, or any failure to take any action, if the director

or had a direct or indirect financial interest in a transaction and the transaction was not approved in accordance with the MBCA and (B) the challenged conduct consisted or was the result of: (i) action not in good faith, (ii) a decision which the director did not reasonably believe to be in the best interests of Peoples or as to which the director was not informed to an extent the director reasonably believed appropriate in the circumstances, (iii) a lack of objectivity due to the director's familial, financial or business relationship with, or a lack of independence due to the director's domination or control by, another person having a material interest in the challenged conduct which relationship or which domination or control could reasonably be expected to have affected the director's judgment respecting the challenged conduct in a manner adverse to Peoples, and after a reasonable expectation to such effect has been established, the director shall not have established that the challenged conduct was reasonably believed by the director to be in the best interests of the corporation, (iv) a sustained failure of the director to be informed about the business and affairs of Peoples, or other material failure of the director to discharge the oversight function, or (v) receipt of a financial benefit to which the director was not entitled or any other breach of the director's duties to deal fairly with Peoples and its shareholders that is actionable under applicable law.

performed the duties of the office in compliance with the following provisions. A director shall discharge all duties as a director, including duties as a member of a committee, (i) in good faith, (ii) with the care an ordinarily prudent person in a like position would exercise under similar circumstances, and (iii) in a manner the director reasonably believes to be in the best interest of the corporation. In discharging such duties, a director is entitled to rely on information, opinions, reports, or statements, including financial statements and other financial data, if prepared or presented by (i) one or more officers or employees of Renasant (or Renasant Bank) whom the director reasonably believes to be reliable and competent in the matters presented, (ii) legal counsel, public accountants, or other persons as to matters the director reasonably believes are within the person's professional or expert competence, or (iii) a committee of the board of directors of which the director is not a member, if the director reasonably believes the committee merits confidence. A director is not acting in good faith if the director has knowledge concerning the matter in question that makes reliance otherwise permitted unwarranted.

Business combination restrictions

Peoples' articles of incorporation contain a "fair price" provision. This provision is designed to deter an unfriendly acquisition. The provision provides that the affirmative vote of the holders of not less than 80% of the outstanding shares of all voting stock of Peoples and the affirmative vote of the holders of not less than 67% of the outstanding shares of voting stock held by stockholders other than the "controlling party" shall be

The TBCA's Investor Protection Act does not apply to Renasant because Renasant is a bank holding company. The TBCA's Business Combination Act does not apply to Renasant because Renasant does not have any shares of stock registered under Section 12(g) of the Securities Exchange Act of 1934, as amended. The TBCA's Control Share Acquisition Act does not apply to Renasant because neither Renasant's

Peoples

required for the approval or authorization of any merger, consolidation, sale, exchange or lease of all of the assets or of assets having a fair market or book value of 25 percent or more of the total assets of Peoples. These provisions only apply if such transaction involves a controlling party. A "controlling party" is a shareholder owning or controlling 20 percent or more of Peoples' voting stock at the time of the proposed transaction. These voting requirements are not applicable in such transactions in which (i) the cash or fair market value of the property, securities or other consideration to be received (which includes common stock of Peoples retained by its existing shareholders in such a transaction where Peoples is the surviving entity) per share by holders of Peoples common stock in such transaction is not less than the highest per share price (with appropriate adjustments for stock splits, recapitalizations and the like) paid by the controlling party in the acquisition of any of its holdings of Peoples common stock in the three years preceding the announcement of the proposed transaction or (ii) the transaction is approved by a majority of the entire board of directors. The affirmative vote of not less than 80% of the outstanding common stock of Peoples is required to amend or appeal this fair price provision.

The MBCA also contains laws relating to business combinations, the Shareholder Protection Act and the Control Share Act. Neither of these laws applies to Peoples because Peoples is a bank holding company.

Vote on extraordinary corporate transactions

Under the MBCA, a merger, share exchange, sale, lease, exchange or other disposal of all or substantially all of Peoples' assets or the dissolution of Peoples is approved if the votes cast in favor of the transaction exceed the votes cast against the transaction.

Renasant

charter nor its bylaws contain an express declaration that control share acquisitions respecting the shares of Renasant are governed by and subject to the provisions of the Control Share Acquisition Act. The TBCA's Greenmail Act does not apply to Renasant because Renasant does not have any shares of stock registered under Section 12(g) of the Securities Exchange Act of 1934, as amended.

Under the TBCA, a merger, share exchange, sale, lease, exchange or other disposal of all or substantially all of Renasant's assets or the dissolution of Renasant must be approved by a majority of all of the outstanding shares of Renasant common stock.

Dissenters' or appraisal rights

The MBCA entitles a shareholder to appraisal rights and to obtain the payment of the fair value of that holder's shares upon the occurrence of certain corporate actions, including a merger or share exchange. However, the MBCA provides that these appraisal rights are not available to the holders of shares which are listed on the American Stock Exchange, such as Peoples' shares. This exclusion does not apply in the following situations: (i) where the corporate action requires the holders of such shares to accept for their shares anything other than cash or shares of any class or series of shares of any corporation, or any other proprietary interest of any other entity, that is either a public company or has at least 2,000 shareholders and the outstanding shares have a market value of at least \$20,000,000, (ii) the shares or assets of Peoples are being acquired by a person owning 20% or more of the voting power of Peoples or having the power to elect 25% or more of the board of directors of Peoples, or (iii) the shares or assets of Peoples are being acquired by a senior executive or director of Peoples who will receive as a result of the corporate action a financial benefit not generally available to other holders of Peoples common stock, with some exceptions.

In the event that any of the above situations occur such that a holder of Peoples common stock is allowed to exercise rights to appraisal, such rights are not materially different from the dissenters' rights afforded a holder of Renasant common stock.

Under the MBCA, an amendment to the charter of Peoples is approved if the votes cast in favor of the amendment exceed the votes cast against the amendment.

Peoples' articles of incorporation, however, contain provisions requiring the affirmative vote of not less than 80% of

The dissenters' rights provided to a holder of Renasant common stock under Chapter 23 of the TBCA are described above under the caption "The Merger – Dissenters' Rights."

Under the TBCA, amendments to the charter must be approved by a majority of the votes entitled to be cast on the amendment by any voting group with respect to which the amendment would create dissenters' rights, with some exceptions for minor amendments.

Peoples Renasant

the outstanding common stock of Peoples to amend or repeal the provisions of the articles of incorporation that establish a classified board of directors or pertain to the fair price provisions of the articles of incorporation.

Bylaw Provisions

The MBCA does not contain any provision permitting the bylaws to increase the quorum or voting requirements for shareholders. Under the TBCA, if expressly authorized by the charter, the shareholders may adopt or amend a bylaw that fixes a greater quorum or voting requirement for shareholders than is required by the TBCA. The adoption or amendment of a bylaw that adds, changes, or deletes a greater quorum or voting requirement for shareholders must meet the same quorum requirement and be adopted by the same vote and voting groups required to take action under the quorum and voting requirement then in effect or proposed to be adopted, whichever is greater. A bylaw that fixes a greater quorum or voting requirement for shareholders under subsection may not be adopted, amended, or repealed by the board of directors.

The Renasant charter contains no such provision.

BUSINESS OF RENASANT BANCSHARES, INC.

Renasant Bancshares

Renasant is a registered bank holding company under the Bank Holding Company Act of 1956, as amended, and owns 100% of the outstanding capital stock of Renasant Bank. Renasant was incorporated under the laws of the State of Tennessee as a Tennessee corporation on September 1, 1998. Renasant's activities are conducted primarily through Renasant Bank.

Renasant Bank

Renasant Bank commenced business operations on May 3, 1999 in a permanent facility located at 2177 Germantown Road South, Germantown, Tennessee 38138. Renasant Bank has one full service branch located in eastern Shelby County and one loan production office located in DeSoto County, Mississippi.

Renasant Bank conducts a full-service banking business in its service area, emphasizing the banking needs of individuals and small to medium-sized businesses. Renasant Bank draws most of its customer deposits and conducts most of its lending transactions from and within a primary service area in the eastern area of Shelby County and western area of Fayette County, Tennessee, as well as the northern area of DeSoto County, Mississippi.

The principal business of Renasant Bank is to accept deposits from the public and to make loans and other investments. The principal sources of funds for Renasant Bank's loans and investments are demand, time, savings, money market deposits, retirement accounts and other deposits (including negotiable order of withdrawal or NOW accounts). In addition, Renasant Bank receives funds from amortization and prepayment of loans, sales to other lenders or institutions of loans or participations in loans, fees received from other lenders or institutions for servicing loans sold to such lenders or institutions, and borrowings. The principal sources of income for Renasant Bank are interest and fees collected on loans including fees received for servicing loans sold to other lenders or institutions, and, to a lesser extent, interest and dividends collected on other investments. The primary expenses of Renasant Bank are interest paid on savings and other deposits (including NOW accounts), interest paid on other borrowings by Renasant Bank, employee compensation, office expenses, and other overhead expenses.

The profitability of Renasant Bank depends primarily on the extent to which Renasant Bank's earnings, primarily interest collected on loans, exceed Renasant Bank's expenses, primarily interest paid by Renasant Bank on deposits used to fund its loans. The extent to which Renasant Bank will be able to produce and maintain a positive spread between its income and expenses will be affected, among other things, by the actions of its competitors and changes in applicable laws and regulations, monetary control policies, general economic conditions, and the extent to which Renasant Bank makes relatively short-term loans or makes long-term loans with variable or adjustable interest rates.

Renasant Bank is authorized to make both secured and unsecured loans to individuals, partnerships, corporations, and other entities. Renasant Bank's lending business consists principally of making consumer loans to individuals and commercial loans to medium-sized business and professional concerns. In addition, Renasant Bank makes real estate related loans, including construction loans for residential and commercial properties, and primary and secondary mortgage loans for the acquisition or improvement of personal residences. Pursuant to the Tennessee Banking Act, loans to any one person or corporation may not exceed certain limitations. The statutory limitations with respect to loans to a single entity restrict the ability of Renasant Bank to originate large commercial or construction and development loans, if it desires to do so. However, such loans may be originated in cooperation with other financial institutions which commit to purchase a participating interest in such loans originated by Renasant Bank.

Renasant Bank may also emphasize the extension of relatively short-term loans or, to the extent it makes long-term loans, the extension of loans with variable or adjustable interest rates, in an effort to retain the flexibility to maintain a positive spread between its interest income and expense. The extent to which Renasant Bank may be successful in implementing this lending policy, however, will depend in part upon consumer acceptance, the actions of Renasant Bank's competitors, and, to some extent, usury laws and other rules and regulations applicable to Renasant Bank.

Renasant Bank also engages in secondary market mortgage activities, which means that Renasant Bank obtains commitments from secondary market mortgage purchasers, such as the Federal National Mortgage Association, to purchase mortgage loans originated by Renasant Bank. Based on these commitments, Renasant Bank seeks to originate mortgage loans on terms corresponding to such commitments, and by this means Renasant Bank has been able and expects to continue to be able to generate fee income to supplement its interest income.

Under statutes and regulations applicable to Renasant Bank, Renasant Bank is authorized to invest without limit in obligations of the United States or any state or territorial government or any agency of such governments or in any securities that are guaranteed as to principal and interest by such governments. Renasant Bank may also invest, subject to certain limitations, in investment grade

securities issued by political subdivisions, and, subject to policies and regulations of the Tennessee Department of Financial Institutions ("TDFI"), in certain types of commercial paper and corporate debt securities. Renasant Bank concentrates its investment activities primarily in government guaranteed securities.

Renasant Bank is authorized to accept and pay interest on deposits from individuals, corporations, partnerships, and any other type of legal entity, including fiduciaries (such as private trusts), subject to the requirements of the TDFI and the Federal Deposit Insurance Corporation ("FDIC") regarding the manner in which such interest may be calculated and paid, and penalties for early withdrawals of deposits. Qualified deposits will be insured by the FDIC in an amount up to \$100,000.

Market Area

Renasant Bank's main office are located in Germantown, Shelby County, Tennessee, and its branch is located in Cordova, Shelby County, Tennessee. Renasant Bank has a loan production office in Hernando, DeSoto County, Mississippi. Renasant Bank's primary service area includes the eastern area of Shelby County and western area of Fayette County, Tennessee, as well as the northern part of DeSoto County, Mississippi. The banking industry has experienced substantial consolidation in the past several years. This consolidation has been reflected in the Memphis area with the merger of both large regional banks and smaller community banks. In addition, the consolidation of the industry has been reflected by ATM's and reduced-service branches in the Shelby County community in lieu of full-service branches.

Competition

Renasant Bank competes with other banks, savings and loan associations, credit unions, finance companies, mortgage companies, and money market mutual funds operating in Renasant Bank's primary service area and elsewhere. Offices affiliated with out-of-state financial institutions have entered Tennessee in recent years to offer limited financial services, including lending and deposit gathering activities. Also, new laws on interstate banking and branching permit banks and bank holding companies headquartered outside Tennessee to acquire Tennessee banks and bank branches and move into Shelby County. Competition for deposit and loan opportunities in Renasant Bank's market area is intense because of existing competitors, the accelerating pace of deregulation, and the geographic likelihood of expansion into the market area by other institutions.

Facilities

Main Office. Renasant Bank owns a two-story commercial facility (approximately 6,300 square feet) located in Germantown, Tennessee, which houses its main office. The facility includes a main banking floor with a drive-in teller area, inside teller spaces, a bookkeeping area and private offices.

Lending Office. Renasant Bank leases a building directly behind the main office. Renasant Bank conducts the following activities in this office: commercial lending, commercial real estate construction lending, residential construction lending, and mortgage loan originations. The building is approximately 3,200 square feet.

Cordova Branch. Renasant Bank's Cordova branch is located in a 4,000 square foot building in Cordova, Tennessee. The branch includes offices, teller stations, drive-in windows, and loan operations. The Company leases the real estate and subleases the space to Renasant Bank.

Loan Production Office. Renasant Bank leases a 1,500 square foot facility for its loan production office on the town square in Hernando, Mississippi.

Deposit Operations Facility. Renasant Bank leases a 2,200 square foot facility to conduct its deposit operations activities. The building is located at 7500 Capital Drive, Germantown, Tennessee 38138.

Employees

As of December 31, 2003, Renasant Bank employed 51 persons on a full-time equivalent basis including approximately 28 officers. Renasant has no employees.

Legal Proceedings

There are no pending legal proceedings to which Renasant or Renasant Bank is a party or of which any of their properties are subject. There are no such proceedings known to Renasant or Renasant Bank to be contemplated by any person or governmental authority.

Liability and Asset Management

Liabilities of Renasant Bank are represented almost entirely by customers' deposit balances. Interest-bearing accounts, including NOW, money market accounts, short-term savings and certificates of deposit totaled approximately 85% of the deposit balances as of December 31, 2003. The remaining deposit balances are in demand deposits (non-interest-bearing). In managing its liabilities, Renasant Bank attempts to attract customers who, assuming rates are competitive, will be inclined to maintain an ongoing relationship with Renasant Bank, and in the case of maturing savings and certificates of deposit, will tend to renew or reinvest with Renasant Bank. While large certificates of deposit (\$100,000 or more) are in certain circumstances issued, Renasant Bank tries to minimize its liability in this area due to the liquidity risk the withdrawal of such deposits can cause.

Assets of Renasant Bank consist primarily of the loan portfolio and investment portfolio. In order to maintain a high degree of liquidity and minimize fluctuations in the interest margin (the difference between interest income and interest expense), management endeavors to maintain loan and investment portfolios which will be generally similar to the maturity distribution of Renasant Bank's liabilities.

Renasant Bank invests the majority of its securities in readily marketable assets, such as Federal Funds and short-to-intermediate term issues of the United States government and its agencies. The majority of Renasant Bank's loans contain either a demand provision or a relatively short-term maturity. Interest rates on many loans are negotiated on a variable rate basis, especially if the loan maturity is in excess of one year. By pricing loans on a variable rate structure and by maintaining loan maturities and investment portfolios relatively short term, Renasant Bank expects to be able to maintain a relatively consistent interest rate margin.

Supervision and Regulation

In addition to the generally applicable state and federal laws governing businesses and employers, bank holding companies and banks are extensively regulated under both federal and state law. As a Tennessee insured state bank, Renasant Bank is under the supervision and regulation of the TDFI and the FDIC. The Company is regulated by the Federal Reserve. With few exceptions, state and federal banking laws have as their principal objective either the maintenance of the safety and soundness of the banking system or the protection of consumers or classes of consumers, rather than the specific protection of the shareholders of Renasant. References under this heading to applicable statutes and regulations are brief summaries, which do not purport to be complete and are qualified in their entirety by reference to

those statutes and regulations. We are unable to predict which fiscal or monetary policies, economic control or new federal or state legislation may impact the business and earnings of Renasant and Renasant Bank in the future. Any change in applicable law or regulation may have an adverse effect on the results of operations and financial condition of Renasant and Renasant Bank.

Renasant is a bank holding company registered under the provisions of the Bank Holding Company Act of 1956, as amended. Registered bank holding companies are subject to examination by the Federal Reserve and are required to file with the Federal Reserve annual reports and other information regarding its business operations and those of its subsidiaries. It is also required to obtain Federal Reserve approval prior to acquiring, directly or indirectly, ownership or control of any voting shares of any bank, if, after such acquisition, it would own or control, directly or indirectly, more than 5% of the voting stock of such bank unless it already owns a majority of the voting stock of such bank.

The Federal Reserve, pursuant to regulation and published policy statements, has maintained that a bank holding company must serve as a source of financial strength to its subsidiary banks. In adhering to the Federal Reserve policy, Renasant may be required to provide financial support to a subsidiary bank at a time when, absent such Federal Reserve policy, Renasant may not deem it advisable to provide such assistance.

The Attorney General of the United States may, within 15 days after approval by the Federal Reserve of an acquisition, bring an action challenging such acquisition under the federal antitrust laws, in which case the effectiveness of such approval is stayed pending a final ruling by the courts. Failure of the Attorney General to challenge an acquisition does not, however, exempt the holding company from complying with both state and federal antitrust laws after the acquisition is consummated or immunize the acquisition from future challenge under the anti-monopolization provisions of the Sherman Act.

Under the Reigle-Neal Interstate Banking and Branching Efficiency Act of 1994 ("Interstate Act"), Renasant or any other bank holding company located in Tennessee is able to acquire a bank located in any other state, and a bank holding company located outside Tennessee can acquire any Tennessee-based bank, in either case subject to certain deposit percentage and other restrictions. The legislation also provides that, unless an individual state has elected to prohibit out-of-state banks from operating interstate branches within its territory, adequately capitalized and managed bank holding companies will be able to consolidate their multistate bank operations into a single bank subsidiary and to branch interstate through acquisitions.

De novo branching by an out-of-state bank is permitted only if it is expressly permitted by the laws of the host state. The authority of a bank to establish and operate branches within a state remains subject to applicable state branching laws. Pursuant to the Interstate Act, the State of Tennessee adopted legislation that authorizes out-of-state banks to operate interstate branches which have been acquired through acquisition within its territory effective June 1, 1997.

On November 12, 1999, the Gramm-Leach-Bliley Financial Services Modernization Act was signed into law. The Financial Services Modernization Act eliminates the barriers erected by the 1933 Glass-Steagall Act and amends the Bank Holding Company Act of 1956, among other statutes. Further, it allows for the affiliation of banking, securities and insurance activities in new financial services organizations.

A dominant theme of that legislation is functional regulation of financial services, with the primary regulator of Renasant or its subsidiaries being the agency which traditionally regulates the activity in which Renasant or its subsidiaries wishes to engage. For example, the Securities and Exchange Commission will regulate bank securities transactions, and the various banking regulators will oversee banking activities.

The principal provisions of the Financial Services Modernization Act will permit Renasant, if it meets the standards for a "well-managed" and "well-capitalized" institution and has at least a "satisfactory" Community Reinvestment Act performance, to engage in any activity that is "financial in nature," including security and insurance underwriting, investment banking, and merchant banking investing in commercial and industrial companies. Renasant, if it satisfies the above criteria, can file a declaration of its status as a "financial holding company" ("FHC") with the Federal Reserve, and thereafter engage directly or through nonbank subsidiaries in the expanded range of activities which the Financial Services Modernization Act identifies as financial in nature. Further, Renasant, if it elects FHC status, will be able to pursue additional activities which are incidental or complementary in nature to a financial activity, or which the Federal Reserve subsequently determines to be financial in nature.

It is expected that the Financial Services Modernization Act will facilitate further consolidation in the financial services industry on both a national and international basis, and will cause existing bank holding companies to restructure their existing activities in order to take advantage of the new powers granted and comply with their attendant requirements and conditions.

A bank holding company which has not elected to become a FHC will generally be prohibited from acquiring control of any company which is not a bank and from engaging in any business other than the business of banking or managing and controlling banks. However, these non-FHC bank holding companies will still be able to engage in certain activities which have been identified by the Federal Reserve to be so closely related to banking as to be a proper incident thereto and thus permissible for bank holding companies.

The permissible nonbanking activities, including those listed below, are left unchanged by the Financial Services Modernization Act which does not prohibit non-FHC bank holding companies from engaging in these activities. Effective April 21, 1997, the Federal Reserve Board revised and expanded the list of permissible non-banking activities, which includes the following activities: extending credit and servicing loans; acting as investment or financial advisor to any person, with certain limitations; leasing personal or real property or acting as a broker with respect to such activity; providing management and employee benefits consulting advice and career counseling services to non-affiliated banks and nonbank depository institutions; operating certain nonbank depository institutions; performing certain trust company functions; providing certain agency transactional services, including securities brokerage services, riskless principal transactions, private placement services, and acting as a futures commission merchant; providing data processing and data transmission services; acting as an insurance agent or underwriter with respect to limited types of insurance; performing real estate appraisals; arranging commercial real estate equity financing; providing check-guaranty, collection agency and credit bureau services; engaging in asset management, servicing and collection activities; providing real estate settlement services; acquiring certain debt which is in default; underwriting and dealing in obligations of the United States, the states and their political subdivisions; engaging as a principal in foreign exchange trading and dealing in precious metals; providing other support services such as courier services and the printing and selling of checks; and investing in programs designed to promote community welfare.

In determining whether an activity is so closely related to banking as to be permissible for bank holding companies, the Federal Reserve is required to consider whether the performance of such activities by a bank holding company or its subsidiaries can reasonably be expected to produce benefits to the public such as greater convenience, increased competition and gains in efficiency, that outweigh such possible adverse effects as undue concentration of resources, decreased and unfair competition, conflicts of interest and unsound banking practices. Generally, bank holding companies are required to obtain prior approval of the Federal Reserve to engage in any activity not previously approved by the Federal Reserve or to modify in any material respect an activity for which Federal Reserve approval has been obtained.

Tennessee statutes and federal law regulate a variety of the banking activities of Renasant Bank, including required reserves, investments, loans, mergers and consolidations, issuance of securities, payment of dividends, and establishment of branches. There are certain limitations under federal and Tennessee law on the payment of dividends by banks. The payment of dividends by any bank is dependent upon its earnings and financial condition and is subject to the statutory power of certain federal and state regulatory agencies to act to prevent what they deem unsafe or unsound banking practices. The payment of dividends could, depending upon the financial condition of Renasant Bank, be deemed to constitute such an unsafe or unsound practice. Federal law prohibits a state bank, the deposits of which are insured by the FDIC, from paying dividends if it is in default in the payment of any assessments due the FDIC. The Federal Reserve Board has the power to prohibit dividends by bank holding companies if their actions constitute unsafe or unsound practices. The Federal Reserve Board has issued a policy statement on the payment of cash dividends by bank holding companies, which expresses the Federal Reserve Board's view that a bank holding company experiencing earnings weaknesses should not pay cash dividends that exceed its net income or that could only be funded in ways that weaken the bank holding company's financial health, such as by borrowing. Furthermore, the TDFI also has authority to prohibit the payment of dividends by a Tennessee bank when it determines such payment to be unsafe and unsound banking practice.

Renasant Bank also is subject to certain restrictions on loan amounts, interest rates, "insider" loans to officers, directors and principal shareholders, tie-in arrangements, and transactions with affiliates, as well as many other matters. Strict compliance at all times with state and federal banking laws is required.

Tennessee law contains limitations on the interest rates that may be charged on various types of loans, and restrictions on the nature and amount of loans that may be granted and on the types of investments which may be made. The operations of banks are also affected by various consumer laws and regulations, including those relating to equal credit opportunity and regulation of consumer lending practices. All Tennessee banks must become and remain insured banks under the Federal Deposit Insurance Act.

Under Tennessee law, state banks are prohibited from lending to any one person, firm or corporation amounts more than 15% of its equity capital accounts, except (i) in the case of certain loans secured by negotiable title documents covering readily marketable nonperishable staples or (ii) with the prior approval of Renasant Bank's Board of Directors or finance committee (however titled), Renasant Bank may make a loan to one person, firm or corporation of up to 25% of its equity capital accounts.

Congress enacted the Financial Institutions Reform, Recovery and Enforcement Act of 1989 ("FIRREA"), which provides that a depository institution insured by the FDIC can be held liable for any loss incurred by, or reasonably expected to be incurred by, the FDIC in connection with (i) the default of a commonly controlled FDIC insured depository institution or (ii) any assistance provided by the FDIC to a commonly controlled FDIC insured depository institution in danger of default. FIRREA provides that certain types of persons affiliated with financial institutions could be fined up to \$1 million per day for each violation of certain regulations related (primarily) to lending to and transactions with executive officers, directors, and principal shareholders, including the interests of these individuals. Other violations may result in civil money penalties of \$5,000 to \$25,000 per day or in criminal fines and penalties. In addition, the FDIC has been granted enhanced authority to withdraw or to suspend deposit insurance in certain cases.

The Federal Deposit Insurance Corporation Improvement Act of 1991 ("FDICIA") substantially revised the depository institution regulatory and funding provisions of the FDIA and made revisions to several other federal banking statutes. Among other things, FDICIA requires the federal banking regulators to take "prompt corrective action" in respect of FDIC-insured depository institutions that do not meet minimum capital requirements.

FDICIA generally prohibits an FDIC-insured depository institution from making any capital distribution (including payment of dividends) or paying any management fee to its holding company if the depository institution would after taking such action be undercapitalized. Undercapitalized depository institutions are subject to restrictions on borrowing from the Federal Reserve System. In addition, undercapitalized depository institutions are subject to growth limitations and are required to submit capital restoration plans. A depository institution's holding company must guarantee the capital plan, up to an amount equal to the lesser of 5% of the depository institution's assets at the time it becomes undercapitalized or the amount of the capital deficiency when the institution fails to comply with the plan. The federal banking agencies may not accept a capital plan without determining, among other things, that the plan is based on realistic assumptions and is likely to succeed in restoring the depository institution's capital. If a depository institution fails to submit an acceptable plan, it is treated as if it is significantly undercapitalized.

Significantly undercapitalized depository institutions may be subject to a number of requirements and restrictions, including orders to sell sufficient voting stock to become adequately capitalized, requirements to reduce total assets and cessation of receipt of deposits from correspondent banks. Critically undercapitalized depository institutions are subject to appointment of a receiver or conservator.

FDICIA contains numerous other provisions, including accounting, audit and reporting requirements, termination of the "too big to fail" doctrine except in special cases, limitations on the FDIC's payment of deposits at foreign branches, regulatory standards in such areas as asset quality, earnings and compensation and revised regulatory standards for, among other things, powers of state banks, real estate lending and capital adequacy. FDICIA also requires that a depository institution provide 90 days prior notice of the closing of any branches.

The FDIC has adopted regulations under FDICIA governing the receipt of brokered deposits. Under the regulations, a bank cannot accept a rollover or renew brokered deposits unless (i) it is well capitalized or (ii) it is adequately capitalized and receives a waiver from the FDIC. A bank that cannot receive brokered deposits also cannot offer "pass-through" insurance on certain employee benefit accounts. Whether or not it has obtained such a waiver, an adequately capitalized bank may not pay an interest rate on any deposits in excess of 0.75% over certain prevailing market rates specified by regulation. There are no such restrictions on a bank that is well capitalized. These brokered deposit regulations may have an affect on the funding or liquidity of Renasant Bank.

Renasant Bank is subject to regulatory capital requirements imposed by the FDIC and the TDFI. The FDIC has issued risk-based capital guidelines for banks which make regulatory capital requirements more sensitive to differences in risk profiles among banks to account for off-balance sheet exposure and to minimize disincentives for holding liquid assets. The Federal Reserve has imposed similar regulations for bank holding companies with consolidated assets of \$150 million or more. Assets and off-balance sheet items are assigned to broad risk categories each with appropriate weights. The resulting capital ratios represent capital as a percentage of total risk-weighted assets and off-balance sheet items.

A banking organization's qualifying total capital consists of two components: Tier I capital (core capital) and Tier II capital (supplementary capital). Tier I capital is an amount equal to the sum of: (i) common shareholders' equity (including adjustments for any surplus or deficit); (ii) non-cumulative

perpetual preferred stock; and (iii) the company's minority interests in the equity accounts of consolidated subsidiaries. Intangible assets generally must be deducted from Tier I capital, subject to limited exceptions for goodwill arising from certain supervisory acquisitions. Other intangible assets may be included in an amount up to 25% of Tier I capital, provided that the asset meets each of the following criteria: (i) the asset must be able to be separated and sold apart from the banking organization or the bulk of its assets; (ii) the market value of the asset must be established on an annual basis through an identifiable stream of cash flows and there must be a high degree of certainty that the asset will hold this market value notwithstanding the future prospects of the banking organization; and (iii) the banking organization must demonstrate that a liquid market exists for the asset. Intangible assets in excess of 25% of Tier I capital generally are deducted from a banking organization's regulatory capital. At least 50% of the banking organization's total regulatory capital must consist of Tier I capital.

Tier II capital is an amount equal to the sum of (i) the allowance for possible credit losses in an amount up to 1.25% of risk-weighted assets; (ii) cumulative perpetual preferred stock with an original maturity of 20 years or more and related surplus; (iii) hybrid capital instruments (instruments with characteristics of both debt and equity), perpetual debt and mandatory convertible debt securities; and (iv) in an amount up to 50% of Tier I capital, eligible term subordinated debt and intermediate-term preferred stock with an original maturity of five years or more, including related surplus. The inclusion of the foregoing elements of Tier II capital are subject to certain requirements and limitations of the FDIC. Investments in unconsolidated banking and finance subsidiaries, investments in securities subsidiaries and reciprocal holdings of capital instruments must be deducted from capital. The federal banking regulators may require other deductions on a case-by-case basis.

Under the risk-weighted capital guidelines, balance sheet assets and certain off-balance sheet items, such as standby letters of credit, are assigned to one of four risk weight categories (0%, 20%, 50%, or 100%) according to the nature of the asset and its collateral or the identity of any obligor or guarantor. For example, cash is assigned to the 0% risk category, while loans secured by one-to-four family residences are assigned to the 50% risk category. The aggregate amount of such asset and off-balance sheet items in each risk category is adjusted by the risk weight assigned to that category to determine weighted values, which are added together to determine the total risk-weighted assets for the banking organization. Accordingly, an asset, such as a commercial loan, which is assigned to the 100% risk category is included in risk-weighted assets at its nominal face value, whereas a loan secured by a single-family home mortgage is included at only 50% of its nominal face value. The application ratios are equal to capital, as determined, divided by risk-weighted assets, as determined.

The FDIC has issued a final regulation requiring banking organizations to maintain minimum Leverage Capital Ratios. The Leverage Capital Ratio is defined as Tier I capital minus intangible assets, divided by total assets. In order for an institution to operate at or near the minimum Tier I leverage capital requirement of 3%, the FDIC expects that such institution would have well-diversified risk, no undue rate risk exposure, excellent asset quality, high liquidity and good earnings. In general, Renasant Bank would have to be considered a strong banking organization, rated in the highest category under the bank rating system and have no significant plans for expansion. Higher Tier I leverage capital ratios of up to 5% will generally be required if all of the above characteristics are not exhibited, or if the institution is undertaking expansion, seeking to engage in new activities, or otherwise faces unusual or abnormal risks.

FDICIA establishes five capital tiers: "well capitalized," "adequately capitalized," "undercapitalized," "significantly undercapitalized" and "critically undercapitalized." Under applicable regulations, a FDIC-insured depository institution is defined to be "well capitalized" if it maintains a Leverage Ratio of at least 5%, a risk adjusted Tier 1 Capital Ratio of at least 6% and a Total Capital Ratio of at least 10% and is not subject to a directive, order or written agreement to meet and maintain specific

capital levels. An insured depository institution is defined to be "adequately capitalized" if it maintains a Leverage Ratio of at least 4% (or 3% subject to the conditions described above), a risk adjusted Tier 1 Capital Ratio of at least 4% and a Total Capital Ratio of at least 8%. An insured depository institution will be considered "undercapitalized" if it fails to meet any minimum required measure, "significantly undercapitalized" if it is significantly below such measure and "critically undercapitalized" if it fails to maintain a level of tangible equity equal to not less than 2% of total assets. An insured depository institution may be deemed to be in a capitalization category that is lower than is indicated by its actual capital position if it receives an unsatisfactory examination rating.

The FDIC rule provides that institutions not in compliance with the regulation are expected to be operating in compliance with a capital plan or agreement with the applicable regulatory authority. If they do not do so, they are deemed to be engaging in an unsafe and unsound practice and may be subject to enforcement action. Failure to maintain capital of at least 2% of assets constitutes an unsafe and unsound practice and may be subject to enforcement action. Failure to maintain capital of at least 2% of assets constitutes an unsafe and unsound condition justifying termination of FDIC insurance.

On October 26, 2001, the President signed the USA PATRIOT Act of 2001 into law. This act contains the International Money Laundering Abatement and Financial Anti-Terrorism Act of 2001 (the "IMLAFA"). The IMLAFA substantially broadens existing anti-money laundering legislation and the extraterritorial jurisdiction of the United States, imposes new compliance and due diligence obligations, creates new crimes and penalties, compels the production of documents located both inside and outside the United States, including those of foreign institutions that have a correspondent relationship in the United States, and clarifies the safe harbor from civil liability to customers. The U.S. Treasury Department has issued a number of regulations implementing the USA PATRIOT Act that apply certain of its requirements to financial institutions such as Renasant and its broker-dealer subsidiaries. The regulations impose new obligations on financial institutions to maintain appropriate policies, procedures and controls to detect, prevent and report money laundering and terrorist financing. The Treasury Department is expected to issue a number of additional regulations which will further clarify the USA PATRIOT Act's requirements. The IMLAFA requires all "financial institutions," as defined, to establish anti-money laundering compliance and due diligence programs no later than April 2002. Such programs must include, among other things, adequate policies, the designation of a compliance officer, employee training programs, and an independent audit function to review and test the program.

Effects of Governmental Policies

Renasant Bank's earnings are affected by the difference between the interest earned by Renasant Bank on its loans and investments and the interest paid by Renasant Bank on its deposits or other borrowings. The yields on its assets and the rates paid on its liabilities are sensitive to changes in prevailing market rates of interest. Thus, the earnings and growth of Renasant Bank will be influenced by general economic conditions, fiscal policies of the federal government, and the policies of regulatory agencies, particularly the Federal Reserve, which establishes national monetary policy. The nature and impact of any future changes in fiscal or monetary policies cannot be predicted.

Commercial banks are affected by the credit policy of various regulatory authorities, including the Federal Reserve. An important function of the Federal Reserve is to regulate the national supply of bank credit. Among the instruments of monetary policy used by the Federal Reserve to implement these objectives are open market operations in U.S. Government securities, changes in reserve requirements on bank deposits, and changes in the discount rate on bank borrowings. The Federal Reserve uses these means in varying combinations to influence overall growth of bank loans, investments and deposits, and also to affect interest rates charged on loans, received on investments or paid for deposits.

The monetary and fiscal policies of regulatory authorities, including the Federal Reserve, also affect the banking industry. Through changes in the reserve requirements against bank deposits, open market operations in U.S. Government securities and changes in the discount rate on bank borrowings, the Federal Reserve influences the cost and availability of funds obtained for lending and investing. No prediction can be made with respect to possible future changes in interest rates, deposit levels or loan demand or with respect to the impact of such changes on the business and earnings of Renasant Bank.

From time to time, legislation is enacted which has the effect of increasing the cost of doing business, limiting or expanding permissible activities, or affecting the competitive balance between banks and other financial institutions. For example, the Depository Institutions Deregulation and Monetary Control Act of 1980 (the "Deregulation Act") provided for the phasing out of restrictions on deposit interest rate ceilings, the authorization of new accounts and related services, and the expansion of the lending authority of savings and loan associations. The Deregulation Act has altered, to a certain extent, the competitive relationship that previously existed among financial institutions, and it may result in a substantial reduction in the historical distinction between the services offered by banks, savings and loan associations, and other financial institutions.

OWNERSHIP OF RENASANT COMMON STOCK BY CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following table sets forth certain information as of April 2, 2004 with respect to ownership of the outstanding common stock of Renasant by (i) all persons known to Renasant to own beneficially more than 5% of the outstanding shares of the common stock of Renasant (ii) each director of Renasant, (iii) each executive officer of Renasant, and (iv) all executive officers and directors of Renasant as a group.

Beneficial Owner	Amount and Nature of Beneficial Ownership ⁽¹⁾	Percent of Outstanding Shares ⁽¹⁾
John A. Bobango	25,833(2)	1.81%
Francis J. Cianciola	60,250(3)	4.16%
James K. Dickinson, P.E.	34,000(4)	2.38%
T. Michael Glenn	20,848	1.46%
Jack C. Johnson	65,023(5)	4.55%
Phillip R. Langsdon, M.D.	34,000(6)	2.38%
Michael J. McCarver	29,850(7)	2.06%
Michael R. Pera	17,636(8)	1.23%
Donald E. Russell	23,500(9)	1.63%
Bruce C. Taylor	10,000	0.70%
All executive officers and directors as a group (10 persons)	320,940	21.67%

Except as otherwise indicated, each person named in this table possesses sole voting and investment power with respect to the shares beneficially owned by such person. "Beneficial Ownership" includes shares for which an individual, directly or indirectly, has or shares voting or investment power or both. The percentages are based upon 1,429,697 shares outstanding plus the shares of common stock that the person has the right to acquire as of such date or within 60 days thereafter. Each individual listed on this table is a director of Renasant.

²⁾ Includes 2,000 shares held in an individual retirement account.

- Also Vice Chairman and Chief Executive Officer of Renasant. Consists of 40,000 shares held in an individual retirement account and 20,250 shares issuable upon the exercise of stock options that are exercisable within 60 days of the date of this proxy statement.
- (4) Includes 12,000 shares held in an individual retirement account.
- Also Chairman of Renasant. Includes 43,723 shares held in an individual retirement account, 4,050 held by a custodian for the benefit of Deanna D. Johnson and 11,900 shares held by Germantown Home Builders, Inc. Retirement Plan. Jack C. Johnson serves as trustee for the Retirement Plan. Jack C. Johnson and Deanna Johnson are married. Jack C. Johnson disclaims beneficial ownership of the shares held for the benefit of his spouse.
- (6) Consists of 34,000 shares held by Phillip R. Langsdon and Carol Langsdon, Tenants by the Entirety.
- (7) Also Secretary of Renasant. Includes 16,200 shares issuable to Michael J. McCarver and 1,350 shares issuable to Jeannine Nelson McCarver, his spouse, upon the exercise of stock options exercisable within 60 days of the date of this proxy statement/prospectus. Also includes 1,800 shares held in an individual retirement account for the benefit of Ms. McCarver. Michael J. McCarver disclaims beneficial ownership of the shares issued for the benefit of or issuable to Ms. McCarver.
- (8) Consists of 17,636 shares held by the Barbara G. Pera Revocable Living Trust. Michael R. Pera and Barbara Pera are married. Michael R. Pera disclaims beneficial ownership of the shares held in the Barbara G. Pera Revocable Living Trust.
- (9) Includes 6,167 shares held in an individual retirement account and 13,500 shares issuable upon the exercise of stock options exercisable within 60 days of the date of this proxy statement.

EXPERTS

Ernst & Young LLP, independent auditors, have audited the consolidated financial statements of Peoples included in The Peoples Holding Company Annual Report on Form 10-K for the year ended December 31, 2003, and 2002, and for each of the years in the three-year period ended December 31, 2003, as set forth in their report, which is incorporated by reference in this proxy statement/prospectus and elsewhere in the registration statement. Such financial statements are incorporated by reference in reliance on Ernst & Young LLP's report, given on their authority as experts in accounting and auditing.

LEGAL MATTERS

Certain legal matters with respect to the legality of the Peoples common stock to be issued pursuant to the merger will be passed upon by Phelps Dunbar LLP, Peoples outside legal counsel. The federal income tax consequences of the merger also will be passed upon for Peoples by Phelps Dunbar LLP. William M. Beasley, a partner of Phelps Dunbar LLP, is the Co-Chairman of the Board of Directors of Peoples. Phelps Dunbar LLP also provides legal advice to Peoples on a regular basis. As of the date of this proxy statement/prospectus, certain members of Phelps Dunbar LLP owned an aggregate of approximately 34,133 shares of Peoples common stock.

Baker, Donelson, Bearman, Caldwell & Berkowitz, P.C. has served as counsel for Renasant with respect to matters involving the merger.

WHERE YOU CAN FIND MORE INFORMATION

Peoples files annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any reports, statements or other information that Peoples files with the SEC at the SEC's public reference rooms at 450 Fifth Street, N.W., Washington, D.C. Please call the SEC at 1-800-SEC-0330 for further information on the public reference room. Peoples' SEC filings are also available to the public from commercial document retrieval services and at the Internet World Wide Web site maintained by the SEC at http://www.sec.gov. Reports, proxy statements and other information concerning Peoples are also found on Peoples' website, www.thepeoplesbankandtrust.com, under the link "Investor Relations".

Peoples filed a registration statement on Form S-4 to register with the SEC the shares of Peoples common stock to be issued to Renasant stockholders in the merger. This proxy statement/prospectus is a part of that registration statement and constitutes a prospectus of Peoples and a proxy statement of Renasant for the Renasant special meeting.

The SEC allows Peoples to "incorporate by reference" information into this proxy statement/prospectus, which means that Peoples can disclose important information to you by referring you to another document filed separately with the SEC. The information incorporated by reference is deemed to be part of this proxy statement/prospectus, except for any information superseded by information contained directly in the proxy statement/prospectus. This proxy statement/prospectus incorporates by reference the documents set forth below that Peoples has previously filed with the SEC. These documents contain important information about Peoples and its business.

Peoples SEC Filings (File No. 1-08918)

- 1. Annual Report on Form 10-K for the year ended December 31, 2003 filed on March 11, 2004;
- 2. Definitive Proxy Statement, filed on March 11, 2004;
- 3. Current Report on Form 8-K filed on January 22, 2004 announcing the financial results of Peoples for the quarter ended December 31, 2003;
- 4. Current Report on Form 8-K filed on February 18, 2004 announcing the signing of the definitive agreement to acquire Renasant;
- 5. Current Report on Form 8-K filed on February 20, 2004 filing the visual materials presented in the webcast on February 18, 2004 concerning the definitive agreement to acquire Renasant;
- Current Report on Form 8-K filed on February 27, 2004 announcing the declaration of a quarterly cash dividend on Peoples' common stock of \$.20 per share; and
- 7. The description of Peoples' common stock contained in Peoples' Registration Statement on Form S-4 filed with the SEC on February 17, 1999.

Peoples also incorporates by reference additional documents that it may file with the SEC between the date of this proxy statement/prospectus and the date of the Renasant special meeting. These include periodic reports, such as annual reports on Form 10-K, quarterly reports on Form 10-Q and current reports on Form 8-K, as well as proxy statements.

You can obtain any of the documents incorporated by reference in this proxy statement/prospectus, other than exhibits to those documents, without charge by requesting them in writing, as follows:

The Peoples Holding Company 209 Troy Street Tupelo, Mississippi 38802-0709 Attention: James W. Gray Telephone: (662) 680-1001

If you would like to request documents, please do so prior to _______, 2004, in order to receive them before the Renasant special meeting.

Renasant has supplied all information contained in this proxy statement/prospectus relating to Renasant, and Peoples has supplied all such information relating to Peoples.

You should rely only on the information contained or incorporated by reference in this proxy statement/prospectus to vote your shares at the Renasant special meeting. Renasant and Peoples have not authorized anyone to provide you with information that is different from what is contained in this proxy statement/prospectus. This proxy statement/prospectus is dated ________, 2004. You should not assume that the information contained in the proxy statement/prospectus is accurate as of any date other than that date, and neither the mailing of this proxy statement/prospectus to Renasant's stockholders nor the issuance of Peoples common stock in the merger will create any implication to the contrary.

ANNEX A-1
AGREEMENT AND PLAN OF MERGER
BY AND AMONG
THE PEOPLES HOLDING COMPANY,
PEOPLES MERGER CORPORATION
AND
RENASANT BANCSHARES, INC.

DATED FEBRUARY 17, 2004

TABLE OF CONTENTS

		Page
ARTICLE I	THE MERGER	5
1.1	The Merger	5
1.2	Effective Time	5
1.3	The Articles of Incorporation and Bylaws of the Surviving Corporation	6
1.4	Directors and Officers	6
1.5	Effect of the Merger	6
ARTICLE II	EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT CORPORATIONS; EXCHANGE OF	
	CERTIFICATES	6
2.1	Conversion of Shares	6
2.2	Exchange of Seller Common Stock Certificates	10
2.3	Dissenting Shares	11
2.4	Additional Actions	12
2.5	Rights as Stockholders	12
2.6	Stock Transfer Records	12
ARTICLE III	REPRESENTATIONS AND WARRANTIES OF SELLER	13
3.1	Corporate Organization	13
3.2	Capitalization	13
3.3	Authority; No Violation	14
3.4	Financial Statements	15
3.5	Absence of Certain Changes or Events	15
3.6	Legal Proceedings	15
3.7	Taxes and Tax Returns	16
3.8	Employee Benefit Plans	17
3.9	Regulatory Reports	19
3.10	Seller Information	19
3.11	Compliance with Applicable Law	19
3.12	Deposit Insurance and Other Regulatory Matters	20
3.13	Certain Contracts	20
3.14	Properties and Insurance	21
3.15	Environmental Matters	22
3.16	Allowance for Loan Losses and Real Estate Owned	22
3.17	Minute Books	23
3.18	Affiliate Transactions	23
3.19	Internal Controls	23
3.20	Risk Management Instruments	23
3.21	Fairness Opinion	23
3.22	Broker Fees	23
3.23	Loans	23
3.24	Investments	24
3.25	Employees; Compensation	24
3.26	Tax and Regulatory Matters	24
3.27	Intellectual Property	25
3.28	Community Reinvestment Compliance	25
3.29	No Existing Discussions	26
3.30	Certain Business Practices	26

3.31 3.32	Continuity of Business Enterprise Full Disclosure	26 26
RTICLE IV	REPRESENTATIONS AND WARRANTIES OF ACQUIROR	26
4.1	Corporate Organization	26
4.2	Merger Sub; No Activities	27
4.3	Authority; No Violation	27
4.4	Financial Statements	28
4.5	Absence of Certain Changes or Events	28
4.6	Legal Proceedings	29
4.7	Acquiror Information	29
4.8	Deposit Insurance and Other Regulatory Matters	29
4.9	Capital; Availability of Funds; Acquiror Shares	29
4.10	Broker Fees	29
4.11	Securities Documents and Regulatory Reports	29
4.12	Compliance with Applicable Law	30
4.13	Full Disclosure	30
RTICLE V	COVENANTS OF THE PARTIES	30
5.1	Conduct of the Business of Seller	30
5.2	Negative Covenants of Seller	31
5.3	No Solicitation	32
5.4	Negative Covenants of Acquiror	34
5.5	Current Information	35
5.6	Access to Properties and Records; Confidentiality	35
5.7	Regulatory Matters	36
5.8	Approval of Stockholders	36
5.9	Further Assurances	37
5.10	Disclosure Supplements	37
5.11	Public Announcements	37
5.12	Failure to Fulfill Conditions	37
5.13	Certain Post-Merger Agreements	37
5.14	Takeover Laws; No Rights Triggered	39
5.15	Preparation of Registration Statement	39
5.16	Affiliates	40
5.17	Adjustment for Changes in Outstanding Shares	40
5.18	Adoption of Accounting Policies	40
5.19	Covenant to Close	40
5.20	Certain Agreements	40
5.21	Hold Harmless	41
5.22	Stock Options	41
5.23	Warrants	41
RTICLE VI	CLOSING CONDITIONS	42
6.1	Conditions to the Parties' Obligations under this Agreement	42
6.2	Conditions to the Obligations of Acquiror and Merger Sub under this Agreement	42
6.3	Conditions to the Obligations of Seller under this Agreement	43
RTICLE VII	TERMINATION, AMENDMENT AND WAIVER, ETC.	43
7.1	Termination	43
7.2	Effect of Termination	45

7.3	Amendment, Extension and Waiver	45
7.4	Termination Fees	45
ARTICLE VIII	MISCELLANEOUS	46
8.1	Expenses	46
8.2	Survival	46
8.3	Notices	46
8.4	Parties in Interest	47
8.5	Complete Agreement	47
8.6	Counterparts	48
8.7	Governing Law	48
8.8	Interpretation	48
8.9	Enforcement	48

Schedules:

Schedule 5.13(b) [attached] Schedule 5.16 [attached] Schedule 5.20-A [see Annex B-1] Schedule 5.20-B [see Annex B-2] Schedule 5.20-C [attached] Schedule 5.23 [attached]

Exhibits:

Exhibit A Plan of Merger [see Annex A-2]
Exhibit B Seller Disclosure Schedule [omitted but available upon request]
Exhibit C Acquiror Disclosure Schedule [omitted but available upon request]

AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER ("**Agreement**"), dated as of February 17, 2004, by and among The Peoples Holding Company, a Mississippi corporation ("**Acquiror**"), Peoples Merger Corporation, a Tennessee corporation ("**Merger Sub**"), and Renasant Bancshares, Inc., a Tennessee corporation ("**Seller**"). Each of Acquiror, Merger Sub and Seller is a party ("**party**") hereto, and one or more of them are parties ("**parties**") to this Agreement as the context may require.

WITNESSETH:

WHEREAS, the Boards of Directors of Acquiror and Seller each have determined that it is advisable and in the best interests of their respective companies and their stockholders to consummate the business combination transactions provided for herein, including the merger of Seller with and into Merger Sub (together, sometimes referred to as the "Constituent Corporations") subject to the terms and conditions set forth herein; and

WHEREAS, the parties desire to provide for certain undertakings, conditions, representations, warranties and covenants in connection with the transactions contemplated hereby.

NOW, THEREFORE, in consideration of the premises and the mutual covenants, representations, warranties and agreements herein contained, and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE I THE MERGER

- 1.1 <u>The Merger</u>. Subject to the terms and conditions of this Agreement, at the Effective Time (as defined in Section 1.2 hereof), Seller shall be merged with and into Merger Sub (the "**Merger**") in accordance with the Tennessee Business Corporation Act (the "**TBCA**") with Merger Sub as the surviving corporation (hereinafter sometimes called the "**Surviving Corporation**") which shall continue its corporate existence under the laws of the State of Tennessee as a subsidiary of Acquiror, and the separate corporate existence of Seller shall terminate. The Merger shall in all respects have the effects provided in Section 1.5.
- 1.2 Effective Time. The Merger shall become effective on the date and at the time that Articles of Merger are filed with the Secretary of State of the State of Tennessee pursuant to Section 48-21-107 of the TBCA, unless a later date and time is specified as the effective time in such documents, provided that the parties shall cause the Merger to be effective no later than the day following the date on which the Closing occurs (the "Effective Time"). A closing (the "Closing") shall take place at 10:00 a.m. on the fifth Business Day (the "Closing Date") following the receipt of all necessary approvals and consents of any governmental or regulatory authority, agency, court, commission or other entity, domestic or foreign ("Governmental Entity") and the expiration of all statutory waiting periods in respect thereof and the satisfaction or waiver, to the extent permitted hereunder, of the conditions to the consummation of the Merger specified in Article VI of this Agreement (other than the delivery of certificates, instruments and documents to be delivered at the Closing), at the offices of Acquiror, or at such other place, at such other time or on such other date as the parties may mutually agree upon. For purposes of this Agreement, a Business Day ("Business Day") is any day that banks located in the State of Tennessee are not permitted or required to be closed, except a Saturday or Sunday. At the Closing, there shall be delivered to Acquiror and Seller the certificates and other documents required to be delivered under Article VI hereof.

- 1.3 <u>The Articles of Incorporation and Bylaws of the Surviving Corporation</u>. The Articles of Incorporation and the Bylaws of Merger Sub shall be the Articles of Incorporation and the Bylaws of the Surviving Corporation until thereafter changed or amended as provided therein or by applicable law.
- 1.4 <u>Directors and Officers</u>. Immediately after the Effective Time, the directors of the Surviving Corporation shall consist of the directors of Merger Sub, in office immediately prior to the Effective Time, until their respective successors are duly elected, appointed or qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and the Bylaws of the Surviving Corporation. The officers of Merger Sub shall, from and after the Effective Time, continue as the officers of the Surviving Corporation until their successors shall have been duly elected, appointed or qualified or until their earlier death, resignation or removal in accordance with the Articles of Incorporation and the Bylaws of the Surviving Corporation.
- 1.5 Effect of the Merger. At the Effective Time, the separate existence and corporate organization of Seller shall cease, and all right, title and interest in and to all real estate and other property owned by Seller shall be allocated to and shall be vested in Merger Sub as the surviving corporation, without reversion or impairment, without further act or deed, and without any transfer or assignment having occurred (but subject to any existing liens or other encumbrances thereon), and all liabilities and obligations of Seller shall be allocated to Merger Sub as the surviving corporation, as primary obligor therefor and, except as set forth herein, no other person shall be liable therefor, and all proceedings pending by or against the Seller shall be continued by or against Merger Sub as the surviving corporation, and all liabilities, obligations, assets or rights associated with such proceedings shall be allocated to and vested in Merger Sub as the surviving corporation.

ARTICLE II EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT CORPORATIONS; EXCHANGE OF CERTIFICATES

- 2.1 <u>Conversion of Shares</u>. All of the shares of Acquiror and Merger Sub issued and outstanding immediately prior to the Effective Time shall remain issued and outstanding after the Effective Time and shall be unaffected by the Merger. The manner and basis of converting the shares of common stock, \$1.00 par value, of Seller (the "Seller Common Stock") upon consummation of the Merger shall be as follows:
- (a) At the Effective Time, by virtue of the Merger and without any action on the part of Acquiror, Merger Sub, Seller or the holders of Seller Common Stock:
 - (i) Subject to the other provisions of this Section 2.1, each share of Seller Common Stock issued and outstanding immediately prior to the Effective Time (excluding any treasury shares, shares held by Acquiror or any of the subsidiaries of Acquiror or Seller (other than in a fiduciary capacity) and shares held by Dissenting Stockholders (as defined below)) shall be converted into the right to receive the Merger Consideration. The "Merger Consideration" shall mean either (a) cash in the amount of the Price Per Share (as defined below), without interest (the "Cash Consideration"), (b) a share of Acquiror Common Stock (as defined below), rounded to the nearest six decimals, equal to the Exchange Ratio (the "Stock Consideration"), or (c) a combination of Cash Consideration and Stock Consideration in accordance with subparagraph (iii) of this Section 2.1. As used in this Agreement, the term "Price Per Share" equals \$36.37, the term "Exchange Ratio" equals 1.117015, and the term "Acquiror Common Stock" means the common stock, \$5.00 per share par value, of the Acquiror.

- (ii) The number of shares of Seller Common Stock to be converted into the right to receive Cash Consideration shall not be less than 45% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be cancelled pursuant to clause (x) of this Section 2.1(a)) (the "Minimum Cash Election Number") and shall not be greater than 50% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be converted pursuant to clause (x) of this Section 2.1(a)) (the "Maximum Cash Election Number"). For purposes of determining both the Minimum Cash Election Number and Maximum Cash Election Number, Dissenting Stockholders shall be deemed to have made a Cash Election (as defined below) unless such Dissenting Stockholders shall effectively withdraw or lose (through failure to perfect or otherwise) his right to payment as a dissenting shareholder under the TBCA at or prior to the Effective Time. The number of shares of Seller Common Stock to be converted into the right to receive Stock Consideration shall be not less than 50% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be cancelled pursuant to clause (x) of this Section 2.1(a)) (the "Minimum Stock Election Number") and not greater than 55% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be cancelled pursuant to clause (x) of this Section 2.1(a)) (the "Maximum Stock Election Number").
- (iii) Subject to the proration and election procedures set forth in this Section 2.1(a), each holder of record of shares of Seller Common Stock (excluding any treasury shares, shares held by Acquiror or any of the subsidiaries of Acquiror or Seller (other than in a fiduciary capacity) and shares held by Dissenting Stockholders) will be entitled to elect to receive (a) Cash Consideration for all such shares (a "Cash Election"), (b) Stock Consideration for all of such shares (a "Stock Election") or (c) Cash Consideration for 45% of such shares and Stock Consideration for 55% of such shares (a "Combination Election"). All such elections shall be made on a form designed for that purpose prepared by Acquiror and reasonably acceptable to Seller (a "Form of Election"). Holders of record of shares of Seller Common Stock who hold such shares as nominees, trustees or in other representative capacities (a "Representative") may submit multiple Forms of Election, provided that such Representative certifies that each such Form of Election covers all the shares of Seller Common Stock held by each such Representative for a particular beneficial owner.
- (iv) Acquiror and Seller shall mail the Form of Election to all Persons who are holders of Seller Common Stock on the record date for the Special Meeting (as defined below), on a date that is not less than 20 Business Days prior to the Effective Time, and thereafter Acquiror and Seller shall each use its reasonable efforts to (x) mail the Form of Election to all persons who become holders of Seller Common Stock during the period between the record date for the Special Meeting and 10:00 a.m., Memphis, Tennessee time, on the date seven Business Days prior to the anticipated Effective Time and (y) make the Form of Election available to all Persons who become holders of Seller Common Stock subsequent to such day and no later than the close of business on the fourth Business Day prior to the Effective Time. A Form of Election must be received by the Exchange Agent in the manner described below no later than by the close of business on the Business Day which is three Business Days immediately prior to the Effective Time (the "Election Deadline") in order to be effective. All elections will be irrevocable.
- (v) Elections shall be made by holders of Seller Common Stock by mailing, faxing or otherwise delivering to the Exchange Agent (as defined below), in a manner acceptable to

- Acquiror, a Form of Election. To be effective, a Form of Election must be properly completed, signed and submitted to the Exchange Agent. Acquiror will have the discretion, which it may delegate in whole or in part to the Exchange Agent, to determine whether Forms of Election have been properly completed, signed and submitted and to disregard immaterial defects in Forms of Election. The decision of Acquiror (or the Exchange Agent) in such matters shall be conclusive and binding. Neither Acquiror nor the Exchange Agent will be under any obligation to notify any Person of any defect in a Form of Election.
- (vi) A holder of Seller Common Stock who does not submit a Form of Election which is received by the Exchange Agent prior to the Election Deadline shall be deemed to have made a Combination Election. If Acquiror or the Exchange Agent shall determine that any purported Cash Election or Stock Election was not properly made, such purported Cash Election or Stock Election shall be deemed to be of no force and effect and the holder of shares of Seller Common Stock making such purported Cash Election or Stock Election shall for purposes hereof be deemed to have made a Combination Election.
- (vii) All shares of Seller Common Stock which are subject to Cash Elections are referred to herein as "Cash Election Shares." All shares of Seller Common Stock which are subject to Stock Elections are referred to herein as "Stock Election Shares." If, after the results of the Forms of Election are calculated, the number of shares of Seller Common Stock to be converted into shares of Acquiror Common Stock exceeds the Maximum Stock Election Number, the Exchange Agent shall, after the Election Deadline but prior to the Effective Time, determine the number of Stock Election Shares which must be redesignated as Cash Election Shares in order to reduce the number of such shares to the Maximum Stock Election Number. All holders who have Stock Election Shares shall, on a pro rata basis, have such number of their Stock Election Shares redesignated as Cash Election Shares so that the Maximum Stock Election Number and the Minimum Cash Election Number are achieved. If, after the results of the Forms of Election are calculated, the number of shares of Seller Common Stock to be converted into cash exceeds the Maximum Cash Election Number, the Exchange Agent, after the Election Deadline but prior to the Effective Time, shall determine the number of Cash Election Shares which must be redesignated as Stock Election Shares in order to reduce the amount of such cash to the Maximum Cash Election Number. All holders who have Cash Election Shares shall, on a pro rata basis, have such number of their Cash Election Shares redesignated as Stock Election Shares so that the Maximum Cash Election Number and the Minimum Stock Election Number are achieved. Notwithstanding the foregoing, no redesignation shall be effected for a holder who has made a Cash Election but, as a result of such redesignation, would receive fewer than 10 shares of Acquiror Common Stock in exchange for all of such holder's shares of Seller Common Stock unless such redesignation is necessary for the Merger to qualify as a reorganization under Section 368 of the Code as described in Section 2.1(a)(viii). In this event, the Cash Election Shares of the remaining holders of shares of Seller Common Stock shall be redesignated on a pro rata basis to achieve the Maximum Cash Election Number and the Minimum Stock Election Number. Holders who make Combination Elections will not be subject to the redesignation procedures described herein unless such redesignation is necessary for the Merger to qualify as a reorganization under Section 368 of the Code as described in Section 2.1(a)(viii). Dissenting Stockholders who are deemed to have made Cash Elections shall not be subject to the redesignation procedure described herein. Acquiror or the Exchange Agent shall make all computations contemplated by this

Section 2.1(a) and all such computations shall be conclusive and binding on the holders of Seller Common Stock.

- (viii) It is the intent of the parties that the Merger qualify as a reorganization under Section 368 of the Code. In order to qualify as a reorganization under Section 368 of the Code, the Merger must satisfy the Continuity of Interest requirement set forth in Treasury Regulation Section 1.368-1(e). If after the Exchange Agent completes the redesignation procedures set forth in Section 2.1(a)(vii) (or if there is no redesignation required by the Exchange Agent after the Exchange Agent notifies the parties that redesignation is not required), the Acquiror determines, in good faith after consultation with its tax counsel, that the value of Acquiror Common Stock included in the Merger Consideration will not be sufficient to satisfy the Continuity of Interest requirement, then, except as provided hereinbelow, Acquiror shall increase the Exchange Ratio and decrease the Price Per Share to the extent Acquiror, in good faith after consultation with its tax counsel, deems necessary to meet the Continuity of Interest requirement. Notwithstanding anything to the contrary in this Section 2.1(a)(viii), if the increase in the Exchange Ratio would cause Acquiror to issue more than 85,000 additional shares of Acquiror Common Stock pursuant to this Section 2.1(a)(viii), then the adjustment described in this Section 2.1(a)(viii) shall be at Acquiror's option and shall not be mandatory. Any adjustment of the Exchange Ratio and the Price Per Share will be done in a manner so that the reduction in Cash Consideration included in the Merger Consideration (prior to any adjustment for fractional shares) due to the adjustment to the Price Per Share shall be equal to the increase in value of Acquiror Common Stock included in the Merger Consideration (prior to any adjustment for fractional shares) due to the adjustment to the Exchange Ratio. Acquiror shall notify Seller of any adjustment pursuant to this Section 2.1(a)(viii) promptly after the calculation of such adjustment but in no event less than one Business Day prior to the Effective Time. The adjustment of the Exchange Ratio and the Price Per Share by Acquiror under this Section 2.1(a)(viii) shall be conclusive and binding on the parties to this Agreement. For purposes of this Section 2.1(a)(viii), the Continuity of Interest requirement is agreed to be at least 40% continuity.
- (ix) After the redesignation procedure, if any, and the adjustment procedure, if any, set forth in Sections 2.1(a)(vii) and (viii), respectively, are completed, all Cash Election Shares and 45% of the shares of Seller Common Stock which are subject to Combination Elections shall be converted into the right to receive the Cash Consideration, and all Stock Election Shares and 55% of the shares of Seller Common Stock which are subject to Combination Elections shall be converted into the right to receive the Stock Consideration. Such certificates previously evidencing shares of Seller Common Stock shall be exchanged for (a) certificates evidencing the Stock Consideration, or (b) the Cash Consideration, multiplied in each case by the number of shares previously evidenced by the cancelled certificate, upon the surrender of such certificates in accordance with the provisions of Section 2.2, without interest. Notwithstanding the foregoing, however, no fractional shares of Acquiror Common Stock shall be issued, and, in lieu thereof, a cash payment shall be made pursuant to Section 2.1(b).
- (x) Each share of Seller Common Stock held in the treasury of Seller and each share of Seller Common Stock owned by Acquiror or any subsidiary of Acquiror or Seller (other than in a fiduciary capacity) immediately prior to the Effective Time shall be cancelled and extinguished without any conversion thereof and no payment shall be made with respect thereto.

(b) No certificates or scrip representing fractional shares of Acquiror Common Stock will be issued as a result of the Merger. In lieu of the issuance of fractional shares pursuant to Section 2.1(a) of this Agreement, cash adjustments (without interest) will be paid to the holder of Seller Common Stock in respect of any fraction of a share of Acquiror Common Stock that would otherwise be issuable to such holder of Seller Common Stock, and the amount of such cash adjustment shall be determined by multiplying the fraction of a share of Acquiror Common Stock otherwise issuable by the average of the closing price of one share of Acquiror Common Stock for the 10 trading days immediately preceding the last trading day immediately prior to the Closing Date as reported by the American Stock Exchange, and no such holder shall be entitled to dividends, voting rights or any other right of stockholders in respect of any fractional share.

2.2 Exchange of Seller Common Stock Certificates.

- (a) Seller and Acquiror shall designate a mutually acceptable person to serve as exchange agent (the "Exchange Agent") in connection with the Merger. The Exchange Agent shall provide appropriate stock certificate transmittal materials to the former holders of Seller Common Stock promptly after the Effective Time (which transmittal materials shall specify that the delivery shall be effected, and risk of loss and title shall pass, only upon proper delivery of a certificate or certificates formerly representing shares of Seller Common Stock to the Exchange Agent) and instructions for use in effecting the surrender to the Exchange Agent of certificates of Seller Common Stock in exchange of the Merger Consideration. After the Effective Time, each holder of shares of Seller Common Stock issued and outstanding immediately prior to the Effective Time (other than shares held by Dissenting Stockholders) shall surrender for cancellation the certificate or certificates representing such shares to the Exchange Agent, together with a letter of transmittal duly executed and completed in accordance with the instructions thereto and any other documents reasonably required by the Exchange Agent or Acquiror, and promptly upon surrender of such certificates and other documents Exchange Agent on behalf of Acquiror shall deliver the Merger Consideration to each such holder in exchange for each such share, without interest. Each certificate for Seller Common Stock so surrendered and all transmittal materials shall be duly completed and endorsed as the Exchange Agent may require. Exchange Agent shall not be obligated to deliver the Merger Consideration to which any former holder of Seller Common Stock is entitled as a result of the Merger until such holder surrenders his certificate or certificates representing shares of Seller Common Stock for exchange as provided in this Section 2.2. After the Effective Time, each certificate that represented outstanding shares of Seller Common Stock prior to the Effective Time shall be deemed for a
- (b) Any Merger Consideration held by an Exchange Agent that remains undistributed to the former stockholders of Seller for six months after the Effective Time shall be delivered to Acquiror upon demand, and any former stockholders of Seller who have not theretofore complied with this Section 2.2 shall thereafter look only to Acquiror for payment of their claims for cash, Acquiror Common Stock, any cash in lieu of fractional shares of Acquiror Common Stock or any dividends or distributions with respect to Acquiror Common Stock (all without any interest thereon).
- (c) None of Acquiror, any subsidiary thereof or the Exchange Agent shall be liable to any former holder of Seller Common Stock for cash, shares of Acquiror Common Stock (or dividends or distributions with respect thereto) or cash in lieu of fractional shares of Acquiror Common Stock delivered to public officials pursuant to any applicable abandoned property, escheat or similar law.

- (d) From and after the Effective Time, the holders of certificates of Seller Common Stock shall cease to have any rights with respect to shares of Seller Common Stock represented thereby except as otherwise provided in this Agreement or by applicable Law. All rights to receive the Merger Consideration issued upon conversion of the shares of Seller Common Stock pursuant to this Article II shall be deemed to have been paid or issued, as the case may be, in full satisfaction of all rights pertaining to such shares of Seller Common Stock.
- (e) Promptly after the Effective Time, Acquiror shall deposit with the Exchange Agent (defined below) (i) cash in an amount equal to the aggregate Cash Consideration and cash sufficient to pay fractional shares and (ii) certificates representing shares equal to the aggregate Stock Consideration (such cash and shares of Seller Common Stock, together with any dividends or distributions with respect to such Seller Common Stock, the "Exchange Fund") for the benefit of Seller's Stockholders.
- (f) If any portion of the Merger Consideration is to be paid to a person other than the person in whose name a Seller Stock Certificate so surrendered is registered, it shall be a condition to such payment that such Seller Stock Certificate shall be properly endorsed or otherwise be in proper form for transfer, and the person requesting such payment shall pay to the Exchange Agent any transfer or other similar Taxes (as defined in Section 3.7(c)) required as a result of such payment to a person other than the registered holder of such Seller Stock Certificate, or establish to the reasonable satisfaction of the Exchange Agent that such Tax has been paid or is not payable. Acquiror or the Exchange Agent shall be entitled to deduct and withhold from the Merger Consideration otherwise payable pursuant to this Agreement to any holder of the Seller Common Stock such amounts as Acquiror or the Exchange Agent is required to deduct and withhold under the Internal Revenue Code of 1986, as amended (the "Code"), or any provision of state, local or foreign Tax law, with respect to the making of such payment. To the extent the amounts are so withheld by Acquiror or the Exchange Agent, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the holder of shares of the Seller Common Stock in respect of whom such deduction and withholding was made by Acquiror or the Exchange Agent.
- (g) After the Effective Time, there shall be no further registration or transfers of shares of the Seller Common Stock. If after the Effective Time, the Seller Stock Certificates are presented to the Surviving Corporation, they shall be cancelled and exchanged for the Merger Consideration in accordance with the procedures set forth in this Article II.
- (h) In the event any of the Seller Stock Certificate(s) shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Seller Stock Certificate(s) to be lost, stolen or destroyed and, if required by Acquiror or the Exchange Agent, the posting by such person of a bond in such sum as either of them may reasonably direct as indemnity against any claim that may be made against it or the Surviving Corporation with respect to such Seller Stock Certificate(s), the Exchange Agent will issue the Merger Consideration deliverable in respect of the shares of Seller Common Stock represented by such lost, stolen or destroyed Seller Stock Certificate(s).
- (i) The Exchange Agent shall invest the cash balances in the Exchange Fund as directed by Acquiror. Any interest and other income resulting from such investments shall be paid to the Acquiror upon termination of the Exchange Fund pursuant to Section 2.2(b). Acquiror shall instruct the Exchange Agent to timely pay the Merger Consideration.
- 2.3 <u>Dissenting Shares</u>. No outstanding share of Seller Common Stock as to which the holder has exercised dissenters rights under the TBCA shall be converted into or represent a right to receive the Merger Consideration, and the holder thereof shall be entitled only to such rights as are granted by the TBCA. Seller shall give Acquiror prompt notice upon receipt by Seller of any such written demands for

payment of the fair value of such shares of the Seller Common Stock and of withdrawals of such demands and any other instruments provided pursuant to the TBCA (any stockholder duly making such demand being hereinafter called a "Dissenting Stockholder"). Acquiror shall direct all negotiations and proceedings with respect to any such demands or notices. Seller shall not, without the prior written consent of Acquiror, make any payment with respect to, or settle, offer to settle or otherwise negotiate, any such demands. If any Dissenting Stockholder shall effectively withdraw or lose (through failure to perfect or otherwise) his right to such payment at or prior to the Effective Time, such holder's shares of the Seller Common Stock shall be automatically converted into a right to receive the Merger Consideration in accordance with the applicable provisions of this Agreement, without any interest thereon, as if such holder had made a Combination Election. If such holder shall effectively withdraw or lose (through failure to perfect or otherwise) his right to such payment after the Effective Time, each share of the Seller Common Stock of such holder shall be automatically converted, on a share-by-share basis, into the right to receive the Merger Consideration, without any interest thereon, as if such holder had made a Combination Election.

- 2.4 <u>Additional Actions</u>. If at any time after the Effective Time the Surviving Corporation shall consider that any further assignments or assurances in law or any other acts are necessary or desirable to carry out the purposes of this Agreement, Seller and its proper officers and directors shall be deemed to have granted to the Surviving Corporation an irrevocable power of attorney to execute and deliver all such proper deeds, assignments and assurances in law and to do all acts necessary or proper to vest, perfect or confirm title to and possession of such rights, properties or assets in the Surviving Corporation and otherwise to carry out the purposes of this Agreement; and the proper officers and directors of the Surviving Corporation are fully authorized in the name of Seller or otherwise to take any and all such action.
- 2.5 Rights as Stockholders. Former stockholders of Seller and any other individuals or entities who or which are entitled to receive Acquiror Common Stock as a result of the Merger will be able to vote after the Effective Time at any meeting of Acquiror stockholders or pursuant to any written consent procedure the number of whole shares of Acquiror Common Stock into which their shares of Seller Common Stock are converted, regardless of whether they have exchanged their Seller Stock Certificates. In addition, whenever a dividend is declared by Acquiror on the Acquiror Common Stock after the Effective Time, the declaration shall include dividends on all shares issuable hereunder, but no stockholder will be entitled to receive his distribution of such dividends until physical exchange of such stockholder's Seller Stock Certificates shall have been effected. Upon exchange of a stockholder's Seller Stock Certificates, any such person shall be entitled to receive from Acquiror an amount equal to all dividends (without interest thereon less the amount of any taxes, if any, that may have been withheld, imposed or paid thereon) declared, and for which the payment has occurred, on the shares represented thereby; provided, however, that former stockholders of Seller shall not be entitled to receive any dividend on their Acquiror Common Stock with respect to any period for which Seller paid a dividend prior to the Effective Time.
- 2.6 <u>Stock Transfer Records</u>. Prior to the Effective Time, Seller shall continue to maintain its stock transfer records and to transfer and replace stock certificates in accordance with its existing policies and past practices with regard to such transfers and replacements. From and after the Effective Time, there shall be no transfers on the stock transfer books of Seller or Acquiror of shares of Seller Common Stock which were issued and outstanding immediately prior to the Effective Time.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF SELLER

Seller makes the representations and warranties to Acquiror contained in this Article III, except as set forth in the disclosure schedule attached hereto as <u>Exhibit B</u> (the "Seller Disclosure Schedule"). The Seller Disclosure Schedule is arranged in sections corresponding to the numbered and lettered sections contained in this Article III. Disclosure in any section of the Disclosure Schedule shall be effectively made whether or not expressly excepted in the corresponding section of this Agreement. The following statements are made as of the date of this Agreement. The phrase "delivered to Acquiror" or "made available to Acquiror" or any phrase of similar import means that Seller has delivered, provided access to or made certain items available for review and copying to Acquiror or its counsel. For purposes of this Article III, the phrase "to the Knowledge of Seller" or any phrase of similar import shall be deemed to refer to the actual knowledge of the officers (i.e., the vice president level and up) of Seller after investigation.

3.1 Corporate Organization.

- (a) Seller is a corporation duly organized, validly existing and in good standing under the laws of the State of Tennessee. Seller has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted and is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not have a material adverse effect on the business, operations, assets or financial condition of Seller and the Seller Subsidiary (as defined in Section 3.1(b) hereof) taken as a whole. Seller is registered as a bank holding company under the Bank Holding Company Act of 1956, as amended. Seller Disclosure Schedule 3.1(a) sets forth true and complete copies of the Charter and Bylaws of Seller and the Seller Subsidiary as in effect on the date hereof.
- (b) The only direct or indirect subsidiary of Seller is Renasant Bank (the "Seller Subsidiary"). The Seller Subsidiary (i) is duly organized and validly existing and in good standing under the laws of its jurisdiction of incorporation, (ii) has the corporate power and authority to own or lease all of its properties and assets and to conduct its business as it is now being conducted, and (iii) is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not have a material adverse effect on the business, operations, assets or financial condition of Seller and the Seller Subsidiary taken as a whole. Other than the Seller Subsidiary, Seller does not own or control, directly or indirectly, a 5% or greater equity interest in any corporation, company, association, partnership, joint venture or other entity.
- 3.2 <u>Capitalization</u>. The authorized capital stock of Seller consists of 2,000,000 shares of the Seller Common Stock, of which 1,429,697 are issued and outstanding and 3,626 shares are held in treasury as of the date hereof, and 1,000,000 shares of preferred stock, no par value, of which no shares are issued and outstanding. All issued and outstanding shares of capital stock of Seller and all issued and outstanding shares of capital stock of the Seller Subsidiary have been duly authorized and validly issued and are fully paid, non-assessable and free of preemptive rights. Except as disclosed in Section 3.2 of the Seller Disclosure Schedule, all of the outstanding shares of capital stock of the Seller Subsidiary are owned by Seller free and clear of any liens, encumbrances, charges, restrictions or rights of third parties of any kind whatsoever; and, except for options to purchase 99,980 shares of the Seller Common Stock which have been granted prior to the date hereof pursuant to the Stock Option Plan (as defined below) and which are outstanding as of the date hereof (the "Outstanding Seller Stock Options"), and warrants to

purchase 120,000 shares of the Seller Common Stock which have been granted prior to the date hereof (the "Outstanding Seller Warrants"), neither Seller nor the Seller Subsidiary has or is bound by any outstanding subscriptions, options, warrants, calls, commitments or agreements of any character calling for the transfer, purchase or issuance of any shares of capital stock of Seller or the Seller Subsidiary or any securities representing the right to purchase or otherwise receive any shares of such capital stock or any securities convertible into or representing the right to purchase or subscribe for any such stock.

3.3 Authority; No Violation.

- (a) Subject to the approval of this Agreement and the Plan of Merger and the transactions contemplated hereby and thereby by the stockholders of Seller, Seller has full corporate power and authority to execute and deliver this Agreement and the Plan of Merger and to consummate the transactions contemplated hereby and thereby in accordance with the terms hereof and thereof. The execution and delivery of this Agreement and the Plan of Merger and the consummation of the transactions contemplated hereby and thereby have been duly and validly approved by the Board of Directors of Seller. Except for the approval of Seller's stockholders of this Agreement and the Plan of Merger, no other corporate proceedings on the part of Seller are necessary to consummate the transactions so contemplated. This Agreement and the Plan of Merger have been duly and validly executed and delivered by Seller and constitute valid and binding obligations of Seller, enforceable against it in accordance with and subject to their terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally, and except that the availability of equitable remedies (including specific performance) is within the discretion of the appropriate court.
- (b) None of the execution and delivery of this Agreement and the Plan of Merger by Seller, nor the consummation by Seller of the transactions contemplated hereby and thereby in accordance with the terms hereof and thereof, or compliance by Seller with any of the terms or provisions hereof or thereof, will (i) violate any provision of the Charter or other governing instrument or Bylaws of Seller or the Seller Subsidiary, (ii) assuming that the consents and approvals set forth below are duly obtained, violate any statute, code, ordinance, rule, regulation, judgment, order, writ, decree or injunction applicable to Seller or the Seller Subsidiary or any of their respective properties or assets, or (iii) violate, conflict with, result in a breach of any provisions of, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of, accelerate the performance required by, require the payment of any termination or like fee, or result in the creation of any lien, security interest, charge or other encumbrance upon any of the respective properties or assets of Seller or the Seller Subsidiary under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which Seller or the Seller Subsidiary is a party, or by which any of their respective properties or assets may be bound or affected. Except as set forth in Seller Disclosure Schedule 3.3(b) and for consents and approvals of or filings or registrations with or notices to the Secretary of State of the State of Tennessee, the Federal Reserve Board (the "FRB"), the Federal Deposit Insurance Corporation (the "FDIC"), the Tennessee Commissioner of Financial Institutions (the "TCFI"), the Mississippi Commissioner of Banking and Consumer Finance (the "MCB"), and the stockholders of Seller, no consents or approvals of or filings or registrations with or notices to any Governmental Entity or non-gove
- (c) Seller has taken all action required to be taken by it in order to exempt this Agreement and the transactions contemplated hereby from, and this Agreement and the transactions contemplated hereby are exempt from, the requirements of any "moratorium," "control share," "fair price,"

"supermajority," "affiliate transactions," "business combination" or other state antitakeover laws and regulations (collectively, "Takeover Laws").

3.4 Financial Statements.

- (a) Seller has previously delivered to Acquiror copies of the consolidated balance sheets of Seller as of December 31, 2002, 2001 and 2000 and the related consolidated statements of income, changes in stockholders' equity and cash flows for the years ended December 31, 2002, 2001 and 2000, in each case accompanied by the audit reports of Reynolds, Bone & Griesbeck PLC, independent public accountants, as well as the unaudited consolidated balance sheet of Seller as of December 31, 2003 and the related unaudited consolidated statements of income, changes in stockholders' equity and cash flows for the three and twelve months ended December 31, 2003 and 2002. The consolidated financial statements of Seller referred to herein (including the related notes, where applicable), as well as the consolidated financial statements contained in the reports of Seller to be delivered by Seller pursuant to Section 5.5 hereof, fairly present or will fairly present, as the case may be, the consolidated financial condition of Seller as of the respective dates set forth therein, and the related consolidated statements of income, changes in stockholders' equity and cash flows (including the related notes, where applicable) fairly present or will fairly present, as the case may be, the results of the consolidated operations, changes in stockholders' equity and cash flows of Seller for the respective periods or as of the respective dates set forth therein (it being understood that Seller's interim financial statements are not audited and are not prepared with related notes but reflect all adjustments which were, at the time, in the opinion of Seller, necessary for a fair presentation of such financial statements).
- (b) Each of the financial statements referred to in this Section 3.4 (including the related notes, where applicable) has been or will be, as the case may be, prepared in accordance with U.S. generally accepted accounting principles ("GAAP") consistently applied during the periods involved. The books and records of Seller and the Seller Subsidiary are being maintained in material compliance with applicable legal and accounting requirements and reflect only actual transactions.
- (c) Except to the extent reflected, disclosed or reserved against in the consolidated financial statements referred to in the first sentence of Section 3.4(a) or the notes thereto or liabilities incurred since December 31, 2003 in the ordinary course of business and consistent with past practice (none of which arises from breach of any contract or agreement, breach of warranty, tort, infringement, violation of any applicable federal, state or local law or ordinance or any litigation or other proceeding or is otherwise a "loss contingency" within the meaning of Statement of Financial Accounting Standards No. 5), neither Seller nor the Seller Subsidiary has any obligation or liability, whether absolute, accrued, contingent or otherwise, material to the business, operations, assets or financial condition of Seller and the Seller Subsidiary taken as a whole.
- 3.5 <u>Absence of Certain Changes or Events</u>. There has not been any material adverse change in the business, operations, prospects, assets or financial condition of Seller and the Seller Subsidiary taken as a whole since December 31, 2003, other than: (i) any change in banking or similar laws, rules or regulations of general applicability to banks or their holding companies or interpretations thereof by Governmental Entities; (ii) changes in GAAP that are generally applicable to the banking industry; (iii) reasonable expenses incurred in connection with the transactions contemplated hereby; and (iv) changes attributable to or resulting from changes in general economic conditions, including changes in the prevailing level of interest rates.
- 3.6 <u>Legal Proceedings</u>. Neither Seller nor the Seller Subsidiary is a party to any, and there are no pending or, to the Knowledge of Seller, threatened legal, administrative, arbitration or other proceedings, claims, actions or governmental investigations of any nature against Seller or the Seller

Subsidiary. Neither Seller nor the Seller Subsidiary is a party to any order, judgment or decree which affects the business, operations, assets or financial condition of Seller and the Seller Subsidiary taken as a whole.

3.7 Taxes and Tax Returns.

- (a) Each of Seller and the Seller Subsidiary has duly filed (and until the Effective Time will so file) all returns, declarations, reports, information returns and statements ("Returns") required to be filed or sent by or with respect to them in respect of any Taxes (as hereinafter defined) and has duly paid (and until the Effective Time will so pay) all Taxes due and payable other than Taxes or other charges which (i) are being contested in good faith (and are set forth on Seller Disclosure Schedule 3.7(a)) and (ii) have not finally been determined. Seller and the Seller Subsidiary have established (and until the Effective Time will establish) on their books and records reserves that are adequate for the payment of all Taxes not yet due and payable, whether or not disputed or accrued, as applicable. Except as set forth in Seller Disclosure Schedule 3.7(a), (i) the federal income tax returns of Seller and the Seller Subsidiary have not been examined by the Internal Revenue Service (the "IRS") (or are closed to examination due to the expiration of the applicable statute of limitations), and (ii) the Tennessee franchise tax returns of Seller and the Seller Subsidiary have not been examined by applicable authorities (or are closed to examination due to the expiration of the statute of limitations), and in the case of both (i) and (ii) no deficiencies were asserted as a result of such examinations which have not been resolved and paid in full. All Returns filed (and until the Effective Time to be filed) are or will be, as applicable, complete and accurate. There are no audits or other administrative or court proceedings presently pending nor any other disputes pending, or claims asserted in writing for, Taxes or assessments upon Seller or the Seller Subsidiary, and no taxing authority has given written notice of the commencement of any audit, examination or deficiency action. Neither Seller nor the Seller Subsidiary has given any currently outstanding waivers or comparable consents regarding the application of the
- (b) Except as set forth in Seller Disclosure Schedule 3.7(b), neither Seller nor the Seller Subsidiary (i) has requested any extension of time within which to file any Return which Return has not since been filed, (ii) is a party to any written or unwritten agreement, arrangement or understanding providing for the allocation or sharing of, or indemnification with respect to, Taxes, (iii) is required to include in income any adjustment pursuant to Section 481(a) of the Code by reason of a voluntary change in accounting method initiated by Seller or the Seller Subsidiary (nor does Seller have any Knowledge that the IRS has proposed any such adjustment or change of accounting method), (iv) has filed a consent pursuant to Section 341(f) of the Code or agreed to have Section 341(f)(2) of the Code apply, (v) has been a "distributing corporation" or a "controlled corporation" in a transaction intended to qualify under Section 355(a) of the Code within the past five years, (vi) has ever been a member of a consolidated, combined or unitary Tax group (other than a group of which Seller is or was the common parent) or (vii) has ever engaged in any "listed transaction" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).
- (c) For purposes of this Agreement, "**Taxes**" shall mean all taxes, charges, fees, levies or other assessments, including all net income, gross income, gross receipts, sales, use, ad valorem, transfer, franchise, profits, license, withholding, payroll, employment (including withholding, payroll and employment taxes required to be withheld with respect to income paid to employees), excise, estimated, severance, stamp, occupation, property or other taxes, customs duties, fees, assessments or charges of any kind whatsoever, together with any interest and any penalties, additions to tax or additional amounts imposed by any taxing authority (domestic or foreign).

- (d) No liens for Taxes exist with respect to any of the assets or properties of Seller or Seller Subsidiary, except for liens for Taxes not yet due and payable.
- (e) Neither Seller nor Seller Subsidiary is, or has been, an S corporation within the meaning of Section 1361(a) of the Code.
- (f) None of the assets of Seller or Seller Subsidiary (i) is tax-exempt use property within the meaning of Section 168(h) of the Code, (ii) directly or indirectly secures any debt the interest on which is exempt under Section 103(a) of the Code or (iii) is property that is required to be treated as being owned by a person (other than Seller or Seller Subsidiaries) pursuant to the provisions of Section 168(f)(8) of the Internal Revenue Code of 1954, as amended, as in effect immediately before the enactment of the Tax Reform Act of 1986.
- (g) Neither Seller nor Seller Subsidiary will be required to include in a taxable period ending after the Effective Time taxable income attributable to income that economically accrued in a taxable period ending on or before the Effective Time as a result of the installment method of accounting, the completed contract method of accounting, any method of reporting revenue from contracts which are required to be reported on the percentage of completion method (as defined in Section 460(b) of the Code) but that were reported using another method of accounting, or any other method of accounting.
- (h) Neither Seller nor Seller Subsidiary is a party to any contract or plan which, individually or collectively with respect to any person, could give rise to the payment of any amount that would not be deductible by Seller or Seller Subsidiary, by reason of Section 162(a)(1), 162(m) or 280G of the Code.
- (i) Neither Seller nor Seller Subsidiary is a partner or a member of any partnership or joint venture, or any other entity classified as a partnership for federal income tax purposes.
- (j) Seller and Seller Subsidiary have disclosed on their federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of federal income tax within the meaning of Section 6662 of the Code.

3.8 Employee Benefit Plans.

- (a) Each employee benefit plan or arrangement of Seller or the Seller Subsidiary which is an "employee benefit plan" within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), is listed in Seller Disclosure Schedule 3.8(a) and each bonus, deferred compensation, pension (including an "employee pension benefit plan" within the meaning of Section 3(2) of ERISA ("Pension Plan")), retirement, profit-sharing, thrift, savings, employee stock ownership, stock bonus, stock purchase, restricted stock and stock option plan, employment or severance contract and all other employee benefit plans, practices or arrangements that cover current or former officers or employees ("Employees") or current or former directors of it and the Seller Subsidiary, whether individually or in the aggregate or by group or class, whether written or unwritten, qualified or non-qualified, including all amendments, supplements or other related documents thereto, are listed in Seller Disclosure Schedule 3.8(a) (the "Seller Plans"). Seller has previously furnished to Acquiror true and complete copies or descriptions of each Seller Plan together, if applicable, with (i) the most recent actuarial and financial reports prepared with respect to any qualified Seller Plans, (ii) the three most recent annual reports filed with any Governmental Entity, and (iii) all rulings and determination letters and a description of any open requests for rulings or letters that pertain to any qualified Seller Plans.
- (b) Each Seller Plan has been operated in compliance in all respects with the applicable provisions of ERISA, the Code, all regulations, rulings and announcements promulgated or issued

thereunder, and all other applicable governmental laws and regulations. Each of its Seller Plans which is intended to be qualified under Section 401(a) of the Code has received a favorable determination letter from the IRS covering all tax law changes prior to the Economic Growth and Tax Relief Reconciliation Act of 2001 or has applied to the IRS for such favorable determination letter within the applicable remedial amendment period under Section 401(b) of the Code. As of the date hereof, there is no pending or, to Seller's Knowledge, threatened claim, administrative proceeding or litigation relating to its Seller Plans except claims for benefits arising in the ordinary course of the administration of the plans. Neither Seller nor the Seller Subsidiary has engaged in a transaction with respect to any Seller Plan subject to ERISA (an "ERISA Plan") that could subject it or the Seller Subsidiary to a tax or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA. Neither Seller nor the Seller Subsidiary has incurred a tax or penalty imposed by Section 4980F of the Code or Section 502 of ERISA. No liability under Title IV of ERISA has been incurred by Seller or the Seller Subsidiary, with respect to a Seller Plan currently or formerly maintained by any of them, or the single-employer plan of any entity ("ERISA Affiliate Plan") which is considered one employer with it under Section 4001 of ERISA or Section 414 of the Code ("ERISA Affiliate"). No plan which is subject to Title IV of ERISA has ever been sponsored, maintained or contributed to by Seller or any ERISA affiliate of Seller. Seller and the Seller Subsidiary have neither contributed to nor been obligated to contribute to any "multi-employer plan" within the meaning of Section 3(37) of ERISA, regardless of whether based on contributions of an ERISA Affiliate.

- (c) All contributions required to be made by Seller or the Seller Subsidiary under the terms of any of its Seller Plans, as of the date hereof, have been timely made or have been reflected on its financial statements referred to in Section 3.4. Neither any Pension Plan nor any single-employer plan of an ERISA Affiliate has an "accumulated funding deficiency" (whether or not waived) within the meaning of Section 412 of the Code or Section 302 of ERISA and no ERISA Affiliate has an outstanding funding waiver. It is not reasonably anticipated that required minimum contributions to any Pension Plan under Section 412 of the Code will be materially increased by application of Section 412(1) of the Code. Neither Seller nor the Seller Subsidiary has provided, or is required to provide, security to any Pension Plan pursuant to Section 401(a)(29) of the Code.
- (d) Except as disclosed in Section 3.8(d) of the Seller Disclosure Schedule, neither Seller nor the Seller Subsidiary has any obligation to provide health and life benefits under any Seller Plan for any period after the termination of employment, except as may be required by Section 4980B of the Code or Section 601 of ERISA. Seller and the Seller Subsidiary may amend or terminate any health or life benefit plan maintained by Seller or the Seller Subsidiary at any time without incurring any liability thereunder other than in respect of claims incurred prior to such amendment or termination.
- (e) Except as disclosed in Section 3.8(e) of the Seller Disclosure Schedule, there has been no amendment to, announcement by Seller or the Seller Subsidiary relating to, or change in employee participation or coverage under, any Seller Plan which would increase the expense of maintaining such Seller Plan above the level of the expense incurred therefor for the most recent fiscal year. Except as set forth in Seller Disclosure Schedule 3.8(e), neither the execution of this Agreement, stockholder approval of this Agreement nor the consummation of the transactions contemplated hereby (individually or in conjunction with any other event) will (i) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits or increase in the amounts payable or result in any other material obligation pursuant to any Seller Plan; (ii) limit or restrict its right or, after the consummation of the transactions contemplated hereby, the right of Acquiror (as defined in Section 5.13(a)(1)) to merge, amend or terminate any Seller Plan; (iii) entitle any Employee to severance pay or any increase in severance pay upon any termination of employment after the date hereof; (iv) result in any payment under any Seller Plan which would not be deductible under Section 162(m) or

Section 280G of the Code; or (v) cause it or any of its Subsidiaries to record additional compensation expense on its income statement with respect to any outstanding stock option or other equity-based award.

(f) Except as set forth on Seller Disclosure <u>Schedule 3.8(f)</u>, with respect to each Seller Plan that is subject to Section 302 of ERISA or Section 412 or 4971 of the Code: (i) there does not exist any accumulated funding deficiency within the meaning of Section 412 of the Code or Section 302 of ERISA, whether or not waived; and (ii) the fair market value of the assets of such Plan equals or exceeds the termination liabilities of such plan.

3.9 Regulatory Reports.

Seller and the Seller Subsidiary have duly filed with the TCFI, the FDIC and the FRB in correct form the monthly, quarterly and annual reports required to be filed under applicable laws and regulations, and Seller has delivered or made available to Acquiror accurate and complete copies of such reports. Seller Disclosure Schedule 3.9 lists all examinations of Seller or the Seller Subsidiary conducted by the applicable bank regulatory authorities since January 1, 2000 and the dates of any responses thereto submitted by Seller. Except as set forth in Section 3.9 of the Seller Disclosure Schedule, in connection with the most recent examinations of Seller or the Seller Subsidiary by the applicable bank regulatory authorities, neither Seller nor the Seller Subsidiary was required to correct or change any action, procedure or proceeding which Seller or the Seller Subsidiary believes has not been now corrected or changed as required.

3.10 <u>Seller Information</u>. None of the information relating to Seller and the Seller Subsidiary to be provided by Seller or the Seller Subsidiary for use in the registration statement/proxy statement on Form S-4 to be filed by Acquiror with the Securities and Exchange Commission (the "Commission") in connection with the solicitation of the approval of this Agreement, the Plan of Merger and the transactions contemplated hereby and thereby by the stockholders of Seller, as amended or supplemented (or on any successor or other appropriate schedule or form) (the "Registration Statement") will, as of the date such Registration Statement is mailed to the stockholders of Seller and up to and including the date of the Special Meeting, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading (except that this representation shall not apply to any information in the Registration Statement supplied by Acquiror for use therein), provided that information as of a later date shall be deemed to modify information as of an earlier date.

3.11 Compliance with Applicable Law.

- (a) Each of Seller and the Seller Subsidiary has all permits, licenses, certificates of authority, orders and approvals of, and has made all filings, applications and registrations with Governmental Entities that are required in order to permit it to carry on its business as it is presently being conducted; all such permits, licenses, certificates of authority, orders and approvals are in full force and effect; and to the Knowledge of Seller, no suspension or cancellation of any of the same is threatened.
- (b) Neither Seller nor the Seller Subsidiary is in violation of its respective Charter or other governing instrument or Bylaws, or of any applicable federal, state or local law or ordinance or any order, rule or regulation of any Governmental Entity, or in default with respect to any order, writ, injunction or decree of any court, or in default under any order, license, regulation or demand of any Governmental Entity; and neither Seller nor the Seller Subsidiary has received any written notice from any Governmental Entity asserting that Seller or the Seller Subsidiary is in violation of any of the foregoing.

Except as set forth in Section 3.11(b) of the Seller Disclosure Schedule, neither Seller nor the Seller Subsidiary is subject to any regulatory or supervisory cease and desist order, agreement, written directive, memorandum of understanding or written commitment (other than those of general applicability to all banks issued by Governmental Entities), and none of them has received any written communication requesting that they enter into any of the foregoing.

3.12 <u>Deposit Insurance and Other Regulatory Matters</u>. The deposit accounts of the Seller Subsidiary are insured by the FDIC to the maximum extent permitted by the Federal Deposit Insurance Act, as amended (the "FDIA"), and the Seller Subsidiary has paid all premiums and assessments required by the FDIA and the regulations thereunder. No action, suit or proceeding is pending or, to the Knowledge of Seller, has been threatened by the FDIC against Seller or the Seller Subsidiary with respect to the termination of such insurance.

3.13 Certain Contracts.

- (a) Except as disclosed in Seller Disclosure Schedule 3.13(a), neither Seller nor the Seller Subsidiary is a party to, is bound or affected by, receives or is obligated to pay benefits under (i) any agreement, arrangement or commitment, including any agreement, indenture or other instrument relating to the borrowing of money by Seller or the Seller Subsidiary or the guarantee by Seller or the Seller Subsidiary of any obligation except for deposit liabilities, federal funds purchased, repurchase agreements, trade payables less than \$5,000.00 per item in the ordinary course of business; (ii) any agreement, arrangement or commitment relating to the employment of a consultant or the employment, retirement, election or retention in office of any present or former director, officer or employee of Seller or the Seller Subsidiary; (iii) any contract, agreement or understanding with a labor union; (iv) any agreement, arrangement or understanding pursuant to which any payment (whether of severance pay or otherwise) became or may become due to any director, officer or employee of Seller or the Seller Subsidiary upon execution of this Agreement or upon or following consummation of the transactions contemplated by this Agreement (either alone or in connection with the occurrence of any additional acts or events); (v) any agreement, arrangement or understanding to which Seller or the Seller Subsidiary is a party or by which any of the same is bound which limits the freedom of Seller or the Seller Subsidiary to compete in any line of business or with any person, or that involve any restriction of the geographic area in which, or method by which, they may carry on their business (other than as may be required by law or any regulatory agency); (vi) any assistance agreement, supervisory agreement, memorandum of understanding, consent order, cease and desist order or condition of any regulatory order or decree with or by the FRB or any other regulatory agency; (vii) any joint venture, partnership or similar agreement, arrangement or understanding providing for the sharing of profits, losses, costs or liabilities by Seller or the Seller Subsidiary with any other person; or (viii) any other agreement, arrangement or understanding to which Seller or the Seller Subsidiary is a party and which is material to the business, operations, assets or financial condition of Seller and the Seller Subsidiary taken as a whole (excluding loan agreements or agreements relating to deposit accounts); in each of the foregoing cases whether written or oral; (each such agreement listed, or required to be listed, in this Section 3.13(a) is referred to herein as a "Seller Agreement"). Seller has no obligation other than as set forth in clause (vii) of Seller Disclosure Schedule 3.13(a) to make any additional capital contributions with respect to any matter described in clause (vii) of Seller Disclosure Schedule 3.13(a).
- (b) Neither Seller nor the Seller Subsidiary is in default or in non-compliance under any Seller Agreement and there has not occurred any event that with the lapse of time or the giving of notice, or both, would constitute such a default or non-compliance. Each Seller Agreement is legal, valid, binding and enforceable against Seller or the Seller Subsidiary and, to the Knowledge of Seller, the other parties thereto in accordance with their respective terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally, and

except that the availability of equitable remedies (including specific performance) is within the discretion of the appropriate court. With respect to each Seller Agreement, such Seller Agreement is in full force and effect in accordance with its terms; all rents and other monetary amounts that may have become due and payable thereunder have been paid; and the Merger will not constitute a default or cause for termination or modification of such Seller Agreement.

3.14 Properties and Insurance.

- (a) All real and personal property owned by Seller or the Seller Subsidiary or presently used by either of them in their respective business is in an adequate condition (ordinary wear and tear excepted) and is sufficient to carry on the business of Seller and the Seller Subsidiary in the ordinary course of business consistent with their past practices. Seller and the Seller Subsidiary have good and, as to owned real property, marketable title to all material assets and properties, whether real or personal, tangible or intangible, reflected in Seller's consolidated unaudited balance sheet as of December 31, 2003 or owned and acquired subsequent thereto (except to the extent that such assets and properties have been disposed of for fair value in the ordinary course of business since December 31, 2003), subject to no encumbrances, liens, mortgages, security interests or pledges, except (i) those items that secure liabilities that are reflected in said consolidated balance sheet or the notes thereto or have been incurred in the ordinary course of business after the date of such consolidated balance sheet; (ii) statutory liens for amounts not yet delinquent or which are being contested in good faith; (iii) such encumbrances, liens, mortgages, security interests, pledges and title imperfections that are not in the aggregate material to the business, operations, assets or financial condition of Seller and the Seller Subsidiary taken as a whole; and (iv) with respect to owned real property, title imperfections noted in title reports delivered to Acquiror prior to the date hereof. Seller and the Seller Subsidiary as lessees have the right under valid and subsisting leases to occupy, use, possess and control all property leased by them in all respects as presently occupied, used, possessed and controlled by Seller and the Seller Subsidiary, and the consummation of the transactions contemplated hereby and by the Plan of Merger will not affect any such right. Seller Disclosure Schedule 3.14(a) sets forth an accurate listing of each lease pursuant to which Seller or the Seller Subsidiary acts as lessor or lessee, including the expiration date and the terms of any renewal options which relate to the same. Neither Seller nor the Seller Subsidiary has any legal obligation, absolute or contingent, to any other person to sell or otherwise dispose of any substantial part of its assets or to sell or dispose of any of its assets except in the ordinary course of business consistent with past practices.
- (b) Seller Disclosure Schedule 3.14(b) sets forth a list of all policies of fire, theft, public liability and other insurance (including, without limitation, fidelity bonds insurance) maintained by Seller and the Seller Subsidiary at the date thereof. The business operations and all insurable properties and assets of Seller and the Seller Subsidiary are insured for their benefit against all risks which, in the reasonable judgment of the management of Seller, should be insured against, in each case, under valid, binding and enforceable policies or bonds issued by insurers of recognized responsibility, in such amounts with such deductibles and against such risks and losses as are, in the opinion of the management of Seller, adequate for the business engaged in by Seller and the Seller Subsidiary. As of the date hereof, neither Seller nor the Seller Subsidiary has received any written notice of cancellation or written notice of a material amendment of any such insurance policy or bond or is in default under such policy or bond, no coverage thereunder is being disputed, and all material claims thereunder have been or will be filed in a timely fashion.

3.15 Environmental Matters. For purposes of this Agreement, the following terms shall have the indicated meaning:

"Environmental Law" means any federal, state or local law, statute, ordinance, rule, regulation, code, license, permit, authorization, approval, consent, order, judgment, decree, injunction or agreement with any Governmental Entity relating to (1) the protection, preservation or restoration of the environment (including air, water vapor, surface water, groundwater, drinking water supply, surface soil, subsurface soil, plant and animal life or any other natural resource); and/or (2) the use, storage, recycling, treatment, generation, transportation, processing, handling, labeling, production, release or disposal of Hazardous Substances. The term "Environmental Law" includes (1) the Comprehensive Environmental Response, Compensation and Liability Act, as amended, 42 U.S.C. Section 9601, et seq.; the Resource Conservation and Recovery Act, as amended, 42 U.S.C. Section 6901, et seq.; the Clean Air Act, as amended, 42 U.S.C. Section 7401, et seq.; the Federal Water Pollution Control Act, as amended, 33 U.S.C. Section 1251, et seq.; the Toxic Substances Control Act, as amended, 15 U.S.C. Section 9601, et seq.; the Emergency Planning and Community Right to Know Act, 42 U.S.C. Section 11001, et seq.; the Safe Drinking Water Act, 42 U.S.C. Section 300f, et seq.; and all comparable state and local laws, and (2) any common law (including common law that may impose strict liability) that may impose liability or obligations for injuries or damages due to, or threatened as a result of, the presence of or exposure to any Hazardous Substance.

"Hazardous Substance" means any substance presently listed, defined, designated or classified as hazardous, toxic, radioactive or dangerous, or otherwise regulated, under any Environmental Law, whether by type or by quantity, including any regulated material containing any such substance as a component. Hazardous Substances include, without limitation, petroleum (including crude oil or any fraction thereof), asbestos, radioactive material and polychlorinated biphenyls.

"Loan Portfolio Properties and Other Properties Owned" means those properties owned, leased or operated by a person or its subsidiaries or those properties which serve as collateral for loans owned by a person or its subsidiaries.

- (a) Except as set forth on Seller Disclosure <u>Schedule 3.15</u>, neither Seller nor the Seller Subsidiary has been or is in violation of or liable under any Environmental Law.
- (b) To Seller's Knowledge, except as set forth on Seller Disclosure <u>Schedule 3.15</u>, none of the Loan Portfolio Properties and Other Properties Owned by Seller or the Seller Subsidiary has been or is in violation of or liable under any Environmental Law.
- (c) Except as set forth on Seller Disclosure Schedule 3.15, there are no actions, suits, demands, notices, claims, investigations or proceedings pending or, or to the Knowledge of Seller threatened, relating to the Loan Portfolio Properties and Other Properties Owned by Seller or the Seller Subsidiary under any Environmental Law, including any notices, demand letters or requests for information from any federal or state environmental agency relating to any such liabilities under or violations of Environmental Law.
- 3.16 Allowance for Loan Losses and Real Estate Owned. The allowance for loan losses reflected on Seller's consolidated balance sheets included in the consolidated financial statements referred to in Section 3.4 hereof was, at the time recorded, or will be in the case of subsequently delivered financial statements, as the case may be, in the opinion of Seller's management adequate in all material respects as of their respective dates under the requirements of GAAP to provide for reasonably anticipated losses on outstanding loans net of recoveries. The real estate owned reflected on the consolidated balance sheets included in the consolidated financial statements referred to in Section 3.4 hereof was, at the time recorded, or will be in the case of subsequently delivered financial statements, as the case may be, carried at the lower of cost or fair value, or the lower of cost or net realizable value, as required by GAAP.

- 3.17 <u>Minute Books</u>. Since January 1, 2000, the minute books, including any attachments thereto, of Seller and the Seller Subsidiary contain complete and accurate records of all meetings and other corporate action held or taken by their respective Boards of Directors (including committees of their respective Boards of Directors) and stockholders.
- 3.18 <u>Affiliate Transactions</u>. Except as disclosed in Seller Disclosure <u>Schedule 3.18</u> and except as specifically contemplated by this Agreement, since January 1, 2000, neither Seller nor the Seller Subsidiary has engaged in or agreed to engage in (whether in writing or orally) any transaction with any "affiliate," as such term is defined in Rule 405 under the 1933 Act.
- 3.19 <u>Internal Controls</u>. Seller and the Seller Subsidiary have maintained a system of internal accounting controls sufficient to provide reasonable assurances that (a) transactions are executed in accordance with management's general or specific authorizations, (b) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles and to maintain accountability for assets, (c) access to assets is permitted only in accordance with management's general or specific authorization, and (d) the recorded accountability for assets is compared with the existing assets at reasonable intervals, and appropriate action is taken with respect to any differences.
- 3.20 Risk Management Instruments. All interest rate swaps, caps, floors, option agreements, futures and forward contracts and other similar risk management arrangements (collectively, the "Risk Management Instruments"), whether entered into for Seller's own account or for the account of one or more of the Seller Subsidiary or its customers, were entered into (i) in the opinion of Seller's management, in accordance with prudent business practices and all applicable laws, rules, regulations and regulatory policies and (ii) with counterparties believed to be financially responsible at the time; and each of the Risk Management Instruments constitutes the valid and legally binding obligation of Seller or the Seller Subsidiary, enforceable in accordance with the terms of such Risk Management Instrument (except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally, and except that the availability of equitable remedies (including specific performance) is within the discretion of the appropriate court), and is in full force and effect. Seller and the Seller Subsidiary have duly performed in all material respects all of their material obligations under Risk Management Instruments to the extent such obligations to perform have accrued; and there are no material breaches, violations or defaults, by any party thereunder.
- 3.21 <u>Fairness Opinion</u>. As of the date of this Agreement, Seller has received the written opinion, dated the date of this Agreement, of its financial advisor, Alex Sheshunoff & Co., to the effect that the Merger Consideration is fair, from a financial point of view, to the holders of Seller Common Stock. Seller has furnished an accurate and complete copy of such opinion to Acquiror.
- 3.22 <u>Broker Fees</u>. Except as set forth in Seller Disclosure <u>Schedule 3.22</u>, none of Seller, the Seller Subsidiary or any of the respective directors or officers of such companies has employed any consultant, broker or finder or incurred any liability for any consultant's, broker's or finder's fees or commissions in connection with any of the transactions contemplated by this Agreement.
- 3.23 <u>Loans</u>. All of the loans on the books of Seller and the Seller Subsidiary are valid and properly documented in all material respects and were made in the ordinary course of business, and the security therefor, if any, is valid and properly perfected in all material respects, and no collateral has been released from the lien granted to Seller and the Seller Subsidiary with respect to any such loans unless approved by Seller and the Seller Subsidiary and documented in their files. Neither the terms of such loans nor any of the loan documentation nor the manner in which such loans have been originated,

administered and serviced nor Seller's procedures and practices of approving or rejecting loan applications violates in any material respect any law applicable thereto. Seller and the Seller Subsidiary have full power and authority to hold such loans and have good and valid title to all such loans, free and clear of any liens, and the principal balance of each such loan as shown on the books and records of Seller and the Seller Subsidiary is true and correct as of the last date shown thereon. Seller and the Seller Subsidiary have complied in all material respects with all of their obligations under such loans as to which compliance is required as of the Effective Time; and to the Knowledge of Seller, each such loan is enforceable against the obligor(s) thereunder in accordance with its terms, in each such case subject to (i) bankruptcy, fraudulent transfer, insolvency, moratorium, reorganization, conservatorship, receivership or other similar laws from time to time in effect relating to or affecting the enforcement of the rights of creditors of FDIC-insured institutions or the enforcement of creditors' rights generally; and (ii) general principles of equity (whether applied in a court of law or in equity). All such loans will continue in full force and effect immediately after the Effective Time, other than any loans that are paid or otherwise satisfied in full after the date hereof and prior to the Effective Time. For purposes of this Section 3.23, the term "loans" includes, without limitation, the documents relating in any way to such loans, including, without limitation, notes, mortgages, security instruments and guaranties.

3.24 <u>Investments</u>. Except for securities pledged for reverse repurchase agreements, interest rate swap, cap and floor contracts or pledged to secure public trust funds and except as set forth on Seller Disclosure <u>Schedule 3.24</u>, none of the investments reflected in the unaudited consolidated balance sheet of Seller as of December 31, 2003 under the heading "Investment Securities," and none of the investments by Seller and the Seller Subsidiary since December 31, 2003, and none of the assets reflected in the unaudited consolidated balance sheet of Seller as of December 31, 2003 or in any unaudited consolidated balance sheet of Seller furnished to Acquiror after December 31, 2003 under the heading "Cash and Due from Banks" is subject to any restriction, whether contractual or statutory, that materially impairs the ability of Seller or the Seller Subsidiary freely to dispose of such investment at any time. With respect to all repurchase agreements of which Seller or the Seller Subsidiary is a party, Seller or the Seller Subsidiary, as the case may be, has a valid, perfected first lien or security interest in the government securities or other collateral securing each such repurchase agreement which equals or exceeds the amount of the debt secured by such collateral under such agreement.

3.25 Employees; Compensation. Seller has disclosed on Seller Disclosure Schedule 3.25 a complete and correct list of the name, age, position, rate of compensation and any incentive compensation arrangements, bonuses or commissions or fringe or other benefits, whether payable in cash or in kind, of (i) each director and employee of Seller and the Seller Subsidiary, (ii) each independent contractor, consultant and agent of Seller and the Seller Subsidiary who has received or is entitled to payments and/or benefits of \$25,000 or more on an annual basis during fiscal year 2003 or who is proposed to receive or be entitled to payments and/or benefits of \$25,000 or more on an annual basis during fiscal year 2004, and (iii) each other person to whom Seller or the Seller Subsidiary pays or provides, or has an obligation, agreement (written or unwritten), policy or practice of paying or providing, retirement, health, welfare or other benefits of any kind or description whatsoever.

3.26 <u>Tax and Regulatory Matters</u>. Neither Seller nor the Seller Subsidiary has taken or agreed to take any action which would or could reasonably be expected to (a) cause the Merger not to constitute a reorganization under Section 368 of the Code or (b) materially impede or delay receipt of any consents of regulatory authorities referred to in Section 5.7 or result in failure of the condition in Section 7.1(c). Neither Seller nor Seller Subsidiary have (a) redeemed any of their respective stock, (b) declared or paid any dividend other than regular dividends, (c) engaged in any spin off or (d) engaged in any other transaction in contemplation of the Merger which would cause Acquiror not to acquire substantially all of the Sellers assets. No Seller Common Stock has been or will be acquired by the Seller Subsidiary prior to and in connection with the Merger. Neither Seller nor Seller Subsidiary has ever made or been required

to make an election under Section 338 of the Code. The exercise price of the Seller Warrants (as defined below) is \$10.00 per share which was the estimated fair market value of Seller Common Stock on each date on which Seller issued each Seller Warrant.

3.27 Intellectual Property.

- (a) Seller has disclosed in Seller Disclosure Schedule 3.27 the Intellectual Property (as defined below) owned by Seller or the Seller Subsidiary, including a complete and accurate list of all material U.S. and foreign (A) trademark or service mark registrations and applications, (B) copyright registrations and applications, and (C) Internet domain names. Neither Seller nor the Seller Subsidiary owns any patents or patent applications. Seller or the Seller Subsidiary owns or has the valid right to use, in each case free and clear of all material liens, all applications, trademarks, service marks, trademark or service mark registrations and applications, trade names, logos, designs, Internet domain names, slogans and general intangibles of like nature, together with all goodwill related to the foregoing, copyrights, copyright registrations, renewals and applications, Software (as defined below), technology, trade secrets and other confidential information, know-how, proprietary processes, formulae, algorithms, models and methodologies, licenses, agreements and all other material proprietary rights (collectively, the "Intellectual Property") used in the business of Seller and the Seller Subsidiary as it currently is conducted. "Software" means any and all (A) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in source code or object code, (B) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (C) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, (D) the technology supporting and content contained on any owned or operated Internet site(s), and (E) all documentation, including user manuals and training materials, relating to any of the foregoing. The ownership or right to use such Intellectual Property or Software of Seller or the Seller Subsidiary (i) has not been challenged in any prior litigation. The consummation of t
- (b) The conduct of the business of Seller and the Seller Subsidiary as currently conducted or planned by Seller to be conducted does not, in any material respect, infringe upon (either directly or indirectly, such as through contributory infringement or inducement to infringe), dilute, misappropriate or otherwise violate any Intellectual Property owned and controlled by any third party.
- (c) To the Knowledge of Seller, no third party is misappropriating, infringing, diluting or violating any material Intellectual Property owned by or licensed to or by Seller or the Seller Subsidiary, and no such claims have been made against a third party by Seller or the Seller Subsidiary.
- (d) Each material item of Software, which is used by Seller or the Seller Subsidiary in connection with the operation of their businesses as currently conducted, is either (A) owned by Seller or the Seller Subsidiary, (B) currently in the public domain or otherwise available to Seller without the need of a license, lease or consent of any third party, or (C) used under rights granted to Seller or the Seller Subsidiary pursuant to a written agreement, license or lease from a third party.
- 3.28 <u>Community Reinvestment Compliance</u>. The Seller Subsidiary is in compliance with all applicable provisions of the Community Reinvestment Act of 1977, as amended (the "**CRA**"), and has received a CRA rating of "Outstanding" in its most recent exam under the CRA. Seller has no Knowledge of the existence of any fact or circumstance or set of facts or circumstances which could be

reasonably expected to result in the Seller Subsidiary failing to be in compliance with such provisions or having its current rating lowered.

- 3.29 <u>No Existing Discussions</u>. As of the date hereof, each of Seller and the Seller Subsidiary is not engaged, directly or indirectly, in any negotiations or discussions with any other person with respect to an Acquisition Transaction (as defined below).
- 3.30 <u>Certain Business Practices</u>. Neither Seller nor the Seller Subsidiary, and (to the Knowledge of Seller) no director, officer, agent or employee of Seller or the Seller Subsidiary, has (a) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (b) made any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns or violated any provision of the Foreign Corrupt Practices Act of 1977, as amended, or (c) made any other unlawful payment.
- 3.31 <u>Continuity of Business Enterprise</u>. Seller operates at least one significant historic business line, or owns at least a significant portion of its historic business assets, in each case within the meaning of Treasury Regulation Section 1.368-1(d).
- 3.32 <u>Full Disclosure</u>. No representation or warranty of the Seller contained in this Agreement, the Seller Disclosure Schedule or in any agreement, document or certificate delivered by Seller to Acquiror or Merger Sub pursuant to this Agreement (a) contains or at the Closing will contain any untrue statement of a material fact or (b) omits or at the Closing will omit to state a material fact necessary to make the statements herein or therein, as applicable, in light of the circumstances under which such statements were or will be made, not misleading.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF ACQUIROR

Acquiror makes the representations and warranties contained in this Article IV, except as set forth in the disclosure schedule attached hereto as Exhibit C (the "Acquiror Disclosure Schedule"). The Acquiror Disclosure Schedule is arranged in sections corresponding to the numbered and lettered sections contained in this Article IV. Disclosure in any section of the Acquiror Disclosure Schedule shall be effectively made whether or not expressly excepted in the corresponding section of this Agreement. The following statements are made as of the date of this Agreement. The phrase "delivered to Seller" or "made available to Seller" or any phrase of similar import means that Acquiror has delivered, provided access to or made certain items available for review and copying to Seller or its counsel. For purposes of this Article IV, the phrase "to the Knowledge of Acquiror" or any phrase of similar import shall be deemed to refer to the actual knowledge of senior executive officers of Acquiror after investigation.

4.1 <u>Corporate Organization</u>. Acquiror is a corporation duly organized, validly existing and in good standing under the laws of the State of Mississippi. Acquiror has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted and is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder. Acquiror is registered as a bank holding company under the Bank Holding Act of 1956, as amended. Acquiror Disclosure <u>Schedule 4.1</u> sets forth true and complete copies of the Articles of Incorporation and Bylaws of Acquiror and Merger Sub as in effect on the date hereof.

4.2 Merger Sub; No Activities.

- (a) Merger Sub (i) is duly organized and validly existing and in good standing under the laws of its jurisdiction of incorporation, (ii) has the corporate power and authority to own or lease all of its properties and assets and to conduct its business as it is now being conducted, and (iii) is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not have a material adverse effect on the business, operations, assets or financial condition of Acquiror and Merger Sub taken as a whole.
- (b) Merger Sub was formed solely for the purpose of engaging in the Merger. Except for obligations or liabilities incurred in connection with its incorporation or organization and the transactions contemplated by this Agreement, Merger Sub does not have any obligations or liabilities of any nature (whether accrued, absolute, contingent or otherwise) and has not engaged in any business activities of any type or kind whatsoever or entered into any agreements or arrangements with any person.

4.3 Authority; No Violation.

- (a) Acquiror and Merger Sub have full corporate power and authority to execute and deliver this Agreement and the Plan of Merger and to consummate the transactions contemplated hereby and thereby in accordance with the terms hereof and thereof. The execution and delivery of this Agreement and the Plan of Merger and the consummation of the transactions contemplated hereby and thereby have been duly and validly approved by the unanimous vote of the Board of Directors of Acquiror and by the written consent of the Board of Directors of Merger Sub and by Acquiror as the sole stockholder of Merger Sub. No other corporate proceedings on the part of Acquiror or Merger Sub are necessary to consummate the transactions so contemplated. Subject to receipt of the regulatory and other approvals described in this Agreement, this Agreement and the Plan of Merger have been duly and validly executed and delivered by Acquiror and Merger Sub and constitute valid and binding obligations of Acquiror and Merger Sub, respectively, enforceable against each of them in accordance with and subject to their terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally, and except that the availability of equitable remedies (including specific performance) is within the discretion of the appropriate court.
- (b) None of the execution and delivery of this Agreement and the Plan of Merger by Acquiror and Merger Sub nor the consummation by Acquiror and Merger Sub of the transactions contemplated hereby and thereby in accordance with the terms hereof and thereof nor compliance by Acquiror and Merger Sub with any of the terms or provisions hereof or thereof will (i) violate any provision of the Articles of Incorporation or other governing instrument or Bylaws of Acquiror, Merger Sub or Acquiror Bank (as defined below) (Merger Sub and Acquiror Bank are referred to herein collectively as the "Acquiror Subsidiaries"); (ii) assuming that the consents and approvals set forth below are duly obtained, violate any statute, code, ordinance, rule, regulation, judgment, order, writ, decree or injunction applicable to Acquiror or any of the Acquiror Subsidiaries or any of their respective properties or assets; or (iii) violate, conflict with, result in a breach of any provisions of, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of, accelerate the performance required by, require the payment of any termination or like fee, or result in the creation of any lien, security interest, charge or other encumbrance upon any of the respective properties or assets of Acquiror or any of the Acquiror Subsidiaries under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which Acquiror or any of the Acquiror Subsidiaries is a party, or by which any of their respective properties or assets may be bound or affected, except, with respect to (ii)

and (iii) above, such as individually or in the aggregate will not have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder. Except for consents and approvals of or filings or registrations with or notices to the Commission, the Secretary of State of the State of Tennessee, the TCFI, the MCB, the FDIC, applicable state securities laws authorities, the American Stock Exchange and the FRB, no consents or approvals of or filings or registrations with or notices to any Governmental Entity or non-governmental third party are required on behalf of Acquiror in connection with (a) the execution and delivery of this Agreement and the Plan of Merger by Acquiror and Merger Sub and (b) the consummation by Acquiror and Merger Sub of the transactions contemplated hereby and by the Plan of Merger.

4.4 Financial Statements.

- (a) Acquiror has previously delivered to Seller copies of the consolidated balance sheets of Acquiror as of December 31, 2002, 2001 and 2000 and the related consolidated statements of income, changes in stockholders' equity and cash flows for the years ended December 31, 2002, 2001 and 2000, in each case accompanied by the audit reports of Ernst & Young, LLP, independent public accountants, as well as the unaudited consolidated balance sheet of Acquiror as of September 30, 2003 and the related unaudited consolidated statements of income, changes in stockholders' equity and cash flows for the three and nine months ended September 30, 2003 and 2002. The consolidated balance sheets of Acquiror referred to herein (including the related notes, where applicable) fairly present the consolidated statements of income, changes in stockholders' equity and cash flows (including the related notes, where applicable) fairly present the results of the consolidated operations, changes in stockholders' equity and cash flows of Acquiror for the respective periods or as of the respective dates set forth therein (it being understood that Acquiror's interim financial statements are not audited and are not prepared with related notes but reflect all adjustments which were, at the time, in the opinion of Acquiror, necessary for a fair presentation of such financial statements).
- (b) Each of the financial statements referred to in this Section 4.4 (including the related notes, where applicable) has been or will be, as the case may be, prepared in accordance with GAAP consistently applied during the periods involved. The books and records of Acquiror and the Acquiror Subsidiaries are being maintained in material compliance with applicable legal and accounting requirements and reflect only actual transactions.
- (c) Except to the extent reflected, disclosed or reserved against in the consolidated financial statements referred to in the first sentence of Section 4.4(a) or the notes thereto or liabilities incurred since September 30, 2003 in the ordinary course of business and consistent with past practice (none of which arises from breach of any contract or agreement, breach of warranty, tort, infringement, violation of any applicable federal, state or local law or ordinance or any litigation or other proceeding or is otherwise a "loss contingency" within the meaning of Statement of Financial Accounting Standards No. 5), none of Acquiror or any of the Acquiror Subsidiaries has any obligation or liability, whether absolute, accrued, contingent or otherwise, material to the business, operations, assets or financial condition of Acquiror and the Acquiror Subsidiaries taken as a whole.
- 4.5 <u>Absence of Certain Changes or Events</u>. Except as set forth in Acquiror Disclosure <u>Schedule 4.5</u>, there has not been any adverse change in the business, operations, assets or financial condition of Acquiror and the Acquiror Subsidiaries taken as a whole since September 30, 2003 that would have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder, other than: (i) any change in banking or similar laws, rules or regulations of general applicability to banks, savings institutions or their holding companies or interpretations thereof by Governmental Entities; (ii) changes in GAAP that are generally applicable to the banking or savings

industries; (iii) reasonable expenses incurred in connection with the transactions contemplated hereby; or (iv) changes attributable to or resulting from changes in general economic conditions, including changes in the prevailing level of interest rates.

- 4.6 <u>Legal Proceedings</u>. Except as set forth in Acquiror Disclosure <u>Schedule 4.6</u>, none of Acquiror or any of the Acquiror Subsidiaries is a party to any, and there are no pending or, to the Knowledge of Acquiror, threatened legal, administrative, arbitration or other proceedings, claims, actions or governmental investigations of any nature against Acquiror or any of the Acquiror Subsidiaries, except such proceedings, claims, actions or governmental investigations which in the good faith judgment of Acquiror will not have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder. None of Acquiror or any of the Acquiror Subsidiaries is a party to any order, judgment or decree which would reasonably be expected to have such a material adverse effect.
- 4.7 <u>Acquiror Information</u>. None of the information relating to Acquiror and the Acquiror Subsidiaries to be provided by Acquiror or the Acquiror Subsidiaries for use in the Registration Statement will, as of the date such Registration Statement is mailed to the stockholders of Seller and up to and including the date of the Special Meeting, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, provided that information as of a later date shall be deemed to modify information as of an earlier date.
- 4.8 <u>Deposit Insurance and Other Regulatory Matters</u>. The deposit accounts of The Peoples Bank & Trust Company, the financial institution subsidiary of Acquiror ("**Acquiror Bank**"), are insured by the FDIC to the maximum extent permitted by the FDIA, and Acquiror Bank has paid all premiums and assessments required by the FDIA and the regulations thereunder.
- 4.9 <u>Capital; Availability of Funds; Acquiror Shares</u>. On the date hereof, the Acquiror Bank is, and on the Closing Date, the Acquiror Bank will be, "well capitalized" as such term is defined in the rules and regulations promulgated by the FDIC, and on the date hereof, Acquiror is, and at the Effective Time, Acquiror will be, "well capitalized" as such term is defined in the rules and regulations promulgated by the FRB. Acquiror will have available to it on and as of the date set forth in Section 2.2(e) sources of capital and financing sufficient and a sufficient number of shares of Acquiror Common Stock to pay the aggregate Merger Consideration and to pay any other amounts payable pursuant to this Agreement and to effect the transactions contemplated hereby. The shares of Acquiror to be issued to pay the aggregate Merger Consideration pursuant to this Agreement will have been duly authorized and, when issued in accordance with the terms of this Agreement, will be validly issued, fully paid and non-assessable.
- 4.10 <u>Broker Fees</u>. Except as set forth in Acquiror Disclosure <u>Schedule 4.10</u>, neither Acquiror nor Merger Sub nor any of their respective directors or officers has employed any consultant, broker or finder or incurred any liability for any consultant's, broker's or finder's fees or commissions in connection with any of the transactions contemplated by this Agreement.
- 4.11 Securities Documents and Regulatory Reports. Acquiror has previously delivered or made available to Seller an accurate and complete copy of each final registration statement, prospectus, annual, quarterly or current report and definitive proxy statement or other communication (other than general advertising materials) filed pursuant to the 1933 Act or the 1934 Act, or mailed by Acquiror to its stockholders as a class since January 1, 2002, and each such final registration statement, prospectus, annual, quarterly or current report and definitive proxy statement or other communication, as of its date, complied in all material respects with all applicable statutes, rules, and regulations and did not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or

necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading; provided that information as of a later date filed publicly or provided to Seller prior to the date hereof shall be deemed to modify information as of an earlier date. Since January 1, 2002, Acquiror and each of the Acquiror Subsidiaries has duly filed with the FDIC and the FRB in correct form the monthly, quarterly and annual reports required to be filed under applicable laws and regulations, and Acquiror has made available to Seller accurate and complete copies of such reports.

4.12 Compliance with Applicable Law.

- (a) Acquiror and each of the Acquiror Subsidiaries has all permits, licenses, certificates of authority, orders and approvals of, and has made all filings, applications and registrations with Governmental Entities that are required in order to permit it to carry on its business as it is presently being conducted and the absence of which could have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder; all such permits, licenses, certificates of authority, orders and approvals are in full force and effect; and to the Knowledge of Acquiror, no suspension or cancellation of any of the same is threatened.
- (b) Except as set forth in Acquiror Disclosure Schedule 4.12(b), neither Acquiror nor any of the Acquiror Subsidiaries is in violation of its respective Articles of Incorporation or other governing instrument or Bylaws, or of any applicable federal, state or local law or ordinance or any order, rule or regulation of any Governmental Entity, or in default with respect to any order, writ, injunction or decree of any court, or in default under any order, license, regulation or demand of any Governmental Entity, any of which violations or defaults could have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder; and neither Acquiror nor any of the Acquiror Subsidiaries has received any written notice from any Governmental Entity asserting that Acquiror or any Acquiror Subsidiary is in violation of any of the foregoing which could have such a material adverse effect. Neither Acquiror nor any Acquiror Subsidiary is subject to any regulatory or supervisory cease and desist order, agreement, written directive, memorandum of understanding or written commitment (other than those of general applicability to all banks issued by Governmental Entities), and none of them has received any written communication requesting that they enter into any of the foregoing.
- 4.13 <u>Full Disclosure</u>. No representation or warranty of Acquiror contained in this Agreement, the Acquiror Disclosure Schedule or in any agreement, document or certificate delivered by Acquiror to Seller pursuant to this Agreement (a) contains or at the Closing will contain any untrue statement or a material fact or (b) omits or at the Closing will omit to state a material fact necessary to make the statements herein or therein, as applicable, in light of the circumstances under which such statements were or will be made, not misleading.

ARTICLE V COVENANTS OF THE PARTIES

5.1 <u>Conduct of the Business of Seller</u>. During the period from the date hereof to the Effective Time, Seller shall, and shall cause the Seller Subsidiary to, conduct its and their businesses only in the ordinary course and consistent with past practice and prudent banking practice or as required hereunder, except with the prior written consent of Acquiror, which consent shall not be unreasonably withheld, conditioned or delayed. Seller shall use its best efforts to (i) preserve its business organization and that of the Seller Subsidiary intact, (ii) keep available to itself and Acquiror the present services of the employees of Seller and the Seller Subsidiary, and (iii) preserve for itself and Acquiror the goodwill of customers and other third parties with whom business relationships exist.

- 5.2 <u>Negative Covenants of Seller</u>. Seller agrees that from the date hereof to the Effective Time, except as otherwise approved by Acquiror in writing or as permitted or required by this Agreement, Seller will not, nor will Seller permit the Seller Subsidiary to, without the prior written consent of Acquiror, which shall not be unreasonably withheld, conditioned or delayed:
 - (i) change any provision of the Charter or other governing instrument or Bylaws of Seller or the Seller Subsidiary;
 - (ii) except for the issuance of the Seller Common Stock pursuant to the present terms of the Outstanding Seller Stock Options and the Outstanding Seller Warrants, change the number of shares of its authorized or issued capital stock or issue or grant any shares of its capital stock or any option, warrant, call, commitment, subscription, award, right to purchase or agreement of any character relating to the authorized or issued capital stock of Seller or the Seller Subsidiary, or any securities convertible into shares of such capital stock, or split, combine or reclassify any shares of its capital stock, or redeem or otherwise acquire any shares of such capital stock;
 - (iii) declare, set aside or pay any dividend or other distribution (whether in cash, stock or property or any combination thereof) in respect of the capital stock of Seller or the Seller Subsidiary;
 - (iv) grant any severance or termination pay (other than pursuant to binding contracts of Seller in effect on the date hereof and disclosed to Acquiror on Seller Disclosure Schedule 3.13(a)), to, or enter into or amend any employment, consulting or compensation agreement with, any of its directors, officers, employees or consultants; or award any increase in compensation or benefits to its directors, officers, employees or consultants;
 - (v) enter into or modify any pension, retirement, stock option, stock purchase, stock grant, stock appreciation right, savings, profit-sharing, deferred compensation, consulting, bonus, group insurance or other employee benefit, incentive or welfare contract, plan or arrangement, or any trust agreement related thereto, in respect of any of its directors, officers, employees or consultants, other than any change that is required by law or that, in the opinion of counsel, is necessary or advisable to maintain the tax-qualified status of any tax-qualified plan; or make any contributions to any other defined contribution plan or any defined benefit pension or retirement plan other than in the ordinary course of business consistent with past practice or plan terms;
 - (vi) sell or dispose of any assets or knowingly incur any liabilities other than in the ordinary course of business consistent with past practices and policies, or acquire in any manner whatsoever (other than to realize upon collateral for a defaulted loan) any business or entity;
 - (vii) make any capital expenditures in excess of \$50,000 in the aggregate, other than pursuant to binding commitments existing on the date hereof, other than expenditures necessary to maintain existing assets in good repair and other than as set forth in Seller Disclosure <u>Schedule 5.2(vii)</u>;
 - (viii) file any applications or make any contract with respect to branching or site location or relocation;

- (ix) make any material change in its accounting methods or practices, other than changes required by GAAP, or change any of its methods of reporting income and deductions for federal income tax purposes, except as required by changes in laws or regulations;
- (x) change its lending, investment, deposit or asset and liability management or other banking policies in any respect except as may be required by applicable laws or regulation;
- (xi) make, change or revoke any material Tax election, amend any material Return or settle or compromise any material liability for Taxes;
- (xii) engage in any transaction with an "affiliate," in each case as defined in Section 3.18 hereof other than as set forth in Seller Disclosure Schedule 5.2(xii);
- (xiii) enter into any leveraged arbitrage programs, any futures contract, option or other agreement, or take any action for purposes of hedging the exposure of its interest-earning assets and interest-bearing liabilities to changes in market rates of interest;
- (xiv) originate or acquire any loans or other extensions of credit except for originations (A) in accordance with existing Seller lending policies and (B) any lending commitments outstanding on the date hereof;
- (xv) knowingly take any action or knowingly fail to take any action that would reasonably be expected to adversely affect or delay the ability of Acquiror or Seller to perform its covenants and agreements on a reasonably timely basis under this Agreement or to consummate the transactions contemplated under this Agreement;
- (xvi) knowingly take any action or knowingly fail to take any action that would reasonably be expected to result in any of its representations and warranties contained in Article III of this Agreement not being true and correct in any material respect at the Effective Time;
- (xvii) knowingly take any action which would be reasonably expected to adversely affect or delay the ability of the Acquiror or Seller to obtain any necessary approvals, consents or waivers of any Governmental Entity required for the transactions contemplated hereby or which would reasonably be expected to result in any such approvals, consents or waivers containing any condition or restriction that would materially impair the value of the transaction to Acquiror;
- (xviii)merge with any other corporation or bank or permit any other corporation or bank to merge into it or consolidate with any other corporation or bank;
- (xviv)knowingly fail to comply with any laws, regulations, ordinances or governmental actions applicable to it or the Seller Subsidiary and to the conduct of the business of the Seller and the Seller Subsidiary in a manner adverse to such business; or
- (xvv) agree to do any of the foregoing.
- 5.3 No Solicitation.
- (a) Neither Seller nor the Seller Subsidiary shall, nor shall Seller or the Seller Subsidiary authorize or permit any of its directors, officers or employees or any investment banker, financial advisor,

attorney, accountant or other representative of Seller or the Seller Subsidiary to, directly or indirectly, solicit or hold discussions or negotiations with, or provide any information to, any person, entity or group (other than Acquiror) concerning any Acquisition Transaction (as defined below); provided, however, that nothing contained in this Agreement shall prevent Seller or the Board of Directors of Seller prior to receipt of approval by the stockholders of Seller of this Agreement and the Plan of Merger from (i) providing information in response to a request therefor by a person who has made an unsolicited bona fide written proposal to engage in an Acquisition Transaction (an "Acquisition Proposal") if the Board of Directors of Seller receives from the person so requesting such information an executed confidentiality agreement on customary terms and conditions; (ii) engaging in any negotiations or discussions with any person who has made an unsolicited bona fide written Acquisition Proposal; (iii) failing to recommend or withdrawing its recommendation of this Agreement to its stockholders and/or failing to hold the Special Meeting (as defined in Section 5.8) to consider this Agreement, or (iv) recommending such an Acquisition Proposal to the stockholders of Seller, if and only to the extent that, in each such case referred to in clause (i), (ii), (iii) or (iv) above, (A) Seller's Board of Directors determines in good faith (after consultation with outside legal counsel) that such action would be required in order for its directors to comply with their respective fiduciary duties under applicable law and (B) Seller's Board of Directors determines in good faith (after consultation with its financial advisor) that such Acquisition Proposal, if accepted, is reasonably likely to be consummated, taking into account all legal, financial and regulatory aspects of the proposal and the person making the proposal and would, if consummated, result in a transaction more favorable to Seller's stockholders from a financial point of view than the Merger. An Acquisition Proposal which is received and considered by Seller in compliance with this Section 5.3 and which meets the requirements set forth in clauses (A) and (B) of the preceding sentence is herein referred to as a "Superior Proposal". Seller will communicate to Acquiror in writing (the "Notice") as promptly as practicable (and in no event more than 48 hours after receipt) the terms of any proposal which it may receive in respect of any Acquisition Transaction (including amendments thereto) and shall provide Acquiror with copies of (x) all such written inquiries or proposals and (y) an accurate and complete written synopsis of all such oral inquiries or proposals. For purposes of this Agreement, "Acquisition Transaction" means any offer or proposal by a person or entity other than Acquiror for (i) a merger, tender offer, recapitalization or consolidation, or any similar transaction, involving the Seller or the Seller Subsidiary; (ii) a purchase, lease or other acquisition or assumption of all or a substantial portion of the assets of the Seller or the Seller Subsidiary; (iii) a purchase or other acquisition (including by way of merger, consolidation, share exchange or otherwise) of beneficial ownership (the term "beneficial ownership" for purposes of this Agreement having the meaning assigned thereto in Section 13(d) of the 1934 Act, and the rules and regulations thereunder) of securities representing 20% or more of the voting power of the Seller; or (iv) any substantially similar transaction.

(b) During the 10 days immediately following the Acquiror's receipt of the Notice, Acquiror shall be entitled to match or better the terms of the Superior Proposal described in the Notice. If Acquiror advises Seller in writing during such 10 day period that it has elected to match or better the terms of the Superior Proposal, Seller and Acquiror within two days of such election by Acquiror, shall amend this Agreement and the Plan of Merger to reflect the matched or bettered terms of the Superior Proposal as proposed by Acquiror. Seller shall not terminate this Agreement, and Seller promptly shall notify the person that made or conveyed the Superior Proposal that Acquiror has matched or bettered terms. From and after the date on which Acquiror executes an amendment which matches or betters the Superior Proposal, Seller shall, and shall cause the Seller Subsidiary to, immediately cease and cause to be terminated, and cause its officers, directors, employees, investment bankers, financial advisors, attorneys, accountants and other representatives to, immediately cease and caused to be terminated, all discussions and negotiations regarding the Superior Proposal that has been matched or bettered by Acquiror with any person other than Acquiror. A person that has made a Superior Proposal that has been matched or bettered by Acquiror may make further Acquisition Proposals. During the period commencing on the date that Seller

determines that the Acquisition Proposal meets the requirements set forth in clauses (A) and (B) above and ending on the eleventh day after Acquiror's receipt of the Notice, Seller and the Seller Subsidiary shall not terminate this Agreement unless Seller receives written notice from Acquiror that it does not intend to exercise its right to match or better the Superior Proposal. If Acquiror does not notify Seller of its election by the eleventh day after Acquiror's receipt of the Notice, Seller may proceed with the Superior Proposal as provided in this Agreement.

- (c) Except as expressly permitted by this Section 5.3 and Section 5.8, neither the Board of Directors of Seller nor any committee thereof shall (i) withdraw, modify or qualify, or propose publicly to withdraw, modify or qualify, in a manner adverse to Acquiror, the approval of the Agreement and the Plan of Merger or the Seller Recommendation (as defined in Section 5.8) or take any action or make any statement in connection with the Special Meeting inconsistent with such approval or the Seller Recommendation (collectively, a "Change in the Seller Recommendation"), (ii) approve or recommend, or propose publicly to approve or recommend, any Acquisition Proposal, or (iii) cause Seller to enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement (each, a "Seller Acquisition Agreement") related to any Acquisition Transaction. For purposes of this Agreement, a Change in the Seller Recommendation shall include any approval or recommendation (or public proposal to approve or recommend), by Seller's Board of an Acquisition Proposal, or any failure by Seller's Board to recommend against an Acquisition Proposal. Notwithstanding the foregoing, the Board of Directors of Seller, to the extent that it determines in good faith, after consultation with outside counsel, that the failure to terminate this Agreement in light of a Superior Proposal would constitute a breach of its fiduciary duties under applicable law, may terminate this Agreement in order to concurrently enter into a Seller Acquisition Agreement with respect to such Superior Proposal, but only at a time that is after the compliance by Seller with the terms of Sections 5.3(a), 5.3(b) and 7.4.
- (d) Upon the execution of this Agreement, Seller shall, and shall cause the Seller Subsidiary to, immediately cease and cause to be terminated, and cause its officers, directors and employees, investment bankers, financial advisors, attorneys, accountants and other representatives to, immediately cease and cause to be terminated, all discussions and negotiations, if any, that have taken place prior to the date hereof with any persons with respect to any Acquisition Proposal and, upon request by Acquiror, shall request the return and destruction of all confidential information provided to any such person.
- 5.4 Negative Covenants of Acquiror. Except as expressly provided in this Agreement, or as set forth in Acquiror Disclosure Schedule 5.4, during the period from the date of this Agreement to the Effective Time, Acquiror shall not, and shall cause each Acquiror Subsidiary not to, (i) other than in connection with this Agreement, and except as would not have a material adverse effect on the ability of Acquiror and Merger Sub to perform their obligations hereunder, fail to conduct its business in the ordinary and usual course consistent with past practices and prudent banking practice, or fail to maintain and preserve intact its business organization, properties, leases, employees and advantageous business relationships and retain the services of its officers and key employees; (ii) knowingly take any action which would be reasonably expected to adversely affect or delay the ability of Acquiror or Merger Sub to perform their respective covenants and agreements on a timely basis under this Agreement or to consummate the transactions contemplated by this Agreement; (iii) knowingly take any action or knowingly fail to take any action that would reasonably be expected to result in any of its representations and warranties contained in Article IV of this Agreement not being true and correct in any material respect at the Effective Time; (iv) knowingly take any action which would be reasonably expected to adversely affect or delay the ability of the Acquiror or Merger Sub to obtain any necessary approvals, consents or waivers of any Governmental Entity required for the transactions contemplated hereby or which would reasonably be expected to result in any such approvals, consents or waivers containing any condition or restriction that would materially impair the value of the transaction to Acquiror; or (v) agree to do any of the foregoing.

5.5 <u>Current Information</u>. During the period from the date hereof to the Effective Time, each party will cause one or more of its designated representatives to confer on a monthly or more frequent basis with representatives of the other party regarding its business, operations, assets and financial condition and matters relating to the completion of the transactions contemplated hereby. As soon as reasonably available, but in no event more than 45 days after the end of each calendar quarter (other than the last quarter of such party's fiscal year) ending after the date of this Agreement, Seller will deliver to Acquiror the unaudited consolidated balance sheet of Seller as of the end of each calendar quarter and the related unaudited consolidated statements of income, changes in stockholders' equity and cash flows of Seller for the three months then ended, and, as soon as reasonably available, but in no event more than 90 days after the end of each fiscal year, will deliver to Acquiror the audited consolidated balance sheet of Seller as of the end of each fiscal year and the related audited consolidated statements of income, changes in stockholders' equity and cash flows of Seller for the 12 months then ended. Within 30 days after the end of each fiscal quarter, Seller shall provide Acquiror with a copy of the FDIC Call Report filed with the FDIC.

5.6 Access to Properties and Records; Confidentiality.

- (a) Seller shall permit Acquiror and its representatives reasonable access to its properties and those of the Seller Subsidiary and shall disclose and make available to Acquiror all books, papers and records relating to the assets, properties, operations, obligations and liabilities of Seller and the Seller Subsidiary, including, but not limited to, all books of account (including the general ledger), tax records, minute books of directors' and stockholders' meetings, organizational documents, bylaws, material contracts and agreements, filings with any regulatory authority, litigation files, plans affecting employees, and any other business activities in which Acquiror may have an interest. Neither Seller nor the Seller Subsidiary shall be required to provide access to or to disclose information where such access or disclosure would violate or prejudice the rights of any customer or would contravene any law, rule, regulation, order or judgment. Seller will use its best efforts to obtain waivers of any such restriction and in any event make appropriate substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply. Seller and the Seller Subsidiary shall make their respective executive officers available to confer with Acquiror and its representatives, provided that such access shall be reasonably related to the transactions contemplated hereby and not unduly interfere with normal operations.
- (b) All information furnished previously by Acquiror or Seller in connection with the transactions contemplated by this Agreement or pursuant hereto shall be treated as the sole property of the party furnishing the information until consummation of the Merger and, if such Merger shall not occur, the party receiving the information shall, at the request of the party which furnished such information, either return to the party which furnished such information or destroy all documents or other materials containing, reflecting or referring to such information; shall use its best efforts to keep confidential all such information; shall use such information only for the purpose of consummating the transactions contemplated by this Agreement; and shall not directly or indirectly use such information for any competitive or commercial purposes. The obligations of Acquiror and Seller to keep such information confidential shall continue for one year from the date the proposed Merger is abandoned but shall not apply to (i) any information which (A) was already in the possession of the party receiving the information prior to the disclosure thereof to it by the party furnishing the information; (B) was then generally known to the public; (C) became known to the public through no fault of the party receiving the information; or (D) was disclosed to the party receiving the information by a third-party not known by the receiving party to be bound by an obligation of confidentiality; or (ii) disclosures pursuant to a legal requirement or in accordance with an order of a court of competent jurisdiction.

(c) Notwithstanding anything contained herein to the contrary, each party is permitted to disclose the tax treatment and tax structure of the transactions contemplated hereby at any time on or after the earliest to occur of (i) the date of public announcement of discussions relating to the transactions contemplated hereby, (ii) the date of public announcement of the transactions contemplated hereby and (iii) the date of execution of this Agreement (it being understood by the parties that a "public announcement" shall mean a public announcement permitted by this Agreement (or otherwise permitted by agreement of the parties) as distinguished from a third-party announcement or a leak). This Agreement shall not be construed to limit in any way either party's ability to consult any tax advisor regarding the tax treatment or tax structure of the Merger. These provisions are meant to be interpreted so as to prevent the Merger from being treated as offered under "conditions of confidentiality" within the meaning of the Code and the Treasury Regulations thereunder.

5.7 Regulatory Matters.

- (a) The parties hereto will cooperate with each other and use their best efforts to prepare all necessary documentation (including the Registration Statement), to effect all necessary filings and to obtain all necessary permits, consents, approvals and authorizations of all third parties and Governmental Entities necessary or advisable to consummate the transactions contemplated by this Agreement as soon as practicable. The parties shall each have the right to review in advance all information relating to the other, as the case may be, and any of their respective subsidiaries, which appears in any filing made with, or written material submitted to, any third party or Governmental Entity in connection with the transactions contemplated by this Agreement.
- (b) Each of the parties will furnish each other with all information concerning themselves, their subsidiaries, directors, officers and stockholders and such other matters as may be necessary or advisable in connection with any statement or application made by or on behalf of them, or any of their respective subsidiaries, to any Governmental Entity in connection with the Merger and the other transactions, applications or filings contemplated by this Agreement.
- (c) Each of the parties will promptly furnish each other with copies of written communications received by them or any of their respective subsidiaries from, or delivered by any of the foregoing to, any Governmental Entity in connection with the Merger and the other transactions, applications or filings contemplated by this Agreement.
- 5.8 <u>Approval of Stockholders</u>. Seller shall, subject to the provisions of Section 5.3 and this Section 5.8, (i) take all steps (including participation in the preparation of the Registration Statement in accordance with all applicable requirements and having the Registration Statement cleared by the Commission and mailed to its stockholders) necessary to duly call, give notice of, convene and hold a meeting of its stockholders (including any adjournments, the "Special Meeting") as soon as reasonably practicable for the purposes of securing the approval of such stockholders of this Agreement and the Plan of Merger, (ii) recommend to its stockholders the approval of this Agreement and the Plan of Merger and the transactions contemplated hereby and thereby (the "Seller Recommendation"), and (iii) use its best efforts to obtain, as promptly as practicable, such approval; provided, however, that the Board of Directors of Seller may fail to hold such Special Meeting, fail to make such recommendation, or withdraw, modify or change any such recommendation, if such Board of Directors, after having consulted with and considered the advice of outside counsel, has determined in good faith that the holding of such Special Meeting, the making of such recommendation or the failure to withdraw, modify or change such recommendation, as a result of an Acquisition Proposal that has not been withdrawn, would constitute a breach of the fiduciary duties of such directors under applicable law. Seller and Acquiror will reasonably cooperate in the preparation of the Registration Statement.

- 5.9 <u>Further Assurances</u>. Subject to the terms and conditions herein provided, each of the parties hereto agrees to use its best efforts to take, or cause to be taken, all reasonable action and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to satisfy the conditions to Closing contained herein and to consummate and make effective the transactions contemplated by this Agreement and the Plan of Merger.
- 5.10 <u>Disclosure Supplements</u>. From time to time prior to the Effective Time, each party will promptly supplement or amend its respective Disclosure Schedules delivered pursuant hereto with respect to any matter hereafter arising which, if existing, occurring or known as of the date hereof, would have been required to be set forth or described in such Schedules or which is necessary to correct any information in such Schedules which has been rendered inaccurate thereby; provided, however, that the inclusion of any information in any such amendment or supplement not included in the original Disclosure Schedules shall not limit or impair any right which a party may have to terminate this Agreement pursuant to Section 7.1(d) due to failure to satisfy the conditions set forth in Section 6.2(a) or Section 6.3(a), as the case may be.
- 5.11 <u>Public Announcements</u>. Prior to the Effective Time, the parties hereto shall approve in advance the substance, and cooperate with each other in the development and distribution, of all news releases and other public disclosures with respect to this Agreement or any of the transactions contemplated hereby, except as may be otherwise required by law or regulation and as to which the parties releasing such information have used their reasonable efforts to discuss with the other parties in advance.
- 5.12 <u>Failure to Fulfill Conditions</u>. In the event that either of the parties hereto determines that a condition to its respective obligations to consummate the transactions contemplated hereby cannot be fulfilled on or prior to August 31, 2004 and that it will not waive that condition, it will promptly notify the other party. Acquiror and Seller will promptly inform the other of any facts applicable to them, or their respective directors or officers, that would be likely to prevent or materially delay approval of the Merger by any Governmental Entity or which would otherwise prevent or materially delay completion of such Merger or the other transactions contemplated hereby.

5.13 Certain Post-Merger Agreements.

The parties hereto agree to the following arrangements following the Effective Time:

(a) Employee Benefit Plans.

(1) Subject to the provisions of this Section 5.13, all employees of Seller or Seller Subsidiary immediately prior to the Effective Time who become employed by Acquiror or any subsidiary or affiliate thereof (as the case may be, "Employer") immediately following the Effective Time ("Transferred Employees") will be covered by the Employer employee benefit plans on substantially the same basis as other employees of Employer performing services primarily on behalf of Employer ("Acquiror Employees") in a comparable position.

Notwithstanding the foregoing, Employer may determine to continue any of the Seller benefit plans for Transferred Employees in lieu of offering participation in the Employer benefit plans providing similar benefits (e.g., medical and hospitalization benefits), to terminate or amend any of Seller's benefit plans, or to merge any such benefit plans with the Employer benefit plans, provided the result is the provision of benefits to Transferred Employees that are substantially similar to the benefits provided to other such Acquiror Employees generally. Except as

specifically provided in this Section 5.13 and as otherwise prohibited by law, Transferred Employees' service with Seller or the Seller Subsidiary shall be recognized as service with the Employer for purposes of eligibility to participate and vesting under the Employer benefit plans, policies or arrangements (including but not limited to, vacation, sick and other leave policies), subject to applicable break-in-service rules. Acquiror agrees that any preexisting condition, limitation or exclusion in its medical, long-term disability and life insurance plans shall not apply to Transferred Employees or their covered dependents who are covered under a medical or hospitalization indemnity plan maintained by Seller or the Seller Subsidiary at the Effective Time and who then change coverage to the Employer medical or hospitalization indemnity health plan at the time such Transferred Employees are first given the option to enroll. Notwithstanding anything herein to the contrary, after the Effective Time, nothing herein shall restrict the ability of the Surviving Corporation, or Employer, to amend or terminate such Seller Plans in accordance with their terms, provided that such amendment or termination affects all covered employees equally.

- (2) As of the Effective Time, Acquiror shall assume and honor and shall cause the appropriate Employer to assume and honor in accordance with their terms all severance agreements existing immediately prior to the Effective Time which are between Seller or the Seller Subsidiary and any officer which have been disclosed in Seller Disclosure Schedule 5.13(a), in the form in effect as of the date hereof.
- (3) Following the Effective Time, Acquiror shall maintain any deferred compensation plan, policy, program or arrangement listed on Seller Disclosure Schedule 3.8(a), provided that Acquiror may determine to continue such plan, policy, program or arrangement, amend any such plan policy program or arrangement, including the cessation of deferrals thereunder or contributions thereto, or merge or consolidate any such plan, policy, program or arrangement with any similar arrangement sponsored by Acquiror, subject to such terms and conditions as Acquiror deems appropriate.
- (4) On the Closing Date, Seller shall terminate all qualified Seller Plans and shall distribute amounts thereunder in accordance with and at the times permitted under such Plans and under applicable law.

(b) <u>Indemnification</u>. Acquiror shall indemnify and hold harmless each present and former director and officer of Seller and the Seller Subsidiary determined as of the Effective Time and identified on Seller Disclosure Schedule 5.13(b) and their heirs, personal representatives and estates (the "**Indemnified Parties**") against, and shall promptly upon request advance or reimburse, any and all costs or expenses (including reasonable attorneys' fees), judgments, interest, fines, losses, claims, damages, liabilities or amounts paid in settlement (collectively, "**Costs**") as they are incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative, arbitrative or investigative ("**Proceeding**"), whether or not the Indemnified Party is a party to such Proceeding, based upon or arising from the Indemnified Party's capacity as an officer or director of Seller and/or the Seller Subsidiary, whether threatened, asserted or claimed prior to, at or after the Effective Time (collectively, "Claims"), to the same extent as the Indemnified Parties would have been indemnified under the Articles of Incorporation and/or Bylaws of Acquiror as such documents were in effect on the date of this Agreement as if the Indemnified Parties were officers or directors of Acquiror at all relevant times. This

indemnity shall be provided for six years following the Effective Time; or if there shall be any proceeding pending or threatened on the sixth anniversary of the Effective Time, such indemnity shall continue in full force and effect until such pending or threatened Proceeding is finally resolved.

The rights to indemnification granted by this Section 5.13(b) are subject to the following limitations: (i) the total aggregate indemnification provided by Acquiror pursuant to this Section 5.13(b) shall not exceed, as to all Indemnified Parties as a group, the sum of \$2,000,000, and Acquiror shall have no responsibility to the Indemnified Parties for the manner in which such sum is allocated among that group; (ii) a director or officer who would otherwise be a member of the Indemnified Parties under this Section 5.13(b) shall not be entitled to the benefits hereof unless such director or officer has executed a Joinder Agreement in the form of Seller Disclosure Schedule 5.13(b) hereto; and (iii) amounts otherwise required to be paid by Acquiror to the Indemnified Parties pursuant to this Section 5.13(b) shall be reduced by any amounts that such Indemnified Parties recover from any third party.

Acquiror agrees that the \$2,000,000 limit set forth in this Section 5.13(b) shall not apply to any damages, liabilities, judgments and claims (and related expenses, including, without limitation, attorneys' fees and amounts paid in settlement) insofar as they arise out of or are based upon the matters for which indemnification is provided in Section 5.21 hereof.

- (c) <u>Insurance</u>. Acquiror shall use all reasonable efforts to cause the Merger Sub or Acquiror to obtain for a period of six years after the Effective Time policies of directors' and officers' liability insurance covering the persons listed on Seller Disclosure <u>Schedule 5.13(b)</u> at no cost to the beneficiaries thereof with respect to the acts or omissions occurring prior to the Effective Time with substantially the same coverage and containing substantially similar provisions and conditions as existing policies; provided, however, that neither Acquiror nor Merger Sub shall be required to pay an aggregate premium for such insurance coverage in excess of 200% of the amount for such coverage set forth in Section 3.14 of the Seller Disclosure Schedule but in such case shall purchase as much coverage as reasonably practicable for such amount.
- (d) <u>Subsequent Events</u>. In the event that Acquiror or any of its respective successors or assigns (i) consolidates with or merges into any other entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any entity, then, and in each such case, the surviving or purchasing entity and the successors and assigns of such entity shall assume the obligations set forth in Sections 5.13(b) and (c), which obligations are expressly intended to be for the irrevocable benefit of, and shall be enforceable by, each of the Indemnified Parties and the persons listed on Seller Disclosure <u>Schedule 5.13(c)</u>, respectively, and shall survive the Closing of this Agreement.
- 5.14 <u>Takeover Laws; No Rights Triggered</u>. If any Takeover Law may become, or may purport to be, applicable to the transactions contemplated hereby, each of Seller and Acquiror and the members of their respective Boards of Directors will grant such approvals and take such actions as are necessary so that the transactions contemplated by this Agreement may be consummated as promptly as practicable on the terms contemplated hereby and otherwise act to eliminate or minimize the effects of any Takeover Law on any of the transactions contemplated by this Agreement.
- 5.15 <u>Preparation of Registration Statement</u>. The parties hereto shall jointly prepare as promptly as practicable a proxy statement to be mailed to the stockholders of Seller who are to vote upon this Agreement and the Plan of Merger in connection with the transactions contemplated hereby and to be part of the Registration Statement to be filed by Acquiror with the Commission pursuant to the 1933 Act with respect to the shares to be issued in connection with the Merger. When the Registration Statement or any post-effective amendment thereto shall become effective, and at all times subsequent to such

effectiveness, up to and including the time of the last stockholder meeting with respect to the transactions contemplated hereby, such Registration Statement and all amendments or supplements thereto, with respect to all information set forth therein furnished or to be furnished by Acquiror relating to Acquiror and by Seller relating to Seller and the Seller Subsidiary, (i) will comply in all material respects with the provisions of the 1933 Act and the rules and regulations of the Securities and Exchange Commission thereunder and (ii) will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements contained therein not misleading. Acquiror will advise Seller promptly after it receives notice thereof of the time when the Registration Statement has become effective or any supplement or amendment has been filed, of the issuance of any stop order, of the suspension of the qualification of the Acquiror Common Stock issuable in connection with the Merger for offering or sale in any jurisdiction, of the initiation or threat of any proceeding for any such purpose, or of any request by the Commission for the amendment or supplement of the Registration Statement or for additional information.

- 5.16 <u>Affiliates</u>. Prior to the Effective Time, Seller shall deliver to Acquiror a letter identifying all persons whom it believes to be "affiliates" of Seller for purposes of Rule 145(c) or Rule 144 (as applicable) under the 1933 Act ("**Affiliates**"). Seller shall use its best efforts to cause each person so identified to deliver to Acquiror prior to the Effective Time a written agreement in substantially the form of <u>Schedule 5.16</u> hereto providing, among other things, that such person will not dispose of Acquiror Common Stock received in the Merger except in compliance with the 1933 Act and the rules and regulations thereunder.
- 5.17 <u>Adjustment for Changes in Outstanding Shares</u>. In the event that prior to the Effective Time the outstanding shares of Acquiror Common Stock shall have been increased, decreased or changed into or exchanged for a different number or kind of shares or securities by reorganization, recapitalization, reclassification, stock dividend, stock split or other like changes in the Acquiror's capitalization, then an appropriate and proportionate adjustment shall be made in the number and kinds of shares of Acquiror Common Stock to be thereafter delivered in connection with the Merger.
- 5.18 Adoption of Accounting Policies. As soon as practicable after the satisfaction or waiver of all conditions to the Closing set forth in Article VI of this Agreement and at such time as consummation of the transactions contemplated by this Agreement seems reasonably assured and in any event prior to the Effective Time (unless this Agreement is terminated pursuant to Article VII hereof), Seller shall, and it shall cause the Seller Subsidiary to, take any and all necessary or appropriate actions to adopt all Acquiror accounting procedures and policies (including, without limitation, those policies pertaining to charged-off and non-accrual assets); provided, however, that no such action taken by Seller or the Seller Subsidiary at the request of Acquiror pursuant to this section shall be deemed to be, or be deemed to cause, a breach of any representation or warranty made by Seller herein.
- 5.19 <u>Covenant to Close</u>. At such time as is deemed appropriate by the parties hereto or as otherwise set forth in this Agreement, and upon satisfaction or waiver of each of the conditions to Closing of the Merger, the parties agree to take such actions as are reasonably necessary or appropriate to effect the Closing and the Merger.
- 5.20 <u>Certain Agreements</u>. As promptly as practicable after the date hereof, Seller shall have each director of Seller who is not an employee of Seller execute a Lock-Up and Non-Competition Agreement in substantially the form of <u>Schedule 5.20-A</u> hereto, have each executive officer of Seller who is also a director of Seller execute a Lock-Up Agreement in substantially the form of <u>Schedule 5.20-B</u> hereto, and have Francis J. Cianciola execute an Employment Agreement in substantially the form of <u>Schedule 5.20-C</u> hereto.

5.21 Hold Harmless. Acquiror will indemnify and hold harmless Seller, the Seller Subsidiary, each of the directors and officers and each person, if any, who controls Seller or the Seller Subsidiary within the meaning of the 1933 Act against any losses, claims, damages or liabilities, joint, several and solidary, to which they or any of them may become subject, under the 1933 Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon an untrue statement or alleged untrue statement of a material fact contained in the Registration Statement, or in any amendment or supplement thereto, or arising out of or based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading, and will pay or promptly reimburse such person for any legal or other expenses reasonably incurred by such person in connection with investigating or defending any such action or claim; provided, however, that Acquiror shall not be liable in any such case to the extent that any such loss, claim, damage or liability (or actions in respect thereof) arises out of or is based upon any untrue statement or alleged untrue statement or alleged omission made in the Registration Statement or any such amendment or supplement in reliance upon and in conformity with information furnished to Acquiror by Seller or the Seller Subsidiary for use therein. Promptly after receipt by an indemnified party hereunder of notice of the commencement of any action, such indemnified party shall, if a claim in respect thereof is to be made against Acquiror under this Section 5.21, notify Acquiror in writing of the commencement thereof. In case any such action shall be brought against any indemnified party and it shall notify Acquiror of the commencement thereof, Acquiror shall be entitled to participate therein; and to the extent that it shall wish to assume the defense thereof, Acquiror shall not be lia

5.22 <u>Stock Options</u>. Seller has issued options (a "**Seller Stock Option**") with respect to the Seller Common Stock pursuant to the Stock Option Plan for Renasant Bancshares, Inc. (the "**Stock Option Plan**"). At the Effective Time, Acquiror shall assume the Stock Option Plan and all of Seller's obligations thereunder. After the Effective Time, each Seller Stock Option shall entitle the holder thereof to purchase a number of shares of Acquiror Common Stock equal to the number of shares of Seller Common Stock specified in such Seller Stock Option multiplied by the Exchange Ratio at an exercise price equal to the quotient of the exercise price specified in such Seller Stock Option divided by the Exchange Ratio.

5.23 Warrants. Seller has issued warrants to purchase Seller Common Stock (a "Seller Warrant"). Prior to the Effective Time, Seller shall cause each holder of a Seller Warrant to execute and deliver an agreement in substantially the form of Schedule 5.23 pursuant to which such holder, at his or her election, agrees (i) to sell to Acquiror, effective at the Effective Time, such Seller Warrant for an amount equal to the number of shares of Seller Common Stock specified in such Warrant times the Price Per Share less the aggregate exercise price for all shares of Seller Common Stock specified in such Seller Warrant, subject to required withholding taxes, if any, or (ii) to amend such holder's Seller Warrant so that from and after the Effective Time the Seller Warrant shall entitle the holder thereof to purchase a number of shares of Acquiror Common Stock equal to the number of shares of Seller Common Stock specified in such Seller Warrant multiplied by the Exchange Ratio at an exercise price equal to the quotient of the exercise price specified in such Seller Warrant divided by the Exchange Ratio.

ARTICLE VI CLOSING CONDITIONS

- 6.1 <u>Conditions to the Parties' Obligations under this Agreement</u>. The respective obligations of the parties under this Agreement shall be subject to the fulfillment at or prior to the Effective Time of the following conditions:
- (a) All necessary regulatory or governmental approvals and consents required to complete the Merger shall have been obtained, all conditions required to be satisfied prior to the Effective Time by the terms of such approvals and consents shall have been satisfied, and all waiting periods in respect thereof shall have expired; and all notices, reports and other filings required to be made with any Governmental Entity in connection with the Merger prior to the Effective Time by Acquiror or Seller or any of their respective Subsidiaries shall have been made and become final.
 - (b) This Agreement and the Plan of Merger shall have been duly adopted and approved by the requisite votes of the stockholders of Seller.
- (c) None of Seller, the Seller Subsidiary, Acquiror or Acquiror Bank shall be subject to any statute, rule, regulation, injunction or other order or decree which shall have been enacted, entered, promulgated or enforced by any governmental or judicial authority which prohibits, restricts or makes illegal consummation of the Merger or the other transactions contemplated hereby.
- (d) All consents or approvals of all persons (other than Governmental Entities) required for consummation of the Merger shall have been obtained and shall be in full force and effect, unless the failure to obtain any such consent or approval would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on Seller or Acquiror.
- (e) The Registration Statement shall have become effective, and no stop order suspending the effectiveness of the Registration Statement shall have been issued, and no proceedings for that purpose shall have been initiated or threatened by the Securities and Exchange Commission.
- (f) Acquiror and the stockholders of Seller shall have received an opinion of Phelps Dunbar LLP, which opinion shall be satisfactory in form and substance to Acquiror and Seller, to the effect that the Merger when consummated in accordance with the terms hereof will constitute a reorganization within the meaning of Section 368(a) of the Code and that the exchange of Seller Common Stock to the extent exchanged for Acquiror Common Stock will not give rise to gain or loss to the stockholders of Seller with respect to such exchange.
- (g) The shares of Acquiror Common Stock issuable to the holders of Seller Common Stock in the Merger shall have been approved for listing on the American Stock Exchange on or before the Closing Date, subject to official notice of issuance.
- 6.2 <u>Conditions to the Obligations of Acquiror and Merger Sub under this Agreement</u>. The obligations of Acquiror and Merger Sub under this Agreement shall be further subject to the satisfaction, at or prior to the Effective Time, of the following conditions, any one or more of which may be waived by Acquiror (on behalf of itself and Merger Sub) to the extent permitted by law:
- (a) (i) Each of the obligations of Seller required to be performed by it at or prior to the Closing pursuant to the terms of this Agreement shall have been duly performed and complied with in all material respects, and (ii) the representations and warranties of Seller contained in this Agreement shall have been true and correct in all respects (Section 3.15(b) shall be read without regard to any qualifications regarding Knowledge) as of the date hereof and as of the Effective Time as though made at and as of the Effective Time (other than any representation or warranty which specifically relates to an earlier date shall be true and correct as of such date), except for any violations or breaches of the representations and warranties of Seller contained in this Agreement which would not have a Material Adverse Effect, and Acquiror shall have received a certificate to that effect signed by the President and Chief Executive Officer of Seller. For purposes of this Agreement, Material Adverse Effect means a

breach of a warranty or representation of Seller contained in this Agreement which involve matters in which the amounts involved exceed \$300,000 in the aggregate.

- (b) All permits, consents, waivers, clearances, approvals and authorizations of all Governmental Entities or third parties which are necessary in connection with the consummation of the Merger shall have been obtained; and none of such permits, consents, waivers, clearances, approvals and authorizations shall contain any terms or conditions which would materially impair the value of Seller to Acquiror.
- (c) Holders of the Seller Common Stock who dissent from the Merger pursuant to Section 48-23-102 et seq. of the TBCA by meeting the requirements set forth therein shall not hold more than 15% of the Seller Common Stock immediately prior to the Effective Time.
- (d) Two qualified people selected by Seller from its Board of Directors and acceptable to Acquiror shall be appointed to the Board of Directors of Acquiror on the Closing Date.
- 6.3 <u>Conditions to the Obligations of Seller under this Agreement</u>. The obligations of Seller under this Agreement shall be further subject to the satisfaction, at or prior to the Effective Time, of the following conditions, any one or more of which may be waived by Seller to the extent permitted by law:
- (a) (i) Each of the obligations of Acquiror and Merger Sub required to be performed by them at or prior to the Closing pursuant to the terms of this Agreement shall have been duly performed and complied with in all material respects, and (ii) the representations and warranties of Acquiror and Merger Sub contained in this Agreement shall have been true and correct in all respects as of the date hereof and as of the Effective Time as though made at and as of the Effective Time (other than any representation or warranty which specifically relates to an earlier date shall be true and correct as of such date), except for any violations or breaches of the representations and warranties of Acquiror contained in this Agreement which would not have a Material Adverse Effect, and Seller shall have received a certificate to that effect signed by the President and Chief Executive Officer of Acquiror. For purposes of this Agreement, Material Adverse Effect means a breach of a warranty or representation of Acquiror contained in this Agreement which involve matters in which the amounts involved exceed \$300,000 in the aggregate.
- (b) All permits, consents, waivers, clearances, approvals and authorizations of all Governmental Entities or third parties which are necessary in connection with the consummation of the Merger shall have been obtained; and none of such permits, consents, waivers, clearances, approvals and authorizations shall adversely affect the Merger Consideration.

ARTICLE VII TERMINATION, AMENDMENT AND WAIVER, ETC.

- 7.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Effective Time, whether before or after approval of this Agreement and the Plan of Merger by the stockholders of Seller:
 - (a) by mutual written consent of the parties hereto;
- (b) by Acquiror (on behalf of itself and Merger Sub) or Seller (i) if the Effective Time shall not have occurred on or prior to August 31, 2004, unless approval by a Governmental Entity pursuant to Section 6.1(a) is pending and has not been finally resolved, in which event such date shall be automatically extended to October 31, 2004; or (ii) if a vote of the stockholders of Seller is taken and such stockholders fail to approve this Agreement and the Plan of Merger at the Special Meeting; unless the

failure of such occurrence shall be due to the failure of the party seeking to terminate this Agreement to perform or observe its agreements set forth herein to be performed or observed by such party at or before the Effective Time or such vote, as the case may be;

- (c) by Acquiror (on behalf of itself and Merger Sub) or Seller upon written notice to the other (i) 30 or more days after the date upon which any application for a regulatory or governmental approval necessary to consummate the Merger and the other transactions contemplated hereby shall have been denied or withdrawn at the request or recommendation of the applicable Governmental Entity, unless within such 30-day period a petition for rehearing or an amended application is filed or noticed, or 30 or more days after any petition for rehearing or amended application is denied; provided, however, that no party hereto shall have the right to terminate this Agreement pursuant to this Section 7.1(c)(i) if such denial or request or recommendation for withdrawal shall be due to the failure of the party seeking to terminate this Agreement to perform or observe the covenants and agreements of such party set forth herein; and in each case the time period for appeals and requests for reconsideration has run; or (ii) if any Governmental Entity of competent jurisdiction shall have issued a final non-appealable order enjoining or otherwise prohibiting the consummation of the Merger or the other transactions contemplated by this Agreement;
- (d) by Acquiror (on behalf of itself and Merger Sub) in writing if Seller has, or by Seller in writing if Acquiror has, breached (i) any covenant or undertaking contained herein, or (ii) any representation or warranty contained herein, which breach would result in the failure to satisfy the closing condition set forth in Section 6.2(a) or 6.3(a), as the case may be, and such breach cannot be or has not been cured within 30 days after the giving of written notice of such breach to the non-terminating party;
- (e) by Acquiror (on behalf of itself and Merger Sub) if (i) Seller shall have failed to make the Seller Recommendation in the Registration Statement, (ii) Seller shall have effected a change in the Seller Recommendation, (iii) Seller shall have breached its obligations under this Agreement by reason of a failure to call or convene the Special Meeting in accordance with Section 5.8, or (iv) Seller shall have approved or recommended, or proposed publicly to approve or recommend, any Acquisition Transaction;
- (f) by Acquiror if holders of more than 15% of Seller Common Stock exercise statutory rights of dissent and appraisal pursuant to Section 48-23-102 et seq. of the TBCA;
 - (g) by Seller in accordance with the terms and conditions of Section 5.3(c);
- (h) by Acquiror or Seller if an adjustment in the Exchange Ratio and the Per Share Price shall be required in accordance with Section 2.1(a)(viii) and Acquiror shall have elected not to make such election; or
- (i) by Seller upon the occurrence of both of the following events: (i) Acquiror shall have made an adjustment to the Exchange Ratio and the Per Share Price in accordance with Section 2.1(a)(viii) and (ii) (a) the quotient obtained by dividing the Average Closing Price by the Starting Price shall be less than (b) the quotient obtained by dividing the Index Value (as defined below) on the Determination Date by the Index Value on the Starting Date and subtracting 0.15 from the quotient in this clause (ii)(b).

For purposes of this Section 7.1(i), the following terms shall have the meanings indicated:

"Average Closing Price" means the average of the daily last sales prices of Acquiror Common Stock as reported on the American Stock Exchange for the 10 consecutive full trading days in which such shares are traded on the American Stock Exchange ending at the close of trading on the Determination Date.

- "Determination Date" means the later of the date on which (i) the consent of the FRB (without regard to any requisite waiting period thereof) to the Merger shall be received by Acquiror and (ii) the Special Meeting occurs.
 - "Index Group" means the NASDAQ Bank Index as reported by Bloomberg, L.P.
 - "Index Value" on a given date shall mean the index value for the NASDAQ Bank Index as reported by Bloomberg, L.P.
 - "Starting Date" means the fourth full trading day after the announcement by press release of the Merger.
 - "Starting Price" shall mean the closing price per share of Acquiror Common Stock as reported on the American Stock Exchange on the Starting Date.
- 7.2 <u>Effect of Termination</u>. In the event of termination of this Agreement by either Acquiror or Seller as provided above, this Agreement shall forthwith become void (other than Sections 5.6(b), this Section 7.2, Section 7.4 and Article VIII hereof, which shall remain in full force and effect), and there shall be no further liability on the part of the parties or their respective officers or directors except for the liability of the parties under Sections 5.6(b), 7.4 and 8.1 hereof and except for liability for any willful breach of any covenant, undertaking, representation or warranty in this Agreement giving rise to such termination.
- 7.3 Amendment, Extension and Waiver. Subject to applicable law, at any time prior to the consummation of the Merger, whether before or after approval thereof by the stockholders of Seller, the parties may (a) amend this Agreement and the Plan of Merger, (b) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (c) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto, or (d) waive compliance with any of the agreements or conditions contained herein (other than required stockholder and regulatory approval); provided, however, that after any approval of the Merger by the stockholders of Seller, there may not be, without further approval of such stockholders, any amendment or waiver of this Agreement or the Plan of Merger which (i) modifies either the amount or the form of the Merger Consideration to be delivered to stockholders of Seller, or (ii) is reasonably likely to materially delay or jeopardize receipt of any required regulatory approvals or materially impair or prevent the satisfaction of any other condition to the obligations of Acquiror or Seller set forth in Sections 6.1, 6.2 and 6.3 hereof or may materially delay the Effective Time. This Agreement and the Plan of Merger may not be amended except by an instrument in writing signed on behalf of each of the parties hereto. Any agreement on the part of a party hereto to any extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party and which has been authorized by or under the direction of its Board of Directors; but such waiver or failure to insist on strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

7.4 Termination Fees.

(a) In the event that (A) a Pre-Termination Takeover Proposal Event (as defined in subsection (c)) shall occur after the date of this Agreement and thereafter this Agreement is terminated by either Acquiror or Seller pursuant to Section 7.1(b)(ii) or by Acquiror pursuant to Section 7.1(d) as a result of a willful breach by Seller and (B) prior to the date that is 12 months after the date of such termination Seller consummates an Acquisition Proposal, then Seller shall, on the date such Acquisition Proposal is consummated, pay Acquiror a fee equal to \$5,000,000 by wire transfer of same-day funds.

- (b) In the event that this Agreement is terminated by Acquiror pursuant to Section 7.1(e) or by Seller pursuant to Section 7.1(g), then concurrently with such termination, Seller shall pay to Acquiror a fee equal to \$5,000,000 by wire transfer of same-day funds, and such termination shall not be deemed effective hereunder until receipt by Acquiror of such fee. In no event shall Seller be required to pay a \$5,000,000 fee under both this Section 7.4(b) and Section 7.4(a).
- (c) For purposes of this Section 7.4, a "**Pre-Termination Takeover Proposal Event**" shall be deemed to occur if, prior to the event giving rise to the right to terminate this Agreement, an Acquisition Proposal shall have been made known to the senior management or board of directors of Seller or has been made directly to its stockholders generally or any person reasonably qualified to consummate an Acquisition Proposal shall have publicly announced an intention (whether or not conditional) to make an Acquisition Proposal, and such Acquisition Proposal or public announcement shall not have been irrevocably withdrawn not less than five Business Days prior to the Special Meeting with respect to a termination pursuant to Section 7.1(b)(ii) or the date of termination with respect to a termination pursuant to Section 7.1(d). Seller acknowledges that the agreements contained in this Section 7.4 are an integral part of the transactions contemplated by this Agreement and that, without these agreements, Acquiror would not enter into this Agreement; accordingly, if Seller fails promptly to pay the amount due pursuant to this Section 7.4 and, in order to obtain such payment, Acquiror commences a suit which results in a judgment against Seller for the fee set forth in this Section 7.4, Seller shall pay to Acquiror its costs and expenses (including reasonable attorneys' fees and expenses) in connection with such suit.

ARTICLE VIII MISCELLANEOUS

- 8.1 Expenses. Except as provided in Section 7.4, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby (including legal, accounting, investment banking and printing expenses) shall be borne by the party incurring such costs and expenses, provided that Acquiror and Seller shall each bear one-half of all costs of printing, mailing and filing the Registration Statement and all filing and similar fees relating to the Merger.
- 8.2 <u>Survival</u>. The respective representations, warranties and covenants of the parties to this Agreement shall not survive (i) the Effective Time or (ii) the termination of this Agreement, but shall terminate as of the Effective Time or such termination, respectively, except for the provisions of (i) Article II, Sections 5.13, 5.21, 5.22 and 5.23 and this Article VIII which shall survive the Effective Time, and (ii) Sections 5.6(b), 5.6(c), 7.4 and this Article VIII which shall survive such termination, respectively.
- 8.3 <u>Notices</u>. All notices or other communications hereunder shall be in writing and shall be deemed given if delivered personally, sent by overnight express or by facsimile addressed as follows:
 - (a) If to Acquiror or Merger Sub, to:

The Peoples Holding Company 209 Troy Street Tupelo, Mississippi 38801 Attention: E. Robinson McGraw Facsimile: (662) 680-1230

Copy (which shall not constitute notice) to:

Phelps Dunbar LLP 20th Floor 365 Canal Street New Orleans, Louisiana 70130 Attention: Mark A. Fullmer

Facsimile: (504) 568-9130

(b) If to Seller, to:

Renasant Bancshares, Inc. 2177 Germantown Road South Germantown, Tennessee 38138 Attention: Francis J. Cianciola Facsimile: (901) 312-4094

Copy (which shall not constitute notice) to:

Baker, Donelson, Bearman, Caldwell & Berkowitz, P.C. 20th Floor
165 Madison Avenue
Memphis, Tennessee 38103
Attention Pobert Waller

Attention: Robert Walker Facsimile: (901) 577-0785

or such other address as shall be furnished in writing by any party, and any such notice or communication shall be deemed to have been given as of the date so mailed.

- 8.4 <u>Parties in Interest</u>. This Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and assigns; provided, however, that neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other party and, except as otherwise expressly provided in Sections 5.13(a)(2), 5.13(a)(3), 5.13(b), 5.13(c), 5.21, 5.22 and 5.23, that nothing in this Agreement is intended to confer, expressly or by implication, upon any other person any rights or remedies under or by reason of this Agreement.
- 8.5 <u>Complete Agreement</u>. This Agreement and the Plan of Merger, including the documents and other writings referred to herein or therein or delivered pursuant hereto or thereto, including the confidentiality agreement by and by and among the parties dated as of January 6, 2004, contain the entire agreement and understanding of the parties with respect to the subject matter hereof and shall supersede all prior agreements and understandings by and among the parties, both written and oral, with respect to such subject matter. There are no restrictions, agreements, promises, representations, warranties, covenants or undertakings by and among the parties other than those expressly set forth herein or therein.

- 8.6 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and each of which shall be deemed an original.
- 8.7 <u>Governing Law</u>. This Agreement shall be governed by the laws of the State of Tennessee, without giving effect to the principles of conflicts of laws thereof.
- 8.8 <u>Interpretation</u>. When a reference is made in this Agreement to sections, exhibits or schedules, such reference shall be to a section, exhibit or schedule of this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and are not part of this Agreement. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."
- 8.9 <u>Enforcement</u>. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, this being in addition to any other remedy to which they are entitled at law or in equity.

IN WITNESS WHEREOF, Acquiror, Merger Sub and Seller have caused this Agreement to be executed by their duly authorized officers as of the day and year first above written.

THE PE	THE PEOPLES HOLDING COMPANY	
Ву:		
_	E. Robinson McGraw, President and Chief Executive Officer	
PEOPLI	ES MERGER CORPORATION	
Ву:		
	E. Robinson McGraw, President	
RENAS	ANT BANCSHARES, INC.	
By:		
Name: Title:		

INDEX OF DEFINED TERMS

Term	Defined in Section No.
Acquiror	First paragraph
Acquiror Bank	4.10
Acquiror Common Stock	2.1(a)(i)
Acquiror Disclosure Schedule	First sentence Article IV
Acquiror Employees	5.13(a)(1)
Acquiror Subsidiaries	4.3(b)
Acquisition Proposal	5.3(a)
Acquisition Transaction	5.3(a)
Affiliates	5.16
Agreement	First paragraph
Average Closing Price	7.1(h)
Business Day	1.2
Cash Consideration	2.1(a)(i)
Cash Election	2.1(a)(iii)
Cash Election Shares	2.1(a)(vii)
Change in Seller Recommendation	5.3(b)
Claims	5.13(b)
Closing	1.2
Closing Date	1.2
Code	2.2(f)
Combination Election	2.1(a)(iii)
Commission	3.10
Constituent Corporations	First whereas paragraph
Costs	5.13(b)
CRA	3.28
delivered to Acquiror	First paragraph of Article III
delivered to Seller	First paragraph of Article IV
Determination Date	7.1(h)
Dissenting Stockholder	2.4
Effective Time	1.2
Election Deadline	2.1(a)(iv)
Environmental Law	3.15
Employees	3.8(a)
Employer	5.13(a)(1)
Environmental Law	3.15
ERISA	3.8(a)
ERISA Affiliate	3.8(b)
ERISA Affiliate Plan	3.8(b)
ERISA Plan	3.8(b)
Exchange Agent	2.2(a)
Exchange Fund	2.2(e)
Exchange Ratio	2.1(a)(i)
FDIA	3.12(a)
FDIC	3.3(b)
	• •

FRB	3.3(b)
Form of Election	2.1(a)(iii)
GAAP	3.4(b)
Governmental Entity	1.2
Hazardous Substance	3.15
Indemnified Parties	
	5.13(b)
Index Group	7.1(h)
Index Value	7.1(h)
Intellectual Property	3.27(a)
IRS	3.7(a)
Loan Portfolio Properties and Other Properties Owned	3.15
made available to Acquiror	First paragraph of Article III
made available to Seller	First paragraph of Article IV
Maximum Cash Election Number	2.1(a)(ii)
Maximum Stock Election Number	2.1(a)(ii)
MCB	3.3(b)
Merger	1.1
Merger Consideration	2.1(a)(i)
Merger Sub	First paragraph
Minimum Cash Election Number	2.1(a)(ii)
Minimum Stock Election Number	2.1(a)(ii)
Notice	5.3(a)
Outstanding Seller Stock Options	3.2
Outstanding Seller Warrants	3.2
party/parties	First paragraph
Pension Plan	3.8(a)
Pre-Termination Takeover Proposal Event	7.4(a)(iii)
Price Per Share	2.1(a)(i)
Proceeding	5.13(b)
Registration Statement	3.10
Representative	2.1(a)(iii)
Returns	3.7(a)
Risk Management Instruments	3.20
Seller	First paragraph
Seller Acquisition Agreement	5.3(b)
Seller Agreement	3.13(a)
Seller Common Stock	2.1
Seller Disclosure Schedule	First sentence Article III
Seller Plans	3.8(a)
Seller Recommendation	5.8
Seller Subsidiary	3.1(b)
Seller Stock Option	5.22
Seller Warrant	5.23
Software	3.27(a)
Special Meeting	5.8
Starting Date	7.1(h)
Starting Price	7.1(h) 7.1(h)
Stock Consideration	2.1(a)(i)
Stock Election	2.1(a)(iii)
Stock Election	-·-(u)(····)

Stock Election Shares	2.1(a)(vii)
Stock Option Plan	5.22
Superior Proposal	5.3(a)
Surviving Corporation	1.1
Takeover Laws	3.3(c)
Taxes	3.7(c)
TBCA	1.1
TCFI	3.3(b)
to the Knowledge of Acquiror	First paragraph of Article IV
to the Knowledge of Seller	First paragraph of Article III
Transferred Employees	5.13(a)(1)

SCHEDULE 5.13(b)

JOINDER AGREEMENT

THIS AGREEMENT is made and executed as of the ______ day of February, 2004 between The Peoples Holding Company, a Mississippi corporation ("Acquiror"), and the undersigned individual officer and/or director ("Renasant Officer") of Renasant Bancshares, Inc., a Tennessee corporation ("Renasant") or affiliates of Renasant.

RECITALS:

WHEREAS, Acquiror and Renasant have entered into an Agreement and Plan of Merger (the "Merger Agreement") pursuant to which Renasant will be merged into Peoples Merger Corporation, a wholly owned subsidiary of Acquiror (the "Acquisition Subsidiary") and the Acquisition Subsidiary shall be the surviving entity on the terms and subject to the conditions set forth in such Merger Agreement; and

WHEREAS, in consideration of the agreements made by Renasant in connection with the Merger Agreement, Acquiror has agreed to indemnify the Renasant Officer under certain circumstances set forth in the Merger Agreement.

NOW, THEREFORE, in consideration of Acquiror's agreement to indemnify the Renasant Officer and the expenses and costs that may be incurred by Acquiror in connection with such indemnification, the Renasant Officer hereby agrees as follows:

1. Assumption of and Cooperation in Defense.

- 1.1 Notice and Assumption of Defense. In the event a claim arises for which indemnification is or may be sought by the Renasant Officer, the Renasant Officer shall promptly notify Acquiror, in writing at the address set forth in Section 3 hereof, of the commencement of such legal action or existence of and facts relating to such claim. Upon receipt of such notice, or at any time thereafter, Acquiror shall be entitled to participate therein and, in its sole discretion, to assume the defense of such claim, with counsel of its choice, subject to the reasonable approval of the Renasant Officer, and to consider and decide on any proposed settlement, subject to the reasonable approval of the Renasant Officer. In any and all events, Acquiror shall have the right to reasonable control over the nature and extent of expenses incurred in connection with such claim(s). Acquiror shall notify the Renasant Officer of its assumption of the defense of such claim; and after such notice from Acquiror to the Renasant Officer, Acquiror shall not be liable to the Renasant Officer for indemnification under Section 5.13(b) of the Merger Agreement for any legal expenses of other counsel or any other expenses of defense subsequently incurred by such indemnified party.
- 1.2 <u>Cooperation in Defense</u>. The Renasant Officer agrees to cooperate in the defense of any action for which indemnification is sought under Section 5.13(b) of the Merger Agreement. Such cooperation shall include, but not be limited to, providing Acquiror and its counsel copies of any and all relevant documents relating to the claim, consulting with Acquiror and its counsel with regard to the claim, providing testimony, either in deposition or at trial or both, regarding the facts relating to the claim, making himself available at reasonable times for consultation, testimony and fact-finding and otherwise furnishing such information to Acquiror and its counsel as the Renasant Officer would provide to his own counsel in the event he were defending the action himself. Except as expressly permitted by Acquiror, the Renasant Officer shall not object to the production or use of any documents heretofore prepared by or of information provided to Acquiror legal counsel on the basis of any claim of privilege that is available

only to Renasant or Acquiror; provided, however, that Acquiror agrees that it will not, without the consent of the Renasant Officer, waive any applicable privilege of the Renasant Officer. Such cooperation shall be provided regardless of whether Acquiror assumes the defense of the action.

- 2. <u>Duplication of Payment; Limitations; Presumptions</u>.
- 2.1 <u>No Duplication of Payments</u>. Acquiror shall not be liable under this Agreement to make any payment in connection with any claim against the Renasant Officer to the extent the Renasant Officer has otherwise actually received payment (under any insurance policy, certificate of incorporation, bylaw provision or otherwise) of amounts otherwise indemnifiable hereunder.
- 2.2 <u>Limitation on Liability</u>. The Renasant Officer hereby expressly acknowledges and agrees that Acquiror shall not be liable in the aggregate for more than \$2 million in connection with its obligations under Section 5.13(b) of the Merger Agreement. The Renasant Officer further acknowledges and agrees that he shall have no claim against Acquiror for any amount that, when aggregated with amounts paid by Acquiror to other Renasant Officers, exceeds \$2 million. Any claim for reallocation of the amounts paid by Acquiror among Renasant Officers shall be made against the other Renasant Officer(s) involved, and Acquiror shall not be liable in any way for the allocation of such amounts among Renasant Officers.
- 2.3 <u>No Presumption</u>. For purposes of this Agreement, the termination of any claim, action, suit or proceeding by judgment, order, settlement (whether with or without court approval) shall not of itself create a presumption that the Renasant Officer did or did not meet any particular standard of conduct or have any particular belief or that a court has determined that indemnification is not permitted by applicable law.
- 3. <u>Notices</u>. All notices or other communications which are required or permitted hereunder shall be in writing and sufficient if delivered personally or sent by a nationally recognized overnight delivery service (receipt requested), at the addresses listed below:

	Mr. E. Robinson McGraw President and Chief Executive Officer The Peoples Holding Company 209 Troy Street Tupelo, MS 38801
If to the Renasant Officer:	

- 4. Execution in Counterparts. This Agreement may be executed in counterparts, each of which shall be deemed to constitute an original. Each such counterpart shall become effective when one counterpart has been signed by each party hereto.
- 5. <u>Governing Law</u>. This Agreement shall be governed by, and interpreted in accordance with, the laws of the State of Tennessee applicable to agreements made and entirely to be performed within such State, except as federal law may be applicable.
- 6. Amendment. This Agreement may only be amended by a written instrument signed by both parties hereto.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written.		
	THE PEOPLES HOLDING COMPANY	
	By:	
	RENASANT OFFICER	

SCHEDULE 5.16

AFFILIATE LETTER

[date]

The Peoples Holding Company 209 Troy Street Tupelo, MS 38801

Gentlemen:

This letter agreement is given in connection with the closing of the merger (the "Merger") of Renasant Bancshares, Inc. ("Seller") with and into Peoples Merger Corporation, a wholly owned subsidiary of The Peoples Holding Company ("Acquiror"). I am aware and acknowledge that, as a member of the Board of Directors or an officer of Seller or the beneficial owner of a substantial amount of the outstanding common stock of Seller, I may be an "affiliate" of Seller as that term is defined in the Securities Act of 1933, as amended (the "Securities Act"), and the regulations thereunder.

I understand the resales or other dispositions of Acquiror's common stock, \$5.00 per share par value ("Acquiror Common Stock"), acquired by me as a result of the Merger may be governed by Rules 144 and 145 of the Securities Act.

On the basis of the foregoing, and in consideration of the delivery to me of the Acquiror Common Stock into which my Seller common stock will be converted, I agree that I will not, directly or indirectly, sell, transfer, pledge or otherwise alienate or encumber any of the Acquiror Common Stock held by me in violation of the Securities Act or the rules or regulations promulgated thereunder. In addition, I expressly agree to the placement of a restrictive legend on any and all certificates representing Acquiror Common Stock of which I am the beneficial owner reflecting the restrictions described above.

SCHEDULE 5.20-C

EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT (this "Agreement") is entered into as of the _____ day of ______, 2004, by and among Francis J. Cianciola ("Employee"), The Peoples Holding Company ("Peoples") and Renasant Bank, a Tennessee banking corporation and an affiliate of Peoples (the "Company").

WITNESSETH:

WHEREAS, the Company desires to employee on the terms and conditions set forth in this Agreement and Employee desires to be employed by the Company on such terms and conditions;

NOW, THEREFORE, in consideration of the respective representations, warranties and covenants hereinafter set forth, the parties agree as follows:

EMPLOYMENT. The Company agrees to employ Employee and Employee agrees to remain in the employ of the Company, upon the terms and subject to the conditions provided herein.

II. POSITION, DUTIES AND RESPONSIBILITIES.

- A. Employee shall serve as President and Chief Executive Officer or in such other capacity or capacities as shall be mutually agreed upon from time to time by Employee and Company. Employee shall report directly to the President and Chief Executive Officer of Peoples (the "Peoples CEO") and to the Board of Directors of the Company.
- B. Employee's duties shall include the responsibility for the operations of the Company consistent with Peoples' policies as they currently exist and as they may change over time and such other duties as may from time to time be delegated to Employee by the Peoples CEO. Employee shall perform services when and as directed by the Company and Peoples and as more fully described below, except with Employee's prior written consent, Employee's assigned duties shall not be inconsistent with his position. Peoples agrees to vote its stock in the Company to elect Employee as a director of the Company during his employment with the Company. Peoples, also, agrees to cause Employee to be nominated to the Board of Directors of Peoples during Employee's employment.
- C. Employee shall devote substantially all of his business time, attention and efforts in the faithful performance of his duties hereunder, provided, however, Employee may continue his current investing in real estate and other investment opportunities as long as such activities do not interfere with Employee's duties and responsibilities under this Agreement.
- D. Subject to the policies, procedures and code of ethics of Peoples as in effect from time to time, the budget of the Company then in effect ("the "Budget") and all applicable laws and regulations, Employee shall have exclusive authority over recruiting, hiring, firing and setting compensation of all Company employees.

- E. Without the consent of Peoples, Employee shall not cause or permit the Company to:
 - i) issue any securities of the Company or any of its affiliates;
 - ii) make any tax election or determine the accounting method to be used by the Company or any of its affiliates;
 - iii) establish a subsidiary or transfer any interest in any subsidiary or merge or consolidate any subsidiary of the Company with any other entity;
 - iv) adopt a budget or make any material deviations from the Budget then in effect;
 - v) sell, exchange, lease, mortgage, pledge, charge or otherwise transfer or encumber all or any portion of the assets of the Company or any subsidiary;
 - vi) enter into any business combination, strategic partnership or joint venture with another individual, partnership, corporation, trust, limited liability company or any other entity (any of the foregoing may be referred to as "Person"), or acquire or dispose of all or any interest in any other Person, merge or consolidate with or into another Person;
 - vii) create, incur, assume or otherwise become liable with respect to any obligation for borrowed money (including a guarantee of, or contingent liability for, the indebtedness or other obligations of any Person), or issue any bonds, debentures, notes or other evidences of indebtedness;
 - viii) enter into any transaction or series of related transactions involving capital expenditures, including incoming lease commitments, purchases of equipment or inventory or other expenditures that are not consistent with the Budget then in effect;
 - ix) authorize or enter into or amend any agreement, commitment or other transaction, or any series of related agreements, commitments or other transactions between the Company and any affiliate; or
 - x) take any action outside the ordinary course of business.
- F. Employee represents and warrants that he is not bound by any employment, consulting, non-competition, confidentiality, finders, marketing or other agreement or arrangement that would, or might reasonably be expected to, prohibit or restrict him from performing his duties and obligations.

III. COMPENSATION AND BENEFITS.

A. The Company shall pay Employee an annual salary of Two Hundred and Ten Thousand Dollars (\$210,000.00), payable in equal installments every two (2) weeks in accordance with the Company's regular payroll policy subject to tax withholding for taxes as required by law. Employee may receive bonuses in accordance with Peoples' policies in effect from time to time and may be eligible for salary increases as may be mutually agreeable from time to time.

- B. Employee shall also be entitled to receive the benefits of Peoples and its affiliates specified in Schedule 3(b). To the extent permitted by law and each applicable benefit plan, all prior years of service with the Company will be counted for vesting and eligibility purposes under all applicable Peoples benefit plans. In addition, upon Employee reaching the age of 55, Employee shall become eligible to participate in Peoples' Early Retirement Medical Benefit.
- C. Employee is authorized to incur necessary and customary expenses in connection with the business of the Company, including expenses for entertainment, trade association meetings, travel, promotion and similar matters, consistent with Peoples' and the Company's policies as in effect from time to time. The Company will pay or reimburse Employee for such expenses upon presentation by Employee of appropriate records to verify such expense.
- D. Employee shall be entitled to the use of a leased or Company-owned six-passenger motor vehicle to include all operating and maintenance expenses.
- IV. <u>TERM.</u> Employee's employment shall commence on ______, 2004, and shall terminate on ______, 2009 unless terminated sooner as provided in this Agreement.

V. <u>TERMINATION</u>.

- A. If Employee becomes physically or mentally disabled, as determined in the good faith judgment of Peoples' Board of Directors, Employee's employment may be terminated upon sixty (60) days written notice.
- B. Employee may be terminated for cause ("Cause") by the Company if:
 - 1. the Employee commits, or is convicted of, or enters a plea of *nolo contendere* with respect to, a felony or a crime involving moral turpitude or any other criminal activity or unethical conduct that, in the good faith opinion of the Board of Directors of Peoples, would seriously impair Employee's ability to perform his duties hereunder or would impair the business reputation of Peoples, the Company or any of their affiliates:
 - 2. in the good faith opinion of the Board of Directors of Peoples, the Employee has failed to perform the duties assigned to him and such failure is not cured within thirty (30) days of receipt by Employee of written notice thereof;
 - 3. the Employee willfully and knowingly violates any statute, rule or regulation under the federal banking laws or the banking laws of any state that, in the good faith opinion of the Board of Directors of Peoples, would seriously impair Employee's ability to perform his duties hereunder or would impair the business reputation of Employee, Peoples, the Company or any of their affiliates; or
 - 4. in the good faith opinion of the Board of Directors of Peoples, the Employee materially breaches any provision of this Agreement.
- C. Employee may terminate Employee's employment upon written notice to the Company and Peoples at any time for "Good Reason," defined to mean (i) a significant diminution

of duties from those assigned to Employee at commencement of this Agreement, (ii) a relocation of Employee outside of Shelby County, Tennessee without Employee's consent, (iii) the failure by the Company and Peoples to elect Employee the President and Chief Executive Officer and a director of the Company throughout the term of this Agreement, or (iv) any material breach of this Agreement by the Company or Peoples which is not cured within thirty (30) days of receipt by Peoples and the Company of written notice thereof. If Employee terminates for Good Reason, then Employee shall receive a payment equal to Employee's annual base salary at the time of termination, payable in a lump sum upon termination. Employee shall, also, receive a payment equal to his average bonus for the prior two (2) years and a payment equal to the then current annual lease payment on the motor vehicle described in Section 3(d). The Company shall, also, pay a portion of Employee's COBRA premium under the same terms as it pays the Company's other employees' health insurance premiums for a period of one (1) year. Peoples and Employee may mutually agree to have Peoples or the Company make a payment equal to Employee's annual base salary at the time of termination plus an amount equal to Employee's average bonus for the two (2) years prior to termination, payable in twelve (12) equal monthly installments beginning on the first anniversary date of Employee's termination (an "Extension Payment").

- D. The Company may terminate Employee's employment without cause upon four (4) weeks notice to Employee. If Employee is terminated without cause, Employee will receive a payment equal to the Employee's annual base salary at the time of termination, payable in a lump sum upon termination. Employee shall, also, receive a payment equal to his average bonus for the prior two (2) years and a payment equal to the then current annual lease payment on the motor vehicle described in Section 3(d). The Company shall, also, pay a portion of Employee's COBRA premium under the same terms as it pays the Company's other employees' health insurance premiums for a period of one (1) year. Peoples and Employee may mutually agree to have the Company or Peoples make an Extension Payment.
- E. If Employee's employment is terminated under Sections 5(a) or (b), then the Company shall pay to Employee his salary through the date of termination as well as any benefits to which Employee may be entitled as of the date of termination under the benefit plans referred to in Section 3.
- F. If a Change in Control (as hereinafter defined) occurs after the date hereof and during the term of this Agreement and within three (3) years after such Change in Control, either the Company shall terminate the Employee's employment without Cause, or the Employee shall terminate employment with the Company for Good Reason, then the Employee shall be entitled to the benefits provided below for the three (3) year period after Change in Control or the remaining portion of said three (3) year period following the date of termination. For purposes of this Agreement, "Change in Control" shall be deemed to have occurred if: (i) Company sells or otherwise disposes of all or substantially all of its assets; (ii) there is a merger or consolidation of Company with any other corporation or corporations, provided that the shareholders of Company, as a group, do not hold, immediately after such event, at least fifty percent (50%) of the voting power of the surviving or successor corporation; (iii) any person or entity, including any "person" as such term is used in Section 13(d)(3) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), becomes the "beneficial owner" (as defined in Rule 13(d-3) under the Exchange Act) of Common Stock of Company representing fifty percent (50%) or more of the combined voting power of the voting securities of Company (exclusive of

persons who are now officers or directors of Company); or (iv) the approval by the shareholders of a liquidation or dissolution of the Company.

- 1. "Accrued Obligations": On the tenth (10th) business day following the date of termination, the Company shall pay to the Employee the sum of (i) the Employee's annual base salary prorated through the date of termination to the extent not previously paid and (ii) any accrued vacation pay to the extent not previously paid.
- 2. "Severance Amount": The "Severance Amount" shall be an amount equal to 2.99 times the Employee's annual base salary in effect on the business day prior to the date of termination plus 2.99 times the Employee's average annual bonus for the two (2) years prior to the Change in Control. The Company shall set aside this amount in escrow for a period of three (3) years and the escrow agent shall pay to the Employee, beginning on the tenth (10th) business day following the date of termination, a monthly amount of one-thirty-sixth (1/36) of the severance amount less any monthly W-2, Schedule C or Schedule F earnings reportable on Internal Revenue Service Form 1040 which are received by the Employee from his present employer or any future employer or employers for a period of thirty-six (36) months or until the earlier exhaustion of the entire severance amount plus interest thereon. In the event of the Employee's death after a termination for which a "Severance Amount" is payable, the escrow agent shall continue to pay to the Employee's spouse or other named beneficiary the remaining obligation owed the Employee under the terms of this Agreement and the escrow agreement. The Company may, however, at its option, elect to pay the Severance Amount to the Employee, or in the event of his death, his spouse or other named beneficiary, in the form of a lump-sum cash payment on or before the date the first monthly payment is due; or the Company, at its option, at anytime during the term of the escrow agreement, can direct the escrow agent to pay the Employee, the Employee's spouse, or named beneficiary, as the case may be, the then remaining balance of the severance amount, plus any accrued and accumulated interest thereon, in the form of a lump-sum cash payment, and the rights and obligations of all parties under both this Agreement and escrow agreement shall be terminated.

In the event, subsequent to the Change in Control, the Employee (i) violates the provisions of Section 6 or (ii) engages in any conduct which would violate the provisions of Section 6 if the Restricted Period (as defined below) were still in effect (if the Restricted Period has expired or expires during such time), then any obligation of the Company under this Section 5(f) is terminated, and the Employee shall not be entitled to any further benefits under this Agreement. If this Section 5(f) is applicable, then the Restricted Period is the period set forth in Section 6(d)(3).

The severance amount set aside in escrow shall be invested according to the provisions of an escrow agreement and the interest earned included in the amount payable to the Employee. Any severance amounts not paid to the Employee shall be returned to the Company at the end of the thirty-six (36) month escrow period, or sooner should the Employee (i)

violate the provisions of Section 6 or (ii) engage in any conduct which would violate the provisions of Section 6 if the Restricted Period (as defined below) were still in effect (if the Restricted Period has expired or expires during such time). All interest earned on the account shall be paid to the Employee following the final severance payment. If this Section 5(f) is applicable, then the Restricted Period is the period set forth in Section 6(d)(3).

- 3. "Other Benefits": To the extent not previously provided, the Company shall timely pay or provide to the Employee and/or the Employee's family any other amounts or benefits including benefits from welfare benefit plans required to be paid or provided for which the Employee and/or the Employee's family is eligible to receive pursuant hereto and under any plan, program, policy or practice or contract or agreement of Peoples as those provided generally to other peer executives and their families during the ninety (90) day period immediately preceding the date hereof or, if more favorable to the Employee, as those provided generally after the date here of to other peer executives of Peoples and their families.
- 4. If termination of the Employee occurs less than three (3) years after such Change in Control, then the benefits provided by this Agreement shall be prorated on the ratio of the remaining portion of said three (3) year period to the full three (3) year period following Change in Control.

"Excess Parachute Payment": Anything herein to the contrary notwithstanding, in the event that an independent accountant shall determine that any payment or distribution by the Company, Peoples or their affiliates to or for the benefit of Employee (whether paid or payable or distributed or distributable pursuant to the terms hereof or otherwise) (a "Payment") would be nondeductible by the Company, Peoples or their affiliates for Federal income tax purposes because of Code §280G or would constitute an "excess parachute payment" (as defined in Code §280G), then the aggregate present value of amounts payable or distributable to or for the benefit of the Employee pursuant hereto or pursuant to any other agreement with the Company, Peoples or their affiliates because of the occurrence of a Change in Control (such payments or distributions are hereinafter referred to as "Agreement Payments") shall be reduced (but not below zero) to the Reduced Amount. For purposes of this paragraph, the "Reduced Amount" shall be an amount expressed in present value which maximizes the aggregate present value of Agreement Payments without causing any payment to be nondeductible by the Company, Peoples or their affiliates because of Code §280G or without causing any portion of the Payment to be subject to the excise tax imposed by Code §4999.

If the independent accountant determines that any Payment would be nondeductible by the Company, Peoples or their affiliates because of Code §280G or that any portion of the Payment would be subject to the excise tax imposed by Code §4999, the Company shall promptly give Employee notice to that effect. The Employee may then elect, in the Employee's sole discretion, which and how much of the Agreement

Payments shall be eliminated or reduced (as long as after such election the aggregate present value of the Agreement Payments equals the Reduced Amount, and shall advise the Company and Peoples in writing of the Employee's election within ten (10) days after the Employee's receipt of such notice. If no such election is made by the Employee within such ten (10) day period, Peoples may elect which and how much of the Agreement Payments shall be eliminated or reduced (as long as after such election the aggregate present value of the Agreement Payments equals the Reduced Amount) and shall notify the Employee promptly of such election. For purposes of this paragraph, present value shall be determined in accordance with Code §280G(d)(4). All determinations made by the independent accountant under this paragraph shall be binding upon the Company, Peoples or their affiliates and the Employee and shall be made within sixty (60) days of a termination of employment of the Employee. As promptly as practicable following such determination and the elections hereunder, the Company, Peoples or their affiliates, as applicable, shall pay to or distribute to or for the benefit of the Employee such amounts as are then due to the Employee hereunder and shall promptly pay to or distribute for the benefit of the Employee in the future such amounts as become due to the Employee hereunder.

As a result of the uncertainty in the application of Code §§280G and 4999 at the time of the initial determination by the independent accountant hereunder, it is possible that Agreement Payments will be made by the Company, Peoples and/or their affiliates which should not have been made ("Overpayment") or that additional Agreement Payments which have not been made by the Company, Peoples and/or their affiliates should have been made ("Underpayment"), in each case, consistent with the calculation of the Reduced Amount hereunder. In the event that the independent accountant, based upon the assertion of a deficiency by the Internal Revenue Service against the Employee, the Company, Peoples or any of their affiliates which the independent accountant believes has a high probability of success, determines that an Overpayment has been made, any such Overpayment shall be treated for all purposes as a loan to the Employee which the Employee shall repay to the Company, Peoples or their affiliates, as applicable,, together with interest at the applicable Federal rate provided for in Code §7872(f)(2); provided, however, that no amount shall be payable by the Employee to the Company, Peoples or their affiliates, as applicable, if and to the extent such payment would not reduce the amount which is subject to taxation under Code §4999 or if the period of limitations for assessment of tax under Code §4999 against the Employee shall have expired. If the Employee is required to repay an amount under this Section, the Employee shall repay such amount over a period of time not to exceed one (1) year for each Twenty-Five Thousand Dollars (\$25,000.00) which the Employee must repay to the Company, Peoples or their affiliates, as applicable. In the event that the independent accountant, based upon controlling precedent, determines that an Underpayment has occurred, any such Underpayment shall be promptly paid by the Company, Peoples or their affiliates, as applicable, to or for the benefit of the Employee

together with interest at the applicable Federal rate provided for in Code §7872(f)(2)(A).

The Company shall bear all costs the independent accountants may incur in connection with any calculations contemplated herein.

If this Section 5(f) applies, Employee shall not be entitled to receive any payments or benefits under Sections 5(c), 5(d) or 5(e).

VI. NON-COMPETITION.

- A. In consideration of the sums paid by Peoples to Employee pursuant to the Agreement and Plan Merger Agreement by and among Peoples, Peoples Merger Corporation and Renasant Bancshares, Inc. dated _______, 2004 (the "Merger Agreement") and the salary and benefit payments to be made to the Employee under the terms of this Agreement, Employee hereby covenants to the Company and Peoples that during the Restricted Period (as hereinafter defined), he shall not, except on behalf of the Company and its affiliates directly or indirectly, in his own capacity or through any other Person, whether as owner, consultant, executive, partner, member, manager, officer, director, venturer, agent, through stock ownership, investment of capital, lending of money or property, rendering of services (including, without limitation, rendering services as an employee) or otherwise, engage in the Business (as hereinafter defined) or any business similar thereto in the Territory (as hereinafter defined).
- B. During the Restricted Period, Employee covenants that he will not, directly or indirectly, in his own capacity or through any other Person (as defined above) (i) solicit or contact for business purposes any existing customer, supplier, or prospective customer or supplier, of the Company, Peoples or any of their affiliates for the purpose of competing with the Business for himself or for any other Person, (ii) induce, or attempt to induce, any employees, agents, consultants or suppliers of or to the Company, Peoples or any of their affiliates, or any other Person to do anything from which Employee is restricted by reason of this Section 6, (iii) interfere with existing or proposed agreements or other arrangements, or knowingly interfere with future agreements or other arrangements, between the Company, Peoples or any of their affiliates on the one hand and any other Person on the other hand or (iv) induce, attempt to induce, solicit, offer or aid others to offer employment or engagement as a consultant or agent to anyone who is an employee, agent or consultant of or to the Company, Peoples or any of their affiliates.
- C. Territory means Shelby and Fayette Counties in Tennessee and DeSoto County, Mississippi, or any other county in which Employee may be assigned to work for the Company, Peoples or their affiliates.

- D. The Restricted Period means:
 - (1) The period beginning with the date Employee commences employment and ending the later of either (i) five (5) years after the date hereof or (ii) one (1) year after Employee ceases to be an employee of the Company, Peoples or any of their affiliates, unless Peoples and Employee agree to an Extension Payment, in which case the period in clause (ii) of this sentence will be two (2) years; however, this Subsection (d)(1) shall not apply if Subsection(d)(2) or (d)(3) apply.
 - (2) If Employee is terminated under Section 5(c) or 5(d) and such termination is not in connection with a Change in Control, then the Restricted Period means the period beginning with the date Employee commences employment and ending the later of either (i) three (3) years after the date hereof or (ii) one (1) year after Employee ceases to be an employee of the Company, Peoples or any of their affiliates, unless Peoples and Employee agree to an Extension Payment, in which case the period in clause (ii) of this sentence will be two (2) years.
 - (3) If the termination occurs pursuant to Section 5(f), then the Restricted Period means the period beginning with the date Employee commences employment and ending three (3) years after the date hereof.
- E. "Business" means commercial banking, financial services or the lending of money.
- F. The Restricted Period shall be extended by the period of time, if any, during which Employee is in violation of the Employee's agreement and obligation. If Employee violates the provisions of this Section 6, then, in addition to, and not in lieu of, any other remedy available to the Company, Peoples or their affiliates, any obligation of the Company, Peoples or their affiliates to make any payment under Section 5 including, without limitation, any Extension Payment shall terminate.
- G. Employee acknowledges that a breach of the covenants contained in this Agreement, including the covenants contained in this Section 6, may cause irreparable damage to the Company, Peoples or their affiliates, the amount of which will be difficult to ascertain, and that the remedies at law for any such breach will be inadequate. Accordingly, Employee agrees, that, in addition to any other remedy which may be available at law or in equity, the Company, Peoples and each affiliate shall be entitled to specific performance and injunctive relief to prevent any actual, intended or likely breach. The parties acknowledge that the time, scope and other provisions of this Section 6 have been specifically negotiated by sophisticated commercial parties and agree that all such provisions are reasonable for a transaction of the nature provided for in the Merger Agreement and payments to Employee under Section 3 of this Agreement.
- H. In the event that the agreements in this Section 6 or any other provision contained in this Agreement shall be determined by any court of competent jurisdiction to be unenforceable such agreements or provisions shall be interpreted to extend only over the maximum period of time for which they may be enforceable and/or over the maximum geographical area as to which they may be enforceable and/or to the maximum extent in all other respects as to which they may be enforceable, all as determined by such court in

- such action so as to be enforceable to the extent consistent with then applicable law. The existence or alleged existence of any claim which Employee may have against the Company, Peoples or any affiliate shall not constitute a defense or bar to the enforcement of any of the provisions of this Section 6 and shall be pursued through separate court action by Employee.
- I. Employee may be permitted to own not more than five percent (5%) of the outstanding capital stock of (i) a publicly-owned corporation engaged in the Business or (ii) other community banks which do not have any offices in the Territory so long as, in each case, Employee is not in control of such corporation; does not serve as a director, officer, employee, agent or consultant to such corporation; or does not otherwise violate the provisions of this Section 6.
- VII. <u>DISCLOSURE OF INFORMATION</u>. Employee shall not, at any time during the term of Employee's employment at the Company, Peoples or at any affiliate or thereafter, disclose to any Person, except as required by law, any non-public information (including, without limitation, non-public information obtained prior to the date hereof) concerning the business, clients or affairs of the Company or Peoples, or any affiliate of the Company or Peoples, for any reason or purpose whatsoever. Employee shall not make any use of any of such non-public information for his own purpose or for the benefit of any Person except the Company. Upon the termination of Employee's employment at the Company, Employee shall return to the Company or Peoples, as appropriate, all property of the Company or Peoples and any affiliate of the Company or Peoples then in the possession of Employee and all books, records, computer tapes, discs or other electronic media and all other material containing non-public information concerning the business, clients or affairs of the Company, Peoples or any affiliate of the Company or Peoples. Employee shall not retain copies of any material required to be returned to the Company or Peoples.
- VIII. **INTELLECTUAL PROPERTY.** Employee shall promptly disclose, grant and assign to the Company and/or Peoples for their use and benefit any and all marks, designs, logos, inventions, improvements, business processes, technical information and suggestions relating in any way to the business conducted by the Company or Peoples, or any affiliate of the Company or Peoples, which he may develop or which may be acquired by Employee during the term of Employee's employment at the Company or Peoples (whether or not during usual working hours), together with all trademarks, patent applications, letters patent, copyrights and reissues thereof that may at any time be granted for or upon any such mark, design, logo, invention, improvement, process or technical information. In connection therewith:
 - 1. Employee shall without charge, but at the expense of the Company, promptly at all times hereafter execute and deliver such applications, assignments, descriptions and other instruments as may be necessary or proper in the opinion of the Company and Peoples to vest title to any such marks, designs, logos, inventions, improvements, business processes, technical information, trademarks, patent applications, patents, copyrights or reissues thereof in the Company and/or Peoples and to enable them to obtain and maintain the entire right and title thereto throughout the world;
 - 2. Employee shall render to the Company and/or Peoples at its expense all such assistance as it may require in the prosecution of applications for said trademarks, patents, copyrights or reissues thereof, in the prosecution or defense of interferences which may be declared involving any said trademarks,

- applications, patents or copyrights and in any litigation in which the Company, Peoples or any of their affiliates may be involved relating to any such trademarks, patents, inventions, improvements, processes or technical information; and
- 3. for the avoidance of doubt, the foregoing provisions shall be deemed to include an assignment of future copyright in accordance with Section 201 of the Copyright Act of 1986 and any amendment or re-enactment thereof relating in any way to the business conducted by the Company or Peoples or any affiliate of the Company or Peoples.
- IX. **HEADINGS.** Section and other headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- X. **INTEGRATED AGREEMENT.** This Agreement constitutes the entire understanding and agreement among the parties hereto with respect to the subject matter hereof, and there are no other agreements, understandings, restrictions, representations or warranties among the parties other than those set forth herein or provided for herein.
- XI. <u>AMENDMENTS</u>. This Agreement may be amended or modified at any time in any or all respects, but only by an instrument in writing executed by the parties hereto.
- XII. CHOICE OF LAW. The validity of the Agreement, the construction of its terms, and the determination of the rights and duties of the parties hereto shall be governed by and construed in accordance with the internal laws of the State of Tennessee excluding conflicts of law principles. Each party irrevocably (i) submits to the exclusive jurisdiction of any Tennessee state or federal court sitting in the Western District of Tennessee, with respect to matters arising out of or relating hereto, (ii) agrees that all claims with respect to such action or proceeding may be heard and determined in such Tennessee state or federal court, (iii) waives to the fullest possible extent, the defense of an inconvenient forum, (iv) waives the right to a trial by jury and (v) agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.
- XIII. NO STRICT CONSTRUCTION. The language used in this Agreement shall be deemed to be the language chosen by the parties to express their collective mutual intent, and no rule of strict construction shall be applied against any Person. The term "including" as used herein shall be by way of example and shall not be deemed to constitute a limitation of any term or provision contained herein.
- XIV. <u>ATTORNEY'S FEES AND COSTS</u>. If an action at law or in equity is necessary to enforce or interpret any provision of this Agreement, the prevailing party shall be entitled to reasonable attorney's fees, court costs and other expenses, in addition to any other relief to which such party may be entitled.
- XV. **NOTICES.** All notices and other communications under this Agreement must be in writing and will be deemed to have been duly given when (a) delivered by hand, (b) sent by facsimile to a facsimile number given below, provided that a copy is sent by a nationally recognized overnight delivery service (receipt requested), or (c) when received by the addressee, if sent by a nationally recognized overnight delivery service (receipt requested), in each case as follows:

If to Employee:

Francis J. Cianciola 2177 Germantown Road South Germantown, TN 38138 Facsimile No.: (901) 312-4098

If to Peoples or the Company:

Mr. E. Robinson McGraw President and CEO The Peoples Holding Company Post Office Box 709 209 Troy Street (38804) Tupelo, MS 38802 Facsimile No.: (662) 680-1234

or to such other addresses as a party may designate by notice to the other parties.

- XVI. <u>ASSIGNMENT</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, assigns, heirs, estate, legatees and legal representatives. The rights and obligations of the Company and Peoples under this Agreement may be assigned to or assumed by any other Person. Employees' rights or obligations hereunder may not be assigned to or assumed by any other Person. Any assignment by Peoples and/or the Company shall not affect the Employee's duties or responsibilities under this Agreement.
- XVII. SEVERABILITY. Each provision of the Agreement is intended to be severable. In the event that any one or more of the provisions contained in this Agreement shall for any reason be held to be invalid, illegal or unenforceable, the same shall not affect the validity or enforceability of any other provision of this Agreement, but this Agreement shall be construed as if such invalid, illegal or unenforceable provisions had never been contained herein.

 Notwithstanding the foregoing, however, no provision shall be severed if it is clearly apparent under the circumstances that the parties would not have entered into the Agreement without such provision.
- XVIII. SURVIVAL. The provisions of Sections 6 through 18 shall survive the termination of the employment period or termination of this Agreement.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written.

Francis J. Cianciola
RENASANT BANK
By:
Name:
Title:

 $Annex\ A\text{--}1-Page\ 67$

Ву:	
Name:	
Title:	

THE PEOPLES HOLDING COMPANY

Name:

Date of this Notice: , 2004

SCHEDULE 5.23

NOTICE OF OFFER TO PURCHASE WARRANTS ISSUED BY RENASANT BANCSHARES, INC.

THIS OFFER HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY APPLICABLE STATE SECURITIES LAW. THIS OFFER IS MADE ONLY TO ACCREDITED INVESTORS, AS DEFINED IN RULE 501, REGULATION D, PROMULGATED UNDER SUCH ACT

THE PEOPLES HOLDING COMPANY NOTICE OF OFFER TO PURCHASE WARRANTS ISSUED BY RENASANT BANCSHARES, INC.

Number of Warrants:	Warrant Exercise Price: \$10.0	00 per share	
Date of Issuance:	Warrant No.:		
Renasant Bancshares	, Inc. ("Renasant") has issued to you warrants to purchas	se shares of Renasant's \$1.00 par value voting common stock, as more f	ull

Renasant Bancshares, Inc. ("Renasant") has issued to you warrants to purchase shares of Renasant's \$1.00 par value voting common stock, as more fully described above (the "Warrants").

Renasant has now entered into that certain Agreement and Plan of Merger by and among The Peoples Holding Company ("Peoples"), Peoples Merger Corporation, and Renasant, dated February 17, 2004 (the "Merger Agreement"), pursuant to which Renasant will be merged with and into Peoples Merger Corporation (the "Merger"). After the consummation of the Merger, Renasant will cease to exist and Renasant Bank will be a wholly-owned subsidiary of an affiliate of Peoples. In connection with the Merger, Peoples now offers to purchase all of your outstanding Warrants, whether or not currently vested or exercisable, and to pay to you the Warrant Consideration (as defined herein) in exchange therefor. If the Merger is not consummated, the tender and assignment to Peoples of your Warrants will be void and of no effect.

In consideration of the tender and assignment of your Warrants to Peoples, as of the date of the completion of the Merger (the "Merger Date"), you will be entitled to payment by Peoples of the Warrant Consideration, which is defined as the excess of (a) the per share cash consideration to be paid to shareholders of Renasant under the Merger Agreement, over (b) the per share exercise price of your Warrants. The Warrant Consideration will be paid to you as of the Merger Date entirely in the form of cash and will be determined and paid with respect to all of your Warrants, whether or not they are presently exercisable or vested.

To tender and assign your Warrants, you must execute and deliver to Peoples Exhibit A hereto, and you must represent that you are an "accredited investor" as defined in Rule 501, Regulation D., promulgated under the Securities Act of 1933, as amended, by execution and delivery to Peoples of Exhibit B hereto. By execution of Exhibit A, you also represent that you have had access to all information related to the Warrants, the Merger, Renasant and Peoples that you reasonably consider important in making a decision to tender and assign your Warrants to Peoples and that you have had ample time to ask questions of Renasant's and Peoples' representatives concerning such matters as you deem relevant to your decision.

If you do not elect to tender and assign your Warrants to Peoples, as provided herein, upon the consummation of the Merger, your Warrants will be amended and converted to warrants to acquire \$5.00 par value voting common stock issued by Peoples. After the consummation of the Merger, your Warrants will entitle you to purchase a number of shares of Peoples common stock equal to the number of shares of Renasant common stock specified in your Warrants multiplied by the Exchange Ratio (as defined in the Merger Agreement) at an exercise price equal to the quotient of the exercise price specified in your Warrants divided by the Exchange Ratio. In all other respects, your Warrants will remain subject to the terms, conditions and limitations set forth in your individual agreement with Renasant. By execution of Exhibit A you acknowledge and consent to the conversion and you agree to the deemed amendment of the documents evidencing your Warrants to the extent necessary to reflect such conversion. You also represent that you have had access to all information related to the Warrants, the Merger, Renasant and Peoples that you reasonably consider important in making a decision to retain your Warrants and that you have had ample time to ask questions of Renasant's and Peoples' representatives concerning such matters as you deem relevant to your decision.

Your election, either to tender and assign your Warrants or to retain them, should be made using the attached form, Exhibit A hereto. If you elect to tender and assign your warrants to Peoples, you must also certify that you are an accredited investor on Exhibit B hereto. Your election and/or certification must be executed, dated, delivered to The Peoples Bank & Trust Company, Attention: Stuart R. Johnson, P.O. Box 709, Tupelo, Mississippi 38802-0709, on or before the Closing Date.

THE PEOPLES HOLDING COMPANY
By:

THE PEOPLES HOLDING COMPANY NOTICE OF OFFER TO PURCHASE WARRANTS ISSUED BY RENASANT BANCSHARES, INC.

EXHIBIT A

ELECTION

By execution below, I acknowledge that I have received from The Peoples Holding Company a Notice of Offer to Purchase Warrants Issued by Renasant
Bancshares, Inc. I also acknowledge that I have had access to all information related to the Warrants, the Merger, Renasant and Peoples that I reasonably consider
mportant in making a decision whether to tender and assign my Warrants and that I have had ample time to ask questions of Renasant's and Peoples'
epresentatives concerning such matters as I deem relevant to my decision. I elect:

Semai	atives concerning such matters as I deem relevant to my decision. I elect.		
	To tender and assign to Peoples all of my Warrants, whether or not presently exercisable and vested, in consideration of the payment to me of the Warrant Consideration more fully described in such notice. I certify that I am an "accredited investor" within the meaning of Rule 501, Regulation D., promulgated under the Securities Act of 1933, as amended, by execution and delivery of Exhibit B to such notice.		
	To retain my Warrants, subject to their conversion to the right to receive \$5.00 par value voting common stock issued by Peoples and to the adjustment of the exercise price thereof and number of shares subject thereto, to the extent necessary to prevent the dilution or enlargement acknowledge that this election shall be deemed to constitute an amendment of the documents evidencing my Warrants to the extent necessar reflect such conversion.		
	Sig	gnature	
	Pri	nt Name:	
	Da	te:, 2004	
	Annov A 1 Dago 71		

THE PEOPLES HOLDING COMPANY NOTICE OF OFFER TO PURCHASE WARRANTS ISSUED BY RENASANT BANCSHARES, INC.

STATUS AS ACCREDITED INVESTOR EXHIBIT B

As a condition of my tender and assignment of the Warrants to Peoples as more fully set forth in that certain Notice of Offer to Purchase Warrants Issued by Renasant Bancshares, Inc., I represent to Peoples that the information set forth below is true and correct:

I had individual gross income (exclusive of any income attributable to my spouse) of more than \$200,000 in each of the most recent two tax years,

<i>y</i> 1	of \$200,000 for the current tax year. (Gross income means adjusted gross etirement plan, and any amount by which income from long-term capital gains		
☐ My spouse and I together had gross income of more than \$300,000 in have gross income in excess of \$300,000 for the current tax year.	My spouse and I together had gross income of more than \$300,000 in each of the two most recent tax years, and I reasonably expect that we will have gross income in excess of \$300,000 for the current tax year.		
\square I have an individual net worth (total assets less total liabilities) or my	I have an individual net worth (total assets less total liabilities) or my spouse and I have a combined net worth in excess of \$1,000,000.		
Name	Name (if more than one)		
Social Security Number	Social Security Number		
Signature	Signature		
, 2004	, 2004		
THE OPER BY REON TO BURCHAGE CERTAIN WAR DANTE LOCKED BY DENIA CAN'T DANCOUADED. INC. HAC NOT DEEN DECICEERED			

THE OFFER BY PEOPLES TO PURCHASE CERTAIN WARRANTS ISSUED BY RENASANT BANCSHARES, INC. HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY APPLICABLE STATE SECURITIES LAW. THIS OFFER IS MADE ONLY TO ACCREDITED INVESTORS, AS DEFINED IN RULE 501, REGULATION D, PROMULGATED UNDER SUCH ACT.

ANNEX A-2

PLAN OF MERGER BY AND AMONG THE PEOPLES HOLDING COMPANY, PEOPLES MERGER CORPORATION and RENASANT BANCSHARES, INC.

This Plan of Merger ("Plan of Merger") is dated as of ______, 2004, by and among The Peoples Holding Company, a Mississippi corporation ("Acquiror"), Peoples Merger Corporation, a Tennessee corporation and a wholly owned subsidiary of Acquiror ("Merger Sub"), and Renasant Bancshares, Inc., a Tennessee corporation ("Seller"). Each of Acquiror, Merger Sub and Seller is a "party" to this Agreement, and one or more of them are "parties" hereto, as the context may require.

WITNESSETH:

WHEREAS, Acquiror, Merger Sub and Seller have entered into an Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"); and

WHEREAS, pursuant to the Merger Agreement and this Plan of Merger, and subject to the terms and conditions set forth therein and herein, Seller shall be merged with and into Merger Sub, with Merger Sub being the surviving corporation in such merger.

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements contained herein and in the Merger Agreement, the parties hereto do mutually agree as follows:

ARTICLE I DEFINITIONS

Except as otherwise provided herein, the capitalized terms set forth below shall have the following meanings:

"Effective Time" shall mean the date and time at which the Merger contemplated by this Plan of Merger becomes effective as provided in Section 2.2 of this Plan of Merger.

"Merger" shall refer to the merger of Seller with and into Merger Sub as provided in Section 2.1 of this Plan of Merger.

"Merging Corporations" shall mean Merger Sub and Seller.

"Seller Common Stock" shall mean the common stock, par value \$1.00 per share, of Seller.

"Seller Dissenting Shares" shall mean the shares of Seller Common Stock as to which dissenters' rights have been asserted and duly perfected in accordance with Section 48-23-102 et seq. of the TBCA (as defined in Section 2.1) and not effectively withdrawn.

"Stockholder Meeting" shall mean the meeting of the stockholders of Seller held pursuant to Section 5.8 of the Merger Agreement.

"Surviving Corporation" shall mean Merger Sub as the surviving corporation of the Merger.

ARTICLE II TERMS OF THE MERGER

- 2.1 THE MERGER. Subject to the terms and conditions set forth in the Merger Agreement, at the Effective Time, Seller shall be merged with and into Merger Sub pursuant to and in accordance with the Tennessee Business Corporation Act (the "TBCA"). Merger Sub shall be the Surviving Corporation in the Merger and shall continue to be governed by the laws of the State of Tennessee. At the Effective Time, the separate existence and corporate organization of Seller shall cease, and all right, title and interest in and to all real estate and other property owned by each of the Merging Corporations shall be allocated to and shall be vested in Merger Sub as the Surviving Corporation, without reversion or impairment, without further act or deed and without any transfer or assignment having occurred (but subject to any existing liens or other encumbrances thereon), and all liabilities and obligations of the Merging Corporations shall be allocated to Merger Sub as the Surviving Corporation, as the primary obligor therefor, and, except as set forth in the Merger Agreement, no other person shall be liable therefor, and all proceedings pending by or against any of the Merging Corporations shall be continued by or against the Surviving Corporation, and all liabilities, obligations, assets or rights associated with such proceedings shall be allocated to and vested the Surviving Corporation.
- 2.2 EFFECTIVE TIME. The Merger shall become effective on the date and at the time that Articles of Merger are filed with the Secretary of State of the State of Tennessee pursuant to Section 48-21-107 of the TBCA, unless a later date and time is specified as the effective time in such documents.
 - 2.3 NAME OF THE SURVIVING CORPORATION. The name of the Surviving Corporation shall be "Peoples Merger Corporation."
- 2.4 ARTICLES OF INCORPORATION. On and after the Effective Time, Articles of Incorporation of the Surviving Corporation shall be the Articles of Incorporation of Merger Sub until amended in accordance with applicable law.
- 2.5 BYLAWS. On and after the Effective Time, the Bylaws of the Surviving Corporation shall be the Bylaws of Merger Sub until amended in accordance with applicable law.

ARTICLE III CONVERSION OF SHARES

- 3.1 CONVERSION OF SHARES. All of the shares of Acquiror and Merger Sub issued and outstanding immediately prior to the Effective Time shall remain issued and outstanding after the Effective Time and shall be unaffected by the Merger. The manner and basis of converting the Seller Common Stock upon consummation of the Merger shall be as follows:
- (a) At the Effective Time, by virtue of the Merger and without any action on the part of Acquiror, Merger Sub, Seller or the holders of Seller Common Stock:
 - (i) Subject to the other provisions of this Section 3.1, each share of Seller Common Stock issued and outstanding immediately prior to the Effective Time (excluding any treasury shares, shares held by Acquiror or any of the subsidiaries of Acquiror or Seller (other than in a fiduciary capacity) and shares held by Dissenting Stockholders (as defined below)) shall be converted into the right to receive the Merger Consideration. The "Merger Consideration" shall mean

- either (a) cash in the amount of the Price Per Share (as defined below), without interest (the "Cash Consideration"), (b) a share of Acquiror Common Stock (as defined below), rounded to the nearest six decimals, equal to the Exchange Ratio (the "Stock Consideration"), or (c) a combination of Cash Consideration and Stock Consideration in accordance with subparagraph (iii) of this Section 3.1. As used herein, the term "Price Per Share" equals \$36.37, the term "Exchange Ratio" equals 1.117015 and the term "Acquiror Common Stock" means the common stock, \$5.00 per share par value, of the Acquiror.
- (ii) The number of shares of Seller Common Stock to be converted into the right to receive Cash Consideration shall not be less than 45% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be cancelled pursuant to clause (x) of this Section 3.1(a)) (the "Minimum Cash Election Number") and shall not be greater than 50% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be converted pursuant to clause (x) of this Section 3.1(a)) (the "Maximum Cash Election Number"). For purposes of determining both the Minimum Cash Election Number and Maximum Cash Election Number, Dissenting Stockholders shall be deemed to have made a Cash Election (as defined below) unless such Dissenting Stockholder shall effectively withdraw or lose (through failure to perfect or otherwise) his right to payment as a dissenting shareholder under the TBCA at or prior to the Effective Time. The number of shares of Seller Common Stock to be converted into the right to receive Stock Consideration shall be not less than 50% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be cancelled pursuant to clause (x) of this Section 3.1(a)) (the "Minimum Stock Election Number") and not greater than 55% of the number of shares of Seller Common Stock outstanding immediately prior to the Effective Time (excluding shares to be cancelled pursuant to clause (x) of this Section 3.1(a)) (the "Maximum Stock Election Number").
- (iii) Subject to the proration and election procedures set forth in this Section 3.1(a), each holder of record of shares of Seller Common Stock (excluding any treasury shares, shares held by Acquiror or any of the subsidiaries of Acquiror or Seller (other than in a fiduciary capacity) and shares held by Dissenting Stockholders) will be entitled to elect to receive (a) Cash Consideration for all such shares (a "Cash Election"), (b) Stock Consideration for all of such shares (a "Stock Election") or (c) Cash Consideration for 45% of such shares and Stock Consideration for 55% of such shares (a "Combination Election"). All such elections shall be made on a form designed for that purpose prepared by Acquiror and reasonably acceptable to Seller (a "Form of Election"). Holders of record of shares of Seller Common Stock who hold such shares as nominees, trustees or in other representative capacities (a "Representative") may submit multiple Forms of Election, provided that such Representative certifies that each such Form of Election covers all the shares of Seller Common Stock held by each such Representative for a particular beneficial owner.
- (iv) Acquiror and Seller shall mail the Form of Election to all Persons who are holders of Seller Common Stock on the record date for the Special Meeting, on a date that is not less than 20 Business Days prior to the Effective Time, and thereafter Acquiror and Seller shall each use its reasonable efforts to (x) mail the

Form of Election to all persons who become holders of Seller Common Stock during the period between the record date for the Special Meeting and 10:00 a.m., Memphis, Tennessee time, on the date seven Business Days prior to the anticipated Effective Time and (y) make the Form of Election available to all Persons who become holders of Seller Common Stock subsequent to such day and no later than the close of business on the fourth Business Day prior to the Effective Time. A Form of Election must be received by the Exchange Agent in the manner described below no later than by the close of business on the Business Day which is three Business Days immediately prior to the Effective Time (the "Election Deadline") in order to be effective. All elections will be irrevocable. As used herein "Special Meeting" shall mean the special meeting of the shareholders of Seller to approve the Merger Agreement, this Plan of Merger and the Merger.

- (v) Elections shall be made by holders of Seller Common Stock by mailing, faxing or otherwise delivering to the Exchange Agent, in a manner acceptable to Acquiror, a Form of Election. "Exchange Agent" shall mean the person mutually acceptable to Acquiror and Seller who is selected to act as exchange agent to perform the services specified in the Merger Agreement. To be effective, a Form of Election must be properly completed, signed and submitted to the Exchange Agent. Acquiror will have the discretion, which it may delegate in whole or in part to the Exchange Agent, to determine whether Forms of Election have been properly completed, signed and submitted and to disregard immaterial defects in Forms of Election. The decision of Acquiror (or the Exchange Agent) in such matters shall be conclusive and binding. Neither Acquiror nor the Exchange Agent will be under any obligation to notify any Person of any defect in a Form of Election.
- (vi) A holder of Seller Common Stock who does not submit a Form of Election which is received by the Exchange Agent prior to the Election Deadline shall be deemed to have made a Combination Election. If Acquiror or the Exchange Agent shall determine that any purported Cash Election or Stock Election was not properly made, such purported Cash Election or Stock Election shall be deemed to be of no force and effect and the holder of shares of Seller Common Stock making such purported Cash Election or Stock Election shall for purposes hereof be deemed to have made a Combination Election.
- (vii) All shares of Seller Common Stock which are subject to Cash Elections are referred to herein as "Cash Election Shares." All shares of Seller Common Stock which are subject to Stock Elections are referred to herein as "Stock Election Shares." If, after the results of the Forms of Election are calculated, the number of shares of Seller Common Stock to be converted into shares of Acquiror Common Stock exceeds the Maximum Stock Election Number, the Exchange Agent shall, after the Election Deadline but prior to the Effective Time, determine the number of Stock Election Shares which must be redesignated as Cash Election Shares in order to reduce the number of such shares to the Maximum Stock Election Number. All holders who have Stock Election Shares shall, on a pro rata basis, have such number of their Stock Election Shares redesignated as Cash Election Shares so that the Maximum Stock Election Number and the Minimum Cash Election Number are achieved. If, after the results of the Forms of Election are calculated, the number of shares of Seller

Common Stock to be converted into cash exceeds the Maximum Cash Election Number, the Exchange Agent shall, after the Election Deadline but prior to the Effective Time, determine the number of Cash Election Shares which must be redesignated as Stock Election Shares in order to reduce the amount of such cash to the Maximum Cash Election Number. All holders who have Cash Election Shares shall, on a pro rata basis, have such number of their Cash Election Shares redesignated as Stock Election Shares so that the Maximum Cash Election Number and the Minimum Stock Election Number are achieved. Notwithstanding the foregoing, no redesignation shall be effected for a holder who has made a Cash Election but, as a result of such redesignation, would receive fewer than 10 shares of Acquiror Common Stock in exchange for all of such holder's shares of Seller Common Stock unless such redesignation is necessary for the Merger to qualify as a reorganization under Section 368 of the Code as described in Section 3.1(a)(viii). In this event, the Cash Election Shares of the remaining holders of shares of Seller Common Stock shall be redesignated on a pro rata basis to achieve the Maximum Cash Election Number and the Minimum Stock Election Number. Holders who make Combination Elections will not be subject to the redesignation procedures described herein unless such redesignation is necessary for the Merger to qualify as a reorganization under Section 368 of the Code as described in Section 3.1(a)(viii). Dissenting Stockholders who are deemed to have made Cash Elections shall not be subject to the redesignation procedure described herein. Acquiror or the Exchange Agent shall make all computations contemplated by this Section 3.1(a) and all such computations shall be conclusive and binding on the holders of Seller Common Stock.

(viii) It is the intent of the parties that the Merger qualify as a reorganization under Section 368 of the Code. In order to qualify as a reorganization under Section 368 of the Code, the Merger must satisfy the Continuity of Interest requirement set forth in Treasury Regulation Section 1.368-1(e). If after the Exchange Agents completes the redesignation procedures set forth in Section 3.1(a)(vii) (or if there is no redesignation required by the Exchange Agent after the Exchange Agent notifies the parties that redesignation is not required), the Acquiror determines, in good faith after consultation with its tax counsel, that the value of Acquiror Common Stock included in the Merger Consideration will not be sufficient to satisfy the Continuity of Interest requirement, then, except as provided hereinbelow, Acquiror shall increase the Exchange Ratio and decrease the Price Per Share to the extent Acquiror, in good faith after consultation with its tax counsel, deems necessary to meet the Continuity of Interest requirement. Notwithstanding anything to the contrary in this Section 3.1(a)(viii), if the increase in the Exchange Ratio would cause Acquiror to issue more than 85,000 additional shares of Acquiror Common Stock pursuant to this Section 3.1(a)(viii), then the adjustment described in this Section 3.1(a)(viii) shall be at Acquiror's option and shall not be mandatory. Any adjustment of the Exchange Ratio and the Price Per Share will be done in a manner so that the reduction in Cash Consideration included in the Merger Consideration (prior to any adjustment for fractional shares) due to the adjustment to the Price Per Share shall be equal to the increase in value of Acquiror Common Stock included in the Merger Consideration (prior to any adjustment for fractional shares) due to the adjustment to the Exchange Ratio. Acquiror shall notify Seller of any adjustment pursuant to this Section 3.1(a)(viii) promptly after the calculation of such

- adjustment but in no event less than one Business Day prior to the Effective Time. The adjustment of the Exchange Ratio and the Price Per Share by Acquiror under this Section 3.1(a)(viii) shall be conclusive and binding on the parties to this Agreement. For purposes of this Section 3.1(a)(viii), the Continuity of Interest requirement is agreed to be at least 40% continuity.
- (ix) After the redesignation procedure, if any, and the adjustment procedure, if any, set forth in Section 3.1(a)(vii) and (viii), respectively, are completed, all Cash Election Shares and 45% of the shares of Seller Common Stock which are subject to Combination Elections shall be converted into the right to receive the Cash Consideration, and all Stock Election Shares and 55% of the shares of Seller Common Stock which are subject to Combination Elections shall be converted into the right to receive the Stock Consideration. Such certificates previously evidencing shares of Seller Common Stock shall be exchanged for (a) certificates evidencing the Stock Consideration, or (b) the Cash Consideration, multiplied in each case by the number of shares previously evidenced by the cancelled certificate, upon the surrender of such certificates to the Exchange Agent, without interest. Notwithstanding the foregoing, however, no fractional shares of Acquiror Common Stock shall be issued, and, in lieu thereof, a cash payment shall be made pursuant to Section 3.1(b).
- (x) Each share of Seller Common Stock held in the treasury of Seller and each share of Seller Common Stock owned by Acquiror or any subsidiary of Acquiror or Seller (other than in a fiduciary capacity) immediately prior to the Effective Time shall be cancelled and extinguished without any conversion thereof, and no payment shall be made with respect thereto.
- (b) No certificates or scrip representing fractional shares of Acquiror Common Stock will be issued as a result of the Merger. In lieu of the issuance of fractional shares pursuant to Section 3.1(a) of this Agreement, cash adjustments (without interest) will be paid to the holder of Seller Common Stock in respect of any fraction of a share of Acquiror Common Stock that would otherwise be issuable to such holder of Seller Common Stock, and the amount of such cash adjustment shall be determined by multiplying the fraction of a share of Acquiror Common Stock otherwise issuable by the average of the closing price of one share of Acquiror Common Stock for the 10 trading days immediately preceding the last trading day immediately prior to the Closing Date as reported by the American Stock Exchange, and no such holder shall be entitled to dividends, voting rights or any other right of stockholders in respect of any fractional share.
- 3.2 EXCHANGE OF CERTIFICATES FOR STOCK AND/OR CASH. After the Effective Time, each holder of a certificate previously representing outstanding shares of Seller Common Stock shall surrender and exchange such certificates for the Merger Consideration in the manner provided in Section 2.2 of the Merger Agreement.
- 3.3 DISSENTING SHARES. No outstanding share of Seller Common Stock as to which the holder has exercised dissenters rights under the TBCA shall be converted into or represent a right to receive the Merger Consideration, and the holder thereof (such holder, a "Dissenting Stockholder") shall be entitled only to such rights as are granted by the TBCA. If any Dissenting Stockholder shall effectively withdraw or lose (through failure to perfect or otherwise) his right to such payment at or prior to the Effective Time, such holder's shares of the Seller Common Stock shall be converted into a right to receive the Merger Consideration in accordance with the applicable provisions of the Merger Agreement and this Plan of Merger as if such holder had made a Combination Election. If such holder shall

effectively withdraw or lose (through failure to perfect or otherwise) his right to such payment after the Effective Time, each share of the Seller Common Stock of such holder shall be converted, on a share-by-share basis, into the right to receive the Merger Consideration in accordance with the applicable provisions of the Merger Agreement and this Plan of Merger as if such holder had made a Combination Election.

ARTICLE IV MISCELLANEOUS

- 4.1 CONDITIONS PRECEDENT. The respective obligations of each party under this Plan of Merger shall be subject to the satisfaction, or waiver by the party permitted to do so, of the conditions set forth in Article VI of the Merger Agreement.
- 4.2 TERMINATION. This Plan of Merger shall be terminated automatically without further act or deed of either of the parties hereto in the event of the termination of the Merger Agreement in accordance with Article VII thereof.
- 4.3. AMENDMENTS. To the extent permitted by the TBCA, this Plan of Merger may be amended by a subsequent writing signed by each of the parties hereto upon the approval of the Board of Directors of each of the parties hereto; provided, however, that the provisions of Article III of this Plan of Merger relating to the consideration to be paid for the shares of Seller Common Stock shall not be amended after the Stockholder Meeting so as to modify either the amount or the form of such consideration or to otherwise materially adversely affect the stockholders of Seller without the approval of the stockholders of Seller.
 - 4.4 SUCCESSORS. This Plan of Merger shall be binding on the successors of Acquiror, Merger Sub and Seller.

IN WITNESS WHEREOF, Acquiror, Merger Sub and Seller have caused this Plan of Merger to be executed by their duly authorized officers as of the day and year first above written.

THE	E PEOPLES HOLDING COMPANY
By:	
	Name:
	Title:
PEC	OPLES MERGER CORPORATION
By:	
	Name:
	Title:
REN	NASANT BANCSHARES, INC.
By:	
	Name:
	Title:

ANNEX B-1

FORM OF LOCK-UP AND NON-COMPETITION AGREEMENT

LOCK-UP AND NON-COMPETITION AGREEMENT

This Agreement is made and executed as of the 17th day of February, 2004, between The Peoples Holding Company, a Mississippi corporation ("Acquiror"), and the undersigned director ("Renasant Official") of Renasant Bancshares, Inc., a Tennessee corporation ("Renasant"), or affiliates of Renasant.

Acquiror and Renasant have entered into an Agreement and Plan of Merger (the "Plan of Merger"), pursuant to which Renasant and Acquiror agree that Renasant will merge (the "Merger") with and into Peoples Merger Corporation, a wholly owned subsidiary of Acquiror (the "Acquisition Subsidiary"), and the Acquisition Subsidiary shall be the surviving entity. In consideration of (i) the payment on the effective date of the Merger of \$250,000.00 by Renasant to the Renasant Official pursuant to the Change of Control Agreement dated as of January 30, 2004 by and among the Renasant Official, Renasant and Renasant Bank, Inc., (ii) the expenses that Acquiror will incur in connection with the transactions contemplated by the Plan of Merger, and (iii) in order to preserve the value of the franchise to be purchased by Acquiror and induce Acquiror to proceed to incur such expenses, the Renasant Official makes the following agreements in favor of Acquiror:

1. Undertakings of Renasant Official.

- 1.1 The Renasant Official agrees and undertakes to vote or cause to be voted in favor of the approval of the Plan of Merger all shares of common stock of Renasant, \$1.00 par value (the "Renasant Stock"), as to which he has voting power (other than shares held in a fiduciary capacity), which amount of shares is shown on the schedule attached hereto and made a part hereof, at any meeting or meetings (including any and all adjournments thereof) held on or before August 31, 2004. The parties hereto acknowledge and agree that nothing in this section or this Agreement is intended to dictate or require that the Renasant Official vote as a director in any manner.
- 1.2 The Renasant Official further agrees that he will not transfer any of the shares of Renasant Stock over which he has dispositive power, which number of shares is shown on the schedule attached hereto and made a part hereof, until the vote upon the Plan of Merger by Renasant's stockholders has been taken or until the Plan of Merger has been terminated pursuant to the provisions thereof, except (i) for transfers by operation of law, and (ii) for transfers in connection with which Acquiror has consented to the transfer and the transferee shall agree in writing with Acquiror to be bound by this Agreement as fully as the undersigned.
 - 1.3 This Agreement shall terminate at such time as the Plan of Merger terminates.

2. Agreement Not to Compete.

The Renasant Official agrees that for a period of five years following the Closing (as that terms is defined in the Plan of Merger), the Renasant Official will not serve as an officer or director, or acquire 5% or more of the outstanding equity securities, of any bank or savings and loan association or bank holding company, or federal or state chartered bank, savings bank, thrift, homestead association, savings association, savings and loan association or cooperative bank that has its principal business location within any of the following counties in Tennessee: Fayette and Shelby and DeSoto county, Mississippi; provided, however, that the Renasant Official shall be entitled to serve as a director of a bank holding company or a bank with assets in excess of twenty billion dollars.

3. Miscellaneous.

- 3.1 The provisions of this Agreement shall be enforceable through an action for damages at law or a suit for specific performance or other appropriate extraordinary relief, the Renasant Official acknowledging that remedies at law for breach or default might be or become inadequate.
- 3.2 The Renasant Official acknowledges and agrees that this Agreement is executed in connection with the sale of all of the business of Renasant. The Renasant Official further acknowledges and represents that the provisions of this Agreement will not work a hardship on him and will not prevent him from engaging in his occupation.
- 3.3 To the extent permitted under applicable law, any provision of this Agreement may be amended or modified at any time, either before or after its approval by an agreement in writing among the parties hereto.
- 3.4 This Agreement may be executed in counterparts, each of which shall be deemed to constitute an original. Each such counterpart shall become effective when one counterpart has been signed by each party hereto.
- 3.5 This Agreement shall be governed by, and interpreted in accordance with, the laws of the State of Tennessee applicable to agreements made and entirely to be performed within such State, except as federal law may be applicable.
 - 3.6 The Renasant Official may not assign any of his rights or obligations under this Agreement to any other person.
- 3.7 This Agreement supersedes any and all oral or written agreements and understandings heretofore made between the parties hereto relating to the subject matter hereof and contains the entire agreement of the parties relating to the subject matter hereof; provided, however, that notwithstanding the foregoing, this Agreement does not modify or amend any stock option agreement, employment agreement, option or similar employee benefit agreement between Renasant or an affiliate of Renasant and the Renasant Official. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors, heirs and legatees.
- 3.8 For a period of two years from and after the effective date of the Merger, Acquiror will cause the Renasant Official to be appointed to the Board of Directors of Renasant Bank unless such Renasant Official (i) violates this Agreement or (ii) in the good faith opinion of Acquiror does not possess the qualifications to serve as a director of Renasant Bank. So long as the Renasant Official remains a director of Renasant Bank during such two year period, he or she shall be entitled to receive director fees in accordance with the policy of Renasant Bank on the date of the Plan of Merger.

IN WITNESS WHEREOF, the parties have signed this Agreement effective as of the date first set forth above.		
	THE PEOPLES HOLDING COMPANY	
	Ву:	
	RENASANT OFFICIAL	

SCHEDULE TO LOCK-UP AND NON-COMPETITION AGREEMENT

Number of shares of common stock, \$1.00 par value, of Renasant Bancshares, Inc. owned by the Renasant Official: ______ shares.

Annex B-1 – Page 4

ANNEX B-2

FORM OF LOCK-UP AGREEMENT

LOCK-UP AGREEMENT

This Agreement is made and executed as of the 17th day of February, 2004, between The Peoples Holding Company, a Mississippi corporation ("Acquiror") and the undersigned individual executive officer and director ("Renasant Official") of Renasant Bancshares, Inc., a Tennessee corporation ("Renasant") or an affiliate of Renasant.

Acquiror and Renasant have entered into an Agreement and Plan of Merger (the "Plan of Merger"), pursuant to which Renasant and Acquiror agree that Renasant will merge with and into Peoples Merger Corporation, a wholly owned subsidiary of Acquiror (the "Acquisition Subsidiary") and the Acquisition Subsidiary shall be the surviving entity. In consideration of the expenses that Acquiror will incur in connection with the transactions contemplated by the Plan of Merger and in order to preserve the value of the franchise to be purchased by Acquiror and induce Acquiror to proceed to incur such expenses, the Renasant Official makes the following agreements in favor of Acquiror:

1. Undertakings of Renasant Official.

- 1.1 The Renasant Official agrees and undertakes to vote or cause to be voted in favor of the approval of the Plan of Merger all shares of common stock of Renasant, \$1.00 par value (the "Renasant Stock"), as to which he has voting power (other than shares held in a fiduciary capacity), which amount of shares is shown on the Schedule attached hereto and made a part hereof, at any meeting or meetings (including any and all adjournments thereof) held on or before August 31, 2004. The parties hereto acknowledge and agree that nothing in this Section or this Agreement is intended to dictate or require that the Renasant Official vote as a director in any manner.
- 1.2 The Renasant Official further agrees that he will not transfer any of the shares of Renasant Stock over which he has dispositive power, which number of shares is shown on the Schedule attached hereto and made a part hereof, until the vote upon the Plan of Merger by Renasant's shareholders has been taken or until the Plan of Merger has been terminated pursuant to the provisions thereof, except (i) for transfers by operation of law, and (ii) for transfers in connection with which Acquiror has consented to the transfer and the transferee shall agree in writing with Acquiror to be bound by this Agreement as fully as the undersigned.
 - 1.3 This Agreement shall terminate at such time as the Agreement and Plan of Merger terminates.

2. [Reserved]

3. Miscellaneous.

- 3.1 The provisions of this Agreement shall be enforceable through an action for damages at law or a suit for specific performance or other appropriate extraordinary relief, the Renasant Official acknowledging that remedies at law for breach or default might be or become inadequate.
- 3.2 The Renasant Official acknowledges and agrees that this Agreement is executed in connection with the sale of all of the business of Renasant. The Renasant Official further acknowledges

and represents that the provisions of this Agreement will not work a hardship on him and will not prevent him from engaging in his occupation.

- 3.3 To the extent permitted under applicable law, any provision of this Agreement may be amended or modified at any time, either before or after its approval by an agreement in writing among the parties hereto.
- 3.4 This Agreement may be executed in counterparts, each of which shall be deemed to constitute an original. Each such counterpart shall become effective when one counterpart has been signed by each party hereto.
- 3.5 This Agreement shall be governed by, and interpreted in accordance with, the laws of the State of Tennessee applicable to agreement made and entirely to be performed within such State, except as federal law may be applicable.
 - 3.6 The Renasant Official may not assign any of his rights or obligations under this Agreement to any other person.
- 3.7 This Agreement supersedes any and all oral or written agreements and understandings heretofore made between the parties hereto relating to the subject matter hereof and contains the entire agreement of the parties relating to the subject matter hereof; provided, however, that notwithstanding the foregoing, this Agreement does not modify or amend any stock option agreement, employment agreement, option or similar employee benefit agreement between Renasant or the Bank and the Renasant Official. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the parties hereto, and their respective successors, heirs and legatees.

IN WITNESS WHEREOF, the parties have signed this Agreement effective as of the date first set forth above.

THE PEOPLES HOLDING COMPANY
By:
RENASANT OFFICIAL

Annex B-2 – Page 2

SCHEDULE TO LOCK-UP AGREEMENT

Number of shares of common stock, \$1.00 par value, of Renasant Bancshares, Inc. owned by the Renasant Official: ______ shares.

Annex B-2 – Page 3

ANNEX C

February 17, 2004

Board of Directors Renasant Bancshares, Inc. 2177 Germantown Road South Germantown, Tennessee 38138

Members of the Board:

You have requested our opinion as to the fairness, from a financial point of view, to the holders of the outstanding shares of common stock of Renasant Bancshares, Inc. ("Renasant") of the Merger Consideration, as defined below, in the proposed merger between Renasant and The Peoples Holding Company, Tupelo, Mississippi (the "Company").

Pursuant to an Agreement and Plan of Merger to be dated on or about February 17, 2004 (the "Merger Agreement"), the Company has agreed to exchange cash and shares of its common stock equal to \$36.37 per share for each of Renasant's outstanding shares of common stock. The options of Renasant outstanding as of the Effective Date shall entitle the holder thereof to purchase a number of shares of Company common stock equal to the number of shares of Renasant common stock multiplied by the exchange ratio at an exercise price equal to the quotient of the exercise price specified in such Renasant option divided by the exchange ratio.

The warrants of Renasant outstanding as of the Effective Date may be sold for an amount equal to the number of shares of Company common stock specified in such warrant times the value per share to be received by the holders of Renasant common stock as of the Effective Date less the aggregate exercise price for all shares of Renasant common stock specified in such warrant or may be converted into the right to purchase a number of shares of Company common stock equal to the number of shares of Renasant common stock specified in such warrant multiplied by the exchange ratio at an exercise price equal to the quotient of the exercise price specified in such warrant divided by the exchange ratio.

The value to be received by holders of each of the several equity interests may be adjusted pursuant to the terms of the Merger Agreement and the interests together are referred to herein as the Merger Consideration. The price of the Company's common stock is subject to market variations. Pursuant to the Merger Agreement, Renasant will be merged with and into Peoples Merger Corporation, a wholly-owned subsidiary of the Company (the "Merger").

As of the date hereof, the Merger Consideration is fair, from a financial point of view.

Alex Sheshunoff & Co. Investment Banking, LP ("Sheshunoff") is regularly engaged in the valuation of securities in connection with mergers and acquisitions, private placements, and valuations for estate, corporate and other purposes. Sheshunoff did not advise Renasant in connection with the proposed merger. The type and amount of consideration and the terms and conditions of the Merger were negotiated directly by and between Renasant and the Company.

In connection with our opinion, we, among other things:

- 1. Reviewed a draft of the Merger Agreement;
- 2. Evaluated Renasant's consolidated results based upon a review of its regulatory reports for the three-year period ending December 31, 2003;
- 3. Conducted conversations with executive management regarding recent and projected financial performance of Renasant;
- 4. Compared Renasant's recent operating results with those of certain other banks in the United States that have recently been acquired;
- 5. Compared Renasant's recent operating results with those of certain other banks located in the Southeast Region of the United States (as defined SNL Financial) that have recently been acquired;
- 6. Compared the pricing multiples for Renasant in the Merger to those of certain other banks in the United States that have recently been acquired;
- 7. Compared the pricing multiples for Renasant in the Merger to those of certain other banks located the Southeast Region of the United States (as defined SNL Financial) that have recently been acquired;
- 8. Analyzed the present value of the after-tax cash flows Renasant could produce through the year 2008, based on assumptions provided by management;
- 9. Reviewed the historical stock price data and trading volume of the Company's common stock and the lack of any active market for the common stock of Renasant; and
- 10. Performed such other analyses as it deemed appropriate.

We assumed and relied upon, without independent verification, the accuracy and completeness of the information provided to us by Renasant for the purposes of this opinion. In addition, where appropriate, we relied upon publicly available information that we believe to be reliable, accurate, and complete; however, we cannot guarantee the reliability, accuracy, or completeness of any such publicly available information.

We did not make an independent evaluation of the assets or liabilities of Renasant or the Company, nor were we furnished with any such appraisals. We are not experts in the evaluation of loan portfolios for the purposes of assessing the adequacy of the allowance for loan and lease losses and assumed that such allowances for each of the companies are, in the aggregate, adequate to cover such losses. We did not conduct an on-site review of Renasant or its operations.

We assumed that all required regulatory approvals will be received in a timely fashion and without any conditions or requirements that could adversely affect the Merger or the Company's operations following the Merger.

Our opinion is necessarily based on economic, market, and other conditions as in effect on, and the information made available to us as of, the date hereof. Events occurring after the date hereof could materially affect the assumptions used in preparing this opinion.

Our opinion is limited to the fairness of the Merger Consideration, from a financial point of view, to the holders of Renasant common stock. Moreover, this letter and the opinion expressed herein do not constitute a recommendation to any stockholder as to any approval of the Merger or the Merger

Agreement. It is understood that this letter is for the information of the Board of Directors of Renasant and may not be used for any other purpose without our prior written consent.

Based on the foregoing and such other matters we have deemed relevant, it is our opinion, as of the date hereof, that the Merger Consideration to be received by the Renasant stockholders pursuant to the Merger is fair, from a financial point of view.

Very truly yours,

/s/ ALEX SHESHUNOFF & CO. INVESTMENT BANKING, LP

ANNEX D

SECTION 48 OF THE TENNESSEE BUSINESS CORPORATION ACT CHAPTER 23 – DISSENTERS' RIGHTS.

PART 1 RIGHT TO DISSENT AND OBTAIN PAYMENT FOR SHARES

48-23-101. Chapter definitions.

As used in this chapter, unless the context otherwise requires:

- (1) "Beneficial shareholder" means the person who is a beneficial owner of shares held by a nominee as the record shareholder;
- (2) "Corporation" means the issuer of the shares held by a dissenter before the corporate action, or the surviving or acquiring corporation by merger or share exchange of that issuer:
- (3) "Dissenter" means a shareholder who is entitled to dissent from corporate action under §48-23-102 and who exercises that right when and in the manner required by part 2 of this chapter;
- (4) "Fair value", with respect to a dissenter's shares, means the value of the shares immediately before the effectuation of the corporate action to which the dissenter objects, excluding any appreciation or depreciation in anticipation of the corporate action;
- (5) "Interest" means interest from the effective date of the corporate action that gave rise to the shareholder's right to dissent until the date of payment, at the average auction rate paid on United States treasury bills with a maturity of six (6) months (or the closest maturity thereto) as of the auction date for such treasury bills closest to such effective date;
- (6) "Record shareholder" means the person in whose name shares are registered in the records of a corporation or the beneficial owner of shares to the extent of the rights granted by a nominee certificate on file with a corporation; and
- (7) "Shareholder" means the record shareholder or the beneficial shareholder.

48-23-102. Right to dissent.

- (a) A shareholder is entitled to dissent from, and obtain payment of the fair value of the shareholder's shares in the event of, any of the following corporate actions:
 - (1) Consummation of a plan of merger to which the corporation is a party:
 - (A) If shareholder approval is required for the merger by §48-21-104 or the charter and the shareholder is entitled to vote on the merger; or
 - (B) If the corporation is a subsidiary that is merged with its parent under §48-21-105;
 - (2) Consummation of a plan of share exchange to which the corporation is a party as the corporation whose shares will be acquired, if the shareholder is entitled to vote on the plan;
 - (3) Consummation of a sale or exchange of all, or substantially all, of the property of the corporation other than in the usual and regular course of business, if the shareholder is entitled to vote on the sale or exchange, including a sale in dissolution, but not including a sale pursuant to court order or a sale for cash pursuant to a plan by which all or substantially all of the net proceeds of the sale will be distributed to the shareholders within one (1) year after the date of sale;
 - (4) An amendment of the charter that materially and adversely affects rights in respect of a dissenter's shares because it:
 - (A) Alters or abolishes a preferential right of the shares;
 - (B) Creates, alters, or abolishes a right in respect of redemption, including a provision respecting a sinking fund for the redemption or repurchase, of the shares:

- (C) Alters or abolishes a preemptive right of the holder of the shares to acquire shares or other securities;
- (D) Excludes or limits the right of the shares to vote on any matter, or to cumulate votes, other than a limitation by dilution through issuance of shares or other securities with similar voting rights; or
- (E) Reduces the number of shares owned by the shareholder to a fraction of a share, if the fractional share is to be acquired for cash under §48-16-104: or
- (5) Any corporate action taken pursuant to a shareholder vote to the extent the charter, bylaws, or a resolution of the board of directors provides that voting or nonvoting shareholders are entitled to dissent and obtain payment for their shares.
- (b) A shareholder entitled to dissent and obtain payment for the shareholder's shares under this chapter may not challenge the corporate action creating the shareholder's entitlement unless the action is unlawful or fraudulent with respect to the shareholder or the corporation.
- (c) Notwithstanding the provisions of subsection (a), no shareholder may dissent as to any shares of a security which, as of the date of the effectuation of the transaction which would otherwise give rise to dissenters' rights, is listed on an exchange registered under § 6 of the Securities Exchange Act of 1934, as amended, or is a "national market system security," as defined in rules promulgated pursuant to the Securities Exchange Act of 1934, as amended.

48-23-103. Dissent by nominees and beneficial owners.

- (a) A record shareholder may assert dissenters' rights as to fewer than all the shares registered in the record shareholder's name only if the record shareholder dissents with respect to all shares beneficially owned by any one (1) person and notifies the corporation in writing of the name and address of each person on whose behalf the record shareholder asserts dissenters' rights. The rights of a partial dissenter under this subsection are determined as if the shares as to which the partial dissenter dissents and the partial dissenter's other shares were registered in the names of different shareholders.
- (b) A beneficial shareholder may assert dissenters' rights as to shares of any one (1) or more classes held on the beneficial shareholder's behalf only if the beneficial shareholder:
 - (1) Submits to the corporation the record shareholder's written consent to the dissent not later than the time the beneficial shareholder asserts dissenters' rights; and
 - (2) Does so with respect to all shares of the same class of which the person is the beneficial shareholder or over which the person has power to direct the vote.

PART 2 PROCEDURE FOR EXERCISE OF DISSENTERS' RIGHTS

48-23-201. Notice of dissenters' rights.

- (a) If proposed corporate action creating dissenters' rights under §48-23-102 is submitted to a vote at a shareholders' meeting, the meeting notice must state that shareholders are or may be entitled to assert dissenters' rights under this chapter and be accompanied by a copy of this chapter.
- (b) If corporate action creating dissenters' rights under §48-23-102 is taken without a vote of shareholders, the corporation shall notify in writing all shareholders entitled to assert dissenters' rights that the action was taken and send them the dissenters' notice described in §48-23-203.
- (c) A corporation's failure to give notice pursuant to this section will not invalidate the corporate action.

48-23-202. Notice of intent to demand payment.

- (a) If proposed corporate action creating dissenters' rights under §48-23-102 is submitted to a vote at a shareholders' meeting, a shareholder who wishes to assert dissenters' rights must:
 - (1) Deliver to the corporation, before the vote is taken, written notice of the shareholder's intent to demand payment for the shareholder's shares if the proposed action is effectuated; and

- (2) Not vote the shareholder's shares in favor of the proposed action. No such written notice of intent to demand payment is required of any shareholder to whom the corporation failed to provide the notice required by §48-23-201.
- (b) A shareholder who does not satisfy the requirements of subsection (a) is not entitled to payment for the shareholder's shares under this chapter.

48-23-203. Dissenters' notice.

- (a) If proposed corporate action creating dissenters' rights under §48-23-102 is authorized at a shareholders' meeting, the corporation shall deliver a written dissenters' notice to all shareholders who satisfied the requirements of §48-23-202.
- (b) The dissenters' notice must be sent no later than ten (10) days after the corporate action was authorized by the shareholders or effectuated, whichever is the first to occur, and must:
 - (1) State where the payment demand must be sent and where and when certificates for certificated shares must be deposited;
 - (2) Inform holders of uncertificated shares to what extent transfer of the shares will be restricted after the payment demand is received;
 - (3) Supply a form for demanding payment that includes the date of the first announcement to news media or to shareholders of the principal terms of the proposed corporate action and requires that the person asserting dissenters' rights certify whether or not the person asserting dissenters' rights acquired beneficial ownership of the shares before that date;
 - (4) Set a date by which the corporation must receive the payment demand, which date may not be fewer than one (1) nor more than two (2) months after the date the subsection (a) notice is delivered; and
 - (5) Be accompanied by a copy of this chapter if the corporation has not previously sent a copy of this chapter to the shareholder pursuant to §48-23-201.

48-23-204. Duty to demand payment.

- (a) A shareholder sent a dissenters' notice described in §48-23-203 must demand payment, certify whether the shareholder acquired beneficial ownership of the shares before the date required to be set forth in the dissenters' notice pursuant to §48-23-203(b)(3), and deposit the shareholder's certificates in accordance with the terms of the notice.
- (b) The shareholder who demands payment and deposits the shareholder's share certificates under subsection (a) retains all other rights of a shareholder until these rights are cancelled or modified by the effectuation of the proposed corporate action.
- (c) A shareholder who does not demand payment or deposit the shareholder's share certificates where required, each by the date set in the dissenters' notice, is not entitled to payment for the shareholder's shares under this chapter.
- (d) A demand for payment filed by a shareholder may not be withdrawn unless the corporation with which it was filed, or the surviving corporation, consents thereto.

48-23-205. Share restrictions.

- (a) The corporation may restrict the transfer of uncertificated shares from the date the demand for their payment is received until the proposed corporate action is effectuated or the restrictions released under §48-23-207.
- (b) The person for whom dissenters' rights are asserted as to uncertificated shares retains all other rights of a shareholder until these rights are cancelled or modified by the effectuation of the proposed corporate action.

48-23-206. Payment.

- (a) Except as provided in §48-23-208, as soon as the proposed corporate action is effectuated, or upon receipt of a payment demand, whichever is later, the corporation shall pay each dissenter who complied with §48-23-204 the amount the corporation estimates to be the fair value of each dissenter's shares, plus accrued interest.
- (b) The payment must be accompanied by:
 - (1) The corporation's balance sheet as of the end of a fiscal year ending not more than sixteen (16) months before the date of payment, an income statement for that year, a statement of changes in shareholders' equity for that year, and the latest available interim financial statements, if any;
 - (2) A statement of the corporation's estimate of the fair value of the shares;
 - (3) An explanation of how the interest was calculated;
 - (4) A statement of the dissenter's right to demand payment under §48-23-209; and
 - (5) A copy of this chapter if the corporation has not previously sent a copy of this chapter to the shareholder pursuant to §48-23-201 or §48-23-203.

48-23-207. Failure to take action.

- (a) If the corporation does not effectuate the proposed action that gave rise to the dissenters' rights within two (2) months after the date set for demanding payment and depositing share certificates, the corporation shall return the deposited certificates and release the transfer restrictions imposed on uncertificated shares.
- (b) If, after returning deposited certificates and releasing transfer restrictions, the corporation effectuates the proposed action, it must send a new dissenters' notice under §48-23-203 and repeat the payment demand procedure.

48-23-208. After-acquired shares.

- (a) A corporation may elect to withhold payment required by §48-23-206 from a dissenter unless the dissenter was the beneficial owner of the shares before the date set forth in the dissenters' notice as the date of the first announcement to news media or to shareholders of the principal terms of the proposed corporate action.
- (b) To the extent the corporation elects to withhold payment under subsection (a), after effectuating the proposed corporate action, it shall estimate the fair value of the shares, plus accrued interest, and shall pay this amount to each dissenter who agrees to accept it in full satisfaction of the dissenter's demand. The corporation shall send with its offer a statement of its estimate of the fair value of the shares, an explanation of how the interest was calculated, and a statement of the dissenter's right to demand payment under §48-23-209.

48-23-209. Procedure if shareholder dissatisfied with payment or offer.

- (a) A dissenter may notify the corporation in writing of the dissenter's own estimate of the fair value of the dissenter's shares and amount of interest due, and demand payment of the dissenter's estimate (less any payment under §48-23-206), or reject the corporation's offer under §48-23-208 and demand payment of the fair value of the dissenter's shares and interest due, if:
 - (1) The dissenter believes that the amount paid under §48-23-206 or offered under §48-23-208 is less than the fair value of the dissenter's shares or that the interest due is incorrectly calculated;
 - (2) The corporation fails to make payment under §48-23-206 within two (2) months after the date set for demanding payment; or

- (3) The corporation, having failed to effectuate the proposed action, does not return the deposited certificates or release the transfer restrictions imposed on uncertificated shares within two (2) months after the date set for demanding payment.
- (b) A dissenter waives the dissenter's right to demand payment under this section unless the dissenter notifies the corporation of the dissenter's demand in writing under subsection (a) within one (1) month after the corporation made or offered payment for the dissenter's shares.

PART 3 JUDICIAL APPRAISAL OF SHARES

48-23-301. Court action.

- (a) If a demand for payment under §48-23-209 remains unsettled, the corporation shall commence a proceeding within two (2) months after receiving the payment demand and petition the court to determine the fair value of the shares and accrued interest. If the corporation does not commence the proceeding within the two-month period, it shall pay each dissenter whose demand remains unsettled the amount demanded.
- (b) The corporation shall commence the proceeding in a court of record having equity jurisdiction in the county where the corporation's principal office (or, if none in this state, its registered office) is located. If the corporation is a foreign corporation without a registered office in this state, it shall commence the proceeding in the county in this state where the registered office of the domestic corporation merged with or whose shares were acquired by the foreign corporation was located.
- (c) The corporation shall make all dissenters (whether or not residents of this state) whose demands remain unsettled, parties to the proceeding as in an action against their shares and all parties must be served with a copy of the petition. Nonresidents may be served by registered or certified mail or by publication as provided by law.
- (d) The jurisdiction of the court in which the proceeding is commenced under subsection (b) is plenary and exclusive. The court may appoint one (1) or more persons as appraisers to receive evidence and recommend decision on the question of fair value. The appraisers have the powers described in the order appointing them, or in any amendment to it. The dissenters are entitled to the same discovery rights as parties in other civil proceedings.
- (e) Each dissenter made a party to the proceeding is entitled to judgment:
 - (1) For the amount, if any, by which the court finds the fair value of the dissenter's shares, plus accrued interest, exceeds the amount paid by the corporation; or
 - (2) For the fair value, plus accrued interest, of the dissenter's after-acquired shares for which the corporation elected to withhold payment under §48-23-208.

48-23-302. Court costs and counsel fees.

- (a) The court in an appraisal proceeding commenced under §48-23-301 shall determine all costs of the proceeding, including the reasonable compensation and expenses of appraisers appointed by the court. The court shall assess the costs against the corporation, except that the court may assess costs against all or some of the dissenters, in amounts the court finds equitable, to the extent the court finds the dissenters acted arbitrarily, vexatiously, or not in good faith in demanding payment under §48-23-209.
- (b) The court may also assess the fees and expenses of counsel and experts for the respective parties, in amounts the court finds equitable against:
 - (1) The corporation and in favor of any or all dissenters if the court finds the corporation did not substantially comply with the requirements of part 2 of this chapter; or
 - (2) Either the corporation or a dissenter, in favor of any other party, if the court finds that the party against whom the fees and expenses are assessed acted arbitrarily, vexatiously, or not in good faith with respect to the rights provided by this chapter.

(c) If the court finds that the services of counsel for any dissenter were of substantial benefit to other dissenters similarly situated, and that the fees for those services should not be assessed against the corporation, the court may award to these counsel reasonable fees to be paid out of the amounts awarded to the dissenters who were benefited.

PART II INFORMATION NOT REQUIRED IN THE PROSPECTUS

Item 20. Indemnification of Directors and Officers.

Mississippi Business Corporation Act

The Mississippi Business Corporation Act ("MBCA") empowers a corporation to indemnify an individual who is a party to a proceeding because he is a director against liability incurred in the proceeding if

- he conducted himself in good faith;
- he reasonably believed in the case of conduct in his official capacity, that his conduct was in the best interests of the corporation, and in all other cases, that his conduct was at least not opposed to the best interests of the corporation; and
- · in the case of any criminal proceeding, he had no reasonable cause to believe his conduct was unlawful.

A corporation may also indemnify an individual who engaged in conduct which broader indemnification has been made permissible or obligatory under a provision of the articles of incorporation as authorized by Section 79-4-2.02(b)(5) of the MBCA. The termination of a proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent is not, of itself, determinative that the director did not meet the relevant standard of conduct.

Unless ordered by a court under Section 79-4-8.54(a)(3) of the MBCA, a corporation may not indemnify a director

- in connection with a proceeding by or in the right of the corporation except for reasonable expenses incurred in connection with the proceeding if it is determined that the director has met the relevant standard of conduct under the MBCA; or
- in connection with any proceeding with respect to conduct for which he was adjudged liable on the basis that he received a financial benefit to which he was not entitled, whether or not involving action in his official capacity.

The MBCA further provides that a corporation shall indemnify a director who was wholly successful, on the merits or otherwise, in the defense of any proceeding to which he was a party because he was a director of the corporation against reasonable expenses incurred by him in connection with the proceeding. Also, a corporation may, before final disposition of a proceeding, advance funds to pay for or reimburse the reasonable expenses incurred by a director who is a party to a proceeding because he is a director. The director must deliver to the corporation: (1) a written affirmation of his good faith belief that he has met the relevant standard of conduct described in the MBCA or that the proceeding involves conduct for which liability has been eliminated under a provision of the articles of incorporation as authorized by the MBCA; and (2) his written undertaking to repay any funds advanced if he is not entitled to mandatory indemnification under the MBCA and it is ultimately determined under the MBCA that he has not met the relevant standard of conduct described in the MBCA. The undertaking required must be an unlimited general obligation of the director. It need not be secured and may be accepted without reference to the financial ability of the director to make repayment.

A corporation may not indemnify a director as described above unless authorized by:

- the board of directors if there are two or more disinterested directors, by a majority vote of all the disinterested directors (a majority of whom shall for such purpose constitute a quorum) or by a majority of the members of a committee of two or more disinterested directors appointed by such a vote;
- special legal counsel selected in accordance with the MBCA; or
- the shareholders, but shares owned by or voted under the control of a director who at the time does not qualify as a disinterested director may not be voted on the authorization.

A corporation may also indemnify and advance expenses to an officer of the corporation who is a party to a proceeding because he is an officer to the same extent as for a director.

The bylaws of Peoples contain the following indemnification provision: "Any person, his heirs, executors, or administrators, may be indemnified by the corporation for reasonable expenses (including judgments and compromise settlements, except where as in a derivative suit situation any judgment in the matter would run in favor of the corporation) actually incurred in connection with any action, suit or proceeding, civil or criminal, to which he was made a party by reason of service as a director, officer, or employee of the corporation, provided, however, that no person shall be so indemnified or reimbursed as to any matter as to which he shall finally be adjudged to have been guilty of gross negligence, willful misconduct or criminal acts in the performance of his duty to the corporation; and, provided further, that no person shall be so indemnified or reimbursed as to any matter in such action or suit which has been as the subject of a compromise settlement except with the approval (1) of a court of competent jurisdiction, or (2) the holders of record of a majority of the outstanding shares of the corporation, or (3) a majority of the corporation's board of directors, excluding members who are parties to the same or substantially the same suit or proceeding. The foregoing right of indemnification or reimbursement shall not be exclusive of other rights to which such person may be entitled as a matter of law. The board of directors of the corporation may, in its discretion, purchase directors' and officers' liability insurance coverage to provide, in whole or in part, for such indemnification or reimbursement."

Peoples maintains an insurance policy insuring the corporation and its directors and officers against certain liabilities.

Item 21. Exhibits and Financial Statement Schedules

(a) Exhibits.

Exhibit No.	Description of Exhibit
2.1	Agreement and Plan of Merger, dated as of February 17, 2004, and related Plan of Merger among Renasant, The Peoples Holding Company and Peoples Merger Corporation, a wholly owned subsidiary of Peoples (attached as Annex A-1 to the proxy statement/prospectus, which is part of this registration statement)
5.1	Opinion of Phelps Dunbar LLP as to the legality of the shares of Peoples common stock to be issued in the merger
8.1	Opinion of Phelps Dunbar LLP as to certain tax matters
23.1	Consent of Ernst & Young LLP
23.2	Consent of Phelps Dunbar LLP (included in Exhibit 5.1)

- 23.3 Consent of Phelps Dunbar LLP (included in Exhibit 8.1)
- 23.4 Consent of Alex Sheshunoff & Co. Investment Bankers, LP
- 24.1 Power of Attorney (included on page II-5)
- 99.1 Form of Proxy Card of Renasant Bancshares, Inc.
- 99.2 Form of Election
- 99.3 Opinion of Alex Sheshunoff & Co. Investment Bankers, LP (attached as Annex C to the proxy statement/prospectus which is part of this registration Statement)

The Company does not have any long-term debt instruments under which securities are authorized exceeding ten percent of the average assets of the Company and its subsidiaries on a consolidated basis. The Company will furnish to the Securities and Exchange Commission, upon their request, a copy of all long-term debt instruments.

(b) Financial Statement Schedules.

All schedules have been omitted because they are either not applicable or the required information has been included in the consolidated financial statements or notes thereto incorporated by reference into this proxy statement/prospectus.

(c) Reports, Opinions or Appraisals.

Opinion of Alex Sheshunoff & Co. Investment Banking, LP (included as Annex C to the proxy statement/prospectus which is a part of this registration statement).

Item 22. Undertakings

- (a) (1) The undersigned registrant hereby undertakes as follows: prior to any public reoffering of the securities registered hereunder through use of a prospectus which is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.
 - (2) The undersigned registrant hereby undertakes that every prospectus: (i) that is filed pursuant to paragraph (1) immediately preceding, or (ii) that purports to meet the requirements of Section 10(a)(3) of the Act and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
 - (3) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of

appropriate jurisdiction the question of whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

- (4) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (b) The undersigned registrant hereby undertakes to respond to requests for information that is incorporated by reference into the proxy statement/prospectus pursuant to Item 4, 10(b), 11, or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statement through the date of responding to the request.
- (c) The undersigned registrant hereby undertakes to supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Tupelo, State of Mississippi, on April 7, 2004.

THE PEOPLES HOLDING COMPANY

by: /s/ E. ROBINSON MCGRAW

E. Robinson McGraw
President and Chief Executive Officer

KNOW ALL MEN BY THESE PRESENTS, that we, the undersigned directors and officers of The Peoples Holding Corporation, a Mississippi corporation ("Peoples"), do hereby name, constitute and appoint E. Robinson McGraw and Stuart R. Johnson, and each or either of them (with full power to each of them to act alone), our true and lawful agents and attorneys-in-fact, for us and on our behalf and in our name, place and stead, in any and all capacities, to sign, execute, acknowledge, deliver, and file (a) with the Securities and Exchange Commission (or any other governmental or regulatory authority), a Registration Statement on Form S-4 (or other appropriate form) and any and all amendments (including post-effective amendments) thereto, with any and all exhibits and any and all other documents required to be filed with respect thereto or in connection therewith, relating to the registration under the Securities Act of 1933 of Common Stock of Peoples to be issued in the merger between Peoples Merger Corporation and Renasant Bancshares, Inc. ("Renasant"), authorized by resolutions adopted by the Board of Directors on February 17, 2004, and (b) with the securities agencies or officials of various jurisdictions, all applications, qualifications, registrations or exemptions relating to such offering under the laws of any such jurisdiction, including any amendments thereto or other documents required to be filed with respect thereto or in connection therewith, granting unto said agents and attorneys, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises in order to effectuate the same as fully to all intents and purposes as the undersigned might or could do if personally present, and the undersigned hereby ratify and confirm all that said agents and attorneys-in-fact, or any of them may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this registration statement or amendment thereto has been signed below by the following persons in the capacities indicated on April 7, 2004.

		George H. Booth, II Director	
Date: April 7, 2004	by:	/s/ George H. Booth, II	
		William M. Beasley Vice Chairman of the Board and Director	
Date: April 7, 2004	by:	/s/ WILLIAM M. BEASLEY	
		Robert C. Leake Chairman of the Board and Director	
Date: April 7, 2004	by:	/s/ ROBERT C. LEAKE	

Table of Contents Date: April 7, 2004 /s/ Frank B. Brooks by: Frank B. Brooks Director Date: April 7, 2004 by: /s/ JOHN M. CREEKMORE John M. Creekmore Director Date: April 7, 2004 by: /s/ MARSHALL H. DICKERSON Marshall H. Dickerson Director by: KAREN S. DIXON Date: April 7, 2004 Karen S. Dixon Senior Vice President and Controller Date: April 7, 2004 EUGENE B. GIFFORD, JR. by: /s/ Eugene B. Gifford, Jr. Director Date: April 7, 2004 RICHARD L. HEYER, JR. by: Richard L. Heyer, Jr. Director Date: April 7, 2004 by: STUART R. JOHNSON Stuart R. Johnson Executive Vice President, Chief Financial Officer and Principal Financial Officer Date: April 7, 2004 /s/ E. ROBINSON McGRAW by: E. Robinson McGraw **President and Chief Executive Officer and** Director Date: April 7, 2004 /s/ J. NILES MCNEEL by:

by:

Date: April 7, 2004

J. Niles McNeel Director

C. Larry Michael Director

/s/ C. LARRY MICHAEL

Date: April 7, 2004	by:	/s/ Theodore S. Moll
		Theodore S. Moll Director
Date: April 7, 2004	by:	/s/ JOHN W. SMITH
		John W. Smith Director
Date: April 7, 2004	by:	/s/ H. Joe Trulove
		H. Joe Trulove Director
Date: April 7, 2004	by:	/s/ ROBERT H. WEAVER
		Robert H. Weaver Director
Date: April 7, 2004	by:	/s/ J. LARRY YOUNG
		J. Larry Young Director

EXHIBIT INDEX

Exhibit No.	Description of Exhibit
2.1	Agreement and Plan of Merger, dated as of February 17, 2004, and related Plan of Merger among Renasant, The Peoples Holding Company and Peoples Merger Corporation, a wholly owned subsidiary of Peoples (attached as Annex A-1 to the proxy statement/prospectus, which is part of this registration statement)
5.1	Opinion of Phelps Dunbar LLP as to the legality of the shares of Peoples common stock to be issued in the merger
8.1	Opinion of Phelps Dunbar LLP as to certain tax matters
9.1	Form of voting agreements (attached as Annex B-1 and Annex B-2 to the proxy statement/prospectus which is part of this registration statement)
23.1	Consent of Ernst & Young LLP
23.2	Consent of Phelps Dunbar LLP (included in Exhibit 5.1)
23.3	Consent of Phelps Dunbar LLP (included in Exhibit 8.1)
23.4	Consent of Alex Sheshunoff & Co. Investment Bankers, LP
24.1	Power of Attorney (included on page II-5)
99.1	Form of Proxy Card of Renasant Bancshares, Inc.
99.2	Form of Election
99.3	Opinion of Alex Sheshunoff & Co. Investment Bankers, LP (attached as Annex C to the proxy statement/prospectus which is part of this registration Statement)

EXHIBIT 5.1

PHELPS DUNBAR LLP COUNSELORS AT LAW

New Orleans, LA

Baton Rouge, LA

Houston, TX

CANAL PLACE
365 CANAL STREET • SUITE 2000
NEW ORLEANS, LOUISIANA 70130-6534
(504) 566-1311
FAX: (504) 568-9130

Tupelo, MS Gulfport, MS

Tampa, FL

Jackson, MS

www.phelpsdunbar.com

April 7, 2004

The Peoples Holding Company 209 Troy Street Tupelo, Mississippi 38802

Re: The Peoples Holding Company Registration Statement on Form S-4

Ladies and Gentlemen:

We have acted as counsel to The Peoples Holding Company, a Mississippi corporation (the "Company"), in connection with the preparation of the above-referenced Registration Statement on Form S-4 and related Prospectus filed by the Company with the Securities and Exchange Commission (the "Commission") on April 8, 2004, under the Securities Act of 1933, as amended, for the purpose of registering an aggregate of 1,478,000 shares of common stock, par value \$5.00 per share of the Company (the "Shares"), to be issued pursuant to that certain Agreement and Plan of Merger dated February 17, 2004 by and among The Peoples Holding Company, Peoples Merger Corporation and Renasant Bancshares, Inc. (the "Merger Agreement").

In rendering this opinion, we have examined and relied upon the original, or a photostatic or certified (or otherwise satisfactorily identified) copy, of such records of the Company and certificates of officers of the Company, and such other documents as we have deemed relevant and necessary as the basis for the opinion set forth below, including without limitation (i) the Articles of Incorporation of the Company, (ii) the Restated Bylaws of the Company and (iii) the Merger Agreement.

In such examination, we have assumed the genuineness of all signatures appearing on all documents, the legal capacity of all persons signing such documents, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as certified, conformed or photostatic copies, the accuracy and completeness of all corporate records made available to us by the Company, and the truth and accuracy of all facts set forth in all certificates provided to or examined by us.

Based upon the foregoing and subject to the limitations, qualifications, exceptions and assumptions set forth herein, and representations made to us by officers of the Company, we are of the opinion that:

- (1) The Company is a corporation duly incorporated and validly existing in good standing under the laws of the State of Mississippi.
- (2) The Shares are duly authorized and, when issued in accordance with the Merger Agreement, will be legally issued, fully paid and nonassessable.

The foregoing opinions are limited to the laws of the State of Mississippi. We express no opinion as to matters governed by the laws of any other state. Furthermore, no opinion is expressed herein as to the effect of any future acts of the parties or changes in existing law. We undertake no responsibility to advise you of any changes after the date hereof in the law or the facts presently in effect that would alter the scope or substance of the opinions herein expressed.

This letter expresses our legal opinion as to the foregoing matters based on our professional judgment at this time; it is not, however, to be construed as a guaranty, nor is it a warranty that a court considering such matters would not rule in a manner contrary to the opinion set forth above.

We consent to the filing of this opinion as an exhibit to the Registration Statement. In giving this consent, we do not admit that we are within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended, and the General Rules and Regulations of the Commission thereunder.

Very truly yours,

/s/ Phelps Dunbar LLP

EXHIBIT 8.1

PHELPS DUNBAR LLP COUNSELORS AT LAW

New Orleans, LA

Baton Rouge, LA

Houston, TX

CANAL PLACE
365 CANAL STREET • SUITE 2000
NEW ORLEANS, LOUISIANA 70130-6534
(504) 566-1311
FAX: (504) 568-9130

Gulfport, MS Tampa, FL

Jackson, MS

Tupelo, MS

www.phelpsdunbar.com

April 7, 2004

The Peoples Holding Company Post Office Box 709 Tupelo, MS 38802-0709

The Shareholders of Renasant Bancshares, Inc. c/o Board of Directors 2177 Germantown Road South Germantown, TN 38138

Re: The Peoples Holding Company – Acquisition of Renasant Bank

Ladies and Gentlemen:

We have acted as corporate and tax counsel to The Peoples Holding Company ("Peoples") in connection with the transaction described below. In that regard, we have been requested to issue our opinion as to certain matters included in the Registration Statement on Form S-4, as filed with the Securities and Exchange Commission (the "Commission") as filed by Peoples on April 8, 2004 (the "Registration Statement"), under the caption entitled "The Merger – Material United States Federal Income Tax Consequences."

In connection with the foregoing, we have examined and are relying upon the following documents: (1) Agreement and Plan of Merger by and among Peoples, Peoples Merger Corporation ("PMC") and Renasant Bancshares, Inc. ("Renasant") dated February 17, 2004 (the "Plan"); (2) the Articles of Merger between PMC and Renasant (the "Merger Certificate"); (3) the Registration Statement; (4) the Articles of Incorporation of Peoples; (5) the Bylaws of Peoples; (6) the Articles of Incorporation of PMC; (7) the Bylaws of PMC; (8) the Articles of Incorporation of Renasant; and (9) the Bylaws of Renasant. All references to capitalized terms, unless otherwise indicated, shall have the same meaning as such terms have in the Registration Statement.

We have made such legal and factual examinations and inquiries as we have deemed advisable and necessary for the purpose of rendering this opinion. We have examined originals or copies of documents, corporate records and other writings that we consider relevant for the purposes of this opinion. We also have discussed such matters as we have deemed relevant to this opinion with certain officers, directors and shareholders of Peoples, PMC and Renasant and, with respect to certain factual

matters, we have relied upon certificates of Peoples and/or Renasant and on representations, warranties and agreements in the Plan. We have assumed without investigation that any representations and warranties on which we relied that were given or dated earlier than the effective time of the Merger continue to remain accurate, insofar as relevant to such opinion, from such earlier date through and including the effective time of the Merger.

The Boards of Directors of Peoples and Renasant have determined that it is desirable and in the best interest of their respective corporations that Renasant be merged into PMC. The Plan provides that Renasant will merge into PMC in accordance with the laws of the State of Tennessee (the "Merger").

The Plan provides that each share of Renasant common stock, other than treasury shares, shares owned by Peoples or any of the subsidiaries of Peoples or Renasant (other than in a fiduciary capacity) or by any person who has perfected dissenters' rights with respect to shares of Renasant common stock, will be converted on the effective time into the right to receive the merger consideration. The merger consideration is either:

- cash in an amount equal to \$36.37, without interest, for each share of Renasant common stock;
- for each share of Renasant common stock, 1.117015 shares of Peoples common stock; or
- cash consideration for 45% of such holder's shares of Renasant common stock and stock consideration for 55% of such holder's shares of Renasant common stock.

Subject to the redesignation and adjustment procedures described below, the holders of record of shares of Renasant common stock ("Holders") may elect to receive all cash, all shares of Peoples common stock or the combination of cash and Peoples common stock described above as consideration in exchange for their shares of Renasant common stock. The Holders will not receive any fractional shares of Peoples common stock if they elect to receive all or a portion of the merger consideration as shares of Peoples common stock. Instead, the Holders will be paid cash in an amount equal to the fraction of a share of Peoples common stock otherwise issuable upon conversion multiplied by the average closing price of one share of Peoples common stock as reported by the American Stock Exchange for the ten trading days immediately preceding the last trading day immediately prior to the closing date of the Merger.

The Plan contains redesignation procedures which may affect Holders' elections. Under the Plan, the number of shares of Renasant common stock to be converted into the right to receive cash must not be less than forty-five percent (45%) nor more than fifty percent (50%) of the total number of shares of Renasant common stock outstanding immediately prior to the effective time of the Merger, and the number of shares of Renasant common stock to be converted into the right to receive shares of Peoples common stock must not be less than fifty percent (50%) nor more than fifty-five percent (55%) of the total number of shares of Renasant common stock outstanding immediately prior to the effective time of the Merger.

If the number of shares to be converted into the right to receive cash exceeds fifty percent (50%) of the outstanding shares of Renasant common stock, then all shares of Renasant common stock to be converted into the right to receive cash will be redesignated on a pro rata basis into a combination of shares to be converted into cash and shares to be converted into shares of Peoples common stock so that the total number of Renasant shares converted into cash does not exceed fifty percent (50%) of the outstanding shares of Renasant common stock. If the number of shares to be converted into shares of Peoples common stock exceeds fifty-five percent (55%) of the outstanding shares of Renasant common

stock, then all shares of Renasant common stock to be converted into shares of Peoples common stock will be redesignated on a pro rata basis for a combination of shares to be converted into cash and shares to be converted into shares of Peoples common stock so that the total number of Renasant shares exchanged for stock does not exceed fifty-five percent (55%) of the outstanding shares of Renasant common stock. Holders who choose to receive a combination of cash and shares of Peoples common stock are not subject to the redesignation procedures. Also, any Holder who elected to receive cash, but after the redesignation procedures would receive less than ten (10) shares of Peoples common stock for his or her shares of Renasant common stock, is not subject to the redesignation procedures.

Under the Plan, the value of the shares of Peoples common stock to be delivered to the Holders of Renasant common stock pursuant to their elections to receive the Merger consideration must have an aggregate value of at least forty percent (40%) of the value of Renasant as a whole. This is required so that the Merger qualifies as a reorganization under Section 368 of the Internal Revenue Code. If, after the redesignation procedures described immediately above are completed, Peoples determines that the total value of its common stock included as part of the Merger consideration does not satisfy the forty percent (40%) requirement referred to above, the amount of cash into which shares of Renasant common stock will be converted will be reduced, and the number of shares of Peoples common stock into which each share of Renasant common stock will be converted will be increased so that the value of Peoples common stock included as part of the Merger consideration will have a value of at least forty percent (40%) of the value of Renasant as a whole. Any adjustments as described in the foregoing sentence will be done in such manner as to ensure that the decrease in the total reduction in the amount of cash to be paid to the Holders of Renasant common stock is equal to the increase in the total value of the shares of Peoples common stock to be delivered to the Holders of Renasant common stock. All Holders of Renasant common stock are subject to these adjustment procedures, except for Holders who perfected their dissenters' rights under Chapter 23 of the Tennessee Business Corporation Act.

The opinions contained herein are based upon the representations and statements contained in the aforementioned documents and are limited in all respects to matters of federal income tax law. In rendering our opinion, we have assumed, with the approval of Renasant and Peoples, the following:

- (a) The due execution of the Plan, the due execution and proper filing of the Merger Certificate and the enforceability, in accordance with its terms, of the Plan and the effectiveness, in accordance with its terms, of the Merger Certificate.
- (b) The fair market value of the Peoples common stock and other consideration received by each Renasant shareholder will be approximately equal to the fair market value of the Renasant common stock surrendered in the Merger.
- (c) Except for Renasant's issued and outstanding capital stock, there are no interests (including, without limitation, warrants, options and debt instruments) which could be classified as stock or equity for federal income tax purposes.

On the basis of our examination of the aforementioned documents, the representations contained therein and the assumptions set forth above, and having considered the applicable federal income tax law as it exists on the date hereof, we are of the opinion that:

(i) Provided the Merger qualifies as a statutory merger under the laws of the State of Tennessee, the Merger will be treated as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code;

- (ii) A Holder will not recognize gain or loss if the shareholder exchanges Renasant common stock solely for Peoples common stock, except to the extent of any cash received in lieu of a fractional share;
- (iii) Where a Holder exchanges Renasant common stock solely for cash in the Merger, such cash will be treated as having been received by such Holder as a distribution in redemption of such Holder's Renasant common stock, subject to the provisions and limitations of Section 302 of the Internal Revenue Code;
- (iv) A Holder will recognize gain (but not loss) (whether the gain is dividend income or capital gain depends upon the application of Section 302 of the Internal Revenue Code relating to stock redemptions) if the Holder exchanges Renasant common stock for a combination of Peoples common stock and cash, and the Holder's gain will be equal to the lesser of:
 - (1) the excess, if any, of:
 - (a) the sum of the cash (excluding any cash received in lieu of a fractional share of Peoples common stock) and the fair market value of the Peoples common stock received (including any fractional share of Peoples common stock which is deemed to be distributed in the Merger and then redeemed by Peoples), over
 - (b) the Holder's tax basis in the Renasant common stock, or
 - (2) the amount of cash received;
- (v) A Holder's tax basis in the Peoples common stock received in the Merger will equal the Holder's tax basis in the Renasant common stock surrendered (less the tax basis allocable to any fractional share which is deemed to be distributed in the Merger and then redeemed by Peoples), increased by (1) the amount which was treated as a dividend, and (2) the amount of gain to the Holder which was recognized on such exchange (not including any portion of such gain which was treated as a dividend), and decreased by the amount of cash received in the Merger (excluding any cash received in lieu of a fractional share of Peoples common stock);
- (vi) A Holder's holding period for the Peoples common stock received in the Merger will include the holding period for the shares of Renasant common stock surrendered, provided that the Renasant common stock was held as a capital asset at the time of the Merger; and
- (vii) The receipt of cash in lieu of fractional shares of Peoples common stock will be treated as if the fractional shares were distributed in the Merger and then redeemed by Peoples. Generally, these cash payments will be treated as having been received as distributions in full payment in exchange for the shares considered redeemed.

This opinion does not address the tax consequences that may be relevant to particular categories of shareholders subject to special treatment under the Internal Revenue Code (including, but not limited to, financial institutions, tax-exempt organizations, mutual funds, insurance companies, S corporations or other pass-through entities, dealers in securities or foreign currency, Holders who exercise dissenters' rights, foreign Holders, Holders who acquired shares of Renasant common stock pursuant to the exercise

of employee stock options or otherwise as compensation or through tax-qualified retirement plans, traders in securities who elect the mark-to-market method of accounting for their securities holdings, Holders subject to the alternative minimum tax, Holders who have a functional currency other than the U.S. dollar, or Holders who hold Renasant common stock as part of a hedge against currency risk, straddle or a constructive sale or conversion transaction). In addition, this opinion does not address the state, local or foreign tax consequences of the merger.

This opinion is expressly limited to the federal income tax consequences of the Merger that are enumerated above. We express no opinion as to any other federal income tax consequences of the Merger or to matters governed by the laws of any state. Furthermore, no opinion is expressed herein as to the effect of any future acts of the parties or future changes in existing law. We undertake no responsibility to advise you of any changes after the date hereof in the law or the facts presently in effect that would alter the scope or substance of the opinions herein expressed.

This opinion is based upon various statutory provisions, regulations promulgated thereunder and interpretations thereof by the Internal Revenue Service and the courts having jurisdiction over such matters as of and effective on the date hereof, any one or more of which are subject to change either prospectively or retroactively. There can be no assurance that contrary positions may not be successfully asserted by the Internal Revenue Service or that a court considering such matters would not hold otherwise. Moreover, no opinion is rendered with respect to the effect, if any, which pending or proposed legislation may have on any of the foregoing matters.

This letter expresses our legal opinion as to the foregoing matters based upon our professional judgment at this time; it is not, however, to be construed as a guaranty nor is it a warranty that a court considering such matters would not rule in a manner contrary to the opinions set forth above.

This opinion is solely for the benefit of Peoples and the Renasant shareholders and may not be otherwise used, quoted, relied upon or referred to, nor may copies be delivered to any other person or entity without our prior written permission.

We consent to the filing of this opinion as an exhibit to the Registration Statement. In giving this consent, we do not admit that we are within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended, and the General Rules and Regulations of the Commission thereunder.

Sincerely,

/s/ PHELPS DUNBAR LLP

EXHIBIT 23.1

CONSENT OF INDEPENDENT AUDITORS

We consent to the reference to our firm under the caption "Experts" in the Registration Statement (Form S-4) and related Prospectus of The Peoples Holding Company and to the incorporation by reference therein of our report dated February 12, 2004, except for Note T, as to which the date is February 17, 2004, with respect to the consolidated financial statements of The Peoples Holding Company included in its Annual Report (Form 10-K) for the year ended December 31, 2003, filed with the Securities and Exchange Commission.

/s/ Ernst & Young LLP
Ernst & Young LLP

Birmingham, Alabama April 2, 2004

EXHIBIT 23.4

CONSENT OF ALEX SHESHUNOFF & CO. INVESTMENT BANKING

We hereby consent to the use of our opinion letter to the Board of Directors of Renasant Bancshares, Inc., included as Appendix C to the Proxy Statement and Prospectus which forms part of the Registration Statement filed April 8, 2004 on Form S-4 relating to the proposed Merger of Renasant Bancshares, Inc., and The Peoples Holding Company and to the references to such opinion therein.

In giving such consent, we do not admit that we come within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended, or the rules and regulations of the Securities and Exchange Commission thereunder, nor do we hereby admit that we are experts with respect to any part of such Registration Statement within the meaning of the term experts as used in the Securities Act of 1933, as amended, or the rules and regulations of the Securities and Exchange Commission thereunder.

ALEX SHESHUNOFF & CO. INVESTMENT BANKING

/s/ Alex Sheshunoff & Co. Investment Bankers, LP

AUSTIN, TX April 5, 2004

REVOCABLE PROXY RENASANT BANCSHARES, INC.

PLEASE MARK VOTE AS IN THIS EXAMPLE

SPECIAL MEETING	OF STOCKHOLDERS
	. 2004

Stock Inc., comr	er of su kholder 2177 C non sto	bstitution, to vote all shares of common stoors of Renasant Bancshares, Inc. to be held or Germantown Road South, Germantown, Ten	ck, \$1.0 n nessee 3	0 par value per share, which the undersigne _,, 2004 at 5:30 p.m., Central 38138, and at any and all adjournments the	ed is entitle time, at the reof. The p	e principal office of Renasant Bancshares,
1.	The proposal to approve and adopt (a) the Agreement and Plan of Merger dated as of February 17, 2004 by and among Renasant Bancshares, Inc., The Peoples Holding Company and Peoples Merger Corporation, pursuant to which Renasant Bancshares, Inc., upon satisfaction of certain conditions, will merge into Peoples Merger Corporation, with Peoples Merger Corporation surviving the merger as a wholly-owned subsidiary of The Peoples Holding Company, (b) the related plan of merger and (c) the merger.					
		FOR		AGAINST		ABSTAIN
2.	To tra	ansact such other business as may properly o	ome be	fore the special meeting or any adjournmen	its thereof.	
The 1	Board	of Directors unanimously recommends a	vote "I	OR" Proposal No. 1.		

THIS PROXY, IF PROPERLY EXECUTED, WILL BE VOTED AS SPECIFIED OR, WHERE NO DIRECTION IS GIVEN, WILL BE VOTED FOR PROPOSAL NO. 1 AND AT THE DISCRETION OF THE PROXIES ON ANY OTHER MATTERS THAT MAY PROPERLY COME BEFORE THE MEETING OR ANY ADJOURNMENT THEREOF.

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS.

Please be sure to date and	sign this Proxy in the box below.		
Stockholder sign above	Co-holder (if any) sign above	Date	
Detach above card,	date, sign and mail in postage paid er	velope provided.	
	1	RENASANT BANCSI	HARES, INC.
Please sign exactly as you	r name(s) appear(s) on this proxy care	d. Joint owners should	each sign. When signing in a representative capacity, please give full title.
	DATE, SIGI	PLEASE ACT PRO N AND MAIL YOUR	OMPTLY PROXY CARD TODAY
IF YOUR ADDRESS HA THE PROXY IN THE EN		THE ADDRESS IN T	HE SPACE PROVIDED BELOW AND RETURN THIS PORTION WITH
		_	
		_	
		_	

FORM OF ELECTION

To be submitted pursuant to an election in connection with the proposed merger of Renasant Bancshares, Inc. into Peoples Merger Corporation, a wholly owned subsidiary of The Peoples Holding Company ("Peoples").

A properly executed Form of Election must be received by the Exchange Agent by 5:00 p.m., Eastern standard time, three business days prior to the closing date of the merger (the "Election Deadline"). If a holder of common stock of Renasant Bancshares, Inc. (a "Renasant Stockholder") does not make an effective election, as determined in the sole discretion of Peoples, which it may delegate in whole or in part to the Exchange Agent, such stockholder will receive a combination of cash and shares of Peoples common stock as described below.

To: REGISTRAR AND TRANSFER COMPANY*

Delivering By Mail:

By Hand & Overnight Delivery:

10 Commerce Drive

Cranford, New Jersey 07016-3572

10 Commerce Drive Cranford, New Jersey 07016-3572

By Facsimile Transmission:

(908) 497-2313

Confirm by Telephone: (800) 456-0596, ext. 2502

* Deliveries made to an address other than any of the addresses stated above will not be acceptable and Exchange Agent will not be responsible therefor.

Ladies and Gentlemen:

In connection with the merger (the "Merger") of Renasant Bancshares, Inc. ("Renasant"), a Tennessee corporation, into Peoples Merger Corporation, a Tennessee corporation, and a wholly owned subsidiary of The Peoples Holding Company ("Peoples "), the undersigned hereby indicates a preference, subject to the terms and conditions set forth below, to have each share of Renasant Common Stock owned of record or beneficially converted into the right to receive either (i) cash in an amount equal to \$36.37, without interest, (ii) 1.117015 shares of common stock, par value \$5.00 per share, of Peoples, or (iii) the cash consideration described in (i) above for 45% of the holder's shares of Renasant Common Stock and the stock consideration described in (ii) above for 55% of the holder's shares of Renasant Common Stock, all as determined in accordance with Section 2.1 of the Agreement and Plan of Merger, dated as of February 17, 2004, by and among The Peoples Holding Company, Renasant Bancshares, Inc. and Peoples Merger Corporation (the "Merger Agreement").

It is understood that the following election is subject to the terms, conditions and limitations set forth in (i) the Proxy Statement/Prospectus relating to the Merger (the "Proxy Statement/Prospectus"), receipt of which is acknowledged by the undersigned, (ii) the Merger Agreement included as Annex A to the Proxy Statement/Prospectus, and (iii) the accompanying Instructions. The undersigned understands that neither certificates nor scrips representing fractional shares of Peoples common stock will be issued

in connection with the Merger, but in lieu thereof each holder of Renasant common stock who otherwise would have been entitled to a fraction of a share of Peoples common stock will be paid cash in an amount equal to the fraction of a share of Peoples common stock otherwise issuable multiplied by the average of the closing price per share of Peoples common stock for each of the ten (10) consecutive trading days immediately preceding the last business day prior to the closing date, as reported by the American Stock Exchange.

PLEASE READ CAREFULLY THE ACCOMPANYING INSTRUCTIONS

A properly executed Form of Election must be received by the Exchange Agent by the Election Deadline. If a Renasant Stockholder does not make an effective election, as determined in the sole discretion of Peoples, which it may delegate in whole or in part to the Exchange Agent, such stockholder will receive a combination of cash and shares of Peoples common stock as described in this Form of Election.

TYPE OF ELECTION (SEE INSTRUCTIONS B, C AND D) (PLEASE CHECK ONE BOX ONLY):
☐ CASH ELECTION
☐ STOCK ELECTION
☐ COMBINATION ELECTION

IMPORTANT: The merger agreement contains redesignation and readjustment procedures which may affect your election.

The merger agreement requires the following: (i) not less than 50% or more than 55% of the aggregate shares of Renasant common stock owned by Renasant stockholders must be exchanged for Peoples common stock; and (ii) not less than 45% or more than 50% of the aggregate shares of Renasant common stock owned by Renasant stockholders must be exchanged for cash.

If the aggregate number of shares elected to be converted into cash exceeds 50% of the outstanding shares of Renasant common stock, then shares of Renasant common stock to be converted into cash will be redesignated on a pro rata basis into shares to be converted into shares of Peoples common stock so that the total number of Renasant shares to be converted into cash does not exceed 50% of the outstanding shares of Renasant common stock. If the aggregate number of shares elected to be converted into shares of Peoples common stock exceeds 55% of the outstanding shares of Renasant common stock, then shares of Renasant common stock to be converted into shares of Peoples common stock will be redesignated on a pro rata basis into shares to be converted into cash so that the total number of Renasant shares to be converted into shares of Peoples common stock does not exceed 55% of the outstanding shares of Renasant common stock.

Holders of shares of Renasant common stock who elect to receive a combination of cash for 45% of their Renasant common stock and shares of Peoples common stock for 55% of their Renasant common stock will not be subject to these redesignation procedures. Also, a holder who has elected to receive cash for all of his or her shares of Renasant common stock and would receive less than 10 shares of Peoples common stock if his or her shares were redesignated is not subject to the redesignation procedures.

The merger agreement also contains readjustment procedures which may affect your election. Under the merger agreement, the value of the shares of Peoples common stock to be delivered to the holders of Renasant common stock pursuant to their elections to receive the merger consideration must

have an aggregate value of at least 40% of the value of Renasant as a whole. This is required so that the merger qualifies as a tax-free reorganization for federal income tax purposes with respect to the shares of Peoples common stock received in exchange for shares of Renasant common stock. If, after the redesignation procedures described immediately above are completed, Peoples determines that the total value of its common stock included as part of the merger consideration does not satisfy the 40% threshold described above, then Peoples will take two actions. Peoples will reduce the amount of cash into which shares of Renasant common stock will be converted, and Peoples will increase the number of shares of Peoples common stock into which each share of Renasant common stock will be converted. The reduction in the cash payable to shareholders of Renasant will equal the increase in the value of the Peoples common stock that will be delivered to the shareholders of Renasant. This adjustment is mandatory unless the adjustment would cause Peoples to issue more than 85,000 additional shares of its common stock and Peoples decides not to issue the additional shares, then either Peoples or Renasant may terminate the merger.

INSTRUCTIONS

A. SPECIAL CONDITIONS.

- 1. <u>Time in Which to Elect</u>. To be effective, an Election (as defined below) on this form or a facsimile hereof, must be properly completed, signed and submitted to the Exchange Agent, at the address set forth above, no later than 5:00 p.m., Eastern Standard Time, three business days prior to the closing date of the merger. Holders whose Forms of Election are not so received will not be entitled to specify their preference and will receive a combination of cash and shares of Peoples Common Stock in the Merger, defined as a Combination Election in Instruction B below.
- 2. <u>Change or Revocation of Election</u>. All elections will be irrevocable. Once a Renasant Stockholder delivers a Form of Election to the Exchange Agent, such stockholder may not sell, transfer or otherwise dispose of his or her Renasant Common Stock or change or revoke this election.
 - 3. Nullification of Election. All Forms of Election will be void and of no effect if the Merger is not completed.
- 4. <u>Stock Certificates</u>. If the merger is completed, the Exchange Agent will send to you a letter of transmittal which you must compete and return to the Exchange Agent with all of your certificates representing shares of Renasant common stock. **Do not send in your certificates representing shares of Renasant common stock with this Form of Election.**

B. TYPES OF ELECTIONS.

Subject to the provisions of Section 2.1 of the Merger Agreement, each share of Renasant common stock issued and outstanding immediately prior to the closing date of the merger shall be converted into the right to receive either (i) cash in an amount equal to \$36.37, without interest, (ii) 1.117015 shares of Peoples common stock or (iii) the cash consideration described in (i) above for 45% of the holder's shares of Renasant common stock and the stock consideration described in (ii) above for 55% of the holder's shares of Renasant common stock.

By properly completing the box marked "Type of Election," each holder may indicate (an "Election") that such holder desires to have all shares of Renasant Common Stock owned by such holder converted into the right to receive either:

• \$36.37 per share in cash, without interest ("Cash Election");

- shares of Peoples common stock, with cash in lieu of any fractional shares ("Stock Election"); or
- cash consideration for 45% of your shares of Renasant common stock and stock consideration for 55% of your shares of Renasant common stock ("Combination Election").

See Instruction D(2) for information concerning the right to make multiple Elections, and Instruction D(3) for information concerning the submission of multiple Forms of Election.

C. ELECTION, REDESIGNATION AND ADJUSTMENT PROCEDURES.

A description of the election, redesignation and adjustment procedures is set forth in the Proxy Statement/Prospectus under "The Merger Agreement — Merger Consideration" and "The Merger Agreement — Election and Election Procedures." A full statement of the election, redesignation and adjustment procedures is contained in Article II of the Merger Agreement. All Elections are subject to compliance with the election, redesignation and adjustment procedures set forth in the Merger Agreement.

In connection with making any Election, each Renasant Stockholder should read carefully, among other matters, the aforesaid description and statement and the information contained in the proxy statement/prospectus under "The Merger – Material United States Federal Income Tax Consequences."

D. GENERAL.

1. <u>Execution and Delivery</u>. This Form of Election must be properly completed, signed and submitted to the Exchange Agent by mailing, faxing or otherwise delivering to the Exchange Agent at the address set forth above.

The method of delivery of all documents is at the option and risk of the stockholder, but it is suggested that documents be sent by mail, registered mail, return receipt requested.

- 2. <u>Multiple Elections Not Allowed</u>. A holder must make a single Election for all of such holder's shares of Renasant common stock. Different elections may not be made for different portions of such holder's shares, except as set forth in paragraph 3 below.
- 3. <u>Multiple Forms of Elections</u>. Holders of record of shares of Renasant common stock who hold such shares as nominees, trustees or in other representative capacities (a "Representative") may submit multiple Forms of Election, provided that such Representative certifies that each such Form of Election covers all the shares of Renasant common stock held by each such Representative for a particular beneficial owner.

All questions with respect to this Form of Election and the Elections (including, without limitations, questions relating to the timeliness or effectiveness of any Election and computations as to redesignation or adjustment) will be decided by Peoples which may delegate such decision making authority to the Exchange Agent, which determinations shall be conclusive and binding.

Additional copies of this Form of Election may be obtained from the Exchange Agent (whose telephone number is (800) 456-0596, ext. 2502).

(Signature of Shareholder)	(Signature of Shareholder)	
Print Name:	Print Name:	
Date:	Date:	

The signature (or signatures) on the Form of Election should correspond exactly with the name(s) as written on the face of the certificate(s) unless the shares of Renasant common stock described on the Form of Election have been assigned by the registered holder(s), in which event the Form of Election should be signed in exactly the same form as the name of the last transferee indicated on the transfers attached to or endorsed on the certificates.

If the Form of Election is signed by a trustee, executor, administrator, guardian, officer of a corporation, attorney-in-fact, or in any other representative or fiduciary capacity, the person signing must give such person's full title in such capacity, and appropriate evidence of authority to act in such capacity must be forwarded with the Form of Election.